# Rescaling in flood risk governance – new spatial and institutional arrangements and structures

Thomas Thaler



Flood Hazard Research Centre School of Science & Technology Eine der vielen Absurditäten, auf die eine nur beschränkte und summarische Arbeit wie die vorliegende verfallen könnte, wäre es, mit einer »Geschichte« zu beginnen, die man dann sozusagen als eine abgetrennte Welt für sich behandelt und nur aus dem Grund beschwört, um die »Genesis« der Probleme verstehen zu können. Romano Alquati 1962/63.

Dedicate this PhD to:

Alan, Asia, Daniel, Ella, Jana, Mara, Sarah, Selma, Tobias and Valentina

#### Abstract

Flood governance and policy in Europe are changing: the role of the state and individual responsibility for risk management are now key contemporary issues in flood policy. The new policy agenda has been implemented to enhance the responsibilities of local authorities in flood risk management and reduce the controlling role of central national governments. The new strategies place the lead responsibility on *local* organisations to determine *local* strategies to manage *local* risks. This thesis examines the new role of these local authorities and organisations in flood risk management as well as how the nature of partnerships are established and operate, focusing especially on the main barriers and challenges. In principle, local authorities should be able to lead partnerships with local stakeholders to ensure effective local flood risk management. However, with current pressures on local authorities to reduce spending and a parallel reduction in the central state's resources, partnership has been seen as a possibility to both increase the value of budgets available.

The central aspects of this thesis are to investigate and to explore the influence of new scales on the interaction and performance of the different actors and their relationships; particularly with regard to power, processes and scaled networks. The research was carried out through a series of semi-structured interviews with both regional and local stakeholders in Austria and England. It demonstrates not only the importance of network connections between actors at the same scale, but refers also to the networks between actors and stakeholders at different scales, especially between local and national level. Local engagement strongly depends on social capacities, such as knowledge, motivation/self-interest, economics, networks, organisation and procedural capacity.

# Acknowledgements

First, I would like to thanks following people:

my Director of Studies Edmund Penning-Rowsell and my supervisors Sally Priest and Meri Juntti, David Ruggerio, Sven Fuchs, Christophe Viavattene, Dennis Parker, Florian Thaler, Hazel Faulkner, Meghan Alexander, Colin Green, Josie Joyce, Monika Stanzer, Carlos Dionisio Perez Blanco, Filippo Vecchiotti, Joanna Pardoe, Rhiannon Niven, Simon McCarthy, Sue Tapsell, Simrat Riyait, Konstantinos Karagiorgos, Meike Levin-Keitl,

in supporting my work with comments, advices, patience, inspirations, thoughts, data, discussions, administrative difficulties, literature, contacts, encouragement, nightmares and headaches.

In particular, I would like to thank my parents, siblings, brothers and sisters-in law, nephews and nieces, friends (especially Klaus, Michael, Jürgen and Daniela, Alexander, Maria, Christoph, Elisabeth and Martin), housemates (Andrea and Sandro) and my girlfriend Birgit in believing and encouraging me to survive the PhD process.

# Table of contents

Abstract									
A	cknowle	edgei	ments	5					
Та	ble of	conte	ents	6					
Lis	ist of figures								
Lis	List of tables								
List of abbreviations									
1.	Intro	ntroduction							
	1.1.	Bacl	kground and problem definition	. 12					
	1.2. Par		nership approaches in other European countries	. 13					
	1.3.	Poli	tical geography in flood risk management discourse	. 15					
	1.4.	Rese	earch aims	. 16					
	1.5.	Stru	cture of the thesis	. 20					
2.	Polit	tics o	f scale: scalar re-arrangement in flood risk management	. 22					
	2.1.	Aca	demic perspective on politics of scale	. 23					
	2.1.	1.	The concept of scale	. 24					
	2.1.	2.	The political economy of scale	. 27					
	2.1.	3.	Changes in scale constructions: consequences and outcome	. 30					
	2.2.	The	oretical perspective on governance	. 32					
	2.3.	Floc	od risk governance	. 37					
	2.4.	Rela	tionship between the different actors and stakeholders	. 41					
	2.4.1.	St	tate-of the Art	. 42					
	2.4.2.	E	ngagement and interaction in between the new scalar arrangements	. 46					
	2.5.	Key	findings	. 48					
3.	Rese	earch	methods	. 51					
	3.1.	Met	hods overview	. 51					
	3.2.	The	sis philosophy	. 54					
	3.3.	Qua	litative research method	. 57					
	3.3.	1.	Policy analysis	. 57					
	3.3.2.		Stakeholder mapping	. 58					
	3.3.3.		Semi structured in depth-interviews	. 60					
	3.4.	The	matic analysis	. 63					
	3.5.	Tria	ngulation research strategy	. 64					
	3.6.	Кеу	findings	. 68					

4.	Cas	e stu	dy approach	69
4	4.1.	Just	ification for selection of case studies and study sites	71
4	4.2.	Sele	ected case studies	76
	4.2.	1.	The Austrian flood legislation framework	76
	4.2.	2.	Austrian organisational level of flood risk management	79
	4.2.	3.	Inter-communal flood risk management in Austria	80
	4.2.	4.	Selected study sites in Austria	82
	4.2.	5.	The English flood legislation framework	94
	4.2.	6.	English organisational structure in flood risk management	99
	4.2.	7.	New partnership funding scheme in English flood risk management system.	. 102
	4.2.	8.	Selected study sites in England	. 115
4	4.3.	Кеу	findings	. 123
5.	Driv	vers a	nd scalar arrangements in flood risk management in the selected case studie	s126
ļ	5.1.	Driv	ers for network creation in flood risk management	. 126
	5.1.	1.	Effect of crises	. 129
	5.1.	2.	The influence of adequate leadership	. 131
	5.1.	3.	Economic rational	. 133
	5.1.	4.	Fair risk-sharing	. 133
	5.1.	5.	Regional embeddeness	. 134
	5.1.	6.	Common political visions and strategies	. 138
ļ	5.2.	Stat	e spatial process	. 139
ļ	5.3.	Кеу	findings	. 146
6.	Imp	acts	and outcomes of new scale constructions	. 148
(	5.1.	Shif	ts in the socio-economic relationships	. 148
(	5.2.	Nev	v governance structure in flood risk management	. 156
	6.2.	1.	New administration and regulative structure	. 158
	6.2.	2.	New planning practices and processes	. 159
(	5.3.	Nev	v network structure in flood risk management	. 162
	6.3.	1.	Formalised network structures	. 167
	6.3.	2.	Types of networks	. 169
	6.3.	3.	Network size and membership	. 171
	6.3.	4.	Interests	. 173
	6.3.	5.	Power	. 176
(	6.4.	The	governance networks in the flood risk management process	. 180

6.5.	Strategic unification	193		
6.6.	Territory and place in flood risk management	196		
6.7.	Key findings	197		
7. Per	Performance of new networks			
7.1. mana	Performance analysis of actors and stakeholder engagement in flood risk nagement policy			
7.1	1. Spatial planning	207		
7.1	2. Structural measures	211		
7.1	3. Stakeholder engagement	213		
7.1	.4. Risk communication	216		
7.2.	Dependency in partnership approaches	220		
7.2	2.1. Power dependency	222		
7.2	2.2. Resource dependency	228		
7.3.	Barriers and limitations of adopting a partnership approach	228		
7.4.	Key findings	230		
8. Cor	Conclusion			
8.1.	Answering the research questions	235		
8.2.	Social impact and further research	243		
Referen	ices	248		
Appendix				
Appendix 1: Details of interviews				
Арј	Appendix 2: EU Floods Directive			
Ap	Appendix 3: Memorandum capital grants			

# List of figures

Figure 2.1: Triangulation of flood risk governance	40
Figure 3.2: Iterative research design	52
Figure 4.3: Legislation structure in flood risk management in Austria	80
Figure 4.4: Overview Austrian study sides	83
Figure 4.5: Overview of Triesting-Tal study site	84
Figure 4.6: The organisational and structural set-up of the Triesting-Tal study site	85
Figure 4.7: Overview of Aist study site	88
Figure 4.8: The organisational and structural set-up of the Aist study site	89
Figure 4.9: Overview of III-Walgau study site	
Figure 4.10: The organisational and structural set-up in III-Walgau study site	
Figure 4.11: Policy responsibility and law structure in English flood risk management	system
	100
Figure 4.12: National and local flood defence funding in England and Wales, 1996-20	11 107
Figure 4.13: National Flood and Coastal Risk Management Grant-in-Aid Funding: 2003	7-08 to
2014-15	111
Figure 4.14: Overview English study sides	115
Figure 4.15: Overview Cockermouth study site	116
Figure 4.16: Overview Morpeth study site	119
Figure 4.17: Overview Bridgwater study site	121
Figure 5.18: Travel to work, Aist study site	135
Figure 5.19: Travel to work: III-Walgau	136
Figure 5.20: Travel to work, Triesting-Tal study site	137
Figure 6.21: Governance structure in Austria	156
Figure 6.22: Governance structure in England	157
Figure 6.23: Stakeholder overview in the Austrian case study	174
Figure 6.24: Stakeholder overview in English case study	175
Figure 6.25: Power and interests positions in Austrian case study	176
Figure 6.26: Power and interests positions in English case study	178
Figure 6.27: Role of national and regional authorities and local actors and stakeholde	rs in
partnership approach	184
Figure 7.28: Integration and network engagement in the selected study sites	207

# List of tables

Table 2.1: Streams in the partnership discussion	. 44
Table 2.2: Types of relationships in relation to commitment and time horizon in flood risk	
management	. 45
Table 3.3: Policy decision levels and geographic domains in flood risk management	. 53
Table 3.4: Research structure	. 57
Table 3.5: Taxonomy of triangulation	. 65
Table 4.6: Logics of politics of scale construction in flood risk management	. 72
Table 4.7: Justification for the selection of the study sites	. 74
Table 4.8: Overview of main legislations affecting Austrian flood risk management system	. 77
Table 4.9: Overview the role of the EA and LLFA under the Flood Risk Regulations 2009	. 98
Table 4.10: Key phases in English flood risk funding scheme	103
Table 4.11: Outcome measures scheme in England and Wales	109
Table 4.12: Local levy contributions by English RFCC	114
Table 4.13: Flood defence funding in Cockermouth	117
Table 4.14: Summary of economic feasibility of Bridgwater flood defence scheme	122
Table 4.15: The levels of contribution set by the tariff	123
Table 5.16: Drivers for network creation	126
Table 5.17: Preconditions in the development of new networks in flood risk management	129
Table 5.18: Current state spatiality in Austria and England	140
Table 6.19: Overview of state spatial process in Austrian and England	154
Table 6.20: Fuzzy boundaries and soft spaces in the selected case studies flood risk	
management policy	160
Table 6.21: Summaries types of relationships in the study sites	170
Table 6.22: Overview of the characteristics from the study sites	172
Table 6.23: Level of activity and self-realisation	182
Table 7.24: Characteristics of different level of engagement in the co-operation process	203
Table 7.25: Risk communication in flood risk management in selected study sites	217
Table 7.26: Power dependency resources and impact in Austria and England	223
Table 7.27: Main questions in the social interaction between the different actors and	
stakeholders	227
Table 8.28: Advantages and disadvantages of co-operations in flood risk management	240

## List of abbreviations

Asfinag – Motorway financing stock corporation (Autobahnen- und Schnellstraßen-Finanzierungs-Aktiengesellschaft)

BWV – Federal Water Engineering Administration (Bundeswasserbauverwaltung)

- CBA Cost-Benefit Analysis
- CEA Cost-Effectiveness analysis
- Defra Department for Environment, Food and Rural Affairs
- EA Environment Agency
- EU European Union
- FAG Flood Action Group
- FDGiA Flood and coastal erosion risk management Grant in Aid
- IDB Internal Drainage Board
- LLFA Lead Local Flood Authority
- MCA Multi-criteria analysis
- NFF National Flood Forum
- NIE New Institutional Economics
- NPM New Public Management
- NPPF National Planning Policy Framework
- ÖBB Austrian Federal Railways (Österreichische Bundesbahnen)
- OM Outcome Measures
- RFCC Regional Flood and Coastal Committee
- RFDC Regional Flood Defence Committees
- WFD Water Framework Directive

WLV – Austrian Service for Torrent and Avalanche Control (Wildbach- und Lawinenverbauung in Österreich)

WWF - World Wildlife Fund

## **1. Introduction**

#### **1.1.Background and problem definition**

Flooding is a central problem in our society. Several floods, like those that occurred in 2000, 2007, 2012 and 2014 in England or 2002, 2005 and 2013 in Austria (such as Danube Flood) caused high damages for the environment, economy and humanity (Görlach et al., 2005; Chatterton et al., 2010). This has focused the attention of policy makers as well as researchers on questions of vulnerability<sup>1</sup> reduction to natural hazards. Key challenges of how to approach the topic of flooding is that (1) not everyone is threatened by flood events, (2) there is a shift towards managing flooding towards acceptable risks rather than eliminating flooding, (3) there is an increased future flood risk, because of climate, demographic and land use change (4) there is a trend to build more and new houses in floodplain areas and (5) changes in consumption based on socio-economic circumstances, which provokes an increasing of the costs of flood damages (Johnson et al., 2007; Evans et al., 2004; Parker, 2007). This has led to an increasing discussion on changes beyond that of vulnerability and flood risk (Fuchs et al., 2013) such as climate change and dynamic forces of exposed societies (Keiler et al., 2010; Birkmann et al., 2013). The context of changing flood risks is driving a transformation in the role of the state in terms of responsibility sharing and individual responsibilities for risk management and precaution (Mees et al., 2014; Adger et al., 2013). Recent developments (in Europe) re-arrange a new role and responsibility for the state and to the individual households for flood risk management<sup>2</sup> (Defra, 2012; Johnson and Priest, 2008). The re-design of the

<sup>&</sup>lt;sup>1</sup> The term vulnerability herein used includes 'the characteristics of a system that describes its potential to be harmed. It can be expressed in terms of all functional relationships between expected damage and system, characteristics (susceptibility, value of elements at risk), regarding the whole range of relevant flood hazards' (Mens et al., 2007: 73). In accordance with the definition of United Nations, I divided the factor vulnerability into four groups: (1) economic, (2) social, (3) physical and (4) environmental (UN/ISDR, 2004).

<sup>&</sup>lt;sup>2</sup> Smith and Petley (2009) described risk management as a 'process through which risk is evaluated before strategies are introduced to manage and mitigate the threat' (ibid. 65). Flood risk management plans select final programmes of measures to reduce the flood impact (pre-flood reduction), but also to manage effectively and efficiently flood events as well as the recovery phase, i.e. psychological and economic assistance, support in recovery management, such as rebuilding, cleaning etc. Flood risk management includes three main components: (1) pre-flood reduction (i.e. structural and non-structural measures), (2) flood event reduction (i.e. emergency management) and (3) postflood reduction (i.e. recovery management).

relationship in flood risk management refers to the aspect of governance. Important issues are the different relationships (horizontal, vertical, diagonal) between the different agents (Steurer et al., 2010). At the European level, the different countries show different types of relationships betwenn the different actors, stakeholders and citizien involved in flood risk management discussion, which will discuss below.

#### 1.2. Partnership approaches in other European countries

In Germany, for example, the Wupperverband in North Rhine-Westphalia was established in 1930 with the key purpose of water management, e.g. water supply and water pollution. Today, the inter-local co-operation covers further tasks and objectives, mainly flood risk management, water quality (waste water treatment and sludge disposal), drinking water supply and ecological restoration of the river embankments. A key objective is the implementation of the EC Floods Directive and EC WFD, like III-Walgau study site (see section 4.2.4). The key tasks in flood risk management are the development and use of inter-local flood alleviation schemes (flood storages across the catchment). Key players have been the large industrial enterprises in the region, but they lost their influence over the last decades as a consequence of socio-economic changes in the region (Moellenkamp et al., 2008). The cooperation is organised in a two-way approach; a general assembly and a council with 15 members selected from different representative groups. The level of contribution is mainly based on the use of co-operation. The co-operation introduced different charges for citizien and businesses to flooding, hydropower, provision of drinking water, stormwater management, water course maintains or water pollution (Wupperverband, 2007). Stakeholder engagement and the influence from the citizens is very low meaning a 'classical' top-down flood risk management system. The democratic structures, legitimacy and accountability are based on the local council democratic processes, because the local authorities send their local politicians as representatives in the co-operation (Moellenkamp et al., 2008).

In the Netherlands, the water boards show similar structures and developments to the German and Austrian inter-local partnership approaches. The water boards play a central role in the Dutch water and flood risk management system. Compared to the Austrian and German examples, the water boards are regional authorities where the members are not public authorities (Moellenkamp et al., 2008; de Leeuw et al., 2012; Ward et al., 2013; Mostert, 2006). Today, the Dutch system includes 25 water boards reduced from 3,500 in 1850 or 2,500 in 1950, which are organised on river basin and regional approaches (Moellenkamp et al., 2008). The first water boards were established in the 1200/1300 AD (ibid.), especially with the first polder boards in 1400 (Mostert, 2006). The water boards have three main responsibilities: (1) water quality, (2) water quantity and (3) flood risk management with also a central role in emergency management planning (de Leeuw et al., 2012; Lamers et al., 2010). Warner et al. (2010) observed the main responsibility of water boards behind the dikes and not outside the dikes. The role of the provinces is to supervise the water boards (Mostert, 2006). Membership is legally obligated, but the representatives are selected by the citizens in the water board area by election. This encourages a higher engagement of the citizens compared to the German or Austrian examples. However, also the engagement of the citizens decreased over the last years (see for further information also Moellenkamp et al., 2008). The representatives were divided into four main groups: citizens, entrepreneurs, land owners and owners of buildings. The level of contribution is also based on these categories: inhabitant tax (citizens), property tax (house owners) and pollution levy (households and enterprises) (ibid.). Compared to the German and Austrian inter-local co-operation, the Dutch water boards show a higher democratic process and citizen engagement, because the members for the general assembly are elected by the citizens in the water board area. In most cases the water boards have no strong linkages with local authorities; an exemption is the City of Rotterdam with the Water Plan Rotterdam (Ward et al., 2013).

Beside the formalised approaches, I found also some European examples with highly informal agreements. Examples are the Weisseritz Regio in Germany or Tweed Forum in Scotland (Wirth et al., 2008; Cook et al., 2013). The Weisseritz Regio co-operation was established after the 2002 flood event; it was strongly enforced as well as organised by the Federal State of Saxony (hierarchical structure). The co-operation includes 15 members from local authorities, Federal State organisations and non-state actors, such as the Farmers association or representatives of land users. The co-operation has a one-page declaration with a general agreement of the aims and objectives (Wirth et al., 2008). The organisation structure is similar to the institutionalised inter-local co-operation in the Netherlands, Germany or Austria with a general assembly, a steering group, working groups and a coordinator (ibid.). The advantages of the informal approaches are the fast creation and rapid response and adaptation to new circumstances and frameworks. Nevertheless, the outcome is very limited compared to the other examples of inter-local co-operation. The key outputs have been the distribution of information and a new forest management concept for the catchments. The co-operation mainly gained from the strong financial contributions and political interests after the 2002 flood event.

## 1.3.Political geography in flood risk management discourse

The outcome of these changes and new social contracts between society and government was new scalar arrangements in the political decision-making practices. The scale concept plays a fundamental element in current flood risk management debates, especially relating to the question of risk- and responsibility-sharing between national and local authorities and nonstate actors<sup>3</sup>. Swyngedouw (2004) described scale 'as something that is 'produced'; a process that is always deeply heterogeneous and contested' (ibid: 34). On this basis, scale is never a priori (Smith, 1990), it is based on social construction and evolution and interaction of territory

<sup>&</sup>lt;sup>3</sup> In this thesis, non-state actors are assumed to be someone who is entirely private; not employed by any governmental organisation (Geeraerts, 1994).

and structure, which organise and rule social relationships (Cox, 1998; Gualini, 2006; Brenner, 2004; Swyngedouw, 1997; Bialasiewicz et al., 2013). Scale has been seen as both an absolute (fixed) term and as a relative (dynamic) process, which depends on the political discourse, conflicts and bargaining (Smith, 1990; Smith and Ward, 1987; Thiel, 2010). A key point is the focus on linkages (relationships, interconnections and interactions) between the different actors over spatial-temporal developments (Massey, 1992; Sayre, 2005; Cox and Mair, 1989), which strongly refers to the aspect of the distribution of power (Allen, 2009; Juntti et al., 2009; Evers and de Vries, 2013). Key points are the political processes and conflicts between the different actors (Cox, 1998; Smith, 1990; Sayre, 2005; Brenner, 2004; Swyngedouw, 2000). The re-scaling processes encourage new definitions and discussion of each actor in the policy discussion and social relation as well as their linkages in the definition and implementation of policy strategies (Reed and Bruyneel, 2010; Cox, 2013; Pugalis and Townsend, 2013). The outcome has been changes in the relationship between the different agents and the actual governance framework (Thiel and Egerton, 2011; Gualini, 2006; Thiel, 2010).

#### **1.4.Research aims**

The overall aim of this thesis is to examine the scalar re-arrangements in Austria and England and the consequences and impact on flood risk management. The project will examine how different institutional and governmental structures in the selected case studies approach flood risk management, with particular attention to the change of scales. Therefore, my assumption is that the politics of scale influences the involved actors<sup>4</sup> and stakeholders as well as their performance.

To address this aim, this research is guided by a number of objectives. The important aspect refers to the question of the selection and approval process of a programme of measures in

<sup>&</sup>lt;sup>4</sup> An actor is defined as someone who is involved and acts in the policy system in terms of defining goals and objectives (primary actor) or providing sub-goals or sub-tasks to achieve the policy goals (supporting actor) (Leroy and Arts, 2006). On the other side, stakeholders are understood as persons interested and actively engaged (in terms of support) in the policy system (Green and Penning-Rowsell, 2010). This differs from citizens, which are those who are not actively involved or interested in the policy system.

the policy discussion. Finally, the participation dimension includes the question of the political space for the mobilisation and discussion process of the different actors in the decision-process. The rescaling process includes a change in the relationship as well as new working relations and interventions between the different actors, e.g. new forms of relationships and interactions (Smith, 1990; Perkmann, 2007; Swyngedouw, 2004; Pugalis and Townsend, 2013).

The research objectives of this thesis are:

To investigate and to explore the influence of new scales (inter-local co-operation or localism debate) on the interaction and performance of the different actors and their relationship; particularly with regard to power, processes and scaled networks.

Therefore, based on the research objective four research questions have been developed to organise the research process; these are:

Question 1: Which mechanism and action influence the change of the actual scalar arrangements?

Scales are never fixed, because they are a product of social processes under a political strategy and ideology. Rescaling processes included three main directions: upwards, such as towards the European level; downwards, such as new localism or regionalism; or sidewards, such as privatisation and outsourcing public tasks to private actors, stakeholders and citizens. Therefore, new scalar re-arrangements have a strong influence on the change of level of activity and self-realisation between the different key actors and stakeholders in the policy decision-making practices.

Question 2: How cross-scale dynamics have influenced national flood risk management strategy?

Scalar re-arrangement cause changes in the actual formal and informal codes and norms as well as changes in actual administration practices, which also includes that actors and stakeholders can change their position in the decision-making practises (change of power relationships) to ensure their interests (scale jumping). Furthermore, scalar re-arrangements have an influence on the definition of the duties and responsibilities between the different political units as well as between state and non-state. This has a central influence in the governance structures. The horizontal level refers mainly to the out-scaling of duties and responsibilities, away from the public administration level towards private actors, stakeholders and citizens.

Question 3: What are the barriers and limitations of new scalar re-arrangements in terms of network governance, and what are their social and political functions and consequences?

Co-operations in flood risk management can be seen as an ideal instrument for the politics of flood risk management. Throughout, with current pressures on local authorities to reduce the spending and in parallel, a reduction in the central state's resources, inter-local co-operation has been seen as a possibility to both increase the 'value' of budgets available and to increase efficiency in using current public funds and resources. In general, a co-operation is not a 'business as usual' situation for the different members due to the high risk for each member (van Huijstee et al., 2007). A key question involves how the nature of the partnerships is established and how partnership is embedded in the political context, meaning the relationship between the different actors and stakeholders at different scales (Somerville, 2011; Glasbergen, 2011). Therefore, in this thesis I analyse the motivation and drivers in the creation of new relationships in flood risk management, focussing especially on their main barriers and challenges.

Question 4: What is the role of proximity in shaping the opportunities and barriers for partnership activities in flood risk management, and what are their social and political functions and consequences in flood risk management?

18

Partnership is seen as one form of interaction between state and non-state actors in policy decision-practices (McCay, 1996). The co-operation process seeks to engage a large number of different stakeholders in the policy discussion at same scale (van Huijstee et al., 2007). The key aim is in consensus building between the different actors, stakeholders and citizens with defined responsibilities, power, rights and functions (Berkes, 2010; Margerum, 2002; 2008; Borrini-Feyerdon, 1996). To analyse the integration of partnership activities, I use the concept of proximity. The engagement between different members in a partnership strongly depends on the distance (proximity) between the actors (Lundquist and Trippl, 2013). In analysing the proximity in flood risk management, this research focuses on following categories: spatial, institutional, social, technology and relational proximity. These five proximities are adopted to examine the partnership approach in flood risk management, such as inter-local co-operation in Austria and national-local co-operation in England. The five proximities are highly relevant for flood risk management, because they incorporate structural and non-structural measures such as spatial planning, emergency management, risk communication, but also the aspect of stakeholder engagement in the decision-making practices. These are used to examine and to analyse the current partnership approaches and relationships in flood risk management.

In this thesis, I analyse fairly new policy processes (designed and implemented in the past 10 years). Therefore, the focus is on the changes in the political processes and not the outcome. This is in the line with Sabatier (2007). Sabatier (2007) saw the evaluation of new policy process extremely complex, which included a wide range of different variables, such as time. Policy process needs to analyse with a time range of at least 20 to 40 years to fairly analyse the impact of new policies. Furhtermore, this thesis does not focus on the aspected of whether the selected flood risk management measures have changed as a consequence of scalar rearrangements.

19

#### **1.5.Structure of the thesis**

The thesis is divided into nine chapters. Chapter 2 provides a critical reflection of the existing literature. A key objective is to provide a critical evaluation of the geographical assumptions and work on scalar arrangements and re-arrangements. The aim is to provide an overview of political geography and politico-economic framework to analyse the transformation in statehood and the impact and consequences of this towards local levels. Furthermore, Chapter 2 proceeds by discussing the impact of new territorial state spaces into networks and cooperation discussion as well as addresses the question of the concept of governance in flood risk management. It highlights the problems and difficulties in conceptualising the term governance in the literature, especially the focus on flood risk management. It demonstrates the different approaches in the understanding of the term governance (discussed in more detail also in chapter 5-7). The next two chapters focus on the research design. Chapter 3 examines a critical reflection of the methodology used. It provides an overview of the research design as well as addressing the limitations in the used methodology as well as practical barriers and difficulties. Furthermore, the chapter provides a brief overview and justification of the selected case studies. Chapter 4 outlines the selected case studies. This thesis selected Austria and England as a primary research focus based on various criteria (matching, qualitative assessment and cross-case test). This chapter primarily examines the policy decision-making process in Austria and England beginning with stakeholder and institutional mapping to provide an overview of the main involved actors and stakeholders in the actual flood risk management policy. The aim is to analyse the position, role and interests of each actor and stakeholder in the actual policy process and decision-making practices as well as to understand their changes.

Chapter 5 analyses the drivers and circumstances in the changes of scalar arrangements, which re-define the governance structures in flood risk management policy. Chapter 6 covers the politics of scale of the selected case studies with focus on state spatial frameworks, new territorial creations, socio-spatial relationships and changes in the social interactions. I have observed that recent developments in the policy on flood risk management in the selected countries have changed. In Austria the new politics of scale foresees a catchment management approach in flood risk management. On the other hand, the English new politics of scale includes a stronger focus towards local solutions. This included a new arrangement of governance in each case study. Chapter 7 concentrates on the analysis of performance and interaction in the new governance (outcome and consequences of new politics of scale). The aim is to show the consequences and impacts of the scalar re-arrangements. It shows the types of relationships and networks in the selected case studies as well as the performance and the role of different actors in flood risk management. Finally, Chapter 8 provides a critical reflection of the theoretical assumption and empirical results in the thesis. It provides my key findings from the selected case studies. Furthermore, it examines some developments in further European countries to provide a general overview of scalar re-arrangements in Europe. The final section comprises some main policy implications and recommendations.

## 2. Politics of scale: scalar re-arrangement in flood risk

#### management

In flood risk management in the past two decades, various scholars have observed a shift in flood risk management policy debates (Kubal et al., 2009; Johnson et al., 2007). Recent developments re-arrange a new role and responsibility to the state and to the individual households for flood risk management (Defra, 2012; Johnson and Priest, 2008). This rearrangement of new roles and responsibilities towards private-public actors and citizens has been driven by various developments including: recent financial and economic crises (Defra, 2012; Stockhammer, 2012), implementation of new legal requirements (EC, 2000; 2007) and recent flood events, e.g. the summer floods in 2007, Danube floods in 2002, or Elbe floods in 2002 (Pitt, 2008; Defra, 2005; BMLFUW, 2010; BMU, 2005). These developments suggest that policy agendas should enhance responsibilities of different actors in flood risk management and reduce the controlling role of central governments. The change implies a new contract in the relationship between the different actors (Pearce and Ayres, 2012; Adger et al., 2013), e.g. in the Flood and Water Management Act (2010) the key issue is to encourage the responsibilities of local authorities. The consequences were various scalar re-arrangements in policy decision-making practices, from planning to maintenance. The new politics of scales encourage the implementation of new scales, which have a strong influence for governance, such as the shift from local flood risk management to catchment-wide management strategies.

The purpose of this chapter is to examine the theoretical discussion on scalar re-arrangements and provide the conceptual setting for the research. Correspondingly, this chapter reviews the political geography literature to gain a clearer understanding of scale and its relevance to governance. The scale concept forms an essential element in relating to the question of riskand responsibility-sharing in flood risk management, such as the implementation of catchment-wide management plans, between national and 'lower' local authorities (such as local, district or regional authorities) and non-state actors (see chapter 2 and 7). Politics of scale defined the governance arrangement and structure in flood risk management. Therefore, new scalar arrangements may condition the networks relationships and power settings.

## 2.1.Academic perspective on politics of scale

Swyngedouw (2004) understood scale as an ongoing socio-political process, a process that is always deeply heterogeneous and contested' (ibid: 34). Scale has been seen as a absolute (fixed) term and as a relative (dynamic) process, which depends on the political discourse, conflicts and bargaining (Smith, 1990; Smith and Ward, 1987; Thiel, 2010; Marston et al., 2005) as well as from the social reproduction and consumption (Marston, 2000). An important issue is the dynamic-evolution perspective of scalar arrangements, because of the ongoing changes in the social structures between public and private actors and stakeholders (Brenner, 1997). This can be understood as a product of social processes (Smith, 1990). In this line, Fainstein (1999) underlined the importance of analysing the social structures of each actor and stakeholder in the ongoing policy discussion with a focus on the relationship between humanenvironment interactions. Following these authors, the term scales is understood as a highly political discussion.

On the other hand, current developments in the academic research understood scales as a depoliticisation debate, especially in the introduction of the scale debate into New Institutional Economics (NIE) and a multi-level governance debate, such as spatial fits and misfits (Moss, 2012). For example, Gibson et al. (2000) defined scale as 'the spatial, temporal, quantitative, or analytical dimensions used to measure and study any phenomenon' (ibid: 218). The authors divided the term into absolute and relative scales. Absolute scale is 'the distance, time, or quantity measured on an objectively calibrated measurement device' (ibid: 218). Conversely, relative scale is 'a transformation of an absolute scale to one that describes the functional relationship of one object or process to another (e.g. the relative distance between two locations based on the time required by an organism to move between them)' (ibid: 218). Additionally, the theoretical discussion of the scale concept can be distinguished between temporal (management cycle) and spatial boundaries (administration, hydrology, ecosystem and economy) (Dore and Lebel, 2010; Cash et al., 2006). Nevertheless, the boundaries interact between each other, so the 'challenge arises out of the incorrect assumption that there is a single, correct, or best characterisation of the scale and level' (Cash et al., 2006: 4). Therefore, the New Institutional Economics School<sup>5</sup> excludes the socio-political developments and impacts in the constructions and arrangements of scale. The outcome is an exclusion of the question of power, democracy, social constructions as well as social justice and equity in their research discussion, which is a central point in this doctoral research. Therefore, the construction of scale depends on socio-historical developments and conflicts which are continuously changing; depending on the political system and alliances between the different actors, stakeholders and citizens (Jones, 1998).

#### 2.1.1. The concept of scale

Since the 1970s-1980s, I observed within the literature a general consent in increasing the individual (outscaling), local (downscaling) or international (upscaling) responsibility (Smith, 1990; Smith and ; Ward, 1987; Lefebrve, 2007; Masula et al., 2008). This highlights also a change in the policy making processes towards broader hybrid decision-making practices (Mees et al., 2012). Scalar arrangements are currently under review (fluid), where those scalar re-arrangements are socially constructed and influenced by the social interaction at geographical spheres (Brown and Percell, 2005; Brenner, 2004). Moreover, this includes also fixed arrangements in terms of pre-defined social structures and power, especially the hegemonic position of elite groups in the policy decision-practices (Brown and Percell, 2005). In this line, scales include a dialectical interplay between fixed and fluid scalar structures, where power and hegemony is a central point in the ongoing policy-decision practice,

<sup>&</sup>lt;sup>5</sup> This image of scale discussion as a 'hipster-styled' discussion (=exclusion of the politics).

especially the 'control over space' (Allen, 2009: 199). The outcome of the scalar rearrangements has led to a new definition of responsibility and power between the actors, stakeholders and civic society, but also to new policy frames, relationships and networks as well as policy intervention (Pugalis and Townsend, 2013; Thiel and Egerton, 2011). The governance networks are particularly important for political collaborations and interaction across scales to ensure individual interests and needs (Paasi, 2004). This includes that the space has to be seen in a broader sense to include socio-spatial relationships (Jessop et al., 2008). A key question is to understand the spaces of engagement, spaces to act and the capacity to act (Perkmann, 2007) as well as changes in the social interactions (Wissen, 2009; Mees et al., 2012; 2014). Consequently, the change of scalar arrangements leads also to a review of the relationship and interplay between the state and the civil society (Swyngedouw, 2011; Amin, 2004; Pellizzoni, 2011; Neal, 2013; Coaffee, 2005). A key question is the space of engagement to ensure the stakeholder needs and interests (Cox, 2013; Perkmann, 2007).

The new politics of scale encouraged local authorities and institutions to deal with certain local risks. The aim is that local institutions take over responsibilities from the central government. A central ideology in the literature is that local decisions are often seen as a 'better' solution in terms of social justice<sup>6</sup>, accountability<sup>7</sup>, legitimacy<sup>8</sup> and transaction costs<sup>9</sup> in compare to

<sup>&</sup>lt;sup>6</sup> Different concepts of social justice exist (Elster, 1992; Mill, 2010). At its most basic level, social justice concerns questions over fairness in the distribution of resources, capital and wealth across different members of society. The different concepts of social justice differ in their interpretations of fair resource distribution (Varian, 1975). For the discussion of social justice and flood protection, not only the actual distribution of flood protection measures is important (distributive justice) (Johnson et al., 2008), but also the way how this allocation is achieved. Social justice also considers the process how a certain distribution was reached (procedural justice) (Walker and Burningham, 2011). Procedural justice includes moral and ethical standards in the policy discussion (Leventhal, 1980), such as fair allocation – equality of opportunity (e.g. information), exclusion of self-interests in policy discussions by policy makers, evaluation and adaptation of policy implementation processes or possibility to be included in the policy discussion process (Green, 2011).

<sup>&</sup>lt;sup>7</sup> Accountability has seen as a dynamic process, where 'decision-makers have to justify and explain their action' (Bäckerstrand, 2006: 295). In traditional understanding, 'accountability has been defined in its principal form as the mediated relationship between governors and governed, whereby the governors have to justify their actions to the governed as result of being granted delegated authority' (Joss, 2010: 409). The term accountability is distinguished into: (1) internal and (2) external accountability. The internal accountability refers to public administration, which has to justify their actions and decisions towards the society, e.g. electorate control mechanism. The external accountability refers to the justification of decision-makers above all to affected stakeholders of their decision affected by policy strategy. Key points of accountability are: (1) monitor policy discussion and actions and (2) sanction decision if they fail (Leventhal, 1980). This relates accountability to formal – delegation relationships

national decisions. However, this depends strongly on the localities. Localities are highly heterogeneous structures, references and developments (Clarke and Cochrane, 2013). This includes that some local authorities have advantages from a strong localism policy, whilst others not. The outcome has been that local authorities have introduced their own policies and strategies and not only have replied to the national targets and policies. Nevertheless, the new scale construction enforced new questions and problems: (1) who is responsible, e.g. individual, collective, private, public (identity<sup>10</sup>) (2) for which tasks and risks (event<sup>11</sup>), (3) when (time) and (4) how (prescriptions<sup>12</sup>), especially relevant in division between private and public discourse. The central Government defines the question of who, what, when and how.

<sup>10</sup> The attribute identity focuses on the individual actor, especially their motivation, role and characterise.

between authorities and their direct agents, e.g. governors and governed, employer and employees, national and local authorities (Blagescu et al., 2005). As a summary, accountability includes four principles (1) transparency, (2) participation, (3) evaluation, and (4) feedback (Whitty, 2008).

<sup>&</sup>lt;sup>8</sup> Political discussion and decisions made based on negotiations and arrangements between the different stakeholder groups. The enforcement of non-state actors, especially as a response to the various crises, has increased the need for legitimacy in policy decision processes. In general, some scholars see legitimacy as a justification of certain actions and decisions (evaluation instrument). While scholars generalised the term legitimacy and tried to include a more holistic evaluation of policy decision and making process, the term became more unclear. Suchman (1995) described legitimacy as 'a generalized perception or assumption that the actions of an entity are desirable, proper, or appropriate within some socially constructed system of norms, values, beliefs, and definitions' (ibid: 574). Mees et al. (2014) divided legitimacy into (1) input, (2) throughput and (3) output legitimacy. Input legitimacy is oriented towards the relationship and co-operation between public and private agents in the policy discussion, e.g. availability and access to information, transparency of discussion, and monitoring process (e.g. representation of various interests). Further, input legitimacy included the aspects of social equity and justice as well as ensures accountability. Instead output legitimacy focuses on efficiency and effectiveness of problem solving, e.g. clear definition of targets and objectives, response towards problems in the implementation process (e.g. acceptance of stakeholders). Output legitimacy is related more to the variable of economic efficiency. Throughput legitimacy refers to quality of participation, in particular, the potential influence of stakeholders in the policy decision-making practices as well as quality of deliberation, which focus on the quality of the discussion within the decision process (Suchman, 1995; Mees et al., 2014).

<sup>&</sup>lt;sup>9</sup> Transaction costs occur, because there are no perfect market conditions under general equilibrium. Therefore, the transaction costs negotiate the terms of the exchange. The most relevant types of costs are: search and information costs, bargaining and decisions costs, control and enforcement costs as well as regulation of externalities (McCann et al. 2005). In public administration, transaction costs can be understood as costs for decision-making processes and implementations (Birner and Wittmer, 2004). In this way, McCann et al. (2005) defined transaction costs inclusive of administrative costs. This included costs of failures or 'wrong' decisions (sunk costs) in policy decision and implementation process for the public administration (Roggero and Fritsch, 2010). Birner and Wittmer (2004) distinguished transaction between those which are necessary for the policy decision process, e.g. research, collection of information, meetings, information exchange etc. and those which are necessary in the implementation process, e.g. monitoring. In respect to time and frequency in public administration, Birner and Wittmer (2004) distinguished between (1) operational decision, (2) day-to-day decisions and (3) strategic decision, e.g. co-ordination between different organisations and agents.

<sup>&</sup>lt;sup>11</sup> Schlenker et al. (1994) summarised under the term event all aspects of action and the response (consequences of the activity).

<sup>&</sup>lt;sup>12</sup> Prescriptions refer especially to the aspect of rules (formal and also informal) for the individual and collective behaviour and interaction.

Pellizzoni (2004) argued the changes in the public administration 'entails the enhancement or a change in the relations of responsibility' (ibid: 542). However, the 'transition' from the bureaucratic state to a more market oriented and civil society state had also led to a strong shift in the responsibility debate. Under the traditional bureaucratic government the main responsibility resides with the central government. In the market and civil society state the public administration encourages a shared responsibility between private and public actors. Dicke and Meijerink (2008) defined the shift of responsibility between public – private as a process 'beyond the gradual changes that we have seen in the past. Many countries feel that their flood management is in need of review because of the many floods in the past decade and the foreseen impacts of climate change' (ibid: 495). The involvement of 'new' actors had a strong influence in the policy discussion and decision with the consequences of a more decentralised flood risk management system.

Consequently, we can see this as a step back in history. Although, key driver in the shift of public – private spheres is the increasing market activities, therefore, the outcome has been an increased individual responsibility in dealing with risks and environmental problems. A central question refers to the definition of public and private.

#### 2.1.2. The political economy of scale

The political economy of scale includes different forms of scales and their impact and consequences for regulation practices and social conflicts (Swyngedouw, 2000). Additionally, Wissen (2009) argues that the rescaling concept is an important framework to understand the process of the new localism in the risk- and responsibility sharing debate. He summarised, that 'the social production of the scale concept tends to sidestep the structuring effects of scalar configurations as well as the social struggles through which these effects may be challenged' (ibid: 886). A further point in the scale debate is the role of public administration in the process of rescaling; there is not less public administration with rescaling tasks upwards, downwards or outwards; only a change in their purposes and objectives. For instance, 'it is largely accepted

that the national state does not simply disappear when 'new state spaces' are produced' (ibid: 886). Other aspects are like places, territories, networks and physical – ecology restrictions (ibid.).

A central tenet is the introduction of socio-spatial relationships in to the analytic research for a more holistic understanding of scale concepts and linkages to policy discussions and their outcomes (Clarno, 2013). Today, based on the work of Neil Brenner, Erik Swyngedouw and others, scholars include a broader view of the scale concept (Brenner, 2004; Swyngedouw, 1997; Jessop et al., 2008; Gualini, 2006). The authors describe the scale concepts as the interaction between human, environment and political discussion in a time and spatial view. To this extent, scholars define the rescaling process as 'the spatial, temporal, quantitative, or analytical dimensions used to measure, or rank, and study any phenomenon and levels as the units of analysis that are located at different positions on a scale' (Dore and Lebel, 2010: 62). Jones (1998) observed the scale arrangements and production based on geographical and cultural variables. These include also 'a contingent outcome of structural forces and practices of human agents' (Cox, 2009: 885), e.g. social justice and equity, power and stakeholder engagement. For example, Jessop et al. (2008) distinguish socio-spatial relationship in four dimensions: (1) territory, (2) place, (3) scale and (4) network. These analyses include historical, political and social development (see chapter 3). In particular, regulatory codes, norms, and institutions are spatially shifted from one scale to another (Glavovic et al., 2010). Consequently, the rescaling process enforces a change of the actual structure and characteristics of governance, which allows more or less actors and stakeholders to get engaged in the decision making (see section 2.2). A key point in this conceptualisation is the focus on linkages (relationships, interconnections and interactions) between the different agents over spatialtemporal developments (Massey, 1992; Sayre, 2005; Cox and Mair, 1989).

Central aspects in scale discussions are the hierarchical nesting of responsibility and power organised and shared between different scales (Brenner, 2004). In line with Howitt (1993), the

hierarchical concept (national-regional-local relationship) is a key attribute of the concept of scales, but the concept of scale 'should not be seen as a simple hierarchical concept' (ibid: 36). Nevertheless, Wissen (2009) mentioned that 'scales can only be understood in their relationship to other scales' (ibid: 888). On this basis, the concept of scale is based on social construction and evolution and interaction of territory and structure, which organise and rule social relationships (Cox, 1998; Gualini, 2006; Brenner, 2001; 2004; Swyngedouw, 1997; Howitt, 1993). Thus Howitt (2003) assumed three key elements, which summarised the term scale:

- size (based on the physical units, e.g. change in political boundaries),
- level (hierarchical relationship) and
- relation and interactions which are socially and politically constructed.

The interactions between scale units are intended to go in different directions (simultaneously up-, side- and downwards), rather than only one-way interactions (top-down). National policies encourage local authorities to take over certain tasks and responsibilities (downwards). Simultaneously, there is an upwards trend towards international organisations, e.g. EU wide policies and regulations. Simultaneously, the interaction between scales is also defined and organised by the actors. In the literature, various scholars see scale jumping as a central issue in the scale discussion (Smith, 1992; Moore, 2008). Jumping between scales is defined as the 'breakout' of actors in their actual scale position. For example, local actors network at national or international level or vice versa to ensure their interests and objectives. This describes the dynamic processes and changes in the social structure (Moore, 2008). In general, jumping scale enforces the change in the boundaries (Smith, 1992). Jumping scale is generally a synonym of the 'ability of social groups and organisations to move from lower to higher levels' (MacKinnon, 2011: 24), mainly to ensure their interests and objectives (Collinge, 1999). The outcome of scale jumping is the change in regulatory approaches, power structure and processes between the actors (Swyngedouw, 1997). A central problem refers to the question of power as an

outcome of scalar re-arrangements. Smith (1990) added the distinction between spatial (or physical) and social concentration and centralisation of power. Power is a crucial element in the interaction of individuals or groups (Foucault, 1982). It is a key attribute in the social relationship and structure (e.g. policy relationship) between different entities. Rothschild (2002) listed power in the policy discussion mainly linked to aspects of lack of transparency, especially asymmetric distribution of information between policy makers and stakeholders, as well as between different stakeholder groups. Power is seen as the possibility to modify the behaviour of actors, stakeholders and citizens in decision-making practices; such as decision or non-decision making. Further, this thesis understands power as well as the domination of power, to influence also against the individual interests, such as moral concepts, objectives or individual attitudes (Lukes, 2005). Therefore, power is embedded in the socio-cultural context, such as the individual habitous. Following Sayer (2004), power has to be seen in the social structures of the actors and not only in the actors themselves, especially the context of using power. Consequently, this thesis views key elements of power in the policy decision making process at different scales as: who decides in the policy processes, how power is used, who are excluded from the process and who is responsible in national, sub-national, or local level for decision making and implementation.

#### 2.1.3. Changes in scale constructions: consequences and outcome

Rescaling debates encourage new definitions or discussions of each agent in the policy discussion as well as their linkages in the definition and implementation of policy strategies (Edwards et al., 2001). The outcome has been changes in the relationship between the different agents (Thiel, 2009; Gualini, 2006; Thiel and Egerton, 2011). Overall, the rescaling process promoted a change in the governance structure (the interaction between the different actors, in terms of planning and managing). It also influenced and enforced new boundaries of interaction, in terms of planning and managing (Metzger and Schmitt, 2012). Allmendinger and Haughton (2009) introduced the terms 'soft spaces' in line with 'fuzzy boundaries' as a key

outcome of the rescaling process (see more detail information in section 6.5). 'Soft spaces' and 'fuzzy boundaries' are informal strategies, such as the introduction of catchment-wide management plans without 'hard-formal' contracts. The motivations are based on rescaling processes (transfer of power and responsibility to new authority structures) or the involvement of new actors or stakeholders in the decision-making process to overcome the official-formal procedure in the decision-making process, which cannot respond adequately to the existing institutional settings. The consequence is the creation of new 'sub-regions' outside the 'hard' political boundaries (Haughton et al., 2013; Allmendinger and Haughton, 2010). The change in the scale arrangements and possibility to move between the scales is often based on a political strategy, especially by actors in an unfavourable position at one scale arrangement. Their objective is to overcome this position in the policy discussion (Brown and Purcell, 2005; Smith, 1992). A key objective is to understand the mobilisation of the local actors and their influence on changing the social structures, especially the shift in the power structure (Cox, 1998; van Schendel, 2002; Smith, 2001a).

This thesis argues that politics of scale effectually defines the engagement of actors and stakeholders across scales in the decision-making processes (see chapter 6). In past years, in several European Member States there has been a significant rescaling processes ongoing on certain duties and responsibilities away from national level to lower or higher political units (European, regional and local level). This has had a central influence to the definition of who is involved in the decision-making process. The horizontal level refers mainly to the out-scaling of duties and responsibilities, away from the public administration level towards private actors, stakeholders and citizens (profit or charity organisations). The co-operation between state and non-state actors and stakeholders plays a crucial role in the governance discussion.

#### 2.2. Theoretical perspective on governance

In the last 30 years, the term governance has become very popular, but at the same time very contested between schools of academic thought. The term itself is not new (Jordan et al., 2005; Jessop, 1998) and there is a general consensus amongst scholars that governance refers to non-hierarchical, flexible structures, that co-ordinate state and non-state actors (Jordan et al., 2005; Watson et al., 2009; Stoker 1998; Moss and Newig, 2010). The term derives from the Latin word *gubernatia*, which means steering, direction, control, management (Jessop, 1998; Le Gales, 1998). 'It originally referred mainly to the action or manner of governing, guiding, or steering conduct and overlapped with 'government'' (Jessop, 1998: 30). In the past years, the term governance was transformed into a broader definition and understanding with the result that the term itself has become increasingly unclear (Davies, 2002). The consequences were that the term lacks a clear critical definition in the literature (Jordan et al., 2005). Today, governance is a very fashionable term, which tries to understand transformations in public administration and the 'new' role of the state. However, there remains the question of whether the transformation has already happened or not; and if governance can be used as a 'general' theory to understand the past and predict the future (Frederickson, 2004).

An important aspect is that governance is not the same as government (Rhodes, 1996; Stoker, 1998; Jordan, 2008) or like governing (Jordan, 2008). Differences between the terms are widely discussed by different scholars. Stoker (1998) underlined that in the Anglo-American language 'government' refers mainly to the aspects of public state as an institution (as an organisation and a structure). In fact, government refers to the monopoly of power to define, implement and enforce rules to achieve defined targets and objectives. Weber (1991) noted government as the organisation which has the 'monopoly of the legitimate use of physical force within a given territory' (ibid: 78). In the classical understanding, a government has the power to define overall goals and assist different actors to achieve these targets (Jessop, 2004).

Treib et al. (2007) added that the 'government' concept is characterised by elected representatives, who define policy objectives and which are implemented by bureaucrats. In particular, scholars often make a clear reference to a strong bureaucracy state (Stoker, 2006; 2011). In sum, the main characteristics of the term 'government' are:

- strong hierarchical organisation structure (top-down decisions, communication, delegation, control, support),
- 'continuity' in career development (payments and position depending on age and working years instead of performance),
- in-personality (equality for everyone), especially to reduce the possibility to favour any agents,
- privileged expertise 'trained to perform their functions and positioned to control access to privileged knowledge' (Watson et al., 2009: 449) and
- strong parliamentary and national sovereignty.

The main criticism of government in the literature refers to issues such as inflexibility, strong hierarchical structures, top-down decisions and discourage of individual responsibility in negative terms (Watson et al., 2009; Weber et al., 2011; Treib et al., 2007; Bache and Flinders, 2004).

On the other hand, governance focuses on 'participation of public and private actors, as well as non-hierarchical forms of decision-making' (Kohler-Koch and Rittberger, 2006: 28), in which, actors and stakeholders from outside the public administration play a crucial role for the development and use of new policy instruments (design and implementation) (Jordan et al., 2005; Lemos and Agrawal, 2006; Weber et al., 2011). Key objectives of governance include organising/co-ordinating the different interests, choices and conflicts between the different actors (Green, 2010; Fatti and Patel, 2013). Thinking this way, Janssen (2008) described governance 'as a form of co-ordination' (ibid: 349). Furthermore, Watson et al. (2009) regards governance as an ideal case of self-organised policy networks, without or with the limited influence of central governments; as well as a key focus on negotiation between state and non-state actors at an equal level within market-based control instruments. In fact, governance defines new roles between the different state and non-state actors (Doak et al., 2004). Leon (2010) underlined that the shift from government to governance is directly correlated with the change of the question "who is responsible for what". With the introduction of a neo-liberalism paradigm in the policy discussion, a central government is encouraged to shift its own duties and responsibilities to non-state actors. This results in an increased share of responsibility between public and private actors, stakeholders and citizens. The redistribution of the tasks – in terms of politics of scale – include three directions:

- upwards to international or transnational organisations,
- downwards to regional or local authorities and
- outwards to non-state actors or quasi non-governmental organisations (*quangos*) (Reed and Bruyneel, 2010; Gibbs and Jonas, 2001).

In this line, the term governance focussed on the introduction of new economic policy instruments (Jordan et al., 2005), such as the use of market-based policy instruments like price, trading schemes, liability instruments etc. (Jordan, 2008). Since 1980s, several public administration reforms are linked to the 'roll-back'<sup>13</sup> and 'roll-out'<sup>14</sup> doctrine (Peck and Tickell, 2002; 2012). Rhodes (1996) summarised under the term governance in particular the discussion from the New Institutional Economics (NIE) with the '3 Es': economy, efficiency, and effectiveness. The focus is more on private methods like management, performance efficiency, using benchmarks and evaluation methods, reducing costs, introducing competition into public administration, with a strong focus on the performance of the output (Bevir et al., 2003). The

<sup>&</sup>lt;sup>13</sup> 'Roll-back': deregulation process in the 1980s-1990s under Margaret Thatcher and Ronald Reagan (Peck and Tickell, 2002).

<sup>&</sup>lt;sup>14</sup> 'Roll-out': adaptation of institution and organisation to neo-liberalism policy in the 1990s under Tony Blair, Bill Clinton or Gerhard Schroeder (Peck and Tickell, 2002).

motivation is to introduce private market instruments into public administration to reduce the costs and improve efficiency. One reason is the lack of accountability-argument in the public administration. Scharpf (1994) noted in his paper that the classical hierarchical model of classical government (see above) understanding does not provide 'all the answer for' to the new socio-economic framework, such as globalisation or new technological and communicative developments. The author mentioned that 'the advantages of hierarchical coordination are lost in a world that is characterised by increasingly dense, extended and rapidly changing patterns of reciprocal interdependence, and by increasingly frequent, but ephemeral, interactions across all types of pre-established boundaries, intra- and interorganisational, intra- and intersectoral, intra- and international' (ibid: 37). A key ideology is that the public administrations in a decentralised approach are appropriate as they will be more up todate and respond better and faster to the inherent complexity in the world. For example, various policy departments collaborate with other departments from relevant sectors (at same or different scale position).

Further, a key discussion in the political economy of scale is the debate about private and public (Brenner, 2004; Swyngedouw, 1997; Cox, 2013). The term 'public' is defined as an open field for everyone (low excludability and rivalry). In contrast, private is understood as a closed or exclusive area (Habermas, 1989). Private refers mainly to the aspect of private autonomy. Private autonomy (*Privatautonomie*) is the key element in liberal economic understanding or modern economic law. The term is based on the liberal principles, which activities and decisions are based on the freewill and action of the individual citizens. The restrictions are based on the general interests. For Habermas (1989) the public authorities 'corresponded more or less overtly to the compromise between competing private interests' (ibid: 132), especially for needs which cannot be satisfied by the market. This intends that 'the public sphere, which now had to deal with these demands, became an arena of competing interests fought out in the coarser forms of violent conflict' (ibid: 132). Main aim of the public sphere is

a balance between the different interests and conflicts of different actors, with the result that the conflicts moved from private to public level. However, the latest developments were the enforcement of private sector to take over tasks, duties and responsibility from the public sector.

The key aim of governance strategies and structures is to define certain policy goals and measures. The current focus is on the changing role of the state, away from central planning and regulating organization to a more a decentralised self-organised organism, which is quite flexible to adapt rapidly new developments and circumstances. The key role of governance processes is to ensure and enable different actors in (horizontal, vertical, diagonal) co-operation process to avoid social exclusion. Important issues are the different relationships between the different agents (Steurer et al., 2010). However, most scholars use governance as the response of government and civil society to new circumstances (response to the change in the society). Bache and Flinders (2004) underlined that 'governance is not simply a new phrase for describing what governments do' (ibid: 96).

This thesis understands governance as undertaken by a group of different people such as actors, stakeholders or citizens, with different background, interests, purposes, and objectives, interacting in the policy decision-practices are therefore simultaneously both outcomes of, and productive to, social relationships. The social relationships between different actors, stakeholders and citizens have to be seen in a very flexible manner. For example, related to flood risk management, social relationship (co-operation) between different actors, stakeholders and citizens do change over time (co-operation between national/regional/local bodies with planners, mediator, farmers, businesses, and private households). An important question is related to power (who posseses it, how and who uses it) (Green and Penning-Rowsell, 2010). In this line, Le Gales (1998) defined the term governance as 'the capacity to integrate and give form to local interests, organisations and social groups and, on the other hand, the capacity to represent them outside, to develop more or less unified strategies

36
towards the market, the state, other cities and other levels of government' (ibid: 90). Therefore, the broad range of different actors and stakeholders involved in the policy process is central to the definition of governance. Within this concept of governance includes the plurality of the society (Virno, 2004; Sorensen, 2006). 'For Spinoza, the *multitudo* indicates a *plurality which persists as such in the public scene*, in collective action, in the handling of communal affairs, without converging into a One, without evaporating within a centripetal form of motion. Multitude is the form of social and political existence for the many, seen as being many: a permanent form, not an episodic or interstitial form' (Virno, 2004: 21). The group (or multitude) is always in the changing process of (also in spatio-temporal aspects and dimension): co-operation between the members depends on specific objectives; new objectives implicitly demand new constellation with different members (membership is very flexible). However, the co-operation between the different actors and stakeholders are not continuous; they are changing over the time based on policy objective and individual interests.

## 2.3.Flood risk governance

To this point, this chapter has reviewed the concept of governance more broadly. The discussion now turns to how governance has evolved concerning flood risk management. As a starting point a central problem in past governance discussions is the missing link to geographical scales (section 2.1), especially with refer to flood risk governance. To overcome this, Fürst (2003) introduced the term governance into the geographical discussion. For him, governance includes two aspects: (1) functional aspects (a focus on problem or themes instead of territorial or natural boundaries) commonly investigated as part of governance and (2) territorial aspects. This latter territorial aspect focuses on the units in a geographical scale (territorial or natural boundaries). A key point here is the spatial orientation of involved actors and stakeholders in policy decision-making practices.

Following the definition of governance from section 2.2, flood risk governance is understood as different actors and stakeholders from differing societal and political backgrounds with diverse purposes and objectives finding a way to work together in a defined territorial framework, such as a catchment where the actors develop together a catchment-wide management plan. This means that the actors are 'acting' and interacting without geographical content, for example see discussion about proximity below (see section 2.4.2). Private actors (with different interests and affected differently (directly or indirectly) by an event) thereby have a more important role in the policy process, e.g. stakeholder engagement, outsourcing, volunteer work, individual responsibility or other forms (Kettl, 2000; Treib et al., 2007; Wiek and Larson, 2012). Sendzimir et al. (2010) suggested flood risk governance to frame as: (1) involvement of multiple actors in the policy discussion, especially outside of public administration (bottom-up), (2) change language towards integrated flood risk management, and (3) social learning, scholars found a strong link between social learning and adaptive governance (Huntjens et al., 2010; 2012). Such developments can be found in various policy strategy papers or introductions of new law standards (Defra, 2010b; BMU, 2005). Scholars observed, flood risk governance often as a network and a co-operation between the different agents, also with different policy planning background, e.g. agricultural policy, spatial and land use management, emergency management (Adger et al., 2013; Razafindrabe et al., 2014; Coulthard and Frostick, 2010; Ingold et al., 2010) or in terms of risk-sharing to decrease vulnerability in specific defined areas (upstream-downstream), e.g. flood storage, removal or reduce flood defence scheme alongside agricultural areas (Erdlenbruch et al., 2009). The relationship between the different public and private actors in flood risk management policies becomes centralised by this debate. Different scholars summarise the importance of a strong co-operative approach between the different actors in the policy discussion (Fatti and Patel, 2013; Razafindrabe et al., 2014), but often describe a lack of co-operation and harmonisation between the different actors and policy documents which can increase the frustration between the different actors. From examples in the literature, the greatest hindrances and challenges in network governance can be found in the role of regulations, unclear responsibility, informal-formal institutions, capacity and resources between the different actors in dealing with flood risk management (Razafindrabe et al., 2014; Hung et al., 2010).

This work integrates the term governance into flood risk management. Flood risk governance is strongly based on the discussion of flood risk management. The interviews showed that the term flood risk governance has a broader understanding as the focus towards the request of an existing infrastructure (structural or non-structural flood risk management measures). Flood risk governance has started already with the planning and assessment discussion as identified in Austrian and England. Differently to Porse (2013) and Coulthard and Frostick (2010); I understood flood risk governance mainly as the result of existing flood alleviation schemes.

Flood risk governance distinguishes between the terms risk, i.e. flood risk assessment (hazard identification, vulnerability assessment and risk estimation), and the term governance, which includes the process of building and acting of a group of different agents (see text above). The term itself incorporates descriptive (conceptual) understandings as well as a normative framework to understand uncertainty, complexity and ambiguity of risks. Consequently, the thesis understands flood risk governance concept as based on people, risk infrastructure and scalar arrangements (see figure 2.1):



Scalar arrangements

Infrastructure, such as structural and nonstructural measures as well as risk perception

# Figure 2.1: Triangulation of flood risk governance Source: own development

First, flood risk governance has a strong dependence on the given infrastructure (Coulthard and Frostick, 2010). For example, Porse (2013) viewed in his example, that 'cities without major stormwater infrastructure typically also lack effective stormwater governance structures, both within and outside of municipal government' (ibid: 34). Furthermore, Porse (2013) identified flood risk governance in the context of flood defence infrastructure (structural as well as non-structural). He described a direct relationship between the flood defence infrastructure and the governance structures and processes. His argument is a lack of infrastructure causes a lack of governance structures. In general, both the terms risk and governance have two different approaches and meanings. However, the different sources of flooding, e.g. river flooding, pluvial flooding, flash flooding, groundwater flooding etc., does not create and define different governance approaches. In England, for example, the implementation of the Water Industry Act (1989) enforced a change in the flood risk governance structure. The Water Industry Act introduced private water industries to take over the responsibility of urban drainage system (see also Coulthard and Frostick, 2010) Second, flood risk governance transfers the flood risk management into politico-economic discussion. Consequently, the term include 'the complex web of actors, rules, conventions, processes and mechanisms concerned with how relevant risk information is collected, analysed and communicated' (Renn, 2008: 9). According to van Asselt and Renn (2011), flood risk governance should be understood as 'more than just a descriptive shorthand for a complex, interaction network in which collective binding decisions are taken around a particular set of societal issues' (ibid: 432). The authors argued that the term incorporates descriptive (conceptual) understandings as well as a normative framework to understand uncertainty, complexity and ambiguity of risks. Flood risk governance 'can be interpreted as a bridge, combining the idea of 'sound' science with that of democratic participation' (De Marchi, 2003: 171). Nevertheless, De Marchi (2003) pointed out that the combination of the two different concepts 'are apparently separate, but belong instead to spheres of investigation and practical interest that are strictly intertwined and partially overlapping' (ibid: 171). The key aspect of the terms are the analysis of flood risks, policy structure and process (descriptive logic) and core elements of mode of governance (accountability, transparency, legitimacy, efficiency and social equity). Therefore, flood risk governance 'can be used as a checklist of how decisions are made (descriptive use), and at the same time functions as a guideline of how to organise the decision process when complex decision have to be made (normative model)' (van Asselt and Renn, 2011: 436).

## 2.4. Relationship between the different actors and stakeholders

The outcomes of the creation of new scalar arrangements (and the possibilities of jumping between scales) are new networks (coalition) between different actors, stakeholders and citizens. Scholars distinguished between three main forms of networks: bottom-up, hierarchical or top-down as two extreme forms and responsible autonomy as the middle way (Kooiman, 2003; Martin, 2011; Haupter et al., 2005; see also section 6.4). Governancenetworks is 'a governing arrangement where one or more public agencies directly engage nonstate stakeholders in a collective decision-making process that is formal, consensus-oriented, and deliberative and that aims to make or implement public policy or manage public programs or assets' (Ansell and Gash, 2008: 544). Influentially, this thesis describes the network and cooperation as a common working process, where the different actors and stakeholders provide a problem definition as well as the policy solutions (Fenwick et al., 2012; Prager, 2010; Somerville, 2011). The term partnership is used to describe the relationship between different actors, e.g. state and non-state actors or between state actors at the same scale like inter-local co-operation approach (Teamey, 2010; Berkes, 2010; Brinkerhoff, 2002; Harrison, 2002).

Historically, since the 1970s the term co-operation and network has become central in the development aid arena, later strongly adopted in the neo-liberalism economic policy discussion, e.g. Private-Public-Partnership. Today, we can find a wide range of partnership approaches considered in various policy discussions (Maxwell and Riddell, 1998; Fowler, 1998; Abrahamsen, 2004; Pitt, 2008).

## 2.4.1. State-of the Art

In the literature, the term partnership includes two main directions (Glasbergen, 2011). The first direction is a focus on the internal structures and processes (internal factors). This includes mainly the question of the internal decision-making process. A second direction is the focus on the external structures and processes (external factors), e.g. partnership approach between two or more actors at different scales. The term also can be said to be the change in the relationship between state and non-state actors (Abrahamsen, 2004). Important other issues include the exchange of information and knowledge between the different stakeholders as well as ensuring a 'fair' relationship between them (Fowler, 1998). In summary, 'partnership was intended to be an equality in ways of working and mutuality in respect for identity,

position and role' (Fowler, 1998: 141). In the literature, scholars distinguished the term partnership in general between three different areas (table 2.1).

First, the normative stream is primarily focused on the non-state actors critical to the 'classical' policy discussion and general public administration practices. The aim is to encourage a large number of actors and stakeholders in the process to 'improve' policy discussion and implement 'successfully' the policy goals and targets. The idea behind this is to increase the accountability and the acceptance of new policy directions. The policy discussion and decision process (governance process) is enforcing the engagement of local scales in the decision-making practices (Berkes, 2010).

Second, the reactive stream argued that 'partnerships are important not so much in themselves but in order to achieve development objectives' (Teamey, 2010: 7). The main focus here appears to be primarily in the public relationship of central government, e.g. publishing reports inclusive project documentation, mission statements, annual/mid-term reports etc. (Brinkerhoff, 2002; Teamey, 2010). The aim is to increase the transparency in the relationship between state and non-state actors.

The third and final stream (instrumental) focuses on the implementation process. The implementation process can be understood as the capacity building (or function) of partnership (Berkes, 2010). Here, the term partnership referred to the achievement of the policy aims and objectives (Brinkerhoff, 2002; Teamey, 2010).

43

#### Table 2.1: Streams in the partnership discussion

Stream	Description	Aim
Normative stream	Primarily critique of policy discussion in the past (top-down approach)	Stakeholder engagement (main principles participation and empowerment) based on knowledge generations, for instead learning from communities about their risk; aim to increase social equity and justice and to ensure accountability and legitimacy.
Reactive stream	Response to the normative stream; justify neo-liberal policy instrument.	To ensure and improve public relationship between private and public sector, e.g. improve access and availability of information (improve transparency).
Instrumental stream	Main focus on the implementation process of the policy guidelines (applied function of partnership).	Achieve the targets and objectives (main principles efficiency and effectiveness).

Source: adapted from Berkes (2010), Brinkerhoff (2002), Teamey (2010)

On one hand, Abrahamsen (2004) characterised the partnership approach very positively, to ensure legitimacy and accountability as well as social justice and equity and the effective implementation of resources. On the other hand, he viewed partnership very critically. The main critique refers to the implementation and use of partnership in policy with the missing transfer of power between the different actors. In a similar way, Fowler (1998) observed partnership often as rhetoric outcome, because the main focus of partnership approach is only on the implementation performance of policy guidelines instead of solidarity and to integrated local knowledge to the policy definition, such as learning effects from previous natural hazards events. The key point is the intended outcome from the partnership approach. Added value can be understood also as additional benefit for involved actors. The definition of value-added indicators is guite difficult, because each partnership project is unique and due to the use of different experimental techniques as well as each project having different goals, partners, challenges etc. However, the question involves how partnership is embedded in the political context, meaning the relationship between the different actors at different scales (Somerville, 2011). An important aspect is the relationship between the different actors, stakeholders and citizens in decision-making practices (see also chapter 7).

The key aim of networks is to ensure long-term goals instead of a short-term benefit maximisation the benefits for all the members of the co-operation instead of individual members (table 2.2) (Rössl, 2005a).

	Time framework	Commitment	Short description	Examples in inter- local flood risk
				management
Competition	Short	Low	Key objective is to gain relative advantages instead of long- term relationship	Local protection schemes without catchment wide management actions
Egoistic	Medium	Modest	Key objective is to gain the absolute advantages, under consideration of the relative advantages	Catchment-wide management system as long as costs and preferences are higher than local protection measures; highly different between 'rich' and 'poor' communities
Co-operation	Long	Strong	Key objective is to gain the absolute advantages with the aim of coproduction between the actors and stakeholders	Inter-local flood risk management with the aim to incur externalities

Table 2.2: Types of	relationships in relation	n to commitment and time	e horizon in flood risk managemen

Source: adapted from Rössl (2005a): 84

Throughout, co-operation should be seen as 'strategic alliances' between two or more agents. How the relationship is designed depends on the arrangement between the agents (formal, informal or both) (Rössl, 2005b; Wirth et al., 2008). The arrangement includes the question of sharing responsibility and power between different actors (Plummer, 2009; McCay, 1996). Moreover, co-operation process is 'not just about managing resources but is also fundamentally about managing relationships' (Graham and Ernstson, 2012: 3). In summary, a co-operation approach refers to the aspect of sharing power and responsibility between the different actors based on formal rules and structures (Sandström, 2009).

# 2.4.2. Engagement and interaction in between the new scalar arrangements

The engagement between different members in a partnership strongly depends on the distance (proximity) between the actors (Lundquist and Trippl, 2013). The concept of proximity refers 'to the degree of closeness of actors' (Huber, 2012: 1171) and stakeholders. This is well known in the regional economy literature (Zeller, 2004; Boschma, 2005; Moodysson and Jonsson, 2007; Lundquist and Trippl, 2013). A partnership approach is mainly is based on interaction between the different actors and stakeholders. The interaction is strongly influenced and defined by the institutional framework and the relationships of exchange between the different actors and stakeholders under competitive or co-operative environment (Torre and Gilly, 2000). In particular, flood risk management bases on a mix of technology and funding strategies, knowledge, spatial and land use management, qualifications and skills need by different organizations involved in the management system. This interplay and interaction is supported by proximity (Asheim, 2012). Further, proximity is a useful framework to analyse and to discuss differences between different modes of partnership examples and their interaction (Lundquist and Trippl, 2013) as well as to include externalities into the research (Massard and Mehier, 2009). In the literature, the term proximity is understood as 'beneficial for the transfer of knowledge' (Moodysson and Jonsson, 2007: 117). In analysing the proximity in flood risk management (see section 7.1), this research focuses on following categories: spatial, institutional, social, technology and relational proximity. These five proximities are adopted to examine the partnership approach in flood risk management, such as inter-local co-operation in Austria and national-local co-operation in England. The five proximities are highly relevant for flood risk management, because they incorporate structural and nonstructural measures such as spatial planning, emergency management, risk communication, but also the aspect of stakeholder engagement in the decision-making practices. These are used to examine and to analyse the current partnership approaches and relationships in flood risk management. The aim is to provide a conceptual framework to analyse the dynamics in spatial development and policy. In particular, proximity allows this thesis to analyse the interaction between the different actors and stakeholders involved in decision-making process.

Torre and Rallet (2005) defined proximity 'not only means being near him/her, but also means having a strong complicity within a person who is geographically distant, whatever that person belongs to the same circle of friends, family, or even to the same network' (ibid: 48). According to the authors, the inter-regional relationships have a less important role in the development of inter-regional co-operations compared to the intra-regional contacts and networks. The literature distinguishes between various types of proximity. Moreover, the term proximity includes more than just a geographical concept (Boschma, 2005). Torre and Gilly (2000) and Torre and Rallet (2005) distinguished between geographical and organisational proximity. The papers defined geographical proximity as the physical distance between the members of the co-operation, e.g. (1) naturally (in terms of km) and (2) based on individual judgement (in terms of individual perception and background). Organisational proximity is defined as the interaction and co-operation between the different members in the co-operation (Torre and Gilly, 2000; Torre and Rallet, 2005). It includes aspects of personal interactions and similarities between the different members, like sharing experiences, language, knowledge, and representatives (ibid.). However, Moodysson and Jonsson (2007) defined these arguments as 'vague and loosely defined' (ibid: 118).

Important contributions were also delivered by Boschma (2005) and Zeller (2004). Boschma (2005) divided the aspect of proximity into cognitive, organisational, social, institutional, and geographical proximity. Boschma's concept is closely related to the work of Torre and Gilly (2000), especially to the aspects of geographical and organisational proximity. However, Boschma (2005) gave a more in-depth analytical structure, because he introduced institutional frameworks and social relationships in the research framework. On the other hand, Zeller (2004) used the categories spatial, institutional, cultural, relational, technological, and virtual

proximity. In contrast to Torre and Gilly (2000), he focussed on the aspects of social capital and institution in the proximity discussion, to provide a more critical politico-economic framework to analyse the question of power and justice. Additionally, a fourth contribution was provided by Moodysson and Jonsson (2007). The authors divided the term proximity into (1) functional proximity (relative distance) and relational proximity (regulative, normative and cognitive aspects). Their research strongly focuses on the aspect of institutions. Furthermore, functional proximity refers more to the aspect to arrange meetings (face-to-face or virtual) during the working time instead of geographical distance. However, Boschma (2005) described the relationship between the various types of proximity as very complex.

## 2.5.Key findings

The consequences and impacts of new scalar re-arrangements have series of consequences in the political debate, e.g. needs of new structures, organisations and institutions to adapt to the new circumstances. This comprises new governance practices and includes new regulations and administration practices (Gualini, 2006; Moss and Newig, 2010; Gibbs and Jonas, 2000; 2001; Perkmann, 2007; see empirical results in chapters 6 and 7). Consequently, a result can be 'crises in the territorial regulation of development processes' (Thiel, 2009: 226). One key outcome is the re-design of the relationship between the different actors and stakeholders. Partnership is seen as one form of interaction between state and non-state actors in policy decision-practices (McCay, 1996). The co-operation process seeks to engage a large number of different stakeholders in the policy discussion at same scale (van Huijstee et al., 2007). The key aim is in consensus building between the different actors, stakeholders and citizens with defined responsibilities, power, rights and functions (Berkes, 2010; Margerum, 2002; 2008; Borrini-Feyerdon, 1996). Furthermore, Borrini-Feyerdon (1996) distinguished between a 'light' and a 'strong' version of collaboration. The 'light' version includes mainly the stakeholder consultation process. The key point is seeking a consensus regarding the problem definition and enabling problem solving. The 'strong' version goes a step further and includes aspect of devolution processes. Here, co-operation processes encourage local authorities and stakeholders to take over certain tasks and duties from national or regional stakeholders (Berkes, 2010).

The key conclusions from this chapter are:

- Scales are never fixed, because they are a product of social processes under a political strategy and ideology. Rescaling processes included three main directions: upwards, such as European level; downwards, such as new localism or regionalism; or sidewards, such as privatisation and outsourcing public tasks to private actors, stakeholders and citizens.
- Scalar re-arrangement cause changes in the actual formal and informal codes and norms as well as changes in actual administration practices, which also include that actors and stakeholders can change their position in the decision-making practises (change of power relationships) to ensure their interests (scale jumping).
- New scalar re-arrangements have a strong influence in the change of level of activity and self-realisation between the different key actors and stakeholders in the policy decision-making practices.
- Governance often refers to non-hierarchical, flexible structures, that co-ordinate state and non-state actors, where the term plays a strong role in the new 'modernisationreforming' discussion.
- Scalar re-arrangements have an influence on the definition of the duties and responsibilities between the different political units as well as between state and nonstate. This had a central influence in the governance arrangements and structures. The horizontal level refers mainly to the out-scaling of duties and responsibilities, away from the public administration level towards private actors, stakeholders and citizens.

- Governance often refers to non-hierarchical, flexible structures to co-ordinate state and non-state actors and to define certain policy goals and measures. It gives a greater focus on non-state actors in the policy discussion. The term recognises the increasing influence of non-state actors in the current policy discussion (Stoker, 1998; Pierre, 2005). The focus is on the changing role of the state, away from the central planning and regulating organisation to a more a decentralised self-organised organism, which is quite flexible to adapt fast new developments and circumstances. A key role of the governance process is to ensure and enable different actors to co-operate. Important issues are the different relationships (horizontal, vertical, diagonal) between the different agents (Steurer et al., 2010).
- An important aspect in the development of these new governance styles and institutions are the social questions of democratic legitimacy, social equity, accountability, equilibrium in the decision process-less top-down approach, fair economic distribution and optimal allocation process of public tasks (Benson and Jordan, 2010; Gibbs and Jonas, 2000).

This chapter focussed on the politics of scale discussion in the literature. Examples of the analysis are presented in the chapters 6 and 7. A central argument is that scale arrangement hugely influences the governance structure and processes in society (Thiel and Egerton, 2011). The aim is firstly to use the framework to analyse and categories the relationship between the different actors and stakeholders in Austria and England and to explain the differences and commonalities between the different selected study sites to provide a more holistic understanding of the interplay between different local and regional actors and stakeholders in inter-local flood risk management co-operations. The next chapter provides an outline of the used research methods in this thesis.

# 3. Research methods

# 3.1.Methods overview

The central aspects of this thesis is to investigate and to explore the influence of new scales (inter-local co-operation or local level) on the interaction and performance of the different actors and their networks; particularly with regard to power, processes and scaled networks (Neumann, 2009). As discussed in chapter 2, scale is seen as a spatial-political construct, which is embedded in the socio-political and historically circumstances (Jones, 1998). Main factors are politico-economic and socio-psychological developments. This epistemological view, the concept of scale has to include socio-historical developments (Brenner, 1999).

The structure of this thesis (Figure 3.2) is using a heuristic – circular perspective with a focus on multi-methodology, based on primary and secondary data collection, multiple case studies as well as multi-analytical methods (Witt, 2001; Novy, 2002).



#### Figure 3.2: Iterative research design

Source: own development

Further, the research strategy distinguishes between three hierarchical levels (table 3.3):

- operational level (micro level),
- collective choice (meso level) and
- constitutional level (macro level) (Gibson et al., 2000; Turnpenny et al., 2008).

The micro-level refers to individual agents involved in the policy discussion. The research analysis is oriented towards the aspect of resources (human, time and financial). In general, the decision (choice) process is directed towards the input and the output of used resources and decisions made by similar organisations (Ciriacy-Wantrup, 1971). 'Operational choices are made by citizens and by public officials carrying out the policies made by diverse collectives' (Evans et al., 2002: 143). The meso-level covers the organisational structure (North, 1990). The policy assessment includes the aspect of organisational norms and cultures, formal and

informal decision rules, leadership and co-operation as well as co-ordination of the involved agents. The collective choice level selects and authorises individuals and organisations to define collective choices, i.e. new policy strategies and guidelines. The purpose is to define a 'general' approach and control instrument (e.g. economic efficiency and effectiveness, social justice and equity) for operational implementation of policy implementation processes (Ciriacy-Wantrup, 1971). The macro-level refers to the legal framework. Here, choices and policy discussion are related to 'the authority of collective actors' (Paavola, 2005: 12). Constitutional rules or policy levels are set up for a long term, so they change relatively slowly. Following Turnpenny et al. (2008), the data collection and analysis is based on three policy levels (micro, meso and macro). This has the advantage of having a holistic view of the social, political and economic relationship in society (Brenner, 2004).

Spatial level	Macro level	Meso level	Micro level
EU	EU directives and regulation, e.g. EU-Flood Directive, WFD, Soil protection directive, Rural development program etc.	Policy making by European Member States, stakeholder engagement	Managing and supervising projects funded by international agencies.
National	Implementation of EU directives and regulation, e.g. national Flood and Water Law	Policy making by national legislatures, executives, courts, stakeholder engagement	Managing and supervising projects funded by national agencies.
Regional	Federal law and norms, e.g. regional flood action plans, inter-regional planning	Policy making by regional legislatures, executives, courts, stakeholder engagement	Flood protection, providing services, monitoring, sanctioning, land use management, managing and supervising projects funded by regional agencies.
Local	Norms and plans, e.g. emergency planning, spatial and land use planning	Policy making by local legislatures, stakeholder engagement	Flood protection, providing services, monitoring, sanctioning, land use management, funded by local agencies
Used method	Policy analysis; stakeholder and institutional mapping; in-depth interviews	Policy analysis; stakeholder and institutional mapping; in- depth interviews	In-depth interviews

Table 3.3: Policy decision levels and geographic domains in flood risk management

Source: adapted from Gibson et al. (2000)

#### **3.2.Thesis philosophy**

In this thesis, I use the concept of 'interpretative social research' (*interpretative Sozialforschung*) to overcome the dichotomy between objectivism and subjectivism (Novy, 2002). The main idea is that each individual interview is different, but I have to take external (exogenous) as well as internal (endogenous) developments and factors into account for analysing the results, such as policy papers, administration practices, formal and informal rules and regulations. However, the research framework seeks to combine historical factors within politico-economic developments (dual analysis of the role of actors in flood risk management). The view on socio-political and historical developments allows the use of a qualitative-interpretative approach. The key aim is to understand and to analyse the structure and the contexts where the actors are embedded. Subsequently, this thesis adopts abductive reasoning to collect and to interpret the data (Arrighi and Ferrario, 2008).

In this PhD, rescaling is seen as how different actors, stakeholders and citizens get involved in the policy decision-making process; how these different groups see and define the different opportunities in the current policy debate, the way of debating, barging and negotiation between the different actors, stakeholders and citizens with same or different scale backgrounds. The aim is to analyse the social-spatial practices (where the social relationships and interactions are predominant) and representation of space (such as knowledge and power) under new scalar arrangements (Lefebvre, 2007). The use of a qualitative research design permits a better understanding of the complex research environment, especially to get access to additional background information for clarification of certain developments and decisions. In particular, Macpherson et al. (2000) underline the advantages of a qualitative research methodology. The authors state that 'qualitative research seeks to interpret social phenomena, to produce a rich understanding of the complex meaning of structures that social actors construct in their specific social environments' (ibid: 50). Also Smith (2001b) has underlined the advantages of qualitative methods, because they are 'a useful way of proceeding when we are interested in a multiplicity of meaning, representations and practices' (ibid: 24). A qualitative approach, however also have limitations. In particular, the qualitative research method does not allow large and randomised samples. This causes difficulties in finding generalised conclusions. Furthermore, the relationship between interviewer and interviewee has to be free of value to secure objectivity (Tashohkori and Teddlie, 1998).

The research method applied in this thesis is centrally focussed on a qualitative research design. The aim is to adopt different theoretical concepts and methods, such as policy analysis, stakeholder and institutional mapping as well as semi-structured in-depth interviews (see next sections below). The different techniques were used as complimentary observations, and each technique was based on the previous methods and techniques.

This thesis uses a comparative approach to analyse the impact of new scalar arrangements and their influence to the politics on flood risk management. A comparative approach allows us to get a new understanding of national policies in a specific field (Chappell, 2010). It provides the 'opportunity to develop a deeper understanding of specific areas' (ibid.: 183). Further, the advantage of the use of a comparative technique is to control and to explain certain developments in the national policies, such as the political processes in flood risk management, and to overcome 'simple' explanations (Sartori, 1970; 1991). The idea behind is 'to identify the different combination of causally relevant conditions linked on an outcome' (Ragin and Strad, 2008: 431), in my case to understand the impact of new scalar arrangements on different actors and stakeholders in the decision-making process. Key advantages to using a comparative research are:

- understand better the roles and interactions between different actors and stakeholders;
- understand the influence of different institutional frameworks to the engagement of local actors and stakeholders in flood risk management; and

55

 understand the relationship between different actors and stakeholders in the decisionmaking process, which is a crucial also in terms of opportunities in the political engagement.

The key objective in this thesis is to generate or to re-define theoretical concepts and discussions and not to test theoretical concepts and ideas (Mabry, 2009). Therefore, this thesis used an interpretative analytical technique to analyse the ex-ante and ex-post policy decisions and directions in flood risk managements in Austria and England (Dunn, 1994). In summary, this thesis uses two main data collection techniques:

- primary data sources (semi structured in-depth interviews) and
- secondary (grey) data sources (Ingstrup and Damgaard, 2013).

The main reason for this approach is to analyse the fluid change in scalar arrangements in flood risk management policies. The interconnections between the different methodological techniques are illustrated in table 3.4. The table highlights the different sampling techniques, data collection and analysing strategy. In general, each different method is based on the results of the previous research method, e.g. policy analysis as well as partly stakeholder and institutional mapping defined the sampling selection of semi-structure in-depth interview.

#### Table 3.4: Research structure

Method	Policy	Data collection	Sampling design	Analysis	Interview
	decision level			-	design
Policy analysis	Macro and meso level	-academic papers -policy documents -legal acts	Analyse all relevant documentations	-critical social research (discourse analysis)	none
Stakeholder and institutional mapping	Macro and meso level	-academic papers -policy documents -legal acts -semi-structured interviews	multilevel sampling design based on : -identification from policy documents and legal acts -snowballing <sup>a</sup>	-thematic coding -critical social research	-open questioning -thematic problem-lead questioning
In-depth interviews	Meso and micro level	-expert targeting semi-structure interviews based on the outcome of policy analysis and stakeholder and institutional mapping	-multi-level sampling design -snowballing <sup>a</sup>	-thematic coding	-open questioning -thematic problem-lead questioning

a: snowball technique were mainly used to select case studies from recommendations from researchers and national authorities.

Source: own development

## **3.3.Qualitative research method**

#### 3.3.1. Policy analysis

The analysis of policy briefs, guidance and strategy documents are in this analysis central to understand the changes in past years and decades. In particular, policy documents provide formal rules in flood risk management policy to show the framework where the different actors and stakeholders can act (Hodgson, 2006). Within the policy documents and legal framework, views about who the main actors and stakeholders can be analysed. However, policy papers and documents significantly change over time. The method<sup>15</sup> of policy analysis aims to provide a critical evaluation and reflection of the current political ideology and direction (Dunn, 1994). The data collection provides an useful understanding of the definition of the general framework of the role of actors and stakeholders (ideal situation), which is socially and practically significant (Wildavsky, 1969). This means that it is more than the collection of random data (ibid.). The analysed technique is focussed on the critical social

<sup>&</sup>lt;sup>15</sup> Following Downward and Mearman (2007), this thesis distinguishes between methodology and method. Method is the general technique in data collection, coding and analysing; it includes also the use of different techniques. On the other hand, methodology includes a three-stages proceed: (1) combination of different methods, (2) implementation at practical level and (3) assessment of the implementation process.

research (Harvey, 1990; Comstock, 1982). Critical social research allows for a conceptual framework within social relations and structures, such as the question of power (Harvey, 1990; van Dijk, 2001). Comstock (1982) highlighted that 'the function of a critical social science is to increase the awareness of social actors of the contradictory conditions of action which are distorted or hidden everyday understandings' (ibid: 371). Following this understanding, the data were analysed in terms of social interactions between actors, stakeholders and citizens in a given field of discussion, which is defined by the scalar arrangements. As a first step, all relevant data like research papers, consultant reports, reports, statement and acts from public authorities, press articles and online websites, e.g. http://www.hws-aist.at, were gathered. The critical social research permits us to analyse and to assess the consequences and impacts of scalar re-arrangements – which are strongly based on national political ideologies or the governance structure and interactions.

#### 3.3.2. Stakeholder mapping

Stakeholder mapping (see also section 5.2.1, 5.2.5, 7.3.4 and 7.3.5) is an important tool in the policy research study, such as to analyse the legal framework and key stakeholders in flood risk management (Aligica, 2006). Morgan and Taschereau (1996) highlighted the importance of stakeholder analysis in policy research. Stakeholder mapping 'is an analytical approach for assessing the potential support or position to an issue among interested parties, including administrators, advisors and local champions. It is a basic first step in capacity assessment' (ibid: 3); by analysing the various stakeholders in the policy field, the conditions of relationship between the different agents at different scale, are highlighted (ibid.). A key objective is to understand the role and interests of the involved actors and stakeholders in the decision-making practices. A second point is to understand the legal limitations and responsibility of each actor and stakeholder to understand what is the role of each actor, what they do and do not. This is essential, because rules (both formal and informal) have a strong influence on the social structures (Hodgson, 2006).

This thesis used the Mendelow-Matrix (Posthumus and Morris, 2007) to identify and to analyse the key stakeholders and their influence in the politics of flood risk management in both countries. The Mendelow-Matrix (mapping technique to demonstrate the role of each stakeholder in policy process) provides a framework tool to classify stakeholders' position in the policy decision process. The aim is to evaluate of stakeholders influence in the ongoing policy decision-making practices. Key variables are the power to influence the policy as well as the interests to engage in the policy debate (Mendelow, 1981). Mendelow (1981) understood the use of power depending on dynamism of the stakeholders. High dynamic stakeholders are more likely to use their power position to influence the policy decision-making practices. This is necessary to identify the main actors and stakeholders involved in the discussion, their role and their responsibilities (Green, 2011). The stakeholder and responsibility mapping help us to identify who are the main actors and stakeholders in the policy discussion process, policymaking structures and decision-making practices. They permit on understanding of which interests and power they follow. Furthermore, stakeholder mapping also allows us to recognise responsibilities and accountabilities. Consequently, these institutional and organisational settings influence the social structures and provide the framework of interactions and influences of the different stakeholders and actors in the decision-making practice (Hodgson, 2006). This method is a key step in the understanding and analysis of governance (Nelissen, 2002; Mees et al., 2014; Driessen et al., 2012). An adavantage of the Mendelow-Matrix, in compare to other social network methods, is the possibility to include the dynamic developments in the ongoing policy discours. Further, the approach focuses on the assessment of the stakeholders' power within the policy decision-making process (Olander and Landin, 2005). Another advantage is the visual presentation to analyse similar patterns of different stakeholders at the same position, such as their willingness to participate and their potential influence (Romanelli et al., 2011; Reed et al., 2009).

59

#### 3.3.3. Semi structured in depth-interviews

The aim of semi structured in-depth interviews is to understand better the current policy documents, especially background information and how they are adopted in practice: both factors are not available from the secondary data sources. 'Typically qualitative methods yield large volumes of exceedingly rich data obtained from a limited number of individuals and whereas the quantitative approach necessitates standardised data collection, qualitative researchers exploit the context of data gathering to enhance the value of the data' (Walker, 1985: 3). The advantages of semi-structured interviews is the flexibility in the data collection, such as the use of 'following-up' questions for further explanations and explorations which helps to understand current policy developments (Mabry, 2009). In addition, qualitative interviews can overcome the reduction of positivism and social constructionism in research as well as to reduce human interactions as numerical references (Marvasti, 2007). Following Marvasti (2007) 'qualitative research provides detailed description and analysis of the quality, or the substance, of the human experience' (ibid: 7). Disadvantages of quantitative research are the pre-defined limited scope in comparison to qualitative research methods, where the aim is to analyse complex themes (Doorewaard, 2010).

However, there are limitations. There is no neutral relationship between both groups, because researchers are not free of value, background knowledge, experiences and opinions (Novy, 2002). In fact, the 'interviews are not neutral tools of data gathering but active interactions between to (or more) people leading to negotiated, contextually based results' (Fontana and Frey, 2000: 62). Additionally, each interview is unique due to the interaction between interviewer and interviewee as well as due to social constructions (Fontana and Frey, 2000). These different factors have an influence on the data gathering as well as in the interpretation of the results. Throughout, additional limitations which need to be tackled include the simplification of the interpretation of the results, because comparison between the different

interviews is not always possible, because of different background of the case studies or different level of knowledge and experiences of the experts.

Based on the previous steps (see also sections 3.3.1 and 3.3.2), an interview guideline document was designed, also to increase the reliability of the research method (Mason, 1996). The interview themes were structured by the conceptual framework. Subsequent to Flick (1998), the process of data collection mainly included three key steps:

- recording of semi-structured interviewees and adding additional notes during and after the interview,
- transcribing of the recorded data and
- categorising and coding the text.

The interviews were conducted face-to-face or by phone, in general in the office of the interviewee, and took between 60 – 90 minutes. Furthermore, additional notes were taken during and after the interview. Towards the end of the interview, interviewees were encouraged to reflect on the dialogue and contribute additional information that they deemed to be relevant (Graham and Ernstson, 2012).

In agreement with Jackson and Trochin (2002), the open-ended questions encourage the provision of further information to explain the policy decision-making practices. The interview guideline document was also kept very flexible to be adapted to the interview partner and to the uniqueness of the case study. Nevertheless, the interview guidance follows the following headings:

- preconditions of partnership approach,
- development stage of partnership approach,
- role of different actors, interaction between different actors, and

legal status of partnership.

The sampling (expert targeting) was concentrated on the results from the policy analysis and stakeholder and institutional mappings. A multilevel sampling design was used throughout this thesis (Onwuegbuzie and Leech, 2007): two or more subgroups were interviewed to understand and to analyse the knowledge and understanding of national, regional and local actors and stakeholders regarding new flood risk policy. Furthermore, the three levels show a hierarchical relationship. The sampling schemes and size between the three levels are not uniform, because lower-levels (such as local) represent a larger number of involved actors and stakeholders (Patton, 1990). This was necessary to interview actors and stakeholders who were involved in the policy definition, implementation and critical reviewing (such as academic or consultant) process. Criterion sampling draws attention 'to understand cases that are likely to be information-rich because they may reveal major system weaknesses that become targets of opportunity for program or system improvement' (ibid: 176+177).

This research study uses a judgemental or purposive sampling design. For the investigations of the network governance structures and processes, this research study primarily focussed on the involved key decision-makers (national, regional and local level) from the case studies (Somerville and Haines, 2008; Kramer and Revilla Diez, 2012). Selected interviewees were found through networks, recommendations from other interviewees and academics and especially from newspaper articles, academic journals and internet websites. They comprised a mix of academic, public authorities and relevant stakeholders. Selected interviewees (total n=49)<sup>16</sup> were selected from national authorities (n=10), regional authorities (n=11), local authorities (n=15), private actors (n=6) as well as from university and academic research institutes (n=7).

<sup>&</sup>lt;sup>16</sup> Details of interviews see appendix 1.

## **3.4.Thematic analysis**

The collected data were analysed using thematic analysis. A key objective of this thesis is to explore an understanding of scalar re-arrangements with the help of case studies. The use of case studies favours the application of a thematic analysis (Mabry, 2009). Mabry (2009) described thematic analysis as 'the identification of emerging patterns and categories from iterative reviews of the dataset, a process which marshals evidence for developing and warranting findings (ibid: 218).

Thematic analysis helps to identify emergent themes from the discussion, based on the literature review. The aim is to find similar patterns in the interviews to increase the robustness of the data (Fereday and Muir-Cochrane, 2006) and to provide normative conceptual models to explain relationships between the dependent and independent variables (Doorewaard, 2010; Marvasti, 2004).

For analysing the research results, this study uses a range of different analytical methods to interpret the data (Ryan and Bernard, 2005). Based on the research design, this thesis adopted a four step approach for analysing the data:

- preliminary model building
- model testing
- data collection
- open-ended interviews (ibid.).

As a first step pre-defined matrices and models were developed to answer my research questions as well as to reduce the amount of text and to organise the first results. This approach allowed a better comparison of the differences and similarities between the different case studies (Jones and McBeth, 2010). However, the disadvantage here is the 'research-driver-classification' restriction in the analysis of the data (Jackson and Trochin, 2002). Second step, a key-words-in-context (KWIC) method was performed to analyse key words in the transcribed text (Kramer and Revilla Diez, 2012), as a method for analysing freeflowing texts from the open-ended questions and to reduce the restriction from the 'researchdriver-classification' (Ryan and Bernard, 2005). The KWIC involves highlighting key words or phrases to analyse their context (ibid.). Third step, a time-series analyses was conducted to understand and to analyse the developments in policy process over a defined time period (discussion, decision and implementation) to study the dynamic developments over time (Boschma and Fornahl, 2011; Yin, 1994). The advantage in using the different techniques tested in the understanding of the integration processes of co-governance approach in the different case studies. An important aspect was the categorisation of the data. Based on the circular research design (figure 3.2), this research study conceptualised ideal models, to analyse the partnership relationship (horizontal and vertical) in flood risk management (see chapters 6 and 7). These models are motivated by work done by Lundquist and Trippl (2013), Plummer (2009) and Ingstrup and Damgaard (2013). Nevertheless, the analysis of the results is based on subjective knowledge with the key aim to understand the structure and context of scale construction and new state spaces (Brenner, 2004) as well as process and performance of the actors (flood risk governance) in the new state spaces.

#### **3.5.Triangulation research strategy**

The use of triangulation is an important objective in this research study. Triangulation is a technique to use different methods, theories or data to analyse a specific research problem. Triangulation is widely accepted in the social science, especially in studying policy interventions or evaluation studies (Blaikie, 1991). The aim is to overcome one-side positions of view in a research (Downward and Mearman, 2007). The main motivation is to test, compare, confirm and control the collected data and the interpreted results from different perspectives with the aim not to simplify them to a single explanation. Underlying these academic positions, 'triangulation implies combining more than one set of insights in an

investigation' (ibid: 80). Overall, the thesis uses three of four main forms of triangulations, namely (1) data source triangulation, (2) method triangulation and (3) theoretical triangulation (table 3.5).

Taxonomy of triangulation	Short description	Thesis
Data source triangulation	Gathering data from different	Data collection from: policy
	sources, times, location and	documents, briefs, legal acts and
	community	semi-structure interviews with
		experts from national, regional and
		local level
Method triangulation	Use of different research methods	Use of three main qualitative
	for data collection, integration and	methods: policy analysis,
	analysis	stakeholder and institutional
		mapping and semi-structured in-
		depth interviews
Theoretical triangulation	Use of more than one theoretical	Combing theoretical concepts from
	perspectives to analyse the research	political geography (section 2.1),
	questions	public administration (section 2.2)
		and economic geography (section
		2.4)
Investigator triangulation	Research team – more than one	None
	researcher dealing with the same	
	research problem	

#### Table 3.5: Taxonomy of triangulation

Source: adapted from Downward and Mearman (2007) and Mabry (2009)

This thesis is using different data sources from written documents and oral interviews. Although the method of triangulation is mainly understood as the use of quantitative and qualitative research methods (Fiedling and Fiedling, 2009), this thesis integrated different qualitative methods to examine the changes in scalar construction in flood risk management. This thesis combines different research methods based on gathering different data sources; not only for data collection, but also in analysing the data. Relevant actors and stakeholders from different positions (national, regional and local) and places (different study sides) were selected for interviews to assess the consequences and impacts of new policy interventions (scalar re-arrangements). Furthermore, this thesis uses more than one theoretical tradition to understand and to analyse the data to support to answear the four research questions (see section 2.5). As previously mentioned, the key theoretical strengths in this thesis are the combination of theoretical concepts used in political, geography and economic sciences. This allows using different views from different research disciplines. A key reason for this is to not limit my view of the data to enhance the validity of the data and results (Mabry, 2009). The key advantage is to understand the research problem from different lenses (Downward and Mearman, 2007). In particular, to close the research gaps in each theory, such as lack of analysing partnership relationships in flood risk management (economic geography) or to understand the interaction of scalar arrangements (politics of scale) and their impact to politics of flood risk management.

In this line, an important aspect is the reliability and validity of the data and results. For ensuring the validity and reliability of the data and results this thesis used four main strategies:

- use of multiple methods,
- interview quotations,
- revisits and verification by interviewees and
- use of theoretical concepts to support the findings (Baxter and Eyles, 1997).

First, each of the semi-structured interviews was recorded by an audio recorder. Furthermore, I took notes during and after the interviews, which were used for the analysis. Each interview tape was fully transcribed into text. I transcribed the Austrian interviews and the English interviews were transferred by an external person. No translator was used for the Austrian or English interviews. However, each externally transcribed transcription was carefully checked for incompleteness and false transcription to avoid misunderstanding and misleading in the interviewee references and quotes were coded in the format of 'i1', 'i2' etc. (Perkmann, 2007). For each interviewee reference I added the acronym of selected countries (eng for England and aut for Austria, for example i1aut). This is a central aspect to conceptualise evidence in the results.

Influentially, with all research a central problem refers to the reliability of the data (Mason, 1996). The transcriptions were returned and discussed with responders to clarify and to verify meanings of interview and interpretations (Baxter and Eyles, 1997). In addition, the thesis used theoretical concepts to support the findings from the empirical results and also to demonstrate rigorous results (Fereday and Muir-Cochrane, 2006). Finally, the thesis used a triangulated strategy for validating the results. Nevertheless, triangulation is no guarantee for full validity, but strongly increases the validity of the research (Fiedling and Fiedling, 2009).

Another critical element refers to my positionality in this research study. My positionality informs the reader of this thesis my relationship to the selected case studies and the influence in analysing the data (Butler, 2007). An insider and outsider position in a research is a critical aspect, in particular to avoid misinterpretation of data (Merton, 1972; Kerstetter, 2012). In the literature, the insider and outsider problem was strongly discussed in the community-based research, such as by Ryan et al. (2011) or by Kerstetter (2012). 'Outsider' is understood as a non-member of a 'group', on the other hand and the 'insider' takes part of the 'group'. The term "group" is here understood as the selected countries. Insiders share the advantages to have privilege access and knowledge, which can enrich the study with knowledge and data. Nevertheless, the insider researcher may be compromised by a stronger subjective perspective on data. Kerstetter (2012) contests that the major challenges and barriers are to distinguish between personal experiences and research work. On the other hand, an outsider acts in a less 'personal' (or emotional) way during the research process as an insider.

In international research studies a central challenge is to have simultaneously an insider and outsider position; to have an insider perspective in one case study and an outsider position in another case study. This would be misleading the interpretation of the data from one case study in comparison to the other. With regard to the in-depth research conducted in Austria and England, I had the same 'cultural outsider' position to analyse the current flood risk management policy. The advantage was to have the independent position in both case studies for analysing the current policy developments and shifts and the impact (Ley and Mountz, 2001).

## **3.6.Key findings**

In this chapter I showed my methodological approach to undertake this thesis. The structure of this thesis is using a multi-methodology, based on primary and secondary data collection, multiple case studies as well as multi-analytical methods (Witt, 2001; Novy, 2002). The main aim is to incorporate socio-political and historical developments to understand and to analyse the structure and the contexts where actors are embedded, as to interpret and to explain the differences and commonalities between the different selected study sites to provide a more holistic understanding of the interplay between different local, regional and national actors and stakeholders in flood risk management policy. The research method applied in this thesis is centrally focused on a qualitative research design, because there exists a lack of sufficient information and data from the secondary data sources, so a qualitative methodology is a useful instrument to collect new data (Walker 1985). The aim is to adopt different theoretical concepts and methods, such as the policy analysis of flood risk management documentation, stakeholder and institutional mapping as well as semi-structured in-depth interviews. Therefore, this thesis used an interpretative analytical technique to investigate the ex-ante and ex-posit policy decisions and directions in flood risk managements in Austria and England.

## 4. Case study approach

The goal of this thesis is to understand and to assess the governance structures in flood risk management, especially in relate to role of local communities in flood risk policy. A case study approach permits exploratory in-depth research which focuses on achieving an understanding of the problem (Eisenhardt, 1989; Yin, 1994; de Leeuw et al., 2012). The thesis uses case studies to provide an exploratory research approach (Yin, 1994; Eisenhardt, 1989) to understand and to evaluate new politics of scale strategy and the impact towards governance structures. In particular, strategies and initiatives for developing partnership approaches in flood risk management will be identified focusing on different temporal and spatial scales. The niche developments will be assessed and new societal contracts in flood risk governance as well as their limitations and barriers in the implementation process will be studied. The use of case studies allows this thesis to investigate how flood risk management policy functions in practices, in particular to explore the consequences and impacts of the changes in the scalar re-arrangements. This thesis used case study research as the main strategy in collecting data and interpretating the results. Case studies are a suitable instrument and technique to generate or to re-define theoretical concepts and discussions (Marby, 2009). The key advantages in using case studies are to clarify or re-frame theoretical concepts and ideas, because of the possibility to test general theoretical concepts (Flyvbjerg, 2006). Furthermore, this technique is suitable in generating new knowledge based on the intensive in-depth analysis of a research problem (ibid.). In order to generalise the results from the case studies, this thesis also uses normative (conceptual) models to interpret the data (see chapter 6 and 7). Influentially the empirical results are used to support and clarify as well as specify the normative choices associated with the different political, geographical and economic theories as well as to provide recommendations for policy actors and stakeholders (Keessen et al., 2013).

The research is concentrated on a complex issue with the aim to 'extend experience or add strength to what is already known through previous research' (Dooley, 2002: 335). Kumar (2008) identified the advantages of using case studies for 'the investigations of a new area, in which there is little information available and thus case studies provide a rich source of ideas' (ibid: 49). The key principle is that the research interests on a specific problem and development and the 'wishes to understand it completely' (Dooley, 2002: 337). Yin (1994) described the goal of using a case study based on qualitative research method to provide a contribution to the theoretical concepts and ideas. The main aim is not to provide universal solutions. Furthermore, the use of multiple case studies can help to reflect more critically on the results (Blatchford, 2005) and especially to overcome the positivist paradigm in the research study (Macpherson et al., 2000). The authors 'advocate case study research that not only produces rich understanding of social sites and the meaning structures created by the actors who operate there. We aim to use this understanding to inform actions that can effect changes to social practices and public policy' (ibid: 54).

However, a case study approach includes limitations, such as the simplification of the research environment. Verschuren (2003) defined the case study research methodology as: 'a research strategy that can be qualified as holistic in nature, following an iterative-parallel way of preceding, looking at only a few strategically selected cases of sub-cases, observed in their natural context in an open-ended way, explicitly avoiding (all variants of) tunnel vision, making use of analytical comparison of cases or sub-cases, and aimed at description and explanation of complex and entangled group attributes, patterns, structures and processes' (ibid: 137). This limitations can be reduced though well-defined (see next section) site selections and drawing also on broader number of examples. This thesis selected six different study sites in two European countries to analyse, to assess and to understand the impacts and consequences of scalar re-arrangements in flood risk management.

70

Following Yin (1994), the selection of case studies should ensure:

- literal replication (similarities between them) and
- theoretical replication (differences between them).

The advantages in using more than a single case study approach is to encourage the empirical investigations and furthermore to develop new aspects and views in the theoretical concept of politics of scale in flood risk management. Multiple case studies provide a broad variety of results, especially to see better the differences and the patterns between the different case studies (Kramer and Revilla Diez, 2012; May, 2007) and to ensure an 'analytic generalisation' (Verschuren, 2003).

#### 4.1.Justification for selection of case studies and study sites

This thesis selected Austria and England as its primary research focus based on various criteria (matching, qualitative assessment and cross-case test) with the aim to generate and test research objectives and questions to provide new explanations. The selection processes include the choice of one or more case studies, where one independent variable as well as one dependent variable shows similarity, where other factors include different values (Gerring, 2006). Seawright and Gerring (2008) emphasised the possibility to mix and match different strategies. Further, the selection processes includes also the recommendations from interviews with national Government and research scholars.

The changes in the politics of scale and the impact in flood risk management policy seem to lead to similar developments in other European countries, especially regarding partnership approaches between inter-local/regional or national-local organisations. For this reason the selection process of the case studies underlined a first screening in different European countries for flood management policy and policy shifts in past years based on the new politics of scale. Based on the literature, I found similar partnership approaches in flood risk management in the European countries Austria, England, Germany, Italy, the Netherlands and Scotland. The table below shows some examples in Europe.

Politics of scale	Examples
Inter-local co-operation	Wupperverband or Weisseritz Regio (Germany)
Inter-regional co-operation	Water boards (the Netherlands)
Regionalism/localism	South Tyrol (Italy), Switzerland
State-non-state co-operation	Tweed-Forum (Scotland)

Table 4.6: Logics of politics of scale construction in flood risk management

Source: own development

In the Netherlands, Scotland and Germany, the literature provides examples of inter-local partnership approaches in flood risk management (Moellenkamp et al., 2008; Mostert, 2006; Janssen, 2008; Wirth et al., 2008; Cook et al., 2013).

In these countries flood risk management has been institutionalised for decades, but with different institutional settings. As a result, multiple dimensions of vulnerability (Cannon and Müller-Mahn, 2010) as well as challenges at the science-policy intersection have been exhaustively studied. This will help to identify wider lessons for other European regions currently working on the implementation of climate change adaptation strategies arising from flood risk (Butler and Pidgeon, 2011; Hanger et al., 2013). Germany and the Netherlands were excluded, because the policy interventions refer to inter-local co-operations as consequences of scalar re-arrangements undertaken too long ago, e.g. Wupperverband established in 1933. The aim of this thesis is to understand the arguments and discussion at a political level, to analyse actors and stakeholders engagements in the policy planning and not only in the implementation processes. Scotland was excluded because the scalar re-arrangements – partnership approaches – show strong similarities and overlaps with the English flood risk management system. Cook et al. (2013) showed various examples of catchment management plans, which include the co-operation between different local authorities and other agents in a
catchment, e.g. Dee Catchment in Scotland or Tweed Catchment alongside the Scottish-English Borderlands. One objective of these co-operations is the reduction of risk in the downstream communities, for example for the town Galashields. However, also in this case the cooperation does not include an implementation of inter-communal flood defence schemes. Overall, the vertical partnership approach is found in only a few cases, because it is not on the national nor local political agendas (i30eng). The project of new partnership funding schemes shows the need of a strong relationship between the different actors, especially between the key drivers in the project for closing the financial gap for the realisation of the flood defence proposal. Finally, Italy was excluded, because of a lack on actual national policy intervention in past years as well as due to the insignificant role of local authorities in flood risk management. Further the Italian legal system includes no compensation for using farmland for flood protection (Zischg et al., 2011). In addition, the country does not show clear rules in responsibility in flood risk management or funding schemes. A first important criterion emerged from the key literature with focus of this research: aspect of co-governance in flood risk management. Key aspects were the gaps in the literature about developing partnership approaches in flood risk management.

This study aims to get a broad range of results, so it was decided to include different types of flooding (especially flash flooding, flooding in combination with debris flow and fluvial flooding), different community size and type (urban and rural locations of varying size, e.g. village to urban areas). Furthermore, a selection criterion was also the legal framework in the different European countries (e.g. Austrian Water Act from 1959 and funding levy as well as United Kingdom Flood and Water Management Act (2010) and Localism Act (2011)). The new strategy included a shift in the lead responsibility towards local organisations to determine local strategies to manage local risks. Table 4.7 summarises the main criteria for the selection of the study sites.

### Table 4.7: Justification for the selection of the study sites

Criteria	Justification		
	Austria	England	
type of flooding: different types of flooding (flash floods, fluvial flooding, debris flow),	Inclusion different type of flooding, mainly flash flooding with the combination of debris flow or fluvial flooding	Mainly fluvial and pluvial flooding	
flood frequency and recent flood history	Recent flood events: 2002 and 2005 – national impact	Recent flood events: 2000 and 2007 – national impact	
previous research results	Yes, Floodrisk I and II project	Not relevant	
different socio-economic structures	Rural vs. urban	Grassroots organisations vs. non- grassroots organisations	
different Federal States	Yes	Not relevant; no partnership funding agreement in Wales	

Source: own development

## Type of flooding and recent flood history

A first criterion refers to the recent flood frequency and history. In the past 20 years, Austria experiences of major flood events occurred in 1991, 1997, 1999, 2002, 2005 and 2009. These events included different flood types, e.g. 2002: fluvial flooding, 2005: flash flooding in combination with debris flow), and affected different areas in Austria, e.g. 2002: Upper and Lower Austria; 2005: Vorarlberg, Tyrol and Salzburg. The selected study sides include areas affected by different flood events in the past 20 years; Triesting-Tal (1991, 1997 and 2002), Aist (2002) and Ill-Walgau (1999 and 2005). The three study sites are characterised mainly by fast running torrent streams. Furthermore, the Aist and Ill-Walgau study sites are characterised by the combination of flooding and debris flow (landslide and woody debris).

The English case study relates to areas, which were affected by several flood events in the past 15 years, especially in the years 2000, 2005 and 2007. The selected study sites show a recent flood history in the past 10 years. Cockermouth, for example, was affected in 2005 and strongly in 2009 and the town Morpeth was strongly affected in 2008. Further, the study sides include also various flood events with a return-period higher than 1:100 (Kundzewicz et al.,

2013), such as Morpeth (2008), Cockermouth (2009), Aist (2002), Ill-Walgau (2005) and Triesting-Tal (1991 and 1997). Furthermore, all three selected study sites show risks from different types of flooding, often in combination. Morpeth, for example, was affected by fluvial, pluvial and drainage flooding in 2008, together with already wet soil levels with no capacity to absorb further moisture (JBA, 2011).

## Previous research study

Second characteristics were previous research studies: Triesting-Tal – Floodrisk I project (finished in 2004) and Aist – Floodrisk II project (finished in 2009). The key aim of this work is to build on these previous studies as well as to bring new perspectives and aspects in this field (see also Rouquette et al., 2011). While previous studies focussed on the pre-conditions for building an inter-local co-operation, the level of co-operation and role of local and regional actors have not yet been studied. In relation to the English case study there was little previous research conducted on the implementation of partnership funding in flood risk management in England.

### Socio-economic structures

An aspect was the different community size and types, especially to understand better the conflicts in the co-operation group (upstream/downstream conflicts or urban/rural conflicts). The three English study sites have a similar community size and type in comparison to the Austrian study sites. The selected study sides are mainly based in rural areas, having in general small to medium sized communities. Morpeth and Bridgwater have an important position for their district, because they are the administrative headquarters. The socio-economic data between the three English study sites are broadly different. The key objective for Bridgwater is the regeneration of the urban area, because of large socio-economic challenges. On the other hand the communities Morpeth and Cockermouth show classical white 'middle' class rural areas. Morpeth shows a large number of people are employed in the public sector and

Cockermouth is a well frequented tourism attraction in the Lake District (i39eng; i43eng; i35eng).

## Political administration (Federal States)

A fourth attribute was the different area/political backgrounds of the study site. This criterion is only relevant for Austria to explore the national framework in practices in the different Federal States in Austria. The Austrian flood risk management system is based on a regional level (Federal State level). There are considerable differences between the different administration units, relating to spatial and land use, emergency management, compensation funding, funding scheme, flood risk management strategies etc. In general, Austria has nine different flood risk management systems, because of the nine different Federal States. In this line, this research study chose three different areas from different Federal States to investigate the differences and similarities regarding inter-local co-operation. Final, the selection of Ill-Walgau study site is based on the type of members in the co-operation. In this study site, the co-operation process includes state and non-state actors and only Federal Water Engineering Administration (BWV) of Vorarlberg (implications in network governance see also chapter 7).

# 4.2.Selected case studies

This chapter presents the selected case studies and study sites for this research. The research selected two European countries – Austria and England – to analyse the scalar re-arrangement and their implications and consequences for national and local authorities in flood risk management.

# 4.2.1. The Austrian flood legislation framework

The Austrian legislation includes a wide range of different laws relating to flood risk management (table 4.8). In general, the Austrian legislation, based on the federal structure, distinguishes between national and regional acts. The main national acts include the Forest Act (1975) (Republik Österreich, 1975), the Water Act (1959) (Republik Österreich, 1959) and the Hydraulic Engineering Assistance Act (1985) (Republik Österreich, 1985). The Austrian flood risk management system goes back to 1884 with the first official act dealing with flood risk management. The act foresees the implementation of flood alleviation schemes in the upper part of the catchments to reduce the vulnerability in the overall catchment (Holub and Fuchs, 2009).

Table 4.8: Overview of main legislation	s affecting Austrian flood	risk management system
---	----------------------------	------------------------

Act	Short description
Water Act (1959)	General rules and framework for Austrian flood risk
	management system, such as legislations about building
	inter-local flood risk co-operations; definition of the role
	of each actor in flood risk management
Forest Act (1975)	Delimitation of hazard zones in the upper part of the
	catchment areas vulnerable to mountain hazards, such
	as mountain torrents, rock falls or snow avalanches
Hydraulic Engineering Assistance Act (1986)	Regulation about the funding scheme in the Austrian
	flood risk management

Source: adapted from Holub and Fuchs (2009)

At federal level, different acts, which set out the rules for actors, in regarding spatial and land use management exists as well as emergency management and disaster compensation (Holub and Fuchs, 2009; Kanonier, 2006). However, the different acts are not harmonised or coordinated between the different Federal States (Holub and Fuchs, 2009). This means that there are a wide range of different interpretation and legislative requirements between the different Federal States (i9aut). In spatial and land use management, for example, we observe a wide range of different legal requirements (Kanonier, 2006). Some of the Federal States follow a quantitative approach, such as land use restriction based on a quantitative restriction, e.g. in areas with a flood return period of 1:30 (Upper Austria), or use qualitative restrictions, such as no new developments in significant risk areas (e.g. Vorarlberg), which includes a wide range of interpretation (ibid.).

Further, the EU Floods Directive<sup>17</sup> influences the Austrian flood risk policy. In Austria the Government updated the existing Water Act from 1959 in 2011 with the main headings and requirements from the directive. In Austria, the key challenges and barriers have been the

<sup>&</sup>lt;sup>17</sup> More detail information about the EU-Floods Directive see appendix 2.

harmonisation of the management approach of both authorities (WLV and BWV). First, the EU-Floods Directive is focussed mainly to the lower part of the catchments, since the focus of the directive is on flood risk management instead of a multi-risk approach. This is a key issue in the upper part of the catchment. The consequence is that the WLV is less involved in the discussion and management planning. The second challenge refers to the different standard scenario. The WLV has designed their standard protection according to a 1:150 expected frequency. On the other hand, the BWV as well as the EU Floods Directive and the Austrian Water Law 1959 use the Central Europe standard probability of 1:100. This provokes conflicts between both organisations in the border area (see also section 7.1.4). Further challenges are the different funding schemes, system maintenance and the use of structural measures. However, since 2010s both organisations organised various working groups to harmonise their different management approaches. Further challenges refer to the definition of terminologies, such as risk or residual risk and their use in the decision process.

In the past decades, the Austrian flood risk management policy already has moved away from a technical management approach. For example, after the 2002 flood event the Federal State Lower Austria strongly invested in the early flood warning system, training of blue-light experts and the development of emergency management plans at local levels or the development of flood risk maps for the Federal State based on different return periods (i1aut; i8aut). Furthermore, since the late 1990s/early 2000s some Austrian Federal States move their spatial focus from local flood alleviation schemes towards a broader focus based on inter-local flood risk management activities. However, there exists various gaps between the legal – theoretical frameworks and the practical implementation. Specifically, there is no common definition of risk, and there are large differences in the flood risk management system (e.g. prioritisation of measures) between the different Federal States. Furthermore, we can find a lack of coordination and co-operation between the different federal states (i26aut). A further key challenge is the development of flood risk management plans. Until now it is unclear who is responsible for the development and the implementation of the flood risk management plans. This includes also strong challenges with the integration with other fields, e.g. the emergency management and spatial and land use management planning. Here, the Austrian policy has shown strong barriers in the collaboration in an interdisciplinary approach (i23aut; i24aut; i25aut; i28aut; i44aut). Furthermore, the role of the citizens in the management process is still unclear and undefined, especially the question how to encourage local citizens to engage in the planning process and what to do with the outcome of the stakeholder process (i2aut).

### 4.2.2. Austrian organisational level of flood risk management

In general, the WLV (Austrian Service for Torrent and Avalanche Control) and BWV (Federal Water Engineering Administration) have the responsibility for preventive flood risk management in Austria. The BWV, in contrast to the WLV, are under direct control of each Federal State, with the consequence that both organisations act under different level (WLV under national level and BWV under regional level). The local authorities have the responsibility for spatial and land use management plans as well as emergency management for local events. However, the Federal State (Department for Water Engineering Administration and Department of Spatial Planning) and the WLV have the legal right to give advice for new residential and non-residential developments. Based on the used governance definition, the figure 4.3 provides an overview of the key legislations with the definition of policy and operational responsibility in the Austrian flood risk management policy discussion.

At national level, the key actors are the WLV and BWV dealing with the problem of flood risk management. Both organisations have the responsibility to define the policy strategy for the country. Furthermore, both organisations have the operational responsibility of the flood policy as well as provide a large contribution to the funding scheme (approximately 80% of the total costs<sup>18</sup>). Regarding non-structural measures, both organisations have a limited role, e.g.

<sup>&</sup>lt;sup>18</sup> The flood risk management funding is based on national and local taxation; nationally guaranteed funding towards regional and local authorities.

development advice to local spatial and land use management plans. In respect of multi-

hazard risk management, only the WLV has the responsibility for avalanche management.



Figure 4.3: Legislation structure in flood risk management in Austria

Source: own development

## 4.2.3. Inter-communal flood risk management in Austria

The Austrian Water Law 1959 (Article 87) permits buildings a formal co-operation under public law. According to the interviewees the legal framework has simplified the co-operation process. Interviews show a high importance in the legal framework as an orientation of development of inter-local co-operations. The governance structure and model in inter-local co-operation remains highly hierarchical organised with the consequences that legislation is a significant stabilising construct in flood risk management policy. The Austrian Water Act guarantee the powers of the involved actors and stakeholders in flood risk management, such as representation, funding, type of organisation and end up the co-operation (Malova and Sivakova, 1996). For example, the Article 88b from the Austrian Water Law (1959) allows the regional authorities to enforce the co-operation between the different stakeholders at local level, e.g. communities, non-state actors. The co-operation is based as a legal co-operation under public law (Austrian Water Law 1959, Article 87). The co-operation with non-state actors (Austrian Water Law 1959, Article 87 (2)) is not given. The Austrian Water Act (1959) provides the legal framework to organise formal co-operation between communities and non-state actors to share the costs.

Since 1990s, Austria introduces a paradigm shift in flood risk management policy. This shift was encouraged with the change of the executive board at regional or field construction management level. Furthermore, since 2000s the Forest Engineering Service in WLV emphase the use of catchment-management plans (regional studies) to achieve a more holistic view of the catchment. However, these instruments were introduced in the Hydraulic Engineering Assistance Act regulation in 1985, but it never found any support in the public administration, because of high transaction costs and conflicts with local authorities. An important aspect is the formal-statutory mandatory agreement between different actors to define different tasks, duties and responsibilities. The consequences are a highly institutionalised co-operation agreement (Wirth et al., 2008; Beutl, 2010). Furthermore, an important stakeholder is the Austrian Federal Forests Ltd. The Austrian Federal Forest Ltd. is responsible for the nationalpublic forest areas. In the three study sites the majority of land is used for agricultural or forestry purposes (Seher and Berger, 2009). The key owner are private actors, however large areas are under the responsibility of the Austrian Federal Forest Ltd. Other stakeholders are several non-governmental organisations, such as WWF or Austrian Nature and Biodiversity Conservation Union. Their key objectives refer to the protection and creation of habitat areas in floodplain areas (Naturschutzbund, 2010).

A second conflict refers to the decisions for ranking the implementation of the flood alleviation schemes. In Triesting-Tal, for example, the inter-local co-operation initiated the first two flood storage projects in the upstream municipalities Altenmarkt an der Triesting and Furth an der

81

Triesting. The key objective was to achieve their support for the future projects in the catchment (coalition building) (i5aut). In the Aist study site it was decided to introduce various flood storage areas across the catchment, mainly based on a 'watering-can' principle. In particular, the first implementation phase showed a high degree of political decision to satisfy the local authorities instead to introduce large flood storages, which have an impact for the whole catchment. The level of conflict (as well as transaction costs) depends on the level of separation of interests, which has a direct influence in the ability of implementation of flood risk management plans, such as the empirical results shows from the Triesting-Tal study site. The various interests between the policy makers from different local authorities delayed the realisation of the inter-local flood risk management plan by around 10 years (i5aut; i7aut).

### 4.2.4. Selected study sites in Austria

From the Austrian case study I selected three different study sides (figure 4.4) in three different Federal States to analyse the differences and commonalities to achieve a broader overview of the development of partnership approaches in flood risk management. One criterion for the selection of the study sites was also the availability of previous studies (pattern matching). Key objectives were to analyse similar patterns as well as new concepts (Eisenhardt, 1989; Yin, 1994). The Floodrisk I project (Habersack et al., 2004), for example, analysed the potential of the use of inter-local co-operations combined with a flood risk management as well as spatial and land use management planning. Nevertheless, the research study analysed only the potentials of inter-local co-operations in the flood risk management without analysing fully the drivers of the creation as well as the integration of the co-operation. Or the study conducted by Seher and Berger (2009) concluded with the foundation act of the inter-local co-operation, but did not provide any results about the performances of the interplay between different local and regional actors and stakeholders. This thesis starts from their results and is focused on the latest developments in the inter-local co-operations as well as on the performance and interplay between regional and local actors and stakeholders.

Besides previous research studies, an important criterion was the selection of different governance settings and network structures. The III-Walgau study site, for example, was selected, since it includes an involvement of non-state actors and stakeholders in the interlocal co-operation.



### Figure 4.4: Overview Austrian study sides

Source: own development; © Lebensministerium (2007)

### Triesting-Tal (Austria)

The first study side selected in Austria was the Triesting-Tal co-operation in the Federal State Lower Austria. The River Triesting (figure 4.5) is the first selected study site as to assess the interaction between national, regional and local authorities in inter-local flood risk management concepts. The Triesting-Tal inter-local co-operation was founded in 1968 with the key objective of maintaining the local flood defence measures with an annual budget of 7,000  $\in$  (i5aut). In the early 2000s, with the personal change in the WLV-authority, the WLV

started with the Regional Study Triesting-Tal. However, the focus was on the torrent river Further Bach, which includes one third of the overall catchment size. The outcome of the Regional Study Triesting-Tal showed the need for a more holistic understanding and view of the Triesting-Tal in respect to flood risk management (i7aut). The study identified the implementation of flood storages in the upstream communities, e.g. Furth an der Triesting, as well as linear flood defence measures in the different communities (i7aut; i8aut). The interlocal co-operation share the responsibility in building flood defence measures between the inter-local co-operation and the local authorities. In general, the local authorities have the responsibility for the implementation of local protection measures. On the other hand, the inter-local co-operation has the responsibility for the implementation of inter-local flood defence measures (i8aut). In this sense, the inter-local co-operation is the key developer of the implementation of catchment-management plan, for example the design contracts with farmers to purchase the necessary land or design contracts with construction companies.



Figure 4.5: Overview of Triesting-Tal study site Source: own development; © Lebensministerium (2007)

The River Triesting has a length of 43.63 km and includes in total 18 communities. The catchment is located in the Federal State of Lower Austria close to the Austrian city capital Vienna (approximately 45 km). The recent flood frequency and history shows various numbers of flood events in the past 20 years, especially in the years 1991, 1997 and 2002 (i4aut). Competence authorities regarding flood control measures are Austrian Service for Torrent and Avalanche Control (WLV) – responsible in the upper parts of the catchment (Altenmarkt an der Triesting and Furth an der Triesting) – and BWV – responsible for the lower parts of the catchment.

The Triesting-Tal inter-local co-operation includes a two-level structure (figure 4.6): (1) general assembly and (2) steering group with support of a director and as administration office, with total annual costs of  $10,000 \in$ . The assembly of members includes all members of the inter-local co-operation. In general, the steering group is responsible with the implementation of Triesting-Tal-regional study.



Figure 4.6: The organisational and structural set-up of the Triesting-Tal study site

#### Source: own development

An important aspect is for consideration the financial contribution of each local authority towards inter-local flood alleviation schemes. The financing key was introduced based on two components: (1) land use based on four classes (grassland, forests, agricultural areas and artificial surfaces) and (2) population.

The Triesting-Tal region shows a long tradition of structural flood defence measures. The key focus was the implementation of local – linear structural measures. After the recent flood history in 1991 and 1997, the main strategy was to repair linear defence measures. Following the change in leadership in the regional authorities and the use of new instruments, e.g. regional studies, the flood risk management strategy shifted towards a more holistic catchment approach. The new flood risk management strategy has included the implementation of inter-local flood defence measures, especially in the implementation of flood storages in the upstream communities to protect the overall catchment. Since 2000s the WLV has enforced the communities in the catchment to use flood storages and natural retention ponds and lakes to reduce the impact of future events. Analysing the catchmentmanagement plan and local project appraisals, the Triesting region follows two main strategies. A first step is the implementation of local flood defence schemes, mainly in the downstream – industrial communities. Although the regional authorities support this decision, this project blocked the implementation process goal of the catchment management plans in the past. The second step includes the implementation of flood storages in the upstream areas. In 2006, the project team updated the study with the result of the implementation of five flood storages with an average dam height of 5 to 8 m in the middle and upper part of the catchment, including the implementation of 2.5 km linear flood defence measures. This doubled the total project costs. However, the communities showed no or only minor interest in the implementation of this catchment management plan. Most of the community are focussed on their individual flood defence schemes. Since 2009/2010 the co-operation got a fresh boost. In March 2012 the Triesting-Tal realised its first flood storage project in the upstream catchment area. Next steps included the construction of a further flood storage near to the community Altenmarkt an der Triesting. Although this project is co-funded by the Triesting-Tal, the effect

for the downstream communities is not given. This project was mainly done to encourage Altenmarkt an der Triesting to join the inter-local co-operation. Finally, the Triesting-Tal management plan has included semi-integrated flood risk management strategy. The key strategies have been the implementation of linear structural measures in the different communities as well as runoff reduction measures in the upstream catchment. Nevertheless, the Triesting-Tal flood risk management strategy has no inter-local post-flooding measures. Here, the inter-local co-operation has still gaps in the implementation process.

## Aist (Austria)

The second area includes the River Aist in the Federal State Upper Austria (figure 4.7), where the study sites besides the co-operation between national, regional and local authorities (such as Triesting-Tal) includes also informal co-operations with public utility companies, such as regional road authorities. In August 2002, the River Aist flooded large rural and urban areas in the catchment, particularly affecting the downstream communities Schwertberg, Gutau, Tragwein and Pregarten. In this context, the national and regional authorities (WLV and BWV) were initiated to collaborate with the downstream communities to organize an inter-local cooperation. The objective was to build a co-operation under the Water Law of 1959. The initial plan included 29 communities from 2 different districts (Freistadt and Perg). In 2007, the regional inter-local co-operation Aist was established with 27 communities. However, two communities did not take part of the co-operation, because fear to lose the identification and self-governing independence and additional costs for the public budget (Wirth and Biwald 2006; Weichart, 2006). Key promoters were the two regional authorities (WLV and BWV) as well as the downstream communities, especially Schwertberg (in the initial phase) and Gutau (second phase).



Figure 4.7: Overview of Aist study site

Source: own development; © Lebensministerium (2007)

The Aist inter-local co-operation includes a two-level structure: (1) the general assembly and (2) the steering group (similar to the Triesting-Tal; figure 4.8). Since May 2011, the Aist co-operation employed an external director to support the steering group (further information see section 6.3).



Figure 4.8: The organisational and structural set-up of the Aist study site

Source: own development

The financial contribution of each local authority shows a different calculation in comparison to the Triesting-Tal, which is distinguished between five components: (1) land use, (2) flood damages in 2002, (3) potential future flood damages, (4) retention capacity and (5) financial capacity.

The Aist catchment, in contrast to the other two study sites, shows a low degree of river regulation. After the flood event of 2002, WLV and BWV as well as local councillors started a draft management plan for the catchment. The stakeholders developed a catchment wide management plan. The strategy introduced a holistic view of the catchment with the key aspects of implementing flood storage and ensures natural retention areas to reduce the flood peak. The study includes a different range of runoff management schemes. The first step includes the implementation of local flood defence measures. Here, the communities are responsible for the implementation. However, the inter-local co-operation is paying the necessary partnership funding. The second step includes the implementation of flood storages in the catchment. The study identified 25 potential flood storage areas with a total retention

volume of ca. 7.5 million m<sup>3</sup> (i12aut). This includes a reduction of the peak flow 350 m<sup>3</sup>s<sup>-1</sup> to 240m<sup>3</sup>s<sup>-1</sup>. The total costs are estimated with 30 million Euros (4 Euros per m<sup>3</sup> retention). However, the greatest benefits will impact the downstream community Schwertberg with a total reduction of the peak discharge of 109 m<sup>3</sup>s<sup>-1</sup> (Puchinger and Henle, 2007). The timeframe is designed for the next 30 years. Further measures refer to the implementation of forestry management concept. The Aist catchment is characterised by more than 68% of the forests are stocked with spruces. In 2002 flood event, a large number of debris jams was caused by spruces. Moreover, the overall goal is to restore the forest close to the river banks by planting flood proofed trees.

## Ill-Walgau (Austria)

The third fieldwork area is in the western part of Austria in III-Walgau (figure 4.9), where also private companies are formally involved in the inter-local co-operation. In the early 2000s, the BWV started to develop a river development plan for the Ill-catchment (i20aut). Based on the first results, the BWV developed hazard maps to identify the 'hot spots' in the catchment. After the flood event in 2005 and results from river development plan and hazard maps, the BWV started to build an inter-local co-operation along the III-Walgau sub-catchment. In the sub-catchment of III-Walgau, there exists a long tradition of inter-local co-operation in respect to the water management (i17aut). In 2010 the regional inter-local co-operation III-Walgau was established with 20 members, among them 12 communities, three infrastructure operator (Highway Road Authority (Asfinag), National Rail Way (ÖBB) and Regional Road Authority (Landesstraßen)) and five utility operators. This co-operation is a result of the integration of formal inter-local co-operation (Bludenz, Bürs and Nüziders) in the upstream area, the informal inter-local co-operation (Satteins, Schlins, Frastanz, Nenzing and Asfinag) as well as other members, like utility providers, ÖBB or Feldkirch (i19aut; i18aut). The strategy identified two key concepts: implementation of the WFD and on increasing the flood protection in the subcatchment (i20aut).

90



Figure 4.9: Overview of Ill-Walgau study site Source: own development; © Lebensministerium (2007)

The III-Walgau inter-local co-operation includes a two-level structure (figure 4.10): firstly a general assembly and secondly a steering group. Furthermore, at operational level eight employees are responsible for the implementation, future maintenance and administration tasks (i18aut).

The general assembly of members is the highest board in the association. It includes all members of the inter-local co-operation. Usually they hold two meetings per year. Further, the key aim is to exchange information between the different members, discuss the general strategy and discuss general questions (i18aut). The second level (steering group) includes a selection of 7 members (3 permanent members and 4 elected members). The steering group includes a mix of small and large members, based on the contribution. The community Feldkirch, Asfinag and utility operator VIW-Walgau are permanent members of the steering group, because they are the main contributor (i19aut; i16aut).



Figure 4.10: The organisational and structural set-up in III-Walgau study site

Source: own development

The level of contribution by each individual member distinguished between local authorities, operators of infrastructure and utility plant operators (i31aut).

- Local authorities are based on: (1) riverbank length, (2) discharge and (3) retention volume.
- Operators of infrastructure are based on: (1) flood traffic route, (2) embankment on the boundary area of a floodplain and (3) bridges across the river III.
- Utility plant operators: (1) influenced river stretch, (2) operating water volume and (3) upsurge (increased by 1.5).

The III-Walgau catchment shows a long tradition of regulation of the river. Since the 1820s, the various communities close to the river changed the river flows for agricultural and industrial purposes. The key objectives were economic growth and wealth. In the past 20 years, the Federal Water Engineering Administration in co-operation with the local authorities started to implement various flood protection measures alongside the river. After the recent flood experiences, especially the 2005 flood event, and the implementation of WFD and EU-Floods

Directive into the Austrian legislation, a change in flood risk management policy has occurred. Furthermore, especially the downstream communities, e.g. Feldkirch, increased further pressure to reduce the flood peak in the lower part of the catchment. In addition, since the 1990s, the Ministry for Environment has introduced river development schemes to achieve a more holistic view of the catchments.

The river development schemes include three key objectives. The first objective refers to the implementation of the WFD. The programme of measures includes various steps to increase the 'river continuity' to achieve the ecological requirements from the directive. Additionally, the river development schemes include the objective to increase the fish passability in the catchment. A second objective is expected to increase the biodiversity in the catchment, e.g. re-naturation of the river banks. The regional authorities, for example, created wetland areas in the community Göfis with the key objectives to increase the biodiversity and to reduce the flood peak for the neighbourhood communities. The third objective refers to the implementation of flood storages in the catchment to reduce the flood peak in the catchment. The river development scheme includes the implementation of five flood storages in the catchment. The first project includes the communities Bludenz and Gais with four flood storages with a total volume of 600,000 m<sup>3</sup>. The second project includes the communities Frastanz, Nenzing, Schlins and Satteins with a total volume of 1.2 million m<sup>3</sup> (i46aut). In summary, the III-Walgau river development schemes shows mainly measures referred to the implementation of the WFD with the goal to achieve multiple benefits (Posthumus and Morris, 2007). The co-operation (similar to the other two examples) is the official client to build the flood storages. Additionally, the inter-local co-operation is the key communication partner for the different regional and national authorities. The implementation of flood storage is planned mostly on farmland. The flood storage schemes include an agreement on purchasing the necessary land to build the embankment as well as compensation mechanisms are needed (similar results found in England (Morris et al., 2004)). In addition, the inter-local co-operation is responsible for future maintain once on inter-local flood defence schemes. However, each community is responsible for the local-linear flood defence schemes.

### 4.2.5. The English flood legislation framework

A first arrangement was introduced with the report by the Royal Commission on Land Drainage in England and Wales in 1927. This document and the following Land Drainage Act from 1930 re-designed the English funding scheme (Royal Commission, 1927). The key aim of the Land Drainage Act (1930) was to secure food production in England as a response to the World War One (WWI) and early 1920s and not caused by a major flood event (ibid.). The act includes three main directions: (1) creation of 46 Catchment Boards (summarised to the EA in 1995 with the Environment Act 1995), (2) transfer the permissive powers to the IDBs for non-major rivers as well as to drain the area and (3) introduced a new funding scheme and ended the 'old' beneficiary to pay principle (Scrase and Sheate, 2005). Today, the EA is responsible for the operational implementation of the policy guidelines and directions defined by Defra. The key tasks of the EA are:

- regulatory role,
- strategic overview,
- permissive power,
- advisory power and
- manage incidents (i30eng).

Firstly, the EA has the role as a regulatory body in the field of water protection. Here, the key tasks are to fulfil the requested requirements from the WFD in England, e.g. the protection from pollution and abstraction. Secondly, the EA has the role to provide a strategic overview of the flood risk management in England. The Flood and Water Management Act (2010) defined the role of the EA as: 'agency shall in relation to England and Wales exercise a general supervision over all matters relating to flood and coastal erosion risk management' (UK Legislation, 2010: 51). The EA supports and supervises other organisations, such as the Lead

Local Flood Authorities (LLFA), Water companies, Internal Drainage Boards (IDB) or Highway Road Authorities, to assess and organise the flood risk for which the organisation is responsible. Thirdly, the EA has the permissive power to manage flood risks from the main rivers<sup>19</sup> and sea. This includes that the EA can develop flood defence schemes to protect residential, nonresidential properties and agricultural areas (i30eng). Therefore, the EA is also a key actor in the distribution of the national funding scheme (Flood and coastal erosion risk management Grant in Aid (FDGiA)). Besides, the Regional Flood and Coastal Committee (RFCC), successor of the Regional Flood Defence Committees (RFDCs), plays an important role in the overall flood risk management process. The RFCC was created under the Environment Act from 1995 with the key objective to work in co-operation with the EA to define the regional flood risk management policies (Brown and Damery, 2002). The acts provide the RFCC the power in raising additionally funds (local levy<sup>20</sup>) and for approving the regional flood defence programs. Also the Flood and Water Management Act (2010) under Article 23 (1a) underlines the cooperation between the EA and the RFCC. The Act says that the EA 'must – (a) consult each Regional Flood and Coastal Committee about the way in which the Agency proposes to carry out its flood and coastal erosion risk management functions in relation to the Committee's region' (UK Legislation, 2010: 15). Furthermore, the Act underlines the permission from the RFCC to take any further measures in the area. The act states that the EA 'may not implement the regional programme without the consent of the Regional Flood and Coastal Committee for the region concerned' (UK Legislation, 2010: 15). Additionally, the EA has responsibility for flood assets and is required to maintain and operate schemes (i38eng). In contrast to the Austrian flood risk management system; the local authorities are responsible for the maintenance. This includes a much stronger role of the local authorities to engage in flood risk

<sup>&</sup>lt;sup>19</sup> Local authorities are responsible for non-main rivers; these are: small, local, urban rivers (Parker and Derrick Sewell, 1988).

<sup>&</sup>lt;sup>20</sup> The local levy is based on the council tax. The expenses are based on the number of local authorities in RFCC area (UK Legislation, 2011) – see also section 5.5.

management. The consequences is that the Austrian local authorities have a more active role in flood risk management policy, whereas in England the central Government might be considered to patronize the local authorities in terms of decision-making practices. Fourthly, under the planning policy framework, the EA has an advisory rather than an executive role in the local spatial and land use management planning process. Lastly, the EA plays a key role in flood incident management. Here, the EA is responsible for providing the flood warning across England and Wales as well as to resolve flood incidents, e.g. clearing channels etc. (i30eng). Within the Water Act from 1989 (UK Legislation, 1989), the public water industry was privatised, inclusive their infrastructure. This was a new stage in the privatisation of public water supply system (Thaler, 2010). Coulthard and Frostick (2010) observed that the Water Act (1989) increased the number of actors in the urban flood risk management policy. Both authors' showed that for the City of Hull three different organisations were responsible for the urban flood risk management. The City of Hull was mainly responsible for the roads and gully drainage, Yorkshire Water was responsible for the water supply and treatment system, and the National River Authorities (became the EA in 1996) responsible for the main river courses across the city. The Water Act (1989) increased the number of actors from one (the local authorities in the years between 1973 and 1989) to three different agents with different backgrounds: (1) Yorkshire Water (private business), (2) City of Hull (local actor) and (3) EA (quango and national actor). The outcomes were different management and maintenance systems and approaches as well as different interests, which ended in different outcomes (ibid.).

In addition, besides the already mentioned organisations, I found several other organisations dealing with flood risk management, such as the Highway Authorities, which are responsible for the flood defences and drainage of highways in England (UK Legislation, 1980). The Highways Act from 1980 – Article 102 (1) gives the Highway Authorities the responsibility to maintain and provide the protection of the highways against flood risks and other

environmental hazards. The defence measures can be close to the highways or also request land close to the highways. For coastal erosion management, key actors are the Maritime Local Authorities. These authorities mainly act under the Coastal Act 1949 and Flood and Water Management Act 2010. The Maritime Local Authorities have the permissive power to take further flood defence measures to protect the coastal area. This large number of different authorities caused several challenges and barriers in the past years, as noticed by the independent Review into the 2007 floods conducted by Sir Michel Pitt Review in 2008 (Pitt, 2008). One of the key aims of the Flood and Water Management Act (2010) has been to clarify the responsibility of each party, especially the role of the local authorities regarding flood risk management. In England, the EU Floods Directive (2007/60/EU) was implemented under the Flood Risk Regulations in 2009. The introduction of the EU-Floods Directive opens a new chapter in the flood risk discussion in England and Wales, especially regarding the 'new' relationship and role of the different actors under the national implementation of the EU Floods Directive (see overview table 4.9). The Flood Risk Regulations (UK Legislation, 2009) provide new definition of who is responsible for what. In particular, the LLFA has become more prominent in the flood risk management system, because these organisations are responsible for surface runoff management. Main differences to the Flood and Water Management Act 2010 are other aspects, such as co-operation between the different organisations and topics, funding, define the role of RFCC. In regard to flood risk governance, the Flood and Water Management Act has more influence, because it circumscribes more broadly the actors and social processes in English flood risk management.

EU Floods Directive	The Flood Risk Regulations 2009		
	LLFAs	EA	
Step 1: Preliminary flood risk assessment	<ul> <li>Preliminary assessment report in relation to risk of flooding in LLFA area; not required source of flooding which are under the responsibility of the EA</li> <li>Can reject the recommendations from the EA which returns to Defra</li> <li>Review preliminary flood risk assessment report every six years, first time by 22<sup>nd</sup> June 2017</li> </ul>	<ul> <li>Preliminary flood risk assessment report and map for the following source of flooding: (1) sea, (2) main rivers and (3) reservoirs</li> <li>Supervises, reviews and provides recommendations to LLFA</li> <li>Provides guidance document for the LLFA</li> <li>Reviews preliminary flood risk assessment every six years, first time by 22<sup>nd</sup> June 2017</li> </ul>	
Step 2: Flood hazard and risk maps	<ul> <li>Provide flood hazard and risk maps in relation to risk of flooding in LLFA area; not required source of flooding which are under the responsibility of the EA</li> <li>Can reject the recommendations from the EA</li> <li>Review flood hazards and risk maps every six years, first time by 22<sup>nd</sup> June 2019</li> </ul>	<ul> <li>Provides risk and hazard maps, which shows the risk of flooding from (1) sea, (2) main rivers and (3) reservoirs</li> <li>Supervises, reviews and provides recommendations to LLFA</li> <li>Provides guidance document for the LLFA</li> <li>Co-ordinates with WFD-requirements</li> <li>Reviews flood hazards and risk maps every six years, first time by 22<sup>nd</sup> June 2019</li> </ul>	
Step 3: Flood risk management plans	<ul> <li>Provide flood risk management plans in relation to risk of flooding in LLFA area; not required source of flooding which are under the responsibility of the EA</li> <li>Can reject the recommendations from the EA</li> <li>Participation with all relevant stakeholders and public</li> <li>Review flood hazards and risk maps every six years, first time by 22<sup>nd</sup> June 2021</li> </ul>	<ul> <li>Provides flood risk management plans for following source of flooding: (1) sea, (2) main rivers and (3) reservoirs</li> <li>Supervises, reviews and provides recommendations to LLFA</li> <li>Provides guidance document for the LLFA</li> <li>Participation with all relevant stakeholders and public</li> <li>Reviews flood hazards and risk maps every six years, first time by 22<sup>nd</sup> June 2021</li> </ul>	

### Table 4.9: Overview the role of the EA and LLFA under the Flood Risk Regulations 2009

Source: UK Legislation, 2009

In overall, the LLFA has become a more important role in the English flood risk management system, but at the moment it is quite early to be analysing these groups in the organising and implementation process. Nevertheless, the EA has a key role in the ongoing discussions on ordinary water courses. The EA provides guidelines and monitors the implementation process of the LLFA.

# 4.2.6. English organisational structure in flood risk management

The flood risk management system in England includes a wide range of different legislations and 'primary' stakeholders as well as 'secondary' stakeholders (figure 4.11). At the English national level, the key actors are the Department of Environment, Food and Rural Affairs (Defra) and the Environment Agency (EA). Defra is responsible for the definition of the policy directions in England.



Figure 4.11: Policy responsibility and law structure in English flood risk management system

Source: own development based on Brown and Damery, 2002; MAFF, 1993

The LLFAs, Internal Drainage Boards (IDB), Water industry and Highway Road Authorities are generally responsible for the ordinary water courses for pre-defined areas, such as Highway Road Authorities are responsible for flood defences and drainage of highways. The management of flooding shows little co-ordination and co-operation between the different operational actors. In urban areas, for example, the English flood risk management system includes three main actors:

- Water industry: responsible for water supply, sewage and treatment
- Local authorities: responsible for road and gully drainage and
- EA: responsible for main river management (Coulthard and Frostick, 2010).

The consequences with the increase of actors (in the past 40 years) in management and governance system are the problem of co-ordination and adjustment between the different management systems. In the urban areas there was the lack of a 'lead' organisation with the consequences of no co-ordination, adjustment and knowledge transfer between the organisations with the outcome of different management and maintenance system and practices or lack of information, because nobody was responsible to provide further information, such as providing a incorporated map of the river connection, land drainage and sewer network (ibid.). Further, the flood risk management system had no clear division of responsibility in pluvial flooding until the Flood Risk Regulations and Flood and Water Management Act 2010 came into force. Coulthard and Frostick (2010) demonstrated the main barriers and negative consequences for the English flood risk management system based on unclear responsibilities and lack of co-operation with the help of the City of Hull. Nevertheless, the Flood and Water Management Act encourages a stronger co-operation between the different actors (UK Legislation, 2010).

The LLFAs were introduced as recommendations of lessons learned from the summer floods 2007 (Pitt, 2008). One of the key recommendations was to empower the local level to deal with local problems. The summer flood in 2007 may be considered to be a 'non-classical' English flood event, because the classical floodplain areas, e.g. Severn or Trent, were not affected, also because of the

high protection standard in both rivers. The main damages were caused by pluvial flooding in localised hot spots (Pitt, 2008; Chatterton et al., 2010). The LLFAs are responsible for all sources from flooding from ordinary water courses, especially for the surface runoff and the groundwater flooding, but outside the Internal Drainage Boards areas. The LLFAs have the permissive power to undertake flood defence measures for their framework. They are responsible for the management of the risk of surface runoff, of groundwater and of ordinary water courses (UK Legislation, 2010). The key points for this change was the awareness and recognition of the surface run-off problematic from the experiences of the summer floods of 2007 (i33eng) as well as the requirement from the EU Floods Directive Article 2 (1), which includes a large number of different types of flooding, e.g. groundwater and urban flooding (EC, 2007).

## 4.2.7. New partnership funding scheme in English flood risk management system

The role of the different actors in flood risk management has changed over the time, especially in relation to who should contribute to funding to the flood risk management system. It has been a discussion between public actors (nationally and locally) or between public and private actors (businesses and/or individuals) (Scrase and Sheate, 2005). In general, the English funding system for flood risk management can be divided in five key time periods (table 4.10).

Table 4.10:	Key phases	in English f	lood risk	funding scheme
-------------	------------	--------------	-----------	----------------

Year	Prime Minister and political party	Policy-legislation	Short description
1930	James Ramsay MacDonald (Labour Party)	Land Drainage Act 1930	Regional and local focus
1998	John Major (Conservative Party)	Priority scoring	Focus on urban areas; prioritisation of flood and coastal erosion risk management
2004	Tony Blair (Labour Party)	Centralised funding	Shift in the funding decision from local towards national; prioritisation based on priority scoring
2008	Tony Blair (Labour Party)	Payment for outcomes	Change of the prioritisation system; holistic view of flood and coastal erosion risk management; including social justice aspects in the decision- making practices; dominant national funding
2011	David Cameron (Conservative Party)	Partnership funding	National and third party funding

Source: own development

### Land Drainage Act 1930

With the Land Drainage Act<sup>21</sup> from 1930 (following the Royal Commission report 1927), the new funding scheme shifted away from the beneficiaries pays principle (landowners) to a broader definition of contributors (nationally wide) to ensure the necessary flood alleviation schemes for food safety. The government introduced 'a complex system of rating and levies between the drainage authorities' (Scrase and Sheate, 2005: 115). The funding was divided between national, regional and local grants and levies (O'Riordan, 1980). It included a strong local and regional focus, where the national government closed the funding gap. The outcome was a strong local oriented flood risk management policy, because of the strong agricultural interests, such as focus on food security, where farmers were highly subsided by the national Government (O'Riordan, 1980; Escobar and Demeritt, 2014). The policy of flood risk management was mainly oriented to the administration boundaries, such as local authorities or water authorities, with the results of a complex system of

<sup>&</sup>lt;sup>21</sup> Originally the act covered all aspects of flood risk management, such as agricultural drainage, coastal defence or tidal flooding (O'Riordan, 1980).

flood risk management strategies with the outcome of un-coordinated and inappropriate flood protection schemes within the country (Parker and Sewell, 1988; Petit, 1999).

Since the 1960s, the English governments restructured and incrementally changed the water and flood risk management policy various times (Parker and Sewell, 1988). Key steps in this process included the Water Resources Acts from 1963, 1973, 1983 and 1989 as well as the government Green Paper in 1984, where the key purpose was to centralise the water and flood risk management system (Petit, 1999; Thaler, 2010; Scrase and Sheate, 2005). With the Water Resource Act from 1973, the national Government centralised the water management system with the introduction of the Regional Water Authorities (RWA) (Parker and Sewell, 1988; Thaler, 2010). The RWA became responsible for sea and tidal flood risk management, but under control and close co-operation with the Regional and Local Drainage Committees (Petit, 1999). The local land drainage organisations kept their position and responsibilities in flood risk management. Main reason was the strong lobbying efforts at national level with the result of the implementation of the Land Drainage Act from 1976 (Parker and Sewell, 1988; Petit, 1999; O'Riorden, 1980). Based on the socio-political and economic changes in the 1970s and 1980s, such as the intervention by the International Monetary Fund in 1976, growing interests and interaction of environmental grassroots organisations at national level or shift from Keynesianism to Neoliberal economic policy under Conservative Government as well as the change in the food supply system (move from food shortage to overproduction), the local level lost slowly their status and power in flood risk management (O'Riorden, 1980; Escobar and Demeritt, 2014; Ballinger, 1999).

# Priority scoring

In April 1998, a new priority scoring system was introduced, mainly in response to the general centralisation process of the flood risk management system. First proposals initially only included the inland rivers across England and Wales. The second centralisation step occurred in 2006, when the EA received the responsibility for using the priority scoring scheme also for coastal flood

104

protection. 'The Environment Agency (EA) was given responsibility for allocating funds direct to local authorities for coastal protection schemes, meaning the EA could objectively appraise, prioritise and allocate funds to flood and coastal erosion risk management activity across England in line with where greatest benefit could be achieved with the funding available' (Defra, 2010b: 9).

Therefore, the new funding system replaced the strong focus on local funding schemes in the flood risk management system. Due to the centralisation of the funding scheme, the key objective of the priority scoring system was to create a consistent national-wide criteria methodology (i47eng). The consequences were the implementation of a strong hierarchical top-down management system, where the local level had little influence in the decision-making practices. The effect was a political incapacitation of the local actors, stakeholders and citizens. The main reasons for the new funding scheme was the reduction of the problem of charity hazards in the country, because of the methodology (based on quantitative data and unique for the whole country), to simplify the administration process in the flood defence funding and to provide a national-wide flood protection standard (i47eng). The applied system ranked the different capital projects based on their benefits (EA, 2008; House of Commons, 1998). Moreover, the priority scoring provides a method to justify the public money spent in flood defence schemes, because the national funding was insufficient to realise all requested flood defence schemes (Johnson et al., 2007).

In this line, the priority scoring followed two main tasks. First, the scoring had to identify, if the project can be realised with the current national funds, especially to ensure the realisation of projects with highest benefits. Second, the priority scoring system was to guarantee an effective use of public money in flood risk management (i29eng). Due to this system, the selections of flood defence schemes were based on a more technical-objective understanding instead of a political understanding, similar to the political and public pressure in Germany after the 2002-Elbe flood event (Schwarze and Wagner, 2007). In this way, the priority scoring ranked the different projects based on economic, social and environmental factors and focussed strongly on businesses

105

protection (max. 20 from 44 points). The consequences were that areas with few business activities were discriminated against economic centres, such as urban areas instead of rural communities with low economic growth. This led to highly unequal flood defence development in the country in privilege 'wealthy' – international competitive communities (i31eng). The main funding sources for flood risk management came from the local authorities. The local authorities contributed in the years 1996 to 2004 approximately 66% to 78% of the total budget for flood defence schemes<sup>22</sup> (figure 4.12). The consequences were that the decision process was strongly oriented towards a local-regional decision process.

# Centralised funding

After the financial year (2004-05), the contribution process totally changed towards a national system of spending with less local control and influence (i47eng; i48eng). This remains to today; approximately 95% of the funding for flood risk management is provided by the central Government (Defra, 2010b). An analysis of the public funding for flood and coastal risk management showed that the public authorities increased their overall funding especially within the larger nationalisation of the funding.

<sup>&</sup>lt;sup>22</sup> The flood risk management funding is generated mainly on national and local taxation. However, decisions were based on local – regional scale how to spend the budget for flood risk management.



Figure 4.12: National and local flood defence funding in England and Wales, 1996-2011

Source: House of Commons (2009)

### **Outcome measures**

The New Labour government re-defined significantly the flood policy and funding in England partly re-designing the funding scheme alongside the introduction of the integrated-holistic view of flood risk management (Defra, 2005). The new government and the introduction of an holistic view approach (Making Space for Water), Defra replaced the priority scoring system with the outcome measure (OM) approach for prioritising flood defence projects (Johnson and Penning-Rowsell, 2010; EA, 2008). Furthermore, key drivers for this change were the introduction of the WFD in 2000 (EC, 2000) as well as the New Labour Third Way ideology (MacLeavy, 2007).

The main reasons for the introduction of a new approach stemmed from the various criticisms of the priority scoring system. A first critique referred to the strong focus on the economic factors and a lesser importance being given to the other two factors; people and environment. In the Priority scoring the selection scheme excluded the aspect of social justice in flood risk management (House of Commons, 2008), on which New Labour Party focussed its policy discussion (Colomb, 2007; MacLeavy, 2007). A second key critique referred to the strong focus towards wealthy-urban areas, also in cases where flood defence schemes in rural areas were economically feasible (ICE, 2008;

House of Commons, 2008). In addition, a third aspect of the change was that the priority score scheme prioritised the projects within its given limited available funding (national investment programme for 2008 to 2011). It produced smaller forecast contributions towards the new targets than a proposed new system based on OMs (i48eng).

The OM approach introduced four elements (1 April 2011 onwards – see also table 4.11) for ranking the project proposals (Defra, 2007). A key change was the introduction of the aspects of social justice into the final decision process. In doing so, the Government gave a higher weight to deprived areas across the country and these areas were seen by central Government as being more worthy to receive funding for flood management measures. Consequently, the payment rates were higher for deprived households. However, the Government distinguished between three different payment rates depending on the number of deprived households in the area. Furthermore, the Government excluded private businesses and infrastructure operators in the calculation, which had to manage their risk by themselves (i29eng). Here we observed a strong shift from an economic priority more towards the focus on people. A further change included the introduction of OM4, which focussed on to the requested requirements from the WFD – the renaturation of the rivers.
## Table 4.11: Outcome measures scheme in England and Wales

OM	Definition	Payment rate
OM1	Average benefit to cost ratio of schemes delivering OMs	5.56 p per £1 of qualifying benefit
OM2	Households* 'better'**	
	protected against flood risk present value of the reduction	45p per £1
	in direct damages to residential	30p per £1
	properties: -in the 20% most deprived areas -in the 21-40% most deprived areas -in the 60% least deprived areas	20p per £1
OM3	Households* 'better'** protected	
	of the reduction in direct damages to residential properties:	45p per £1
	-in the 20% most deprived areas	30p per £1
	-in the 21-40% most deprived areas -in the 60% least deprived areas	20p per £1
OM4	Statutory environmental obligations	
	met:	
OM4a	Hectares of water-dependent habitat	£15,000 per ha
OM4b	Hectares of inter-tidal habitat created	£50,000 per ha
OM4c	Kilometres of protected river improved	£80,000 per ha

\*Households must be at direct risk of damage from flood damage or coastal erosion and have been build or converted into

housing before January 2012 to be counted (flood damages) or qualified (coastal erosion).

\*\* change in the risk category (very significant risk, significant risk, moderate risk and low risk)

Source: EA (2012)a and Defra (2011)

The OM approach introduced a new methodology for the decision making process (Johnson and

Penning-Rowsell, 2010). The calculation of the OM was based on the sum of each OM divided by the

present value of the total life costs (PV WLC) or the costs for approval of the asset (EA, 2012a):

FDGiA = Qualifying benefits under OM1 \* payment rate + Qualifying benefits under OM2 for each level od deprived to the second second

\* relevant payment rate + Qualifying benefits under OM3 for each level of deprivation

\* relevant payment rates + Payment made under OM4

and

Raw score = FDGiA contribtuion ÷ PV WLC or costs for approval

However, the OM approach strongly focuses on the results from the CBA (similar to the priority scoring system) and thereby many of the limitations of this approach remained (Hanley and Spash, 1994). The outcome was that the OM approach led to changes in types of scheme funded. In

particular, OM favoured projects in areas with socio-economic problems, such as high deprivation. Nevertheless, the main limitations refer to the methodology to assess the benefits, such as intangible and indirect costs (Penning-Rowsell et al., 2005).

# Partnership funding going back to the local

The latest step in the English flood risk management funding regime included the introduction of the partnership funding scheme in April 2011 (Defra, 2012). There were multiple drivers for change and a move back towards locally-based flood risk management. Key factors included the results from the EA study *Investing for the Future* published in 2009 (EA, 2009a) and the recommendations from the Pitt Review (Pitt, 2008). The former study recognised the need of an increase of the annual budget of approximately £20 million, exclusive of the annual rate of inflation (EA, 2009a). However, based on the Government spending review (October 2010) and with the result of the current pressure on public spending and resources, the central Government reduced the public national funding in the following years (figure 4.13). The spending review foresaw a reduction of the capital works from £354 million (2010-11) to £259 million (constant over the next spending review period) as well as the revenue funding (maintained work, emergency response, hazard and risk mapping and modelling) from £275 million (2010-11) to £226 million (2014-15).



Figure 4.13: National Flood and Coastal Risk Management Grant-in-Aid Funding: 2007-08 to 2014-15 Source: House of Commons (2013b): 8

One reason for this budget reduction is the fiscal squeeze and state budget deficit. However, the Government (2012) announced that they would increase additional £120 million for new flood defence buildings in the next two years, with a key objective of economic growth (House of Commons, 2013a)<sup>23</sup>. The focus on economic growth shows a political step backwards towards the 1980s political ideology (Haughton et al., 2013), which refers to the question of the role of deprived and rural areas. The new direction however shows similar concerns that I found under the priority scoring (see text above). In sum, the new Coalition Government has taken over certain aspects from the New Labour policy. First, the Coalition Government includes the social justice discussion in flood risk management; for example, the partnership funding calculator included the outcome of measures calculation scheme. Second, the Coalition Government shows also some strong top-down direction, like the introduction of the new infrastructure levy and the level of national contribution (see also section 6.5). The discussion of a change of the national flood risk management funding regime to open it to third actors is not new (Sugden, 2004; Corkindale, 2007) with the shift to a

<sup>&</sup>lt;sup>23</sup> The spending round 2013 foresees to increase the budget of flood defences to £370 million in 2015-16 (HM Treasury, 2013). However, the key argument lies with economic growth instead of sustainable development.

broader partnership-oriented governance arrangement (Allmendinger and Haughton, 2012). In the past years, caused by a lack of national funding, the EA tried to encourage private and local actors to contribute to the flood defence schemes (i37eng). Under the 'old' funding scheme, the national government normally provided approximately 95 to 100% to the flood defence scheme. The numbers vary widely depending on the contribution from the local levy or private actors, e.g. the £150,000 contribution from Tesco as well as small contribution from Carlisle City Council for Carlisle or the East Lane sea defences in Bawdsey, where private actors largely contributed to the overall defence project (i37eng; i31eng). The contribution from third parties was approximately £13 million in the period between 2008 and 2011 (i48eng); approximately £10 million from local authorities and approximately £3 million from private actors (CCC, 2012).

First, the Government gives the possibility to realise flood defence schemes, even if they do not match the necessary requirements of the 100% funding. Currently, approximately 75% of the projects are fully funded by the national Government; the other 25% of the projects need additional funding from other parties (i29eng). An important aspect of the 'new' funding regime is the possibility for third parties to contribute to close the funding gap, if the project satisfies the minimum criteria (the benefits of the project have to exceed the costs).

In general, partnership funding is based on the methodology of OMs calculation. In comparison to other European countries (such as Germany<sup>24</sup> or Austria<sup>25</sup>), there are no fixed percentages from the national Government (EA, 2013). In this example, the national Government contribute (financially) between 50% and 100%. Furthermore, the Government has adapted the weights in the methodology. The partnership funding approach (as compared to OM) has a higher weighting in favour of protecting people in risk areas within deprived areas, where they are least likely to be insured, less able to afford local protection measures and least likely to be able to recover following

<sup>&</sup>lt;sup>24</sup> 50% Federal State and 50% Laender (Garrelts and Lange, 2011).

<sup>&</sup>lt;sup>25</sup> For example in Carinthia: 40% national contribution, 40% regional contribution and 20% contribution from local authorities and private stakeholders and citizens (i24aut).

a flood event without additionally welfare support, such as Kingston Upon Hull (i29eng). However, partnership funding mainly focuses only on flood defence scheme aspects; where the EA and not other organisations are responsible.

Second, the Flood and Water Management Act (2010) re-designed the use of the local levy, which was understood as a supplement funding source after 2004. The Act foresaw the stronger role of local authorities in the current flood risk management discussion. In this line, the Act included a stronger localism approach also within the local levy. Under the new Act, the RFCC – which also represents local authorities within RFCC unit – could use the local levy as an additionally funding source. The allocation was strongly based on the national prioritisation format, or it was able to use it independently to promote locally preferred schemes (Devon County Council, 2012). The levy was in most cases based on the council tax (i36eng). The level of contribution requested the agreements of each member of the RFCC, because it was valid for everyone in the RFCC area. In general, the level of contribution between the different RFCC was widely different (table 4.12), because each RFCC defined the level of contribution individually.

RFCC – name	2009-10	2010-11	2011-12
Anglian Central	971,958	1,020,556	1,020,556
Anglian Eastern	2,059,654	2,486,008	2,486,008
Anglian Northern	1,547,086	1,547,086	1,547,086
Northumbria	1,850,443	1,905,956	1,982,194
Yorkshire	1,127,750	1,500,000	1,500,000
North West	3,567,000	3,638,000	3,638,000
Southern	1,153,830	1,176,907	1,176,907
Severn Trent	3,075,000	3,000,000	3,000,000
South West	484,472	496,584	546,242
Wessex	3,090,155	3,167,409	3,167,409
Thames	10,000,000	10,000,000	10,000,000
Total	28,927,348	29,938,506	30,064,402

## Table 4.12: Local levy contributions by English RFCC

Source: Devon County Council (2012)

In sum, partnership funding has two main innovations in comparison to the OM funding approach. First, partnership funding prioritises stronger areas within deprived areas, because of long-term socio-economic effects (Rawls, 1972). Second, partnership funding encourages actors, stakeholders or citizen outside from the national Government to contribute (financial) towards new flood alleviation schemes. Nevertheless, the latest House of Commons review (2013) mentioned that the FDGiA still provides approximately 93% of the flood defence budget. These numbers show a very small contribution by local and non-state actors, stakeholders and citizens. Consequently, localism might be seen more as a rhetoric discussion, especially in the assumption that the level of contribution defines the level of participation in the final decision-making practices. A large contribution from national government includes a strong top-down policy. Similar developments can be observed with the new community infrastructure levy and implementation of EU Floods Directive in relation to surface runoff, where the national government dictated the policy and implementation process.

# 4.2.8. Selected study sites in England

From the English case study three different study sides has been selected (figure 4.14) with different

forms of partnership funding to analyse and to understand the influence of new politics of scale.





Source: own development; © EA (2009)b; © Edina Digimap Ordinance Survey Service (2012)

## Cockermouth (England)

The Cockermouth study site in England has been selected because of two main criteria. First, the Cockermouth study site introduces one of the first the new partnership funding policies in England. In addition, the study site includes local grassroots organisation as a key actor, which is strongly involved in the implementation process of new policy. The key aim of this study is to assess the role and the interaction of local grassroots organisation in the local flood risk governance.

The civil parish of Cockermouth is located in the highly frequented tourist area in Lake District in the Borough of Allerdale, with the two rivers the Cocker and Derwent confluence close to the town centre. Cockermouth was flooded three times within less than four years – in 2005, 2008 and in 2009 (i35eng; i36eng). The 2009 event affected the whole of Cumbria as well as parts of Ireland and Scotland, but the Derwent and Cocker valley were particularly hit in 2009 (i35eng; i36eng). In Cockermouth, for example, more than 693 residential and 225 businesses were affected (Cumbria Resilience, 2011). Flooding occurred due to the high peak flow from the River Derwent (figure 4.15). Cockermouth includes a large number of rural oriented small-medium businesses, with more than 80% of them affected by flooding. Implications were that small-medium businesses engage with the Cockermouth Flood Action Group (FAG) as well as the ongoing partnership funding scheme.



Figure 4.15: Overview Cockermouth study site

Source: own development; © EA (2009b); © Edina Digimap Ordinance Survey Service (2012)

In the study site of Cockermouth, the partnership approach between the community and the EA began mainly as a result of the 2009 floods. Between the years 2005 and 2008, Cockermouth was affected partly by two small flood events. After the 2005 floods, some residents of the town organised for their area a first local self-help group. However, the organisation had no wider diffusion to the whole town. The 2008 flood event re-enforced the creation of a second local self-help group in Cockermouth, to solve the surface runoff problem in the Gote Road. The Gote Road organisation started an intensive relationship with local and national politicians as well as the EA regional office (i36eng).

The Cockermouth partnership funding scheme included a wide range of different actors and stakeholders involved in the decision-making process. However, the key actors were the EA and the local FAG. The original flood alleviation project proposal foresees a total cost of £ 5.3 million. Within the new funding scheme, the payment foresees a funding for £ 3 million from national sources, which is not acceptable by the citizens because of the low degree of flood defence schemes (i37eng). First, the EA re-organised their approach to reduce the total costs to £4.4 million, such as selecting different contractors. Second, the additional funding gap was partly closed by local stakeholders and citizens, such as the temporarily increase of the local council tax (i36eng). At the end, the final project proposal had total costs of £ 4.4 million, with £ 3.3 million from national sources (FDGiA) and £ 1.1 million from external contributors (table 4.13).

Funding source	£
FDGiA	£4,400,000
Local Levy	£100,000
Cockermouth Flood Action Group	£215,000
Cockermouth Town Council	£120,000
Allerdale Borough Council	£100,000
Cumbria County Council	£700,000

#### Table 4.13: Flood defence funding in Cockermouth

Source: i36eng; i37eng

A key contributor for collecting fund raised money was the Cumbria Floods Foundation. Most of the financial resources were used for households without insurances to re-build their properties. The further money was used for the Cockermouth flood defence schemes (i37eng). Further key actor was the Cockermouth FAG in the overall process (i37eng; i35eng). The Cockermouth FAG collected up to £400,000, e.g. by temporarily increasing the Council Tax for three years (FAG). However, the local actors were not able to close the financial gap. The shortfall of £700,000 was closed by the Cumbria County Council (i37eng). The FAG played an important role in the relationship with other local Government bodies and with the citizens for convincing them to contribute to the scheme. Third, based on the previous flood event in 2009, a strong incentive was the problem obtaining insurance (see above), which encouraged the local actors to contribute to the funding. Fourth, the socio-economic structure of Cockermouth played an important role, especially regarding business contributions to the funding scheme. Cockermouth's businesses are mainly characterised by smallmedium businesses with a strong focus on local and regional markets. Cockermouth has been a strong market town, where the tourism sector plays an important role in the local economy. The two factors (insurance and locus to local economy) increased the fear of economic decline in the community. This fear stimulated the business owners and citizens to contribute to the funding scheme to reduce the risk of future flood hazard events as well as to buy an insurance bill.

# Morpeth (England)

The second study site Morpeth is the administrative head of the Northumberland County Council. The motivation for selecting Morpeth as a study sites was similar to that for Cockermouth. First, Morpeth was one of the first areas, which implemented the new partnership funding policy in England. Second, Morpeth has like Cockermouth a strong local grassroots organisation, which has been intensively involved in the policy discussions and implementations of the new funding system.

The county town has had a long flooding history at various levels: e.g. in 1963, 1967 and 2008. Key problems in the towns are fluvial flood hazards from the River Wansbeck as well as three smaller

rapid response catchments (Cotting Burn, Church Burn and Postern Burn) which run through the town (i42eng). Furthermore, Morpeth is affected by problematic surface runoff (i43eng). After the 1963 flood event, the public administration carried out flood defence schemes, mainly by improving and providing new river defences in the town, but for only some parts of the town (i43eng). High Stanners and Mitford Road (figure 4.16), for example, had no flood defence schemes. The rest of the town had a flood defence scheme of approximately 1:50 year standard of protection, with the exception of the Middle Green area, which had a higher defence standard – approximately 1:75 year flood defence. The heights of flood defence scheme were based on the flood depth from the 1963 flood event (i38eng). In 6<sup>th</sup> and 7<sup>th</sup> of September 2008, approximately 964 properties were affected by the flood event (JBA, 2011).



#### Figure 4.16: Overview Morpeth study site

Source: own development; © EA (2009b); © Edina Digimap Ordinance Survey Service (2012)

Nevertheless, after the 2008 flood event the EA started to re-design the business cases for a higher standard of protection (from 1:100 to 1:137). The introduction of the partnership funding scheme and the shortfall of the national contribution (fixed FDGiA of £9.6 million) for the original proposal needed a contribution from third actors. Here flood appraisal foresees the implementation of flood storage in the upper part of the county town with a storage capacity of 1:50.

The Morpeth study side includes a cooperative approach between three main actors (EA, Northumberland County Council, Morpeth FAG). Since the approval of the scheme, conflicts were mainly associated to the surface runoff problematic. The 2008 flood event was a combination between fluvial and pluvial flooding. In Morpeth, the Northumberland County Council as LLFA is responsible to manage the problems with surface runoff and risk from ordinary watercourses. The main issues refer to the funding of further measures in relation to reduce the risk of the surface runoff. Especially the EA focussed their funding only to the main river and not to further issues.

## Bridgwater (England)

The study site Bridgwater has been selected being an opposite example compared to Cockermouth and Morpeth. Bridgwater has in contrast to the other two study sites no official local grassroots organisations, different socio-economic frameworks as well as no final – official contracts between the national and local authorities. Furthermore, the time frame is totally different compared to Morpeth and Cockermouth. The aim was to understand the impacts of no local grassroots organisation and different socio-economic frameworks in the policy discussion and implementation.

The town Bridgwater (figure 4.17) is located in Somerset county council and is a key priority area for future economic development strategies (i40eng). Key problems in terms of flooding of Bridgwater are: (1) the tidal flooding risk from the River Parrett and (2) a sea level rising of 978 mm by 2,108 (Sedgmoor, 2009a).





Source: own development; © EA (2009b); © Edina Digimap Ordinance Survey Service (2012)

The last tidal flooding affected the town in 1977 and 1981 (ibid.). In response to the 1977 tidal flooding, the public administration built hard defences (embankments and walls) from the entire tidal reach of the River Parrett until to Oath Lock (few kilometres inland). The flood defence schemes have a 1:200 year standard, albeit with some exemptions (i39eng). The current flood defence schemes protect approximately 8,400 houses and 800 businesses in Bridgwater (Sedgemoor, 2009a). Already the do-nothing scenario estimated a poor condition of the current flood defence schemes by 2030. Furthermore, with the forecast of climate change the sea level in the Seven Estuary will increase by 1 metre by 2100, which includes that the town centre needs flood defence walls up to 2 metres (currently 1 metre), which is socially and politically not acceptable as well as not technically feasible (i39eng). The actual proposals foresee the implementation of a surge barrier downstream of

the town, with total costs of  $\pounds$  24.6 million and additional  $\pounds$  8 million for the improvement of downstream flood defence schemes (table 4.14).

Option	Present Value Flood Damage £ million	Present Value Flood Benefits £ million	Present Value Costs £ million	Whole Life Costs (without discounting) £ million
Do nothing	1,699	-	-	-
Do minimum	915	784	0.63	2.1
Re-Engineer Existing Defences	Negligible	1,699	24	94
New Tidal Barrier	Negligible	1,699	18	54

Table 4 14: Summary	/ of	economic	feasibility	of Brid	gwater	flood	defence	scheme
Table 4.14. Summary	/ 01	economic	reasibility		gwater	11000	uerence	scheme

Source: Sedgemoor, 2009a: 27

The current flood defences can achieve the fit-for-purpose as well as the realisation of the current regeneration plan. The Bridgwater regeneration plan foresee new properties development by 7,700 new properties by 2036 (6,200 new homes within the existing urban area, further 1,500 new homes north of Bridgwater). New developments in Flood Zone 3a foresee 3,677 new properties by 2026 and further 2,000 by 2026-2036 (Sedgemoor, 2009b). Sedgemoor County Council developed the Bridgwater Strategic Flood Defence Tariff SPD adopted by September 2009. The proposal foresees the contribution of 40% of the total costs of the barrier (£ 9.84 million to £ 24.6 million). To close this financial gap, the public administration in co-operation with the EA introduces a 'roof tax' on new developments (table 4.15).

### Table 4.15: The levels of contribution set by the tariff

Development type / flood vulnerability	Flood Zone	Tariff
Housing (i.e. houses, flats, basement	3	£1,500 per 1 and 2 bedrooms
homes)		£1,750 per 3 bedrooms
		£2,000 per 4+ bedrooms
	1 and 2	£525 per 1 and 2 bedrooms
		£613 per 3 bedrooms
		£700 per 4+ bedrooms
Essential infrastructure (i.e. transport	3	£2,000 per 100 sp.m of net additional
and utilities)		maximum of £100,000
	1 and 2	No contribution required
Highly vulnerable (i.e. police, ambulance, fire station and caravan sites)	3	£2,000 per 100 sp.m of net additional floorspace per application to a maximum of £100,000
	1 and 2	No contribution required
More vulnerable (i.e. hospitals, pubs,	3	£2,000 per 100 sp.m of net additional
nightclubs, hotels, schools, medical centres, camping and caravan sites)		floorspace per application to a maximum of £100,000
	1 and 2	No contribution required
Less vulnerable (i.e. shops,	3	£1,000 per 100 sp.m of net additional
general industry, storage and		maximum of £100,000
	1 and 2	No contribution required

Source: Sedgemoor, 2009b: 21

In general, the proposal includes also a financial contribution from the new developments in Flood Zone 1 and 2, in total approximately 35% of the £9.84 million (Sedgemoor, 2009c). The proposal foresees that new developments outside of Flood Zone 3a have a discount rate up to 50%. However,

the proposal includes a maximum cap of  $\pm 100,000$ .

# **4.3.Key findings**

This chapter introduces the findings from the policy analysis and stakeholder and institutional mapping. I found a wide range of different stakeholders and actors in the policy interaction in the studied countries. The number of stakeholders and actors are strongly based on the institutional framework in both countries. For example, the privatisation of the water supply-system in England

or the different responsibility between upper and lower part of the catchment cause different governance structures and management systems.

The key findings from this chapter are:

- In Austria the main national acts include the Forest Act (1975), the Water Act (1959) and the Hydraulic Engineering Assistance Act (1985)
- In England the main national acts include the Flood and Water Management Act (2010) and the Floods Risk Regulations (2009) as well as Royal Commission on Land Drainage in England and Wales from 1927
- Austrian system include a Federal State system, where at federal level, different acts in regarding spatial and land use management exists as well as emergency management and disaster compensation.
- In Austria the key actors are the WLV and BWV dealing with the problematic of flood risk management. Both organisations have the responsibility to define the policy strategy for the country. Furthermore, both organisations have the operational responsibility of the flood policy as well as provide a large contribution to the funding scheme (approximately 80% of the total costs).
- In England the key actors are Defra and the EA. The EA supports and supervises other organisations, such as the LLFA, Water companies, Internal Drainage Boards (IDB) or Highway Road Authorities, to assess and organise the flood risk for which the organisation is responsible. Further, the EA is also a key actor in the distribution of the national funding scheme (Flood and coastal erosion risk management Grant in Aid (FDGiA).
- Key objectives in the selected countries are: (1) implement inter-local flood risk management concepts (Austria) and (2) the introduction of the partnership funding scheme in April 2011 to generate financial contribution from local and non-state actors, stakeholders and citizens.

Funding rules in Austria are (in generally): 40% national; 40% regional and 20% local oragnised in contrast to England, where the national Government contribute (financial) between 50% and 100%.

A key difference between both European countries is the flood risk management strategy regarding to measures within the flood plain. The Austrian flood risk management strategy follows an increasing emphasis upon reducing the runoff adopting a catchment-wide management plan. On the other side, the English strategy focuses on coping with the flow at a local point. A consequence is the different scalar arrangements and governance networks in both countries (Swyngedouw, 2004; Cox, 1998, Brenner, 2004). The next chapter focuses on the scalar arrangements in the selected countries. The aim of this chapter is to provide the actual state spatial processes, construction of new territorial scales and their socio-spatial relations as well as the reason for scalar re-arrangements.

# 5. Drivers and scalar arrangements in flood risk management in the selected case studies

**Research question:** How scalar re-arrangements have been influenced by flood risk management? What mechanism and action have been used to change the policy?

This chapter explores the drivers in the new scalar arrangements in flood risk management policy. From the discussion in the chapter 2, the scale concept plays a fundamental element in current flood risk management debates, especially relating to the question of risk- and responsibility-sharing between national and local authorities and non-state actors. This chapter focuses on the causes, conditions and circumstances in which scalar re-arrangements develop and is the first chapter of the empirical results.

# 5.1.Drivers for network creation in flood risk management

In the partnership approach the first question refers to drivers for the creation of networks (Waddock, 1989; Plummer and FitzGibbon, 2004; Pomeroy et al., 2001). In particular, various scholars emphasise the preconditions or antecedents to build a co-operation process between various stakeholders relating to environment management (Waddock, 1989; Plummer and FitzGibbon, 2004; Pomeroy et al., 2001; Selin and Chavez, 1995). Drawing from previous studies, I have identified six general drivers from the literature that encourage stakeholders to co-operate (table 5.16).

## Table 5.16: Drivers for network creation

Driver	Origin
Real or imagined crisis	Regional / national level
Intervention of a third party	Regional / national level
Legally mandate	Regional / national level
Common vision, definition or understanding	Local level
Existing networks	Local level
Leadership	Local level

Source: adapted from Waddock (1989); Plummer and FitzGibbon (2004); Pomeroy et al. (2001)

The empirical results show various mechanisms and actions encourage a change of socio-economic relationships in flood risk management. The Austrian system provides formalised rules to develop and to organise inter-local co-operations. In England the new funding policy changes the behaviour and interaction of the national authorities towards local actors and stakeholders. The involvement of local actors and stakeholders in the planning process shows an important step in the 'successful' creation of a partnership approach between the different local and national actors and stakeholders. In Austria an important development was the implementation of a 'fair' co-operation model between the different members involved as well as the involvement of 'key' actors in co-operation. Examples are the utility companies in III-Walgau co-operation. The co-operation model foresees the implementation of a combination of polluter-pays and beneficiary pays principles, where the downstream municipalities largely contribute to the financial costs of flood alleviation schemes. The key issue is the fair sharing of financial and risk burdens between the different members. The interlocal co-operation process is not a win-win solution between the different members. An important aspect is to reduce the negative aspects of an inter-local co-operation, e.g. regarding financial burdens. The key point in the negotiation process is to define a fair financial key between the different members (i19aut). There is a high fear in failure of the individual expectations into the inter-local co-operation. Lastly, the inter-local co-operation is normally built up after a major flood event or recent numbers of various flood events, for example Aist after the Danube flood event in 2002 (i9aut; i10aut; i11aut), or Weisseritz after the Elbe flood event in the same year (Wirth et al., 2008). The different study sites show that the inter-local co-operation was developed as direct reactions to recent flood events. However, the result from the interviews shows that other factors and developments (before the flood event) have played an important role. Nevertheless, in many cases the major flood event has been an eye opener for many local policy makers.

In England as for the 'old' funding system, the EA follows a strong top-down approach in the decision-making practices. This research study showed a more equal treatment between the different actors. The idea is to support the local decision process, because if the project costs are not

127

fully covered the local stakeholders have to close the additional financial gap. The quote below highlights the change in the EA.

Participant: 'When I first started working with the Environment Agency I found them quite defensive and a little bit hostile towards us but when they realised that we were prepared to assist in a businesslike manner and negotiate with them and find a solution they started to kind of realise that if we all worked as a team we could get somewhere and with, you know, we're now coming up to the third anniversary and the flood defences are nearly completed' (Local authority, England).

In Cockermouth, the community encouraged the EA to change their flood defence scheme, especially the scheme along the Rubby Banks Road, where the citizens were very protective in respect to an "open view" of the river. The community was quite adamant in the decision process. The outcome of the discussion was the re-design of the flood defence scheme in Cockermouth with higher costs (i37eng). Overall, the partnership funding scheme moved the financial risk towards the local actors from the EA (i43eng; i37eng; i38eng) with the result that the EA are forced to work more closely with the different actors (i38eng). In comparison, the involvement of local stakeholders was very low in the decision process under the 'old' funding system. However, with the new partnership funding scheme the voice of the local actors is much larger in compare to the previous system.

The local FAG has been shown to be mainly as a strong lobby-pressure group. Addressing the financial contribution by the Northumberland County Council it has shown a strong interplay between the local actors and the Government body. The limitations highlighted that further projects cannot be completely funded anymore by the Northumberland County Council. In this case, the county council requested the contribution from third actors (i43eng). This results that there will be conflicts and lack of understandings by the local agents, especially in areas with a much higher socio-economic problematic.

Analysing the interview results, this study found some additional components (table 5.17). The components of precondition are divided into two sources (internal and external). The internal aspects refer to the members of inter-local co-operation; external refers to the aspects of the outside development and pressure. Each of the elements in the table will now be discussed.

128

Precondition	Description	Origin	Case studies
Effect of crises	Flood event	Internal	Austria
	Security approach	Internal	<ul> <li>Austria and</li> </ul>
			England
	Pressure	<ul> <li>Internal</li> </ul>	Austria
	EU Floods Directive	External	Austria
The influence of	Promoter	Internal	<ul> <li>Austria and</li> </ul>
adequate			England
leadership	Political support	• External	<ul> <li>Austria and England</li> </ul>
	Social capital	Internal	<ul> <li>England</li> </ul>
	Cultural capital	Internal	England
Economic rational	Positive net present value	Internal	Austria
	Public contribution	External	<ul> <li>Austria and</li> </ul>
			England
	Pooling monetary resources	<ul> <li>Internal/external</li> </ul>	<ul> <li>Austria and England</li> </ul>
	Funding	External	Austria
Fair risk-sharing	Distributional effects	Internal	Austria and
			England
Regional	Common historical	Internal	Austria
embeddeness	developments and networks		
	Commuting	Internal	• Austria
Common political	Long term strategy	External/internal	Austria
vision and strategy	<ul> <li>Solid management plan</li> </ul>	External/internal	<ul> <li>Austria</li> </ul>
	0 1	•	

#### Table 5.17: Preconditions in the development of new networks in flood risk management

Source: own development

## 5.1.1. Effect of crises

The first forces are relating to the aspect of crises. In general, the partnership process 'needs' a recent flood history or increase risk awareness at local level. The study sites in Austria and England show a policy change at local level after a recent major flood event. However, the need for a major flood event is more important at local level as an 'eye opener', less than as a catalyst for introduction of new instruments and policies (Penning-Rowsell et al., 2006). Analysing the interviews showed a confession of capitulation by the local authorities. In addition, the results show a high pressure from the citizens and local businesses to reduce the impact of future flood events. A key role played by private businesses in the partnership discussion. In the study sites, businesses stressed the possibility of de-localisation their branches to other regions. However, these industries were strongly affected by the last flood events and show an important interest to achieve flood defence measures to reduce the impact of future events. This was a key driver for the local

municipalities to start a regional discourse on flood risk management. On the other hand, the communities need additional strains from the outside. For the implementation of the EU-Floods Directive into national law, the Austrian government requested a risk assessment of new flood defence measures. Importantly, it should be ensured that the new flood defence scheme is not increasing the risk of neighbourhood and/or downstream communities. Consequently, large numbers of new local flood defence measures were not accepted by the regional water authorities or downstream communities. Due to the implementation of EU-Floods Directive, communities in high risk areas need a broader flood risk management plan. Further, in the policy on flood risk management an important incentive is the expectations of safety. Similar to De Marchi and Scolobig (2012), the empirical results noticed the importance of the guarantee of safety for residents. In particular, in the study sites which showed several flood events in a short time the demand for safety was a central argument to get engaged in the policy discussion. Influentially, a central concern was towards the insurance aspect (i36eng; i37eng; i38eng; i43eng). A key aspect, mainly in the study sites Morpeth and Cockermouth, was the fear of no longer being able to secure house insurance or insurance for businesses (i37eng; i38eng), with the risk of failure to obtain a bank loan, or the insurance bill becoming disproportional by expensive (i30eng). This aspect was a key driver for local stakeholders to get engaged in the political discussion. In contrast in Austria where flood insurance has no significant influence in flood risk management policy, because the flood losses compensation scheme is based on public tax and mainly excluded from property insurance policies (i26aut). Nevertheless, also in the Austrian study sites the demand for safety was important mainly in terms of not losing enterprises in the region (i10aut; i11aut). The interconnection between the private insurance system and the public policy of flood risk management can be understood as a dialectical interaction, where the private interests of householders and businesses – affordable insurance bills and access to mortgage – have become part of the general interests (request of flood alleviation scheme funded by the taxpayers) (Habermas, 1989). Therefore, Habermas (1989) sees the role of the public authorities satisfied the needs of the actors, stakeholders and citizens which cannot be satisfied by the market.

## 5.1.2. The influence of adequate leadership

A second group refers to the need of local/regional leadership in the co-operation process. The key task was to mobilise the other members to join and to ask their contribution in the co-operation. All interview partners emphasised the importance of leadership in the discussion. However, most of the leadership reportedly came from the regional authorities rather than locally. In the English case study, the key leadership proceeded the FAG in participation with the EA; except where no local grassroots organisation exists the EA was the representative leader in the decision-making process.

A central debate is the educational and professional background as well as habitus of the residents. In the literature, Bourdieu (1986) described these circumstances as cultural capital. He distinguished between three main types of cultural capital (embodied state, objectified state and institutional state). Embodied state refers to the aspect of habitus of the people. Here cultural capital can be accumulated through cultural environment (home, friends, school, and colleague). Bourdieu (1986) emphasised that accumulation depends on the time period, the socio-economic framework and the social class. The second type (objective state) understood cultural capital as objects (e.g. paintings, books, or sculptures). Finally, the institutional states see cultural capital base on academic and professional education, e.g. academic qualifications, certificates. These three different types of capital influence the individual behaviour within the society.

The interviews highlighted the importance of social and cultural capital in the discussion between the different actors. Morpeth and Cockermouth have a very different sort of social class composition to other English areas. Both communities show a low level of deprivation in their administration unit. In Morpeth, for example, the members of the local FAG are well educated (e.g. lawyers, or architects). Moreover, they have a good network to the various political actors in the ongoing flood risk policy discussion as well as they can argue in a suitable way in sense of lobbying (i38eng). Similar

131

observations are found in the work of McCarthy (2004). However, in contrast these aspects play less important factor in the study site of Bridgwater. This town shows a different socio-economic structure in comparison to Morpeth and Cockermouth (see section 4.2.8). Here, the key actors have been the County Council and the EA. In sum, local leaders were able to engage and influence the decision-making practices based on their local capacities (Bourdieu, 1986; Kuhlicke et al., 2012). These local leaders achieved a more vocal than others and 'hijacked' the agenda to ensure their interests and objectives.

While analysing the cultural capital, especially the two study sites Cockermouth and Morpeth show clear advantages over Bridgwater or other areas in England (i38eng). In both study sites we can find well-educated and well-informed local residents. The members of the FAG have a high level of educational and professional background (institutional state). Furthermore, in both areas we observe strong concerns about the cultural heritage in the town (objective state). Especially in the study site of Cockermouth, the discussion is also focussed on the cultural heritage in the town. The local residents forced the EA to re-design the flood defence scheme to fit in the townscape (i36eng). A third type of cultural capital can be strongly observed in the FAG of Morpeth and Cockermouth. In Morpeth, for example, the interviews show well-educated and well-informed members of the FAG. The key problem is that partnership funding encourages inequality within the country. The policy envisions certain local groups (see also Featherstone et al., 2012). Featherstone et al. (2012) described this process as a 'roll-out' neoliberalism thinking and acting with the consequences of the move away from a nearly homogenic policy of flood risk management. The key problems are disadvantaged communities with no resources, expertises, social and cultural capital to engage in the decision-making process. In particular, communities, which suffered from former process of decentralisations, such as Bridgwater, are more vulnerable to the new funding system (Featherstone et al., 2012).

## 5.1.3. Economic rational

Further important aspects were economic arguments. Some interview partners stressed the need of individual/local benefits (i19aut; i6aut; i14aut). In most cases, the communities joined the cooperation because of a positive benefit for the community. In the economic literature, these councillors represent the principle of NIE (Weichhart, 2006; Niskanen, 1973). The theory emphasises the behaviour of the policy makers as self-interested agents. Their main purpose is to maximise their individual benefit. However, the implementations of inter-local flood defence measures were not only win-win solutions for all members. In particular, upstream communities achieve no or minimal benefits from the inter-local flood defence schemes. Nevertheless, not all communities follow the *homo economicus* principle. In the Aist study site, altruistic behaviour, i.e. where local authorities gain no benefits from the co-operation, was also observed and considered to be a reason for joining the partnership. However, this is more an exception than regularity. In all three study sites the aspect of solidarity played a role in joining the co-operation is mentioned by many interviews (i17aut; i12aut; i13aut; i15aut).

## 5.1.4. Fair risk-sharing

An important aspect refers to the social justice debate (similar results can be found in Johnson et al (2007) and Walker and Burningham (2011)) in the initiation of co-operations in flood risk management. In Austria, the central problem is that the inter-local co-operation includes a safety benefit transfer from upstream communities to the downstream communities. The partnership funding scheme allows sharing the local contribution between the members of co-operation. For some communities this is only possible to get flood defence schemes. To reduce this negative impact, most co-operations introduced a 'solidarity' financing key. This means a monetary transfer from downstream communities to upstream communities (see also section 4.2). Furthermore, the contribution of private actors in the inter-local co-operation process shows a higher acceptance for the local authorities to join the co-operation, e.g. Ill-Walgau study site (i17aut).

Important development refers to the positive answer of 'big' players in the region. The acceptance of 'large' communities and private actors to take part of the inter-local co-operation increases the acceptance of 'smaller' players to join the co-operation. Furthermore, an important aspect is the social impact in the region (Latane, 1996). The representatives from communities know each other very well and were not able to reject the offer. The study sites show a high social influence pressure between the different members.

## 5.1.5. Regional embeddeness

Regional and local networks are usually embedded in a local-regional environment, which influence the social capital within the region and therefore the human behaviour (Rutten and Boekema, 2007). From this perspective, the thesis constructs regional embeddeness primary on common historical developments (path dependency) and economic relationship, such as commuter flow, within the catchment. Several communities had past experiences with inter-local co-operations in flood risk management, e.g. Ehbach-Nafla Wasserverband (communities Göfis and Feldkirch) and Ill-Wasserverband (upper part of the sub-catchment) with Bludenz, Bürs, Nüziders, National Highway Road Authorities, National Railway and Regional Road Authorities; or the informal flood risk management partnership between four communities (Satteins, Schlins, Frastanz and Nenzing) and the Austrian Highway Road Authorities (i45aut). In summary, the region shows a long tradition in inter-local co-operations, which positively influence the creation of the Ill-Walgau co-operation. Therefore, the historical factor – working together in the past and knowing each other –influences the creation of a partnership approach.

In general, analysing the regional embeddeness between the different municipalities, I observed a polycentric structure and spatial development at the catchment level. Similar results were detected in the travel to work distribution (see following figures below).

134



Figure 5.18: Travel to work, Aist study site

Source: own development; data Statistik Austria (2004)



Figure 5.19: Travel to work: Ill-Walgau

Source: own development; data Statistik Austria (2004)



## Figure 5.20: Travel to work, Triesting-Tal study site

Source: own development; data Statistik Austria (2004)

In the Aist study site, for example, I observed a mono-centric region, where Linz (capital of the Federal State Upper Austria) is the dominating centre with its industry and public administration headquarters. Linz and the surrounding area is a high attraction for the region. Most of the citizens from the Aist area travel to work into the Linz area (figure 5.18). In particular, the inhabitants from upstream communities are more likely to work in Linz than in the downstream communities. A key reason is the given infrastructure between Freistadt and its surrounding municipalities and Linz. There exist direct railway and highway roads connections, which reduced the individual transport costs. Consequently, the existing infrastructure interconnections exclude a direction connection between the upper and lower part of the catchment, with the results of a low economic integration process between the upper and the lower part of the catchment. This finding implies that lack of a common economic region complicated the creation of an inter-local co-operation; mainly the upstream communities see no benefit in the co-operation also in terms of their citizens working in the downstream area. With regard to flood risk management analysing the empirical results, in

particular upstream communities blocked the creation of an inter-local co-operation, because of the lack of a common economic region (i14aut; i13aut, i15aut). Nevertheless, a common economic region does not automatically lead to a situation where there is an inter-local co-operation (i49aut).

I also found a similar situation in two other 'smaller' study sites. The III-Walgau study (figure 5.19) shows two main polycentric areas at the whole catchment – in the upper part of the sub-catchment (Bregenz, Nüziders and Bürs) and at downstream area with the municipalities Feldkirch, Frastanz, Satterns, Schlins and Rankweil (which is not a member of the III-Walgau, but shows a key economic role in the region). Furthermore, the region shows a strong cross-border economic relationship, where a large number of Austrian citizens are working across the border in Switzerland. The situation in the Triesting-Tal is slightly different compared to the other two study sites. Here, I identified a stronger interrelationship between the municipalities in the upper and lower part of the catchment (figure 5.20). However, the travel to work structure shows more a one-way direction. Most employers from the upstream area are working in the industrial centres (e.g. Pottenstein or Berndorf) in the lower part of the catchment, but not vice-versa. A second direction, I observed especially in the lower part of the catchment. Here, a large number of employers choose to work in the large urban areas, e.g. Vienna, Mödling, Baden or Traiskirchen. A key reason is the welldeveloped transport infrastructure, e.g. train and highway, towards these urban areas (i5aut). Generally, the empirical results do not show any clear evidence that a common historical development and strong economic interactions between the different local authorities promote a partnership approach between different actors and stakeholders in flood risk management. This is in contrast to the economic geography literature; in this case a common historical development has been an important factor for corporate culture to cooperate and to promote the regional competitiveness and innovation (Gärtner, 2011).

## 5.1.6. Common political visions and strategies

Further another important issue is a common vision and a solid concept for flood risk management. The long term strategy had to define the inter-local co-operation process. The main argument was to

138

know what the purpose of the inter-local co-operation is and why the communities should join it. The concept should include: where are the problems, what are the targets and objectives and how do we achieve these targets. Here, the national and regional authorities play a particular role. However, the co-operation between the different actors is very difficult. In general, the co-operation process includes a high risk of failure. The main problem is the gap between policy guidelines and the implementation process at local levels (Cowell and Owens, 2006) with the result being misunderstandings and conflicts between different actors about their interests and understanding (Mason and Muller, 2007). The result of new scalar re-arrangements can be policy failures and social protests (MacLeod and Goodwin, 1999). The interviews underlined the important aspects of a continuous interaction between the different agents (mainly face-to-face) to reduce the risk of failure. Furthermore, co-operation between communities should be stable based on a strong legal basis and increased frequency of meetings and interactions.

# 5.2. State spatial process

The analysis of the rescaling process (new space creation) and their socio-spatial relationships follows the theoretical frameworks developed by Brenner (2004), Jessop et al. (2008) and Perkmann (2007). The integration of the main research interests – power and interests of each actor, stakeholder and citizen, their interaction and performance in the scalar debate – allows us to overcome the limitations and traps in following only one research dimension (Brenner, 2004; Jessop et al., 2008). To provide a more holistic view, I have elaborated a stepwise analytical approach in the assessment of scalar constructions, their socio-cultural structures and their influence and consequences for the relationship between the different actors, stakeholders and citizens under the new regime. The analysis of the rescaling process (new space creation) and their socio-spatial relationships follows the theoretical frameworks developed by Brenner (2004). Brenner (2004) combined the theoretical concept of Henri Lefebvre's (2007) *State Space* and Bob Jessop's (1990) *State Theory*. The combinations incorporate the transformation of national-state level and the

impact towards regional development (Klink, 2013; Grimm, 2013). Brenner (2004) developed an analytical framework to assess and to interpret changing governance structures. This is highly relevant to understand central – local and state – non-state relationship (Li and Wu, 2012). According to Brenner (2004), scalar re-arrangements are mainly influenced by the general interactions (strategy) by the state as well as the activities resulting from the new scalar arrangements (projects). Table 5.18 summarises the basic characteristics of politics of scale in the selected case study and study sites.

Table 5.18:	Current	state	spatiality	in	Austria	and	England
-------------	---------	-------	------------	----	---------	-----	---------

	Austria	England
State spatial projects	Upscaling of administration, planning and decision-making practices towards inter-local co-operations; uniform and standardised bureaucratic exposure; regional framework	Devolution to new scales (local); downscaling and outscaling of decision-making practices
State spatial strategies	Multi-scalar activities; favour inter- local scale for flood risk management	Introduction of multiplicity scales; uneven flood risk management planning to allow localities to implement flood alleviation schemes; central still highly coordinated flood risk policy

Source: own development

Partnership funding, for example, foresees a stronger engagement of local actors and stakeholders in the flood risk management discussion. Further, the English study sites show a change in the responsibility in terms of design and financing. In terms of partnership funding, local actors and stakeholders have gained an important role in funding and in organising external contributions to close the financial gap.

Participant: 'the [name] Flood Action Group worked very hard to secure funding they pulled in money from all sorts of different sources (...) So it's been a mixture of funding from householders, businesses, Flood Action Group, the Charitable Flood Foundation and local Government' (National Authority, England).

In the three study sites the level of local contribution is between 25%-50%<sup>26</sup>. Thus, the three study sites show different approaches to coping with the financial gap (see section 4.2). Partnership funding encourages local actors to get actively involved in the decision process, which is a central

<sup>&</sup>lt;sup>26</sup> Bridgwater: 40%; Cockermouth: 25%; Morpeth: 50%

objective of new funding regime to encourage local actors, stakeholders and citizens to engage in flood risk management. In particular, a central objective is to encourage local level to take over responsibility from the national Government (i38eng). This finding is examined in following interviewee statement.

Participation: 'Now there was also quite another notable difference I think came about from the community had such a stake in the scheme and paying a substantial proportion towards the cost of it, they wanted a very big say in what was built, and...I mean we always do consult with the local community about all development proposals and take their views on board, but in this case there's a stretch of the scheme along [name], where people were very protective about the open view of the river (...) and they didn't want to lose that (...) we of course would want to take the more conservative approach, or certainly amongst our operations team there was a fear of something with moving parts and the potential for it to go wrong, so yes, traditional flood defence wall and, of course, that's the cheaper option. But the community were quite adamant (...) The outcome of all the consultations we had was, it become very clear that anything other than the self-raising barrier (...) but yes, that was definitely a difference that the local funding of the defence scheme brought about, this demand for a real say in how the money was spent' (National Authority, England).

In Cockermouth, for example, the local actors have already influenced the decision process to ensure their interests. They encouraged the EA to re-design the businesses proposal to include the needs and interests of the town (economic benefits) and individual householders (aesthetic benefits), such as self-raising barrier without losing the view to the river (i36eng; i37eng). An important aspect for the self-organisation of local stakeholder groups was the lack of political representatives, especially their lack of support and interest after the flood events with the result of frustration and mistrust of the local residents regarding the public administration. So, scale jumping can take place in response to the mistrust to national authorities. The 2009 flood caused a large impact on the whole town with the creation of the Cockermouth FAG. Furthermore, the role of the Cockermouth FAG shifted from a lobby group to a professional partner in the discussion. The already existing experiences from the previous flood events as well as through broader interaction in the community, recent the FAG could attract new members with different backgrounds and knowledge (i30). Furthermore, an important aspect was that the local groups already could orient and request

support to the existing structures of the National Flood Forum (NFF)<sup>27</sup>. These two aspects could increase the willingness of engagement in the community.

For the EA these events were considered to be more minor problems when compared to other hotspots in the region, e.g. Carlisle in 2005. The following 2009 flood event caused large pressure from the local stakeholders in the local flood policy discussion. The interviews showed a first key development affected directly the FAG. The 2009 flood event caused a large impact on the whole town with the creation of the Cockermouth FAG. Secondly, there were already existing clear linkages and relationships between the different key actors, which supported the initiation of the discussion. Third, the role of the Cockermouth FAG shifted from a lobby group to a professional partner in the discussion. One reason was the change in the interaction of the FAG with the EA. The already existing experiences from the previous flood events as well as broader interaction in the community, the FAG could encourage new members with different backgrounds and knowledge (i30eng).

A further development was the collaboration of the FAG with an external consultant company to present their first flood risk management plan for the town on a professional basis. At the beginning, the EA was reported to be mainly reserved and defensive relating to co-operation with FAG (i36eng). However due to increase of professionalism of the local FAG, the EA opened the door to the FAG to be part of the planning process; this was a shift from a top-down relationship to a more equal partnership between the different actors. In summary, a key aspect was the change in the local FAG by incorporating internal and external knowledge, such as recruitment of specialists and collaboration with private consultant groups (i36eng; i37eng; i38eng). Here I observed a strong leadership position by the Cockermouth FAG both to start and to organise the interaction between both sides (local-national). Subsequently, the steering group of the FAG addresses flood problematic towards the different Government levels and organisations.

<sup>&</sup>lt;sup>27</sup> The National Flood Forum (NFF) was established in 2002 in England as a bottom-up grassroots organisation to support to support victims of flood events, in particular for the recovery as well as prevention (NFF, n.d.).

The involvement of local actors and stakeholders in the decision-making practices demonstrated a shift in power between national and local level. Nevertheless, the new scalar arrangements have not changed the power inequalities within the country. One of the consequences of the rescaling process was the downscaling of power towards local actors and stakeholders. In particular, the use of power (Sayer, 2004), indicates a change in the power settings based on partnership funding as shown for Morpeth and Cockermouth. The local level has become more capable and dominant in the decision-making practices based on their need of financial contribution towards the flood alleviation scheme. However, the empirical results manifested that also the local capacity is strongly necessary to change the power settings and especially the use of power. Similar results were identified in Morpeth (i38eng), where partnership funding delegated power towards local scale (Cook et al., 2013). However, the Bridgwater study site was different in comparison to the other two English study sites with no direct influence of local actors in the designing and planning process, because the lack of local capacity to engage in the decision-making discourse. Consequently, the EA developed and led the proposal and implementation for the flood alleviation scheme, where the local level was excluded from the decision-making practices.

The empirical results show a shift from a centralised approach towards a more decentralised state, especially to overcome the fiscal squeeze and cuts in the actual spending review period (2010-2015). The transfer includes the change of various regulatory tasks from the national towards a local level. Examples are the organisation of local financing and planning processes, which strongly depends on the EA – relationship with local actors – and if the local actors are able and willing to respond to the new tasks and duties.

Participant: We didn't really apportion cost on [name of the town] and who was going to contribute what. It was very much a ... this is where we are, this is where we need to be, and where could we source that funding? And it really was...we didn't assign that you get his much and you get that much, it was very much a...how we can maximise the income and we were chasing a target' (National Authority, England).

This included a top-down rescaling process, initiated and encouraged by the national Government. The rescaling process included mainly two main directions: downscaling from national to local level with the consequences that the local level had to outscale the additional tasks and duties to nonstate actors and stakeholders. The main reason for this was the lack of capacity by local authorities to deal with the new responsibilities. But whereas in Cockermouth this resulted in significant rescaling where local actors were able to influence (scale jumping) the planning outcomes down to the level of aesthetic detail, in Bridgwater it was a different story. In sum, the new scalar arrangements empowered a specific group of citizens (see also chapter 6).

The new policy framework includes a re-orientation in terms of financing, development and implementation of flood risk management. Nevertheless, the local impact on the actual policy discussion is strongly limited; approximately 93% of the financial contribution is still nationally organised (FDGiA) (House of Commons, 2013b). Interestingly, partnership funding did not provide any evidence of a broader empowering of local levels (key objective of localism debate). The outcome was that the government still had to define rules and objectives, but mandated other actors and stakeholders to realise the national goals and objectives, such as costs savings or implementation of 'hard defenses' instead of adopt a sustainable flood risk management system.

Additionally in the English case study, I identified a strong differentiation and concentration of flood risk management activities and investments. With the introduction of the priority scoring system (in 1998), the different flood risk management projects have been made to compete against each other. A key focus was the aspect of economic growth and the increase of competition for the attraction of local, national and international capital (i33eng).The key objective is to invest in the most competitive space (Brenner, 2004) with the consequences of an uneven development of the country's flood risk management schemes. The introduction of partnership funding has increased the divergence between the local communities based on local resources, such as economic, social and cultural capital. The interviews have shown differences in the development of pathways in the implementation of partnership funding, which are also based on networks and local interactions in the policy discussion (see section 6.4).

144
In particular, the English flood risk management system has increased customisation in arrangements between the different actors. The flood risk management system has been adapted to the local contribution and the influence of local actors in the negotiation process with national actors (i38eng). Furthermore, the English partnership funding schemes include no consistent standard in relation to the amounts of contributions of each partner and the negotiation process. According to the empirical results, this customised national administration standards increase the local specific arrangements according to place-specific policy, such as the funding arrangements from each study sides. The consequences were the development of different local institutions, which emerge from different responses to the new policy. According to Brenner (2004), this politics of scale has increased the unevenness in the country where the national state focussed on specific localities. This led to a change in the relationship between different authorities (national and local) as well as between state and non-state actors and stakeholders with the outcome of hollowing-out of the state (McCarthy and Prudham, 2005) as well as critical legitimacy problems, because of local grassroots organisations in the decision-making practices (Allmendinger and Haughton, 2009). Influentially, the new scalar arrangements increased the interaction and resources (time and costs) between the different scales - especially between national authorities and local actors and stakeholders (i37eng).

Differences were observed in the Austrian flood risk policies, where the regional actors have introduced a more centralised system (based on catchments). The policy shift foresaw the upscaling of responsibility and duties from local authorities to the inter-local co-operation. The shift of responsibility was from local to inter-local levels, e.g. in the case of negotiation with local land owners, financing and management. The inter-local co-operations are mainly responsible for the implementation, financing and maintenance of the assets. Their key task is the negotiation process with the local land owners, where the inter-local co-operation plays a crucial role in the whole process. Normally, the inter-local co-operation leads this process in collaboration with regional farmer representatives and local councillors (i12aut; i20aut). In some of the cases the local

councillors have not been involved in the negotiation process, depending on his/her relationship with local landowners (i19aut; i5aut; i12aut). Moreover, the inter-local co-operation undertakes additional management strategies, like the inter-local forest management strategy in the Aist study site. The development and implementation of multi-objectives encourage the co-operation strength and provide an important stabilising factor. A second additional task is the communication process with local citizens and regional authorities. In the Aist study site, especially the inter-local cooperation is responsible for the communication and interaction with local citizens. In summary, the interaction of the co-operations in flood risk management includes an upscaling process from the local municipalities towards an inter-local spatial scale. The inter-local co-operations take over tasks, such as communication and maintenance, from the local municipalities. Overall, from empirical data I identified a two-way rescaling process: (1) from local authorities to the inter-local co-operation (upscaling)(see above), but (2) also from the regional authorities to the inter-local co-operation (downscaling), such as planning and implementation of structural and non-structural measures. The direction and range of scaling depends on the self-interest and capability of the inter-local authorities (see section 6.4). Inter-local co-operation with strong capacity adopted more tasks and responsibilities from regional and local authorities. Similar developments can be identified in the politics of scale settings, where both countries show changes of the scalar structures based on new national policies (top-down change) (Jessop et al., 2008; Cox, 2013). In contrast to the literature (Delaney and Leitner, 1997; Leitner and Miller, 2007), focussed on a strong nested scale idea, the empirical results show further developments like bottom-up changes in scalar arrangements. The study sites show also bottom-up incentives to change the scalar arrangements, such as shown in the Aist or Cockermouth study sites. Nevertheless, the empirical results show a strong top-down change of scalar arrangements.

# **5.3.Key findings**

In this chapter, I provide the results of analysing the drivers and current scalar arrangements in flood risk management as a central framework for nature of governance and their structures in flood risk management. Results indicated changes in scalar arrangement were based on different political developments at different levels, in particular international, national and local. The main findings are:

- In England, the empirical results indicated a shift from a highly centralised decision-process towards a broader decentralised approach, where the national Government transfers various regulatory tasks, duties and responsibility from national level to the local authorities. The new policy framework includes a re-orientation in terms of financing, development and implementation of flood risk management.
- The English flood risk management system has increased customised relationships between the different actors. The flood risk management system has been adapted to the local contribution and the influence of local actors in the negotiation process with national actors.
- The Austrian flood risk policies, where the regional actors have introduced a more centralised system (based on catchments). The policy shift foresaw the upscaling of responsibility and duties from local authorities to the inter-local co-operation. The inter-local co-operations take over tasks, such as communication and maintenance, from the local municipalities and planning and implementation tasks of flood alleviation schemes from the national and regional authorities.

The next chapter will demonstrate the outcomes and consequences of new scalar arrangements in flood risk management policy. In particular, the chapter focuses on the influence of rescaling processes to policy design in both countries, especially their impacts towards policy networks.

# 6. Impacts and outcomes of new scale constructions

**Research questions:** How cross-scale dynamics have been influenced national by flood risk management strategy? And what are the barriers and limitations of new scalar re-arrangements in terms of network governance, and what are their social and political functions and consequences?

This chapter explores the scalar arrangements in flood risk management policy. The aim is to understand the scalar re-arrangements in flood risk management policy and their impacts in the decision-making practices as well as the social structures of the key actors and stakeholders. From the discussion in the chapter 2, the scale concept plays a fundamental element in current flood risk management debates, especially relating to the question of risk- and responsibility-sharing between national and local authorities and non-state actors. Rescaling processes encourage new definitions and discussion of each actor and their role in policy discussions and social relation as well as their linkages in the definition and implementation of policy strategies (Reed and Bruyneel, 2010; Cox, 2013; Pugalis and Townsend, 2013). The outcome of rescaling is manifested as changes in the relationship between the different agents and actual governance framework (Thiel and Egerton, 2011; Gualini, 2006; Thiel, 2010). The academic debate in the scalar re-arrangements is strongly focussed on causes and consequences of the construction of new territorial scales as well as on historical developments and conditions which influence them. Recent research works are focussed on the socio-cultural relationships between the different actors, stakeholders and citizens (Jessop et al., 2008). A key focus has been the aspects of power in the interaction between the different actors, stakeholders and citizens in policy-decision practices (Allen, 2009). In the second sense, scales have a strong influence on the interactions as well as performances of actors and stakeholders in the policy discussion.

## 6.1.Shifts in the socio-economic relationships

The new scalar arrangements lead also to change in socio-economic relationship between the different actors, stakeholders and citizens within new network structures. The analysis revealed that in the English flood risk management system, a shift in the socio-economic relationship within the

country occurred with the introduction of the partnership funding; with the aim of empowering the local level (Allmendinger and Haughton, 2012). The new funding strategy in England, for example, favours local actors to get more involved in the actual flood risk management discussion. However, the interviews revealed that the local level is seen as a key factor in the overall discussion (i38eng). The new direction devolves power and responsibility towards local actors, such as grassroots organisations like FAG, which need adequate structures and resources to use this empowerment. The outcome is increased competition between the different local communities and a consequent shift to exclusive flood risk management policies. The policy shifted away from understanding flood risk management as a public good towards a narrow club good for privileged places (i33eng; i34eng). The objective is to realise flood alleviation schemes in the areas, where anybody can pay and/or lobby for them. Overall, partnership funding has increased the place-specific governance structures resulting in an intensification of uneven developments between the different areas. An interesting observation from these interviews is that partnership funding allows place-specific decisions in the design and level of standard protection, positively for localities with adequate financial and cultural background; and negative consequences for communities without adequate social and cultural background (lack of interaction at political level) or financial background (not necessary financial resources to realise higher flood protection standards). Additionally, the final decision has shifted from a technical-objective towards a more political-subjective approach. Key objectives were the economic viabilities of the places which are stabilising norms and values in partnership development (see also section 7.2) as well as also ensuring a political engagement by local actors and stakeholders: for Bridgwater re-generation, for Cockermouth and Morpeth especially affordability insurance payments. This enhances differences in the local institutions forms and settings with the result of uneven local policy regimes (Brenner, 2004). In sum, the English policy opens a wide range of possibilities and different approaches, which strongly depend on local actors and their interests and social capacity. Therefore, the local actors have more responsibility and space compared to those in the 'old' funding regime. The new funding policy provides these new opportunities for local actors,

but this requests also the mobilisation of local actors (Cox, 1998; van Schendel, 2002; Collinge, 1999). Primarily, the members from local FAG in Morpeth and Cockermouth mobilised at local scale to use this possibility to change within and across scale their flood risk management plan. Here, the local social groups moved to a higher level (national) of interaction (space of engagement). For example, the local actors use their opportunity to secure their interests. In the selected study sites, I identified that local actors are not only active at local level, but also at national level, where they build and operates networks beyond local scale (Kythreotis and Jonas, 2012). In both study sites, the FAGs have built a relationship towards national actors and used the space of engagement to influence the policy discourse at national scale (Cox, 1998; Kythreotis and Jonas, 2012). Interestingly in terms of power (see section 2.1.2), the local groups influenced particularly the first and second dimension of power relationship (Lukes, 2005). The local grassroot organisations, such as Cockermouth or Morpeth FAG, modified the behaviour of the national Authorities, such as the EA, or other local actors, such as the County Council, within the decision-making process, mainly concerning the change of the design of flood alleviation schemes (first dimension). Furthermore, the local groups had the ability to shape the whole policy agenda. In the Morpeth study, for example, the interaction with local grassroots organisations modified the policy framework for the County Council to close the funding gap in Morpeth and to avoid additional fully funded projects within the County (second dimension). Here, the FAG controlled the agenda of the County Council. On the contrary there are the Austrian study sites, where the new scalar arrangements also influenced the third dimensions, such as flooding agricultural land in the upper part of the catchment to protect downstream communities (see section 7.2).

The consequences of these new scalar re-arrangements were new governance structures, policy outcomes and power geometry between different scales initiated by public and private actors and stakeholders (Gibbs and Jonas, 2000; 2001; Jordan et al., 2005; Jordan, 2008) to promote resilient flood risk management (Pitt, 2008). Their key aim was to ensure strategic advantages compared to other local authorities and their interests, where these multiscalar governance networks influenced

the decision-making process. In sum, the policy problem is localised oriented, but where the FAG utilised political networks to ensure their needs and interests. In the literature, these developments were described as politics of scale 'from below' (Kythreotis and Jonas, 2012). The FAG has been a powerful vehicle for the mobilisation at local level, as observed in Cockermouth and Morpeth and found during the interviews. At national level, the FAG interact with national politicians (i.e. Member of Parliament), especially in the cases of Cockermouth and Morpeth. The empirical results show that not everyone from the local level is able to jump to a higher level. However, an important issue refers to the adequate background of the local actors at national and regional level to change the social interactions and power structures, such as local grassroots organisations in the study sites Cockermouth and Morpeth. The jumping scale includes a change in the regulatory processes and approaches, as already mentioned by Swyngedouw (1997). The new scalar arrangement enforced a new regulatory processes and approaches in the flood risk management system. The implementation of the partnership funding included a wide range of changes in the current approach. Nowadays, the different actors have to follow new regulatory processes introduced by the national Government to satisfy the realisation of the project, especially aspect of level of contribution, accountability or sharing risk burdens (see also appendix 3). Furthermore, partnership funding introduces new regulatory approaches, e.g. interaction between the different actors or partnership funding calculator. The change in the scalar arrangements includes a stronger position towards local actors in the decision-making process. Following Smith (1992), the new scalar arrangement and political strategy was strongly followed by the marginalisation of actors in the flood risk management system. In the 1990s, the English flood risk management system shifted its focus from a strong rural basis to a stronger focus towards urban areas (see also section 4.2.10). Therefore, rural communities did not benefit from these new political circumstances. The partnership funding engages with local actors in terms of funding. Subsequently, the local actors demand to influence the current local flood risk management discussion (see above). This explains the strong presence of rural-local actors in the new policy discussion. The jumping scale allows

marginalised actors to use this prospect to ensure their interests. However, the analysis of the interviews revealed that only a few local actors have the capability to use this opportunity (i38eng). Within the scale jumping, the FAG created 'parallel' networks to interact at national level (Allmendinger and Haughton, 2009).

Differently in the Austrian case study, where local actors and stakeholders are jumping between multi-scale directions (Kythreotis and Jonas, 2012). In the Austrian case study, the scale jump was observed of local actors to an inter-local level – mainly to members in the steering group to interact at local (downwards) and regional (upwards) scale. This foresees that the scalar arrangements do not only depend on top-down policy interventions. The local and regional actors introduced a new scale arrangement, where local actors also get more involved in the flood risk management system. The scale jump shifted the relationship between the different local and regional actors, for example in the Aist study site. Here, the local actors benefited from the new administration structure, especially in the interaction with the regional actors. They exploit the opportunity to ensure their interests and objectives to overcome a 'classical' top-down approach. The inter-local co-operation has changed the boundaries and scalar arrangements in the management system: away from a local perspective towards at catchment-wide management plan. In particular, the networks between the different local municipalities enable them to realise catchment-wide flood defence schemes. The interaction of local actors mainly focuses at a local scale compared to the English case study. A key reason is the different administration structure between both countries. However, the development and the motivation are similar between both countries. However, also here I observed differences between the different study sites. A key aspect is the mobilisation of local actors. The mobilisation is focussed in most of the study sites on local politicians, in contrast to the scale jump in Morpeth and Cockermouth. The process of scale jumping is based more on external drivers as already Peck (2002) noticed. The three Austrian study sites were strongly pushed by regional authorities to organise and to interact amongst themselves at a meta-governance structure.

Austrian flood risk management follows a relatively uniform structure. Based on the interviews, the Austrian approach has established a uniform administration practices with local adaptive forms. The overall goal of all three study sites has been the implementation of inter-local flood defence schemes – mainly flood storages – and local linear measures to reduce the impact of future events (i9aut; i7aut; i11aut; i19aut; i20aut). Differences were identified in the detailed management planning, like forest management concepts in Aist or ensuring of natural retention areas to increase biodiversity in III-Walgau. The national Government introduced a standardised approach in the organisation structure of the inter-local co-operations, but also similar-fixed rules for administration processes and approaches, especially in relation to planning, approval and funding (i3aut; i26aut). Partnership funding contributions, for example, include a relative uniform standard for all projects in each Federal State<sup>28</sup>, which is different to the English partnership funding scheme. Influentially, the Austrian policy tries to redistribute the resources in flood risk management in order to equalise the distribution of investments (independently between rural or urban areas). In summary, the aim is to provide an equal-similar security-approach between the different regions (for example 1:100 standard protections for most of Austrian flood alleviation schemes) (i26aut).

In sum, as highlighted by the table 6.19 the Austrian flood risk management system includes the introduction of multiple spatial scales with the introduction of inter-local scales. However, the policy does not differ between local or inter-local co-operations in the distributing of funding or final approval (i24aut). Here, the public administration does not distinguish between projects at local municipality of inter-local, rural or urban areas. Thus, the policy attempts to introduce a more equal-uniform system between the different areas. Interview participants highlighted that the national scale promotes a fundamental unit in contrast to the English policy. The result is that the system depends less on elite groups. Also the use of the CBA as a final selection tool shows a minor importance in the decision-making process (i24aut; i26aut). In particular, rural areas have a strong position in the actual flood risk management debate (i23aut), also when the different actors and

<sup>&</sup>lt;sup>28</sup> National: 40%; Regional: 40%; Local and third actors: 20%

stakeholders consider the interplay between the memberships of municipal and regional councils. The current relationship between national-regional-local in the selected study sites are summarised in table 6.19.

		State spatial projects: State spatial strategies:		
		(Question of responsibility)	(Question of power)	
Scalar dimension:	England	Decentralised part of state functions: Policy objective is to downscale responsibility and tasks, such as funding and decision-making practices towards localities; responsibility shift from national to local level	Privileged local scale by national state to engage actors and stakeholders to design flood alleviation schemes, in particular based on their financial contribution; multiscalar state power (EU, national and local), where national level has become less privileged	
	Austria	Decentralisation of responsibility from regional authorities, such as bottom- up initiatives. Further re- centralisation process from local to inter-local level, e.g. responsibility of maintenance; state administration and regulatory capacity are centralised	Inter-local scale is favour (not privileged) by national and regional authorities; inter- local scale is encourage to take certain tasks from local and regional authorities, such as administration, communication, negotiation; state power has become multiscalar (EU, national, regional, inter-local and local), especially in terms of regulation and funding; limited power transfer to inter-local co-operations	
Territorial dimension:	England	Customization: create new administration boundaries; change of administrative structure according to localities; customised defined responsibility- sharing (depends on local capacity); area-specific organisational relationship and administrative procedure Relative uniform administration and political	Concentrations: promoting schemes with the highest value from economic analysis or contribution of local actors and stakeholders (state and non-state); concentration flood alleviation schemes; splintering infrastructure in flood risk management Redistributive policies, such as funding or	
		structure and interactions based on the Austrian Water Law (1959). This include standardised rules for the creation of inter-local co- operations	implementation of flood alleviation schemes, are organised to equalised between the country in terms of new flood protection projects – no differences between inter- local or local based flood alleviation schemes	

Table 6.19: 0	Overview of	state spatia	process in	Austrian a	and England
10010 01201 0		beare oparia	p. 000000		and England

Source: own development

Administration centralisation means a concentration of policy definitions towards national authorities. In contrast, administration decentralisation refers to the shift of power, tasks, responsibility and policy definition towards a wide range of different actors, such as public administration units (regional or local ones) and non-state actors (Berkes, 2010; Brenner, 2004; Larson and Soto, 2008; Gualini, 2006). Further an equal spatial strategy foresees that the public administration includes national-wide approaches to define its investments and activities. The encouragement of a central approach includes the shift from a generalised public good approach towards a more single-individual club good approach to promote certain areas within the territory. In the politico-economic discussion, the literature observed a shift from equal towards a concentrated approach in the change of economic policy goals (from Keynesian welfare state towards more cluster-based regional policy) (Meijerink and Dicke, 2008; Brenner, 2004; Thaler, 2010; Swyngedouw, 2004). Influentially, different processes include more individual solutions for the different administration units. The arrangements depend on area-stakeholder specific circumstances (Brenner, 2004; Stoker, 2006). This multiple administration framework encouraged non-state actors to get involved in the ongoing policy discussion (Stoker, 2006) with the consequences of placespecific decisions, especially to secure the local needs and interests (Brenner, 2004). In a more plural-multiple form of scale construction the national government transfer tasks, duties and responsibilities from a national perspective towards higher or lower administrations levels (Martinez-Vazquez and Timofeev, 2008; Larson et al., 2007; Larson and Soto, 2008). In this form, scale is perceived as (relationships, interconnections and interactions) between the different actors over spatial-temporal developments (Massey, 1992; Sayre, 2005; Cox and Mair, 1989), through which power is manifested (Allen, 2009). This stipulates the political processes and interactions between the different actors as central in the production of scale (Cox, 1998; Smith, 1990; Sayre, 2005; Brenner, 2004; Swyngedouw, 2000).

# 6.2.New governance structure in flood risk management

In this thesis, I assume that politics of scale directly has influences and defines the governance structure (see chapter 2). All interview partners stated that the rescaling process changed the governance structure in the different study sites (see figure 6.21 and figure 6.22).





Source: own development



Figure 6.22: Governance structure in England

Source: own development

Old

In particular, the rescaling process increased the number of actors involved in the policy discussion. This had a strong influence on the interaction between the different actors. The relationship includes a wide range of different actors, such as technocrats, politicians, businesses (e.g. Ill-Walgau) and private householders (e.g. Morpeth or Cockermouth). In the Ill-Walgau study site (see section 4.2), for example, a large number of different non-state actors have an important position in the design of new policy instruments with or without consultation with the national government. The Austrian governance structure has a strong focus on technocrat relationships between the different actors. The English study sites include the change in the governance structure mainly for the planning, decision and implementation processes. The relationship includes only short-term and temporal frameworks with the key focus on funding and planning. The EA have taken over the responsibility for the implementation and future maintenance. The consequences for governance building are that

engagement with local scale with their actors and stakeholders is only given for the planning and implementation process.

### 6.2.1. New administration and regulative structure

Further, the rescaling process leads to a new administration and regulative structures, such as a new chairman and directors in the inter-local co-operation. All interview partners stated that the decision-making is mainly organised by project boards with selected key actors. The Austrian approach includes a formalised administrative structure also for the decision process. Each interlocal co-operation contract includes a regulatory part for the decision process. In particular, the contract incorporates tasks, which can be decided by the steering group and which tasks need a general agreement from all members. The assembly of members includes all those of the inter-local co-operation. Usually two meetings per year are held. In the general meetings the steering group has to ensure the internal accountability of the inter-local co-operation. Further, the key aim is to exchange information between the different members, discuss the general strategy and discuss general questions. The second level (steering group) includes a selection of members from the interlocal co-operation (elected by general assembly). Their key aim is the exchange of information between the different stakeholders (regional and local level), implementation of non-local flood protection measures (mainly flood storages, inclusive negotiation process with local farmers), financial support of local protection measures and negotiation with potential new members. This included a power transfer from general assembly to the steering group to increase the efficiency and effectiveness in planning and management process. Moreover, to achieve these tasks, there is a shift of power from general assembly to the steering group. The steering group meets regularly, approximately every six weeks with representatives from regional authorities. In these meetings they exchange information about the progress of implementation and define the next steps. The steering group is responsible for the implementation of the defined program of measures.

On the other hand, the English study sites include a more flexible-informal and non-statutory administration process and unit. A key reason is the lack of a nationwide regulatory approach. The

overall idea is to build a project board, where the different members design a partnership agreement (i40eng; i39eng; i38eng; i37eng). The EA has the leading role in the overall processes, also referring to the decision-making. The involvement of local actors depends strongly on their resources and availability (see also section 6.4).

### 6.2.2. New planning practices and processes

A central outcome of new scalar arrangements was the introduction of new planning practices. As discussed previously, the Austrian study sites include the creation of new 'sub-regions' in organising flood risk management. They include an up-scaling process from local municipalities towards the inter-local authorities (or also catchment / sub-catchment authorities). Simultaneously, a downscaling process from regional authorities towards the catchment authorities can be observed. The rescaling processes from both sides include a transfer of power, duties and responsibility from local and regional authorities towards the catchment authorities, with the outcome of the creation of new spatial imaginaries – new identification platforms for policy makers, stakeholders and citizens, such as the Ill-Walgau co-operation instead of administrative ('hard' political) boundaries (Boudreau, 2007; Allmendinger and Haughton, 2010; Heley, 2013). The new politics of scale in the selected countries introduced new state spaces and governance structures (table 6.20).

Table 6.20: Fuzzy	/ boundaries and soft	spaces in the se	elected case studies	flood risk r	management po	olicy
-------------------	-----------------------	------------------	----------------------	--------------	---------------	-------

Level	Authorities	Management plans	Fuzzy boundaries	Results from the case studies
Supra-national	EU		International river basin authority	Danube and Rhine (transboundary river basin district)
National	National authorities			
		River Basin Management plan; inter-regional catchment management plans	River Basin Authorities	Northumbria River Basin District (Morpeth); North West River Basin District (Cockermouth); South West River Basin District (Bridgwater); Danube River Basin District (Aist and Triesting-Tal); Rhine River Basin District (Ill- Walgau)
Regional/county	Regional authorities; County Council; LLFA			
		Inter-local catchment management plans; shoreline management plans; regional studies	Inter-local catchment authorities; community based approach	Aist Triesting-Tal and III-Walgau
Local	Local authorities; LLFA	Local flood risk management plan		Bridgwater, Morpeth and Cockermouth

Source: own development

As discussed previously, the Austrian study sites include the creation of new 'sub-regions' in organising the flood risk management. They include an up-scaling process from local municipalities towards the inter-local authorities (or also catchment / sub-catchment authorities). Simultaneously, a downscaling process from regional authorities towards the catchment authorities can be observed. The rescaling processes from both sides include a transfer of power, duties and responsibility from local and regional authorities towards the catchment authorities, with the outcome of the creation of new spatial imaginaries (Allmendinger and Haughton, 2010; Heley, 2013).

A second result was that the new flood risk management system in both countries enforced stronger networks (coalition). The coalitions between the actors at different scalar construction are an important condition in both countries. Based on the interviews, the results show not only the importance of network connections between actors at the same scale, but refer to the networks between actors, especially with members at local and national level. The local FAG in Cockermouth and Morpeth, for example, developed direct relationships with their national Members of Parliament, who intervened at a national level in their favour (i38eng). The Austrian networks show a strong orientation towards local and regional levels.

A second important network refers to the non-state actors, e.g. citizens or businesses (in both countries). The size of the networks differs in the selected study sites, from small number of actors (Bridgwater) to a large number of actors (Aist, Cockermouth or Morpeth). Furthermore, the networks can be distinguished between formal or informal organisations. Informal networks were identified in the Aist study site, where local municipalities collaborate with the local enterprises to ensure their needs and interests (see below). In this line, the external agendas of economic development and competition play an important role. Lobbying has an important role in the decision-making. The informal-lobbying activities have been especially identified at local levels, where private businesses encourage politicians to provide flood alleviation schemes to reduce the impact of future events. In Aist the motivation of local municipalities in the downstream area was strongly driven by the local enterprises, which were in turn strongly affected by 2002 flood (i14aut). Similar results were identified in England with the local FAG, which had a strong influence in the decision-making (see also section 7.2). Overall, the strong focus on economic growth enforces a higher competition between the different new state spaces (see for example Bridgwater). This includes an informal management and planning process between state and non-state actors (Haughton and Allmendinger, 2007).

A third category includes the delivery framework (Allmendinger and Haughton, 2009). The study sites show that besides official planning documents the different actors developed shadow strategies and planning documents. The Aist study site, for example, introduces in addition to the official catchment management plan (Aist regional study) shadow strategies like the Aist forest management plan. The interviews emphasized informal strategies as very helpful and useful, especially in the development of flood storages. The interviewees explicitly mentioned the difficulties in the negotiation process with the land owners. During the planning process the authorities have had to adjust the planning proposal various times, because land owners were not willing to sell their property and the Austrian legislation allows no compulsory acquisition in contrast to England (McCarthy et al., 2013). Specifically, the lack of compulsory instruments in Austria, means that the negotiation process require higher resource input in comparison to England as well as the higher need of flexible planning instruments. In most of the cases, regional and local authorities offered additionally benefits to achieve the consents from the land owners (i5aut; i20aut; i38eng). The outcomes from these flexible arrangements were later translated into the flood risk management plans. Additionally, I also identified also a 'bottom-up' planning approach. The rescaling process encourages the shift in the planning approach. In the Cockermouth study site, for example, the FAG has created a local flood risk management plan. Similar developments were discovered in the Aist study site, where the catchment-authority developed individual solutions and ideas (i11aut). The aim of these management plans are mainly delivery focussed and allows the actors a greater flexibility in the implementation process. The official planning procedure is different which includes a long preparation time and inflexible operations.

## 6.3.New network structure in flood risk management

Networks play a critical role in the overall scale constructions and interactions. The two selected countries showed the creation of new network structures along the scalar re-arrangements as a consequence of new scalar arrangements. Further, the rescaling process includes three main forms: up-scaling, downscaling and out-scaling (chapter 2) as well as not always initiated as a top-down

development, for example evidence can be found in the Aist or Cockermouth study site. In the Austrian study sites, the introduction of the inter-local co-operation mainly refers to an upscaling process from the local municipalities towards the inter-local co-operation, and a downscaling process from the regional towards the inter-local level (see also section 4.2.3). The up-scaling process includes the shift of responsibility and regulatory approaches (e.g. maintenance, communication and implementation) towards the inter-local co-operation level, for III-Walgau partly towards the regional authority, especially for the negotiations with local land owners (i20aut). The downscaling process allows the inter-local authorities to get more involved and engaged in the planning process. Nevertheless, only the Aist study site uses this opportunity (see section 6.4). For the English study sites, the introduction of partnership funding had a large impact towards the local scale creations. The downscaling process includes mainly changes in the planning and funding process. The local actors are using the opportunity to influence the planning and decision-making process. Nevertheless, the interviewees show that partnership funding required strong support from higher administration authorities, such as County Council and EA in terms of funding and technical knowledge. Furthermore, the countries include also an outscaling process, which refers to the greater involvement from non-state actors, such as citizens and business companies under the umbrella of FAG, in the current flood risk management discussion. The different FAGs have played a crucial role in organising the funding for the realisation of the local flood risk management proposals. Furthermore, both FAGs show a strong influence also in the planning process (see section 6.2.2). The social processes of scaling influenced the negotiation in the different study sites. In the Austrian study sites, the regional authorities and local actors used a trans-municipal level to implement interlocal flood defence schemes. On the other hand, the English study sites focussed their social process mainly on the local communities. The English partnership funding includes a hierarchical relationship between the different actors (vertical modes of co-governance). While in the UK the inter-communal co-operation in the field of flood risk management, for example, is given between some communities across the country. The focus of the policy is mainly based on local solutions. In

contrast to the Austrian results, here the public authorities try to implement inter-communal flood defence schemes. The core of these activities are focused mainly to share resources (human, software), less in joining common flood defence schemes. Nevertheless, Defra announced the introduction of catchment-wide management plans across the country in 1993, which are based on the principles of flood risk management from the Land Drainage Act 1930 (i30eng). During the 1990s, the efforts were oriented towards Shoreline Management Plans (Evans et al., 2002) as well as Catchment Management Plans for the large river basin in England and Wales, e.g. Thames, Humber, Severn, or Trent (i30eng). Yet the catchment management plans resulted in a more general planning document, instead of a strategic co-operation between the different agents. One of the central points of the catchment management plans has been to provide a summary from the different local flood risk management strategies (i30eng) rather than to coordinate the different local flood defence schemes in the catchment. The main explanations for this development are downstreamupstream conflicts, objectives of growth of each individual community (especially today in the ongoing financial and economic crises), or also the lack of political willingness and interest to cooperate between different communities in the catchment (i33eng; i30eng) - at local and national level. Meanwhile as first results show that the new partnership funding created in fact 'more space for non-state actors' (Deas, 2013: 73).

The Austrian inter-local co-operations are defined on a voluntary partnership approach, although the interaction is based on formalised (pre-defined) rules, such as organisation structure, foundation, contract design, defined under the Austrian Water Law 1959. In the Aist and Ill-Walgau study sites, the networks include in addition to the local authorities various non-state actors (business companies and engineering consultants). The outcome processes for governance include a wide range of different people involved in the decision-making practises (see also figures 6.21 and 6.22). On the other hand, the inter-local municipalities have built vertical networks with regional authorities. This relationship is also characterised by a technocratic nature, since civil servants have the main responsibility and work load in this relationship (Perkmann, 2007). The differences

between the two types of networks are that vertical networks are less focussed on the aspect of bargaining and more oriented towards the problem-solving (Rhodes, 1990). A second key vertical network is the collaboration with the regional farmer unions. The farmers union plays a central role in the negotiation process and agreement with local landowners (i5aut; i12aut). It has a close wellestablished relationship with the local farmers. Furthermore, the farmer union has knowledge about the legal impact of the flood storages for the farmers in terms of income and pension (i5aut). A further vertical network was identified in the Aist study site for its funding. The Aist co-operation uses European funding resources (Leader program) to realise its flood risk management plan. This relationship was established after the creation of a co-operation, as requested by the European Commission (i9aut). This also enforced the shift of power from national and regional levels towards international scales (Späth and Rohrbacher, 2013).

The three English study sites show broad local networks, especially in the two parish councils Morpeth and Cockermouth. These two examples have given a strong weight of non-state (civil society) actors (ibid.). The non-state actors established various local and horizontal networks towards private consultant businesses and national actors, like Members of Parliaments (i36eng). Additionally, the different study sites show further networks at local, regional and national level. The most important network is that with the County Council. In Cockermouth, the FAG established a local network with the EA. The relationship with the Cumbria County Council is given by the EA, where the relationship shows the technocratic nature of the co-operation. Here, the main workload is organised and managed by civil servants (Perkmann, 2007). The consequences were that these network relationships between civil servants (technocratic networks) are more stable in contrast to networks with politicians and non-state actors, stakeholders and citizens (Rhodes, 1990). In addition, both FAGs developed vertical networks mainly with Members of Parliaments and the NFF. However, the empirical data demonstrated that the quality of networks depends strongly on the interests of the local FAG. In Morpeth, for example, the local FAG uses the national level to intervene for the realisation of the local flood defence scheme. The main networks in the study sites are established

with technical experts from the public administration. The aim is to provide a strong technical support in the management process and -in a limited form- of knowledge transfer. Nevertheless, the leadership in the overall discussion are still organised by technical experts (EA and regional authorities). As a drawback, the networks often lack a formal conflict resolution mechanism. The networks play an important role in the participatory process in the planning and decision-making. First, the networks give different local actors a common voice to satisfy their interests and needs. Second, the local networks provide to the local actors technical knowledge and expertise to interact more successfully with the public administration. Third, the building of local networks replaces in part the missing conflict resolution mechanism.

The disadvantages of the involvement of local actors in the planning and decision-making process are the additional costs and resources required, because of the longer negotiation and communication processes (Irvin and Stansbury, 2004). For example, the Triesting-Tal negotiation process took more than 10 years to organise a common inter-local flood risk management concept. The main barriers were: first the parochial politics of the local authorities and second the regional BWV counteracted to the project (i7aut; i8aut). The outcome of the negotiation process was an increasing of transaction costs without improving the management concept. Similar problems are observed in the Aist study sites with the result of free riders in the study site (see section 4.2.4). The costs strongly depend on the number of actors in the process, e.g. costs of co-ordination, joint decision-making etc. based on the Scharpf's law (Birner and Wittmer, 2004).

Further networks were identified between the local actors with universities and private consultant groups, especially in the cases of Cockermouth and Aist. Both study sites collaborate with higher education institutes or consultant group for knowledge transfer and to increase the reliability of their local initiatives. The universities and private consultant groups have a significant role in gathering and distributing information and technical knowledge towards the local actors. For example, the Aist co-operation developed their forest management concept based on the

recommendation of the University of Natural Resources and Life Sciences of Vienna. Similar cooperations were established in the re-evaluation of the flood risk management concept in 2012. In this instance, the universities and consultant groups demonstrate strong trustworthy competences. Finally, the local actors use their vertical and local networks to ensure their interests. Here, especially the scale jumping allows local actors to change the power relationships. In most of the study sites, I identified more horizontal linkages than vertical, mainly when local actors are strongly involved in the planning and decision process. This includes a change in the decision-making power towards local actors, too (Allen and Cochrane, 2007).

## 6.3.1. Formalised network structures

The type of network organisations ranges from highly formalised co-operations based on legal contracts under public or private law to informal meetings and agreements, such as exchange of experiences and knowledge. The strategy of the inter-local co-operation is the use of mainly formalised contracts between the different members of the co-operation. The reasons for selecting formal arrangements are based on various considerations, like time framework, objective and aim, and legal framework. Further, the inter-local co-operation included new administrative structures, inclusive juridical structures (Medeiros, 2011; Wirth et al., 2008). The consequences are that the highly formalised co-operations are strongly regulated between the different members. Here the local actors sign legal contracts (under private or public law) to ensure the sharing of tasks, duties and responsibilities between the different members. In terms of organisation, the co-operation contracts include various defined standards and rules, e.g. annual numbers of general assembly meetings, sharing costs between the members or voting systems. In between these two extremes, there are various different approaches, like common arrangements between local authorities, or sharing resources, or development of common joint projects, inclusive of common funding. The high degree of formal arrangements provides legal security for all involved members in the co-operation process as well as to ensure accountability and legitimacy. The key work is the purchase of land, defining and signing of formal contracts with construction companies and farmers

The English interviews also showed mainly formal arrangements between the different actors. The highly formalised co-operations are mainly to ensure the legal requirements (EA, 2012b) as well as the aspect of accountability (i40eng). The essential points are that the contracts define each actor's rights and duties, especially relating to their role and deliveries as well as the financial contributions of each agent. Furthermore, the contract ensures and defines all deliveries by each stakeholders as well as their role in the process. Under these regulatory conditions, the various actors build a formalised project board (similar to the 'old' system) to plan, decide and implement the flood defence scheme. In the Bridgwater study site, the partnership agreement contract is more complex compared to the other two study sites. Nevertheless, the partnership approaches includes loose and quite flexible relationship, such as no regular meetings between the different actors and stakeholders, no rule about representations without a clear labour division or responsibility-sharing between the members of the co-operation. This type of relationship has been understood as a flexible construction to arrange common activities between the different actors. The regulation process however as well as type of sanction is quite weakly organised and defined. These observations indicate that the partnership approach is mainly organised by funding and not in a broader sense of a regulatory meeting board, such as the Austrian case study. As observed in the empirical results, the partnership funding changed the regulatory (formally and informally) duties and norms.

Overall, the politics of scale in the English's study sites acts mainly on its political-artificial boundaries, e.g. the town council. This has a strong implication for governance structures. Governance, which are defined by people and their interaction (see chapter 2), include a much lower number in comparison to the Austrian case study, where the national and regional authorities have to collaborate with more than one local authority (see also table 6.20). As further outcome is a different type of partnership approach, instead of a strong horizontal partnership approach (Austrian case study), the co-operation is mainly focussed on vertically relationship between national and local level. The institutions and organisations are strongly based on local units rather

than at cross-boundary policies (Cochrane, 2012; Harrison, 2012). I found a strong competition between the local authorities for national funding does exist that in the flood risk management debate at regional and local level (i30eng). On the other hand, the English results show multi-scalar approaches, mainly in the planning and the financing process in flood risk management. The two study sites Cockermouth and Morpeth show the interaction of multiple actors.

### 6.3.2. Types of networks

The study sites show various types of relationships and network power (Allen, 2009) between different private-public actors and stakeholders (table 6.21). The following table summarises the main relationships between state and non-state actors and stakeholders present in the study sites (see following quote).

Participant: 'They have now: a tax advisor for the financial problems, they have an external director and they have a private planning office' (National Authority, Austria).

While examining the interviews, the participant described a wide range of different relationships between the main actors and stakeholders in the study sites. Further, each study sites show more than one type of relationship, based on the interests, objective and tasks. The Cockermouth study site, for example, has various networks with different actors and stakeholders, such as with private consultant groups (market relation) to develop business appraisal and informal relations to Member of Parliament to ensure their needs and interests (scale jumping). The distinction between the mechanism and relationships are 'ideal types', which in reality are generally more complex (Tödtling et al., 2006; 2012; Trippl et al., 2009). Tödtling et al. (2006) use two main dimensions to characterise the relationships and the mechanism within a co-operation. The first dimension includes the interaction between informal and formal relationships, such as state-non-state co-operation in the Aist study site or relationship between FAG and national and local authorities in the Morpeth study site (see also section 6.3.1). The second element includes the mechanism of interaction (static or dynamic). A static mechanism includes the knowledge transfer from an external source (e.g. consultant or academic institute) to the local actors, such as inter-local co-operations or local FAG

whereas a dynamic process has been seen as a more interactive approach within the co-operation, especially for the formal and informal networks of each actor.

	Static	Dynamic			
Formal	<ul> <li>Market relations (1)</li> <li>All study sites with private consultant or research institutes</li> </ul>	<ul> <li>Formal networks (2)</li> <li>Aist, Triesting-Tal, Ill-Walgau, Cockermouth and Morpeth: collaboration contracts</li> <li>Austrian study sites: shared resources of facilities (e.g. administration)</li> </ul>			
Informal	<ul> <li>Knowledge transfer/spillovers (3)</li> <li>Aist: recruitment of a specialist (executive director)</li> </ul>	<ul> <li>Informal networks (4)</li> <li>Aist with Regional Highway authority</li> <li>Cockermouth and Morpeth: FAGs with Members of Parliament</li> <li>Morpeth: FAG with EA and Northumberland County Council</li> <li>Bridgwater: Sedgemoor County Council and the EA</li> </ul>			

## Table 6.21: Summaries types of relationships in the study sites

Source: adapted from Tödtling et al. (2006): 1038

Market relationships are mainly present between the local FAG and inter-local co-operation. This consists of mainly 'buying' knowledge and expertise from consultant groups or academic organisations or of 'buying' service tasks. Examples of this in the study sites are construction works or supports for the negotiation process with local land owners (see III-Walgau study site). Most study sites displayed some evidence of a market relationship. In particular the Austrian study sites show a close relationship with consultant groups and academic institutes as well as private construction companies. Main reason is that the inter-local co-operations have the responsibility for the implementation and the future maintenance of the assets, which is different in the case of England, where the EA is responsible for the future maintenance of the assets. Furthermore, the Austrian study sites show knowledge transfer, mainly technical knowledge such as types of flood alleviation schemes, return periods, type of flooding and impact etc., between the regional authorities towards the inter-local co-operations (i9aut), especially a one-way process (mainly hierarchical organised). Among the market relations only the Aist study site recruited an external specialist in monitoring

and supporting the management process. The overall goal of the executive director is to support the local politicians in the management process. Furthermore, this person is involved in the negotiation process with local land owners and construction companies. The advantage is the technical and knowledge support for the local policy makers, with the consequences that inter-local co-operation developed individual concepts for flood risk management (i9aut). This step includes also a step forward in the professionalism in the inter-local co-operation and a better position in the management process.

Networks (formal and informal ones) are very important in the overall management process (see also section 6.2.2). In summary, the different study sites show a large variety of different interconnections with local, regional and national state and non-state actors. Austrian's study sites include a highly formalised network system between the different members in the inter-local cooperation (see section 6.3.1). Apart from the formal contracts, the three study sites also include sharing information, skills, personnel (mainly administration structures), software programs and data. The empirical results also include informal interactions among the different actors. The Aist study site shows an informal linkage within the Regional Highway Authority. The network is mainly based on a common interest and informal financial agreements. In Morpeth and Cockermouth key links are with the national authorities (see section 4.2.10).

#### 6.3.3. Network size and membership

Membership is a central factor to assess and to understand the policy network. The size of the networks strongly influences the grade of formalisation, organisation of structures, resource inputs and policy outcomes (Hutter, 2014). The empirical results show that larger networks tend to a higher grade of formalisation compared to small networks (see section 6.3.1). Similar outcomes are found in the organisation structures; also here larger networks tend to develop steering groups. In particular, the Austrian study sites mainly constructed steering groups to increase the efficiency and effectiveness within the network in terms of defining and delivering policy goals. On the other side, larger networks create higher transaction costs, particularly due to increase of communication

processes between the members of the co-operation (see also section 7.1.4). Finally, the network size also influences the expectation of the output (ibid.). However, the empirical results manifested no evidence that larger networks improve the flood risk management plan (see also 7.1). The networks different study sites were outlined in table 6.22.

	Study site A: Triesting-Tal	Study site B: Aist	Study site C: III-Walgau	Study site D: Bridgwater	Study site D: Cockermouth	Study site D: Morpeth
Number of communities	13	29	12	1	1	1
Number of members (co- operation)	12	27	20	2	5	4
National/regiona l organisations	WLV/BWV	WLV/BWV	BWV	EA	EA	EA
Initiative	Top-down	Bottom- up/top-down	Top-down	Top-down	Bottom-up/top- down	Bottom-up/top- down

#### Table 6.22: Overview of the characteristics from the study sites

Source: own development

First, the key differences refer to the number of communities and involved members. The study sites include one to 29 local authorities in the partnership approach. Moreover, most of the examples demonstrate linkages to non-state actors and stakeholders, such as small-business companies, utility companies or private householders. On the other side, the Aist and Triesting-Tal study sites show the challenge of non-members (free riders) within the inter-local co-operation process with the consequences that the non-members get the benefits from the co-operation without contributing (financially) to the costs (Baumol, 1952). Second, key differences between the three study sites are the involvement of regional organisations in the inter-local co-operation. A third aspect refers to the initiation process. The Aist, Cockermouth and Morpeth study sites, in contrast to the other three study sites, includes a strong bottom-up approach in the development of the inter-local co-operation. The main reason is the strong engagement of local grassroots organisations and policy makers in decision-making practices.

#### 6.3.4. Interests

Interests play a crucial role in the policy process and policy networks, also because all involved actors follow different ideas about how to solve a certain problems. Therefore, main attributes are (1) conflicts between the different agents or groups and (2) questions of power between the different actors (Lukes, 2005). Interests can be understood as a dependent variable (normative judgements with a moral and political character (ibid.)) – depending on the individual socio-economic structures. Following this line, interests are highly dependent on social construction and should not be seen in absolute terms. The consequences were conflicts between the different individual interests. In respect to flood risk management each stakeholder group follows different interests, such as structural flood defence measures (e.g. dams and dykes – also for legitimacy reason) to reduce the vulnerability of a flood event, or in general to build new and more houses in floodplain areas. In the same time, with the implementation of WFD, European countries have to ensure the good ecologic status of each water body until 2015, e.g. re-naturation of river banks and restrictions in land use management.

Based on the work published by Posthumus and Morris (2007), this thesis classified actors, stakeholders and citizens into five groups of interests (figure 6.23):

- Maintaining economic livelihood,
- reduce probability of future flood events,
- reduce future damages,
- protection and creation of habitat areas and
- reduce flood risk to people.



#### Figure 6.23: Stakeholder overview in the Austrian case study

#### Source: own development

In reality some of the stakeholders may follow more than one interest. As in Austria an important agent is the landowner and the residents who are living in risk areas or have a property in risk areas. This group is mainly affected by any flood defence discussion and who are interested to have a flood defence scheme to protect their property or to restrict the land use by the stakeholders from the policy or operational responsibility level. The involvement of landowners strongly depends on their interests, but in general they have a high power to stop or to change flood defence schemes in Austria, because the lack of compulsory purchase (see section 6.2.2). In the English case study, the interviews emerged that key players are local grassroots organisations, mainly FAG (see figure 6.24).



Figure 6.24: Stakeholder overview in English case study

Source: own development, based on Posthumus and Morris (2007).

FAGs are organised by private households or local businesses. The different FAGs are supported by the NFF with the aim to support private households in the recovery phase after a flood event. The interviewees mentioned the important role of the FAGs also in the assessment and mitigation phase. In the local flood risk management discussions the FAG has had a strong role in the overall debate. Today, the FAGs can be seen as a lobby and pressure group, depending on its members. Their objective has often been the implementation of high flood defence standards in their community. Other important stakeholders are private businesses, such as developers. In the study site of Bridgwater, the private businesses are the primary funding resources for closing the funding gap. The results show that private businesses play a critical position in respect to partnership funding policy (further discussion see section 7.1.3).

### 6.3.5. Power

The figure below summarises the results from the Austrian case study.



#### Figure 6.25: Power and interests positions in Austrian case study

#### Source: own development

The key players in the three study sites are the WLV and the BWV. Both organisations are leading flood risk management policy. They also have the knowledge, expertise and power to influence the current policy discourse. The local authorities also have an important role to play. However, their interests and contributions depend strongly on, whether they are involved in the decision process or not. Nevertheless, the result shows also here a broad range between the three study sites. The engagement and influence of inhabitants at risk depends on the individual awareness (risk and environment). If the inhabitants show high risk awareness, they are more likely to participate in the process. It was observed from the interviews, that participation engagement often starts with concerns about the flood defence measures and not solely based on the flood risk (i12aut; i13aut; i15aut). However, the interviews mentioned a general low stakeholder engagement in the three study sites, with some exception of in the Aist study site (see section 6.4). Further central stakeholders are business companies (i10aut).

Participant: 'The two companies [name, name] were strongly affected by the 2002 flood event. However, they have received massive support in terms of financial compensation. So, their pressure were limited, but still a strong motivator for the major to be active' (Regional Authority, Austria).

This quote indicates that generally, businesses demonstrate a high interest and influence in the policy decision-practice until their needs are fully satisfied. In the study sites, private companies show a strong pressure after frequent flood events or restrictions in business investments. In particular, in the Aist study site private businesses had a leading role in the creation of the inter-local co-operation (see section 4.2.4). However, they were not a member or did not influence the catchment-management plan. Their key purpose was the implementation of flood alleviation schemes to reduce the vulnerability in the catchment, specially to protect their plants. Generally, the companies have a strong power in the region, because in most cases they are the largest employer and taxpayer in the region. A similar development can be observed with the implementation of flood storage on farmland. Farmers have the power to block the implementation process. However, their interest depends strongly on whether they are directly affected by flood defence measures. In the study sites, the inter-local co-operation have to re-design frequently the catchment-management plan, because the lack of support from farmers; use of soft spaces in planning processes (Allmendinger and Haughton, 2009). In sum, the power and influence of farmers is very high (see following quote).

Participant: 'With the small flood storage projects we had only a few problems, especially with some detail questions, but in the whole it was not the big problem. The approvals we had relatively quickly in an open and transparent procedure. We didn't use more than three meetings: prior information, an update meeting and final meeting with the clear message if there is consent or not. With some farmers we singed directly on the table, with some others we had further negotiations and then we got the necessary signatures. With the large flood storage projects we followed the same process, but

in the final meeting we had three agreements and five against the project, and with this number you cannot realise the project. We accepted the final decisions, we weren't angry with anyone, but the expectation was that all was dead now, no it isn't dead, we will think about how to organise and to deal with this situation' (Local authority, Austria).

Participant: 'obviously there were variants already planned, but there is always that one plan does not come through and there we have to adapt it, but until now we have still managed in a slimmed-down version to achieve our targets (Regional Authority, Austria)

Based on the figure 6.26, literature and from the interviews, this study distinguished the different

stakeholders into four categories of the power-interest matrix.



#### Figure 6.26: Power and interests positions in English case study

## Source: own development

From the interviews it emerged that the key players are the EA and partly the RFCC and Defra. Defra has an overall key position, because it defines the current policy direction and provides the funding scheme from the national government. The RFCC has a strong power in the current flood risk management discussion; mainly the committee has the power to reject flood defence schemes. Furthermore, they provide a small contribution in the overall flood defence schemes (with the local levy). The EA has an important role in the overall discussion, i.e. the negotiation with third parties and technical expertise. Apart from this role, the EA is a key actor, because it owns the flood defence schemes and is responsible for maintaining them. From the different study sites, the results show that different stakeholders are involved in the discussion or have different positions in the power-influence matrix. An important agent is the FAG, especially in the Morpeth and Cockermouth study sites. In both study sites they show a high interest in the development and realisation of a new flood defence schemes for its community. For example, the Cockermouth FAG organised external funding of £400,000 to close the funding gap (i36eng). In the case of Morpeth, the FAG plays a dominate role in the discussion because of its network and interactions with different public authorities (locally and nationally) (i38eng). The formation of the FAG is often based on recent flood histories, especially as a reaction of frustration in well educated and wealthy areas (i36eng; i34eng). 'Being heard' is a central problem in flood risk management (see following quote), where local level have the feeling of not been listen by policy makers (similar problems can be found in spatial planning, see also Allmendinger and Haughton, 2012).

Participant: 'And again we didn't see anybody, there was no help, and that's when I decided, that's it, not standing for this anymore, so I created the [name]f because I felt we were unique in our problems and we needed to highlight' (Local actor, England).

The role of the land owners is a key aspect in the Morpeth case. The landowner forced the EA to change the 'original' proposal with the result of various restrictions of the EA in the use of the flood storage and the area across the dam (i42eng). The public authorities, especially the County Councils have a strong position in the actual policy discussion. In the two study sites (Morpeth and Bridgwater) the key leaders in the policy discussion are the EA and County Councils. The interviews showed that the position of Sedgemoor Country Councils have a stronger role in the overall discussion, since there is no local FAG. In the Cockermouth study site, the County Council had a less important role, because it was replaced by FAG. Here the County Council provides a large contribution to the realisation of the flood defence scheme. Analysing the interviews from the

Cockermouth study site, the County Council has more a passive role in the overall discussion. The role of private businesses largely depends on the case study. For example in Cockermouth study site, the private businesses show a higher interest and contribution in the local flood risk management debates. A large number of businesses take part of the Cockermouth FAG. One reason is the business structure, mainly local small-medium businesses, as well as the socio-economic structure of the town council (i36eng). In Bridgwater, private businesses have an important role in the current local flood risk management debate (i40eng). However, the literature already indicates a lack of a 'leading agency' in flood risk managements in England, especially relating to the maintenance of flood defence schemes (Coulthard and Frostick, 2010).

## 6.4. The governance networks in the flood risk management process

Scalar arrangements have not only a direct influence tothe governance, but also influence the task and responsibility-sharing between the different scales. To apply the theoretical discussion from chapter 2 in flood risk management this thesis analysed the changes in governance as the change of level of activity and self-realisation between the different key actors and stakeholders at national, regional and local level in the policy decision-making practices. This has a direct influence in the network relationship and level of engagement and interacting between these different groups. By analysing the literature (Fairtlough, 2007; Ingstrup and Damgaard, 2013) and the empirical results I distinguished four key characteristics to analyse the changes in the role of various actors and stakeholders in the new scalar arrangements (table 6.23) under three ideal types of organisations (hierarchical and heterarchical structure and responsible autonomy), which will now be discussed in more detail.

*Hierarchical structure*: regional authorities play a crucial role in the organisation of the partnership approach. A key aim for inter-local co-operations, for example, is to develop the co-operation and initiate the project proposals for flood defence schemes (develop a project appraisal, ensure funding from regional and national sources, organise all approval needs for the realisation of the project). During this stage the national and regional authorities are the project leaders, who also define the
conditions with the members, e.g. costs sharing between the members. The focus is to find and integrate potential members (state and non-state actors) into the partnership approach. Building trust is important, especially between members who have never collaborated in the past. The overall tasks are to organise seminars and meetings, especially to communicate the potential hazards and risks, define problems and find common objectives and measures. Furthermore, key tasks are to organise and ensure financial support from regional and the national authorities. Since the regional authorities are foreseen as project leader and manager, the key requirements are managerial skills at regional level.

*Heterarchical structure*: there is a shift of tasks and competences from national and regional authorities to the local actors and stakeholders. The key role of national and regional authorities is to build and ensure the relationship between the members, especially to avoid a disintegration of the co-operation. An important aspect is the shift in relation to the aspect of trust (from trust building to trust expansion). Ingstrup and Damgaard (2013) defined this shift as an important growing phase in the inter-local co-operation (from actor bonds to social bonds and later to professional bonds). The role of the national and regional authorities shifts from project initiator to project supporter. The shift in competences refers mainly to aspects of soft skills, e.g. communicator or problem solver. Relating to tasks, the national and regional authorities are focussed on the project monitoring and controlling instead of project planning at this stage.

*Responsible autonomy*: consists of an additional shift of tasks and duties from national and regional authorities to the members of the co-operation. The key focus of national and regional authorities is to support the co-operation. However, their key tasks are still project funding and controlling (financial and implementation).

#### Table 6.23: Level of activity and self-realisation

National and regional authorities role	Hierarchical structure (Type 1) • Networker and project initiator and manager	Heterarchical structure (Type 2) • Relationship builder and project manager and supporter	Responsible autonomy (Type 3) • Integrator and project supporter
National and regional authorities focus	<ul> <li>Project leading</li> <li>Framework conditions</li> <li>Trust building</li> <li>Locate and integrated new actors in inter- local co-operation process</li> </ul>	<ul> <li>Trust expansion</li> <li>Project support</li> </ul>	<ul> <li>Supporting</li> <li>Trust exploitation</li> <li>Project support</li> </ul>
National and regional authorities competencies	<ul> <li>Communicator</li> <li>Networker</li> <li>Technical expertise</li> <li>Managerial skills</li> </ul>	<ul> <li>Analyser</li> <li>Problem solver</li> <li>Communicator</li> <li>Technical expertise</li> <li>Networker</li> <li>Managerial skills</li> </ul>	<ul> <li>Communicator</li> <li>Technical expertise</li> <li>Managerial skills</li> </ul>
National and regional authorities task	<ul> <li>Project planning and appraisal</li> <li>Project monitoring and controlling</li> <li>Funding</li> <li>Branding</li> <li>Seminars and network events</li> </ul>	<ul> <li>Project monitoring and controlling</li> <li>Funding</li> <li>Seminars and network events</li> <li>Branding</li> </ul>	<ul> <li>Project monitoring and controlling</li> <li>Funding</li> <li>Seminars</li> </ul>

Source: adapted from Ingstrup and Damgaard (2013): 569

The EA is the key actor in the English flood risk management discussion. In the past years, the EA changed their policy discussion. The previous EA approach foresaw a strong top-down approach in the decision process with no or minor engagement by local agents (i31eng). Main reason was the fear to lose power with the consequences of no pro-active relationship and engagement with the local scale. However, the new scalar arrangements changed not only the position of local scale but also of the national scale. The engagement of local scales forced national authorities to co-operate with local actors and stakeholders (i38eng; i37eng; i36eng). In Austria, the regional authorities (WLV and the BWV) play a crucial role in the development and management of the inter-local co-operation process. Firstly, the national and regional authorities have the technical expertise, the regulatory power (e.g. they approve the flood defence scheme) and the financial resources to realise flood defence measures. Based on the characteristics of different levels of engagement (see chapter

7), the study is using a conceptual framework to understand and to evaluate the roles of national and regional authorities over the different stages and maturity of the inter-local co-operations.

The national and regional authorities have a dominant role in the flood risk management process in the three different study sites in Austria. The inter-local co-operations are generally passive actors in the regional flood risk management process. Key reasons are the lack of technical knowledge and experiences as well as missing interest. The emphasis of the inter-local co-operation is primarily the operational implementation of the catchment-management plan, e.g. negotiation with farmers and signing contracts with construction companies. The Aist study site, on the contrary, shows a more active role in the flood risk management discussion, because of the high interest of the local leader (i12aut; i15aut; i14aut). However, the steering group has the key role in respect to strategies or processes. These efforts are strongly supported by the regional authorities, e.g. through technical knowledge transfer or financial support. Nevertheless, the regional authorities are still dominant in the processes. In summary, two study sites (III-Walgau and Triesting-Tal) are shown to be at an early stage in the inter-local co-operation life cycle framework. In contrast to them, the Aist study site can be described in the second stage. However, the regional authorities have still to provide an enormous input into the discussion. Behind the technical knowledge, their focus is on the problem solving, communication and networking. I observe a change in the role of the regional authorities from engineer experts to project managers.

Figure 6.27 provides an overview of the results from the two countries. In the Aist and Cockermouth study site, I observed sign of a bottom-up management system. In contrast to the other study sites, where the national or regional authorities provide the flood risk management plans – top-down management system. Additionally, an increasing maturity within the III-Walgau study site emerged during the interviews, especially for the next years. The Triesting-Tal study site, in contrast to the other two study sites, shows a low degree of maturity of the inter-local co-operation. In the English case study, the two FAG communities show a high degree of interests, experience and knowledge in the ongoing policy discussion, especially the Cockermouth study site.

183



Maturity

# Figure 6.27: Role of national and regional authorities and local actors and stakeholders in partnership approach Source: own development

In summary, four of six study sites are characterised by the early stage (type 1) of the partnership life cycle (Martin and Sunley, 2011). The lack of self-responsibility is the result of a strong role of the national and regional authorities.

The Cockermouth partnership funding scheme is characterised by high level of self-realisation, management and initiation by local actors. Local actors show a high interest, knowledge and resources in the developing of a flood defence scheme in their town. In particular, the key local actors are proceeding through the local FAG. In the past years, they changed their relationship and co-operation with the EA – from a lobby group towards an equal partner. This point is essential in a partnership agreement. Although the EA is still a key player in the discussion, for example, in the negotiation with different actors like Cumbria County Council. Challenges were the lack of trust and concerns between both actors (i37eng; i36eng). The result was a common negotiation and conversation about particularly flood defence schemes in the town between the EA and local citizens. In the planning process, concerned citizens were strongly involved in the design of the flood defence

scheme. The EA proposal initially included to implement a 1.6 metre wall close to the river to protect the households at risk. However, the Cockermouth FAG enforced the EA to change their original plan to implement a self-raising barrier and/or windows in the wall (i36eng; i37eng). The results showed that the total costs of the scheme increased after the stakeholder engagement process.

As identified from the data collected here, the FAG in Cockermouth replaced the role of the local authority in the interaction with EA. In fact they closed a political vacuum at the local level, because nobody identified the responsibility to engage in the flood risk management discussion. In particular, interviewees showed that a key reason for this was the citizens' dissatisfaction with the work and results of the EA and local authorities. This dissatisfaction tended the local actors and stakeholder to get engaged in the interaction with the EA (i36eng).

Participant: 'Payment for Outcomes, it's a different process for us to go through, often you've got, it's a different skill set, building up a funding package for a scheme, than perhaps the traditional project management that we were doing because a lot of that was around, if you like, people with an engineering type background. So at the moment it seems to be we're working hard to secure what we can get and we're organising ourselves differently, using people with the right sorts of skills because some are much more effective than others to have those sorts of negotiations really' (National Authority, England).

From this quote, despite this new co-operation approach the EA shifted away from being the engineer (designing and building flood defence schemes) towards more project management tasks. Above all, the consequences for the EA were a stronger focus on communication with local actors and stakeholders as well as stronger focus on project management tasks, such as budgeting and administrating the contribution of the different stakeholders. In line with Jackson (2001) and Clifford (2013), partnership funding needs a new organisational culture, skills, knowledge and competences.

The interview data for the Morpeth study site showed a shift of the relationships, activities and selfrealisations among the different agents. First, the 2008 flood event and its consequences, e.g. insurance problems, increased the interest of the local citizens. They created the Morpeth FAG shortly after the flood event. Second, the partnership funding scheme opened the door towards the local authorities. The actors were partly involved in the planning and decision process of the flood defence scheme to reduce the risk of future flood events from main rivers. Nevertheless, the different actors (Northumberland County Council and FAG) played a minor role in the design and planning process in Morpeth. Main reasons for this were that the businesses case and proposal of the defence scheme were already designed under the 'old' scheme. The key driver for the citizens to get involved was to satisfy the requirements from the insurance industry. Flood insurance plays a central role in the English flood risk management system (Lammond et al., 2009); not only in terms of compensation. A central policy is the needs of insurance cover where obtaining a mortgage and therefore has implications for selling the property (i37eng). The availability to cover depends on the coverage of acceptable risk, which is distinguished between three main groups:

- Properties (private householders or businesses) in areas with a flood protection standard of
   1:75 or less are available; the premium depends on the level of risk
- Properties (private householders or businesses) in areas with a flood risk of 1:75 or more, were only covered, if the national authorities plan a new or improve existing flood protection standards within 5 years
- Properties (private householders or businesses) in areas with a flood risk of 1:75 or more, cannot guarantee to be covered by an insurance cover, if the national authorities plan no further flood alleviation schemes in the area (ABI, 2005).

Furthermore, an important aspect was the high level of knowledge and expertise by the local actors: e.g. Northumberland County Council has four engineers in their team (i43eng). However, their influence in the planning process was mainly concentrated on the re-design of the flood storage scheme, especially for the change of the locations (i42eng; i38eng; i43eng). The key reason for their limited influence was the already existing flood risk management appraisal, which was designed in the beginning of 2000s but never achieved the national funding requirements to realise the scheme. Further the involved actors and stakeholders wants not spend too much time to re-design the flood alleviation scheme. Also in contrast to the Cockermouth example, the Morpeth FAG expresses lower technical knowledge and expertise (i38eng). Indeed, the main interest was to realise the flood alleviation scheme as soon as possible. An advantage in the strong cooperation with the different local actors was the quick response of the planning permission. In summary, the research study showed a co-operative interaction between the EA towards local stakeholders in Cockermouth and Morpeth. The example of Cockermouth confirmed coproduction decision-making practices, where the local scale influenced the design and implementation of management plan (Albrechts, 2013).

On the other hand, in the Bridgwater partnership funding scheme, the Sedgemoor County Council and the EA are key actors in the overall discussion, planning and management. Similar to the other two study sites, the town council played a minor role in the ongoing debate in Bridgwater. The town council shows a passive role in planning, management and implementation process as well as in respect to the definition and introduction of the 'roof tax'. A key reason is the lack of local capacity to engage in the decision-making practices, such as a local FAG as a megaphone. One explanation for the absence of a local FAG is the lack of recent flood events in the town, which in general influence the risk perception and the interest in flooding (Kuhlicke et al., 2012). Another interpretation is that the creation of local grassroots organisations, such as FAG, needs also socio-economic structures, e.g. cultural capital, income and interest. In addition, the Bridgwater town shows different socioeconomic situation compared to the other two study sites. First, the town shows a much higher level of deprivation in comparison to Morpeth or Cockermouth. For example the regeneration plan includes 30% affordable houses (approximately 1,600 new residential properties). Second, the level of knowledge, experiences, wealth and contacts to different Government organisations is widely different. Morpeth and Cockermouth show a wide network with different policy makers from different Government level. Furthermore, both towns show a large number of well educated, middle class, citizens. These aspects are less present in the Bridgwater town with the result of a very low involvement of the citizens and businesses in the discussion. For example, the County Council started a consultation process (six weeks) about the implementation of the 'roof tax' to contribute to the future flood defence scheme. Based on the consultation process, the Council considered the

minor comments and adapted the original scheme (i40eng). The interviews showed that mainly developers contribute in the consultation process, but not by the citizen of the town (i39eng). The relationship between the Sedgemoor County Council and the EA is characterised by a relative low local engagement and management by the County Council to take over tasks and responsibility from the EA. The Sedgemoor County Council understands the problems, but has not shown own initiatives in the developing of the flood defence scheme or question of budgeting and funding. The EA strongly leads the overall discussion, such as the design of contribution (i39eng; i40eng).

Participant: 'Under the old system we'd get 100% funding so we'd do a project appraisal and develop a business case and we would decide what the best solution was and, yes, we'd talk to the residents but most of the time we're kind of saying this is what we think is the right solution for your community, and they'd agree or disagree and sometimes, if they disagreed, we'd go ahead if we thought it was the right thing to do, but we'd try to get their buy into it and we'd deliver that scheme. Now because we can only fund so much; and some projects do get 100% funding from us, but especially when they don't and partnership funding really kicks in, we need to do a lot of work around obviously getting the extra funding in, a lot of working much more closely with those partners because if they're putting the money in they have a right to say what comes out of it, and also the outcome can be different. So the outcome can be much less or much more because they have to put more or less in' (National Authority, England).

This quote indicated under the previous system, the EA focussed their work with the community mainly to show them their best solutions, strongly top-down management system. The management of flooding has been a strong top-down bureaucratic approach, with a strong focus on box-ticking indicators technique. The new schemes encourage a changed approach of the EA. Today, the EA negotiate with the different actors more intensively and they also take the local views, ideas and concerns into consideration. At the same time, the local actors have to take part in the discussion process. In fact, the local actors have become more responsible and active in the flood risk management discussion, especially for the design of the new defence scheme. Similar results were found in the Austrian study sites, where only the Aist example show a strong engagement in the design and implement local ideas, strategies and concepts.

The Triesting-Tal study site is characterized by a passive role in the management process. The key actors are the regional authorities (WLV and BWV). Both are the important drivers in the strategic planning and initiation of new projects. The key tasks of regional authorities are primarily the

planning and the implementation of flood defence measures. Additionally, the key task is to ensure the funding for the realisation of the flood defence schemes. The regional authorities have to monitor and control the implementation process. An important aspect is the organisation of meetings, seminars or events with local actors in order to increase their risk awareness and explain the importance of the inter-local co-operation and the trust building between the different members of the co-operation. The local actors are responsible for the implementation of flood strategy (operative responsibility). A key observation from this analysis is that the inter-local co-operation focussed mainly the implementation of pre-defined tasks provided by the WLV and BWV. The main work refers to the negotiation with farmers about compensation agreements or purchase of farmland. From 2002 to mid-2000s an important actor was the district level, especially for bringing together the different actors and to promote the inter-local flood risk management strategies. Currently, the district no longer has a key role in the inter-local co-operation because of the change of interests. Also, the regional authorities, especially the WLV, have played a very important role in the identification of new members and their joining the co-operation (especially in the case of Furth an der Triesting and Altenmarkt an der Triesting). The co-operation secretary and also the chairman of the co-operation are based in Leopersdorf. The secretary is financed by the members of the Triesting-Tal. Their key tasks are the organisation of the ongoing administrative work.

The Aist study site can be generally described as a partial shift of the activity and self-realisation from regional authorities to the Aist co-operation. Although the key players are still the WLV and BWV (regarding planning, initiator, monitoring, controlling and instructing local actors), the Aist inter-local co-operation introduces its own ideas into the projects. The steering group plays an active role in the overall discussion with regional authorities in respect to the implementation process, but also participates in the planning process for defining solutions and ideas. However, the technical expertise is still controlled by regional authorities and external consultants. They are responsible for the technical solution and the feasibility of the ideas. At operational level the inter-local cooperation is responsible for the implementation of inter-local flood defence schemes, e.g. flood storages. Here, the regional authorities provide a strong support in the prioritisation of flood defence projects. In summary, the regional authorities are important drivers of the general process. The inter-local co-operation is focused on the organisation of social activities (e.g. meetings, seminars and events) between the different members as well as with local citizens to encourage the stakeholder to engage in the overall process. At the beginning, the regional authorities led this task, but then shifted it in co-operation with the foundation. Furthermore, an important aspect refers to the branding of the inter-local co-operation (the key actor is the Aist inter-local co-operation). The regional authorities played an important role in identifying and incorporating new members into the inter-local co-operation. However, this task has been mainly shifted from the regional authorities to inter-local co-operation, especially in the case of the two non-members communities Bad Zell and Weitersfelden.

The III-Walgau study site is characterized by the active role of the regional authority and a rather passive role of the local authorities. The key actor is the BWV. This actor is the important driver in the overall flood risk management process. Key activities are the strategic planning and initiation of new projects. The emphasis is to define the general framework conditions of the overall project. The first activity was to locate and integrate new actors into the inter-local co-operation process (completed by the end of 2009). Additionally, the regional authority ensured the funding for the realisation of the flood defence schemes and was also responsible for the project controlling and monitoring of the implementation process. An important aspect was the organisation of meetings, seminars or events with local actors to increase the risk awareness, explain the importance of the inter-local co-operation and the trust building between the different members of the co-operation. In comparison to the other two study sites, the regional authority still plays an important role in the operational process, especially in the negotiation with local farmers. Moreover, the regional authority remains the key actor in the negotiation in collaboration with the III-Walgau. In the overall strategy plan, the regional authority is encouraging the III-Walgau inter-local co-operation to take over certain tasks from the regional authority (beginning with the second phase). In future, the inter-

local co-operation shall become the key player in the process. The regional authority will limit their key tasks to e.g. funding, project monitoring and controlling, or reports of the experts. At the moment the inter-local co-operation is only responsible for the administration of the co-operation, e.g. in organising meetings and events, administration accounting and emergency management. In the future the inter-local co-operation should take over also the project management.

The national and regional authorities play the central role in flood risk management policy. They are the key actors in the policy discussion and definition. The main reasons for this are their technical knowledge, their central role in funding and their permission power. Overall the local actors and stakeholder strategically depend on the national and regional authorities. The national and regional authorities are leading the design and development of flood alleviation schemes. However, their role strongly depends on the input from the local actors and stakeholders, as seen for example in the Aist study site. In general, we find local actors and stakeholders with a high interest and interaction (maturity). The leadership of the national and regional authorities has changed from promoting, networking and leading towards more project manager, analyser and problem solver. The empirical results show strong correlation between leaderships of national and regional authorities and capacity of local actors and stakeholders. The study sites with high local capacity and existing of space of engagement at regional or national level show a strong leadership at local level and bottom-up concepts and ideas. The outcome leads to a change in the responsibility between scales. Nevertheless, local knowledge and experiences play a role in the planning and implementation of flood risk management strategies. The local involvement in the discussion and decision-process depends on the local capacity (capacity to act), such as resources (knowledge, financial, time), interest, social and cultural capital. It strongly depends on the fact, if localities are able to ensure their interest and needs at higher level (e.g. regional, national). In general, an encouragement of local engagements in the policy discussion and decision process requires 'space of engagement'. The empirical results show different ranges of local involvement, although the local level has mostly a central role in the flood risk management system. In England the focus is on the funding and

planning process, more than in the Austrian case study, where the focus is on funding, planning and future maintenance, strongly depending on the study sites. Overall, the countries encourage local actors and stakeholder to get stronger involved in the process. However, the interviews have shown a power distribution manoeuvre from public authorities towards local elite groups, primarily private citizens and stakeholders.

Due to the recent developments, the role of the state and individual responsibilities for flood risk management are now under review (Defra, 2012). This review has been driven by various developments: financial and economic crises (Defra, 2012; Featherstone et al., 2012), implementation of new legal requirements (EC, 2000; 2007) and recent flood events in England and Wales (Pitt, 2008; Defra, 2005). The new policy agenda tried to enhance responsibilities of different actors in flood risk management and reduce the controlling role of central governments. This implies a change in the relationship between the different actors (Pearce and Ayres, 2012) including the creation of a 'new state space' in the policy discussions (Brenner, 2004). Furthermore, the scalar rearrangement encouraged a new organisational culture, skills, knowledge and competences (Clifford, 2013). The remaining question is if the new territorial scale in flood risk management is more than only a decentralisation of responsibility. In sum, the new territorial scale encourages local actors to get involved in the ongoing decision-process. This included a new space creation for local actors, which included a shifting of power towards non-state actors such as local grassroots organisations (Allen and Cochrane, 2010): in our case towards the FAG. Subsequently, partnership funding and the new localism policy foresee the promotion of wealthy-rural areas, where local actors want and have the experience and capacity to wield it. From the empirical results this discussion is far more relevant in England. Nevertheless, this study shows that the 'successful' implementation of new policy strategies refers to the questions of local capacity and legitimacy of the involved actors, especially in reference to the FAG (McCarthy and Prudham, 2004). Communities who fail to achieve these criteria will lose out from this new policy (MacLeod and Goodwin, 1999).

### **6.5.Strategic unification**

The rescaling process refers to the new scalar construction (Perkmann, 2007). The scalar reorganisation is partly based on politico-territorial intervention from national or regional authorities, but also local groups have initiated to re-organise the scalar fixes (see chapter 2). These strategic interventions are based on a structural analysis of the current situation (i.e. regional studies in Austria) and a concrete action program (i.e. catchment management plan). The creations of the scales in both countries are mainly influenced by previous developments. The river development concept, for example, has a strong influence in the development of catchment-wide management plans in Austria, such as III-Walgau catchment management plan (i16aut; i18aut; i19aut; i20aut). Furthermore, in the late 1990s and early 2000s the flood risk management discussion changed towards an integrated flood risk management approach with a key focus on catchment approach (White and Howe, 2002). Differences between both study sites were the statutory implementation process in Austria (see section 6.3.1). On the other hand, the introduction of the partnership funding and the need of additionally funding sources from local actors, effectively allowed realisation of the Cockermouth and Morpeth scheme as both schemes were designed under the previous funding regime and never satisfied the requirements of the ranking-system (i37eng; i38eng). Based on the flood risk management appraisals, the different study sites developed a programme of measures to reduce the impact of future flood events. The development of the action programme strongly depends on the availability, interests and resources of the local actors (see section 6.4).

Participant: 'certainly some people are better able to take advantage of those mechanisms than others, it's a two-way thing, it's about what's provided, what opportunities there are and then what capacities there are in communities to take those opportunities and sort of make them work in their interests' (Academic, England).

An analysis of the interviews showed a strong top-down approach in the areas where local actors show no or marginal interests (see sub-section individual/regional performance). However, also the study sites with a high influence of local actors show often a lack of co-ordination with other benefits with the consequences of additional costs or opposed management plans. In summary, the action programme follows mainly the goal of flood risk management instead of a holistic spatial development concept. The Austrian study sites showed a long-term planning prospective mainly developed by regional authorities and a low impact by local authorities. On the other hand, the English case study shows in two out of three study sites a stronger involvement by local actors, but also here I observed a strong leading position by the EA and their top-down management approach, except in the case of Cockermouth and to some extend in Morpeth.

Indeed, the re-scalar arrangements had empowered local actors to get involved and have influence in the planning process and decision-making. However, the participants mentioned that the rescaling process only empowers certain groups. Key aspects are the preference of adequate institutional and organisational settings to use these opportunities (see also section 6.4). Without these the local actors and stakeholders were not able to respond to the new responsibilities, tasks and opportunities with the result that the local authorities were not able to handle with the new environment.

Besides partnership funding, the new localism approach in England shows strong restrictions and limitations in the freedom of local decisions. For example the introduction of a 'roof tax' as a financial instrument includes various national restrictions. The implications are that the new localism policy hinders bottom-up developments. The national Government limits the possibilities of local authorities, which cause a mismatch between 'policy' and 'rhetoric' of local engagement in flood risk management. The consequences are disempowered communities with the results of a strong top-down dependency.

Participant: '(...) this community infrastructure levy has been introduced and by Spring of 2014 will no longer be able to take contributions from more than five separate developments, you won't be able to pool...at the moment we're pooling them all together, so from April 2014, something like this flood tariff roof tax scheme wouldn't be legal (...) It has to be based on economic viability and it will be independently examined and, again, doing it at the moment, economic viability is very poor, so it's suggesting quite low levels of contribution' (Local Authority, England).

The introduction of the new community infrastructure levy in England, for example, requires an independent review of the planned local levies and their future use, which causes new restrictions at

local level instead of empowering the local actors. The fear is that the new policy direction leads only to a rhetoric empowerment of local actors and stakeholders (Haughton et al., 2013).

Fuzzy boundaries and soft spaces play an important role in the actual flood risk management discussion in both countries. The study sites show the use of fuzzy boundaries and soft spaces in the planning and decision process. The use of fuzzy boundaries created the local scales a new space of opportunities to develop flood risk management plans (Allmendinger and Haughton, 2012). The consequences were the introduction of new administration practices and regulatory rules, often outside the formal legislation framework. A key motivation is the aspect to include all relevant actors in the planning process as well as flexibility, but also the aspect that flood issues are not oriented towards artificial borders. The Austrian catchment authorities go a step further, where the rescaling process produced a metagovernance structure (Jessop, 2004) or a new 'sub-region' (Haughton and Allmendinger, 2008) to organise more efficiently flood hazard aspects. Nevertheless, the local management plans have to adapt to national and international strategies. Interviews with local and regional authorities in Austria have shown the limited use of soft spaces in the policy discussion. In particular, the selected three study sites show no or only little interaction with other planning instruments and authorities like spatial and land use management or emergency management. The spatial and land use management plans or emergency management plans are strongly focussed on artificial borders (see also section 4.2.1). 'Soft spaces' have been mainly used in the negotiation process between the local municipalities and local land owners to adapt the planning system. The interviews show the high importance of flexibility in the planning process, because of difficulties in the agreement with local land owners or local authorities (i12aut; i13aut). Nevertheless, the Austrian study sites show a process of 'hardening' of the catchment-wide boundaries (Metzger and Schmitt, 2012; Haughton et al., 2013; Heley and Jones, 2012). The different collaborations show a high degree of formalisation between the different actors.

### 6.6. Territory and place in flood risk management

Territory and place play a particular role in the scalar re-arrangement (Jessop et al., 2008; Jones and Jessop, 2010; Brenner, 2004). The aspect of territory in flood risk management is often linked to the policy objective of reducing damages caused by flood events in defined boundaries, such as hydrological or political boundaries. The local municipalities use their territorial power to restrict the land use and to give permission to build flood defence schemes. The local authorities, especially in Morpeth and Cockermouth, use their power to adapt the flood alleviation scheme to their needs and interests, in order not to lose the tourism attraction of the town (i43eng; i37eng). A second aspect refers to the link between funding and territorial power (Späth and Rohrbacher, 2013). The empirical results reported a shift in the funding scheme, which also included a shift in territorial power towards local levels. With the introduction of the partnership funding in England the local contribution and influence increased over the past years. The main reason for this is the access to local financial contributions, which have an importance similar to national resources: In Bridgwater and Morpeth local actors overall contributed more than 40% to the local flood alleviation scheme. The local actors (such as Sedgemoor County Council and the local FAGs in Morpeth and Cockermouth) interact very closely with the citizens in their community. Therefore these local actors have also become more important in the discussion and decision-making (i38eng). In the case of Aist, I identified that beyond the local and national resources, international resources (e.g. Europe government level) play a crucial role in the territorial power relationship. The EU funding shows an equal importance compared to the national and regional funding sources. This has an impact on the multilevel governments and institutions, which decide on the local flood risk management planning (Späth and Rohrbacher, 2013; Jessop et al., 2008). The local authorities use in this case this opportunity to influence the planning process and change partly the power structure between local and regional – national authorities. Nevertheless, in most of other cases the local contributions are only understood as an additional funding source, rather than an alternative to the national sources.

Second, the location is a key role in the political mobilisation of local actors. The different countries are characterized by strong local attachment by local actors, stakeholders and citizens. The consequences were a strong focus on 'place-based' flood risk policy (Hildreth and Bailey, 2013). The different inter-local co-operations, for example, label themselves towards third actors to promote the co-operation. They use the inter-local co-operation as the official 'branding' name in communication processes. Furthermore, the different partnership approaches have designed individual logos and websites to increase their awareness as well as to disseminate the project. However, these efforts demonstrated a limited success (i26aut). Identification plays an important role for promoting financial contributions and supports by local actors (i36eng). Particularly, the English study sites demonstrate a strong interest for the local actors for 'their' community scheme (strong local attachment). Local attachment has also played a key role in the development of the local FAG. The FAG are established and defined for the needs and interests for the community level (i36eng). I identified a similar outcome in the Austrian study sites, where a key conflict was the contribution (e.g. land, financial) of each member in the co-operation. Nevertheless, the regional authorities in co-operation with the local leaders established adequate rules for individual funding (see section 4.2).

# **6.7.Key findings**

The findings of this chapter show the impacts of scalar re-arrangements in the governance structure in flood risk management. The results show that the re-scaling process enforced a change in the social structure and interaction between the different actors with new working relations and interactions. First, the study sites show an increase in the number of actors in the ongoing policy discussion (equalling a new governance framework). Second, the new scalar re-arrangements empower local actors. Third, the local actors are more involved in the decision-making process. Nevertheless, I have showed that the outcomes of this strongly depend on local capacity to use these opportunities. In England, for example, I have identified that especially the middle class professional groups with expertise and skills (e.g. planners or lawyers) have gained from the new policy direction. In summary, the scale jump empowers local actors to get stronger involved in the policy discussion. However, this strongly depends on the local interests, identification with the project and idea and resources (capabilities). This can be observed in both countries. The escapes from the scalar fixes put the local actors in a strong position in the management discussion (i9aut). However, the co-operations are not fully using this power and position. One reason is the interdependences between the different actors regarding interests, objectives and capability as well as the problems of parochial and particularised local policy directions (Kitchin and Wilton, 2003). On the other hand, Morpeth and Cockermouth are using the scale jumping in reorganising of social interaction to gain power from national level for their interests. The actors show a highly active role in the actual policy interaction. The implications are more and new actors and stakeholders were involved in the policy and flood risk management planning discussion and decision-making. The aim of the new group of people is to engage in the process to ensure their interests also in terms of changing political order (Cox, 1998). Therefore, the new politics of scale empower local actors, stakeholders and citizens to engage in the political process, although local capabilities are necessary (see above). The consequences are that these communities will benefit from the new scalar structures.

The main findings are:

- The Austrian governance structure has a strong focus on a technocrat relationship between the different actors. The English study sites include the change in the governance structure mainly for the planning, decision and implementation processes. The relationship includes only a short-term and temporal framework with the key focus on funding and planning.
- The rescaling process leads to a new administration and regulative structure, such as a new chairman and director in the inter-local co-operation. The Austrian approach includes a formalised administrative structure also for the decision process. Each inter-local co-operation contract includes a regulatory part for the decision process. On the other hand,

the English study sites include a more flexible-informal and non-statutory administration process and unit. A key reason is the lack of a nationwide regulatory approach.

- Networks play a critical role in the overall scale constructions and interactions. The two
  countries showed the creation of new networksalong the scalar re-arrangements as a
  consequence of new scalar arrangements.
- The national and regional authorities play the central role in the policy on flood risk management. They are the key actors in the policy discussion and definition. Main reasons are their technical knowledge, their central role in funding and their permission power. Overall the local actors and stakeholder strategically depend on the national and regional authorities. The national and regional authorities are leading the design and development of flood alleviation schemes. However, their role strongly depends on the input from the local actors and stakeholders.
- In England the focus is on the funding and planning process, more than in the Austrian case study, where the focus is on funding, planning and future maintenance, strongly depending on the study sites.
- General implications of new scalar arrangements are the development of new governance constructions as well as their social practices.

The next chapter will focus more on the analysis of spatial development dynamics and policies. In particular, the focus lies on analysis of the interaction between different actors and stakeholders at different scales as well as the management of the relationship between them.

# 7. Performance of new networks

**Research question:** What is the role of proximity in shaping the opportunities and barriers for partnership activities in flood risk management, and what are their social and political functions and consequences in flood risk management?

This following chapter examines the findings to addresses the interaction of new networks in flood risk management. The aim of this chapter is to examine the different stages in the integration of partnership approaches in flood risk management. As discussed previously (chapter 6) the new scalar re-arrangements enforced new relationships between the different actors and stakeholders.

# 7.1.Performance analysis of actors and stakeholder engagement in flood

## risk management policy

To apply the proximity discussion in this thesis, I have constructed a three-stage framework (table 7.24) to translate and analyse proximity within flood risk management. The conceptual framework orients to the previous work by Lundquist and Trippl (2013) and is composed of a stepwise model including three main stages: stage 1 (inadequate integration), stage 2 (semi-integration) and stage 3 (full integration). Stage 1 is characterised by low co-operation between the different actors and stakeholders. The main reasons for this are the lack of flood risk awareness in the communities, strong parochial policies and the fear of losing competences and power. The members have not acknowledged the benefits from co-operation between the different members, actors and stakeholders. The asymmetric levels decrease between the members. However, there still exist many obstacles and barriers for a fully successful co-operation. Finally, a strongly integrated co-operation includes the 'optimal' solution. The co-operation is characterised as an optimal interaction between the actors, stakeholders and citizens. There is a symmetrical flow of information, experiences and ideas among the different members of the co-operation, where the processes are created by fair power, risk- and responsibility-sharing (Innes and Booher, 2004). The ideal is the

introduction of coproduction in flood risk management and planning system (Albrechts, 2013) – stage 3. 'Co-production means delivering public services in an equal and reciprocal relationship between professionals, people using services, their families and their neighbours' (Boyle and Harris, 2009: 11). Each stage will now be discussed in more detail.

*Stage 1 (inadequate integration)*: weak interaction is displaced between the different members in a flood risk management partnership. Key problems include the physical distance between the different actors (functional proximity), a low degree of trust between the actors (social proximity) or lack of benefits from the co-operation. A key barrier in the inter-local co-operation, for example, is lock-in (decision-making) at local level (see also section 7.3), the fear to lose competences, power and limitations of self-governing (Weichart, 2006). Further hindrances are the high degree of competition between the different communities and regions, especially in relation to public funding and new business investments (Hagspiel, 2001; Amdam, 2001). Additionally, often there is no clear definition of how to share responsibility between the different actors as well as unequal – asymmetric power sharing (Hagspiel, 2001). In addition, a key barrier is the uncertainty regarding future developments.

For example, from previous examples in inter-local co-operations, in general, 'large' communities (in Austria mainly communities in the lower part of the catchment) are more likely to co-operate with other communities, because of the cost savings based on the economics of scale effect as well as due the possibility to get a flood defence scheme without increasing the role of the neighbourhood communities. On the other hand, studies show that 'small'-peripheral communities react more sceptically towards an inter-local co-operation process, because they fear losing the identification and self-governing independence (Wirth and Biwald, 2006; Weichart, 2006). Besides, rural (periphery) communities are often too 'weak' to start an inter-local co-operation process as they fear additional costs to the public budget (Weichart, 2006). The consequences are that local authorities define and follow different objectives and goals instead of a common vision. The results are strong conflicts between the different actors. This includes a relatively low willingness for co-

201

operating (Lundquist and Trippl, 2013). In general, harmonisation and the use of non-structural flood defence measures are secondary goals, such as inter-local spatial and land use management strategies. However, the different actors recognise the individual benefits of the inter-local co-operation.

*Stage 2 (semi-integration)*: goes a step further. The physical distance is less important, because the different actors and stakeholders know each other. However, the harmonisation between the different actors and stakeholders is not fully integrated, in particular the spatial and land use management or emergency management plans. In this stage of co-operation decisions are based on given political – institutional barriers. Regarding non-structural measures, I observe a first harmonisation of the management plans between neighbourhood communities or at the district level, e.g. through exchange of information. Furthermore, in inter-local co-operation in flood risk management, for examples focuses to the implementation of inter-local structural measures, e.g. runoff management in the catchment.

*Stage 3 (full integration)*: includes the full integration between the different actors and stakeholders in a partnership approach. I assume this as the final step in the partnership process. The focus in inter-local co-operation, for example, is on the implementation and maintenance of structural measures as well as full harmonisation of non-structural instruments, e.g. spatial governance approach and catchment-based emergency management plan. The different actors and stakeholders exchange information (e.g. round tables, meetings) about their developments, objectives and concepts relation to flood risk management. A key objective is the use of a mix of different management instruments to reduce the impact of future flood events. A further important issue is the need to avoid contradicting policy directions in the inter-local co-operation. Throughout, an important consideration is the social learning process between the different actors. In the ideal case, the partnership approach includes a fully democratic and transparent system to ensure accountability and legitimacy.

202

Table 7.24 proposes a framework to analyse the engagement of the different actors, stakeholders and citizens in various themes in flood risk management. In total, I selected four criteria, which have key functions to flood risk management, to compare the selected study sites.

	Weakly integrated	Semi-integrated	Strongly integrated
	(Stage 1)	(Stage 2)	(Stage 3)
Spatial planning	No/medium harmonisation	Medium/high harmonisation	Spatial governance
Structural measures	No co-operation, e.g. preventive local-linear structural measures	Medium/high co- ordination between communities, e.g. implementation of flood storages	High/fully co-ordination between communities, e.g. implementation of common catchment management schemes, such as to improve infiltration rates (cross-field mould ploughing, improve soil structure, install water boreholes), to reduce flow connectivity (buffer strips, afforestation, flood storages)
Stakeholder engagement	No co-operation between private-public actors and stakeholders	Informal co-operation between private-public actors and stakeholders, e.g. asking for funding scheme	Full co-operation between public-private actors and stakeholders (companies and households), funding and steering group
Risk communication	No communication between public authorities and non- state actors and stakeholders	Communication within the members of the partnership approach	Communication with internal as well as external stakeholders and citizens

Table 7.24.	Change stanistics of	different forces of	Company and the sheet	and an exception in a second
Table 7.24:	characteristics of	different level o	r engagement in the	co-operation process

Source: own development

### Spatial planning

Spatial planning has become popular in the flood risk management discussions. Its role is to coordinate the different interests and objectives of actors, stakeholders and citizens. The term 'spatial planning' itself including all different actors involved in the decision-making practices, so we underlined that spatial planning has to be seen as a socio-spatial process. In this line, spatial planning is a 'political project' (Allmendinger and Haughton, 2009b: 2547). As such, spatial planning includes a 'strategic perspective on land development, environmental concerns, resource use, transport, economic development, social infrastructure, and similar concerns' (Allmendinger and Haughton, 2007: 1478). Spatial planning and land use management never act independently; instead, different interests have to be co-ordinated (Allmendinger, 1998; Neuvel and van der Knapp, 2010; Baker and Wong, 2013). It includes a wide range of different actors involved in the decision making process, e.g. housing, transport, economy, health sector, education and environment (Allmendinger and Haughton, 2009a). Consequently, spatial planning is the field in 'which people shape and govern spaces and takes into account social, economic, and environmental issue' (van Assche et al., 2013: 179), and where the social structures influence spatial planning process and the outcomes. Nevertheless, spatial planning has often been criticised for not responding to the actual level of hazards (Howe and White, 2004; Fuchs et al., 2013) or causing conflicts between the different actors and their interests (Neuvel and van der Knapp, 2010; Pardoe et al., 2011). There exists a central conflict between economic growth and the reduction of future losses caused by natural hazards events; this is of particular importance in European regions due to the scarceness of areas suitable for development activities. A key focus is related to the prevention of future natural hazards events, e.g. local protection and adaptation measures such as the designation of retention areas, flood proofing of new buildings or implementation of sustainable urban drainage measures (Sapountzaki et al., 2011; Holub et al., 2012).

#### Structural measures

Traditionally, flood risk management focused on local protection measures, such as building embankments or straightening the watercourse. The integrated flood risk management approaches recognise a broader view of the management of a catchment as an integrated management system. The extensive use of the space for highly economically valuable activities (agriculture, industry and settlement) in the downstream catchment limits the availability of land for temporal flood storage area. On the other hand the availability of land upstream with no major economic use provides the potential for large water storage capacity. The needs for flood storages in the upper river catchment for the benefits of downstream settlements require an inter-local approach to reduce vulnerability in the catchment. Therefore, a clear statement is the implementation of catchment-wide management plan (Wilkinson et al., 2010). This includes a broader co-operation between local authorities, especially in rural areas. The key aim is the consensus building between the different groups (Berkes, 2010; Margerum, 2008). Key aspects are 'inclusion, power-sharing and joint decision-making' (Berkes, 2010: 492) as well as 'an interaction of equals, rather than a subject-object relationship' (ibid: 492).

Throughout, with current pressures on local authorities to reduce spending and in parallel, a reduction in the central state's resources, catchment flood risk management plans have been seen as a possibility to both increase the 'value' of budgets available and to increase efficiency in using current public funds and resources. Furthermore, they may allow direct and indirect benefits including harmonisation of spatial and land use management plans. From the interviews the most mentioned advantages are access to additional resources (e.g. funding, land).

#### Stakeholder engagement

Stakeholder engagement is a core aspect in an integrated flood risk management (Renn, 2008). Scholars defined stakeholder engagement as a social process; to work together to find a collective solution for a certain problem (Green and Penning-Rowsell, 2010). In the policy discussion, stakeholder engagement is often initiated by political parties/leaders (e.g. mayors) or by the public administration. Key arguments are to increase trust and legitimacy of citizens into public administration. In the literature, scholars often mentioned the inflexibility of public administration to react to the outcome of public participation process (Tseng and Penning-Rowsell, 2012). Other problems relate to the lack of institutional support about how to organise/deal with a stakeholder engagement process, lack of communication and information-sharing and especially lack of resources, in particular in respect to large participation processes. Key problems are related to the different interests and views of each stakeholder groups, e.g. political leaders or public administration try to increase accountability and legitimacy, people in risk areas ask for structural measures to reduce vulnerability etc. Throughout, stakeholder engagement is an important process for the successful implementation of certain policy strategies.

205

#### **Risk communication**

Risk communication is the core aspect in the risk governance framework. It is an important issue, because it influences various aspects, i.e. individual mitigation behaviours (Soane et al., 2010), emergency management processes or warning processes. Risk communication includes the investigation of 'how expert assessments could be communicated to the public best, so that the tension between public preceptors and expert judgement could be bridged' (Renn, 2008: 201). The key aspect is to help stakeholders in understanding technical-scientific results set out by the final objectives and how to achieve these targets (Renn, 2008; IRCG, 2008). Filatova et al. (2011) described the risk communication process as a process of exchange of information and knowledge. IRCG (2008) defined risk communication to enable and engage different stakeholder groups to understand the results from the various different steps (risk appraisal, risk framing and risk management) to support the final decision to reduce negative impact of flood events. Moreover, risk communication shows also different points of view and experiences of the different agents (Lofstedt, 2003).

Figure 7.28 provides an overview of the results of the two countries. None of the selected study sites have fully achieved the 'optimal' level of integration between the different members in their partnership approach. The differences will be described in the following sub-sections.

	Weakly integrated (Stage 1)	Semi-integrated (Stage 2)	Strongly integrated (Stage 3)
Spatial planning	BW CM MP		
Structural measures	BW		Aist
Stakeholder engagement	Aist	$\sim$	
Risk communication		IW BW Aist	

Figure 7.28: Integration and network engagement in the selected study sites

Source: own development\*

\*Abbreviations: BW – Bridgwater CM – Cockermouth IW – III-Walgau MP – Morpeth TT – Triesting-Tal

The following sections are dedicated to offering some key conclusions of the level of engagements in the selected countries.

#### 7.1.1. Spatial planning

Based on the shift in the natural hazards discourse, away from the engineering standard of technical protection schemes towards a broader integrated management, spatial planning has become popular in the flood risk management discussions (Wiering and Immink, 2006; Hutter, 2007; Glavovic et al., 2010; Porter and Demritt, 2012). In general, spatial planning and land use management focus to the prevention of future natural hazards events, e.g. local protection and adaptation measures such as the designation of retention areas, flood proofing of new buildings or implementation of sustainable urban drainage measures (Sapountzaki et al., 2011; Holub et al., 2012). A central point refers to the aspect of the governance structure, the relationship and the interaction of the different

actors in the planning discussion. Spatial planning never acts autonomously and independently from other policy goals and developments (Neuvel and Van der Knapp (2010) or Pardoe et al. (2011)). However, often a missing link is to manage the actual and the potential flood risk in the catchment including a broader view of the catchment.

The patterns observed in the figure 7.28 show no clear integration between the different study sites. Neither the Austrian nor the English study sites could be considered as fully integrated in spatial planning. Flood risk management has no or little impact in the design of local land and spatial planning plans and strategies. The key objective is the economic growth within the administrative boundaries. Further, the land use management and spatial planning plans have no link to their neighbourhood communities or local authorities. The plans have no clear vision how to integrate and to solve the problematic of flood risk management in land use management and spatial planning. For example, analysing the Austrian interviews, the local authorities within the Austrian study sites show a low interest in co-operating within the field of spatial and land use management planning, because of the local parochialism and fetish of economic growth contrary to a regional governance solutions (Fürst, 2003; Wirth et al., 2008). The three study sites show weak preparations for coordination or harmonisation of local spatial and land use management plans. Further, the empirical results show that all three study sites have little interests in closing this gap, because of the increasing tax competition between the local authorities, which favours competition instead of rational collaboration (i3aut). An exception is III-Walgau study sites, where the regional authorities introduced the blue zone concept. Since 2012 the Federal State Vorarlberg has introduced the planning concept of the 'blue zones'. The 'blue zones' define water relevant areas; mainly run-off area of small flood events. The key idea is to restrict the land use management in these areas, e.g. ensure natural retention areas. The definition of 'blue zones' is a top-down approach, where the Federal State defined the water relevant areas. Furthermore, the concept is inter-local and defined by the regional authorities. The consequences were that the study sites show no shifts in the spatial planning process from local based approaches to a catchment-based level. Further, a clear lack is the

openness of the spatial planning thinking to a broader-more holistic view of the space as well as strategic – long-term vision for the catchment in referring to the impacts of climate change. However, also the institutional frameworks have not solved this problem, the lack of formal and informal rules and norms. For example, a remaining obstacle is the low involvement of the Federal State in local spatial planning concepts in an advisory role to control the spatial and land use management acts for flood risk management. On the basis of empirical results, the Federal States in Upper and Lower Austria provides no pressure on the local authorities to develop catchment-wide spatial and land use management plans. Key reasons for this are the political overlapping of local politicians at Federal State level. In Lower Austria and Upper Austria more than 78%<sup>29</sup> of the members of the Federal State have been representative of a local authority (current or past legislation), e.g. mayor, deputy mayor or members of the municipal council, with the effect of a strong position of local interests in the ongoing political discussion at regional level. This observation has strong implications for flood risk management, because the effects from the Federal State towards floodplain development controls are strongly limited. Consequently, the local authorities.

Similar results can be found in England with the introduction of new Town and Country Planning Act (UK Legislation, 2012), which foresees the implementation of neighbourhood area plans in close cooperation with local stakeholders. The consequences were that the neighbourhood planning has abolished the regional spatial strategy plan (UK Government, 2004) by the current government. The outcome was a downscaling process of the planning documents to parish councils or urban boroughs (i40eng). Analysing the neighbourhood plans from the English study sites, only Bridgwater has not developed a neighbourhood plan yet. The neighbourhood plans from Cockermouth and Morpeth were oriented towards housing, local economy, heritage and local environment. In general, the aspect of floods plays a minor role in the overall plans. In Cockermouth, the town council focus their neighbourhood plans towards main street improvements (Cumbria County Council, 2013). In

<sup>&</sup>lt;sup>29</sup> Own calculation: data based on OÖ (2013) and NÖ (2013).

Morpeth, the neighbourhood plan foresees to restrict land use, especially new residential areas. However, the plan has no concrete strategy and advice how to control and to organise flood risk in the town (Northumberland County Council, 2011). Consequently, both neighbourhood plans include no reference to the local flood risk problems. These patterns are in contrast to the National Planning Policy Framework (NPPF, 2012), which request an 'integrated spatial and land use management planning' approach to take into consideration the different aspects, e.g. flood, climate change, housing, or infrastructure (CLG, 2012). Furthermore, the neighbourhood planning include no catchment-wide land use management concepts.

Therefore, the ideal situation would be the introduction of a catchment-wide spatial planning strategy enforces the creation of new state spatialities and processes in the on-going decisionmaking process. The new concept shifts the planning process and discussion from local based approaches towards the natural-surface catchment boundaries. Important aspects are the engagement of new actors in the planning discussion to integrate different aspects in the decisionmaking process. These include a wider range of different actors with different interests. Further, a key point is a strategic – long-term vision from local-based towards catchment-level approach. This comprises a new spatial imagination for local and regional actors to influence management plans and regional defined regulatory, like regulatory zones and building control instruments. The implications of a new spatial planning approach is the introduction of the concepts of soft spaces and fuzzy boundaries (Allmendinger and Haughton, 2009; Haughton and Allmendinger, 2013). The use of soft spaces has become an issue in flood risk management as the strategy of flood risk management moved from local-based approach to a catchment-wide approach. The focus on catchment-wide management strategies request the construction of a network between the different local actors and stakeholders, also in terms of providing bottom-up initiatives and informal strategies to overcome delays and closeness in the public administration. Further, the creation of inter-local co-operations are a response to the institutional limitations in flood risk management, because administrative borders cannot prevent flooding.

### 7.1.2. Structural measures

Traditionally, flood risk management focused on local protection measures, such as building embankments or straightening the watercourse. The integrated flood risk management approaches recognise a broader view of the management of a catchment as an integrated management system. Looking at the strategy to implement flood defence schemes in the study sites, it has become clear that the English and the Austrian study sites are represented by a significant degree of heterogenity. The Austrian study sites focus in the implementation of inter-local flood risk management concepts, such as flood storages across the catchment.

Also in relation to structural defence schemes, the three English study sites show mainly local flood defence schemes, such as embankments. Here the main hindrances are a lack of inter-communal cooperations: (1) there are no legal frameworks or incentives that would encourage the local authorities to co-operate and to work together; (2) the inclination on economic growth: there is no win-win situation between the different partners in the co-operation between upstream and downstream communities, especially the upstream communities have to contribute, e.g. by restricting the land use in its administration unit. Also the current intensive focus on economy and competition between the different local authorities influences the spatial and land use management plans. The drive of economic growth is not at regional level, but on the local level (i33eng); (3) lack of political willingness for inter-communal co-operation: while analysing the interviews a lack of political willingness was found to push and to encourage local authorities to work together; especially local authorities from different political parties. The inter-local co-operation can be seen as an ideal instrument for the implementation of flood storages. Throughout, with current pressures on local authorities to reduce their spending and in parallel, a reduction in the central state's resources, inter-local co-operation has been seen as a possibility to both increase the 'value' of budgets available and to increase efficiency in using current public funds and resources. Furthermore, they may allow direct and indirect benefits including harmonisation of spatial and land use management plans. From the interviews, the most frequently mentioned advantages are the

access to additional resources (e.g. funding, land) and the effects of economies of scale. Here, interlocal co-operation has the possibility to ensure natural retention or flood storage areas in the upstream communities, which is less expensive than individual solutions for each community.

The three study sites in Austria follow a clear inter-local flood defence strategy in contrast to the English empirical results. Analysing the catchment management plans, they encourage the achievement of multiple benefits from the strategic proposals, e.g. by increasing biodiversity or by improving the river continuity. In respect to flood risk management, the overall goals are the implementation of flood storages in the catchment. There are significant differences between the three study sites in Austria in the extent to which temporary flood storage has been adopted. Implementing flood storage areas involved the temporal and unpredicted (in terms of which year and frequency) flooding of large land areas often use for other purposes. The implications are that larger networks create higher transaction costs, particularly due to the increase of communication processes between the members of the co-operation. They also may generate problems related to a fair sharing of power between the different actors. A third aspect includes the risk of a failure of the negotiation process, especially for the re-organization phase. The key barrier is the different interests (flood prevention benefit transfer to downstream communities and financial transfer to upstream communities) of each stakeholder which may complicate the decision process when ranking different options, e.g. ranking the implementation of the flood alleviation schemes. A key problem is that this policy option includes a safety benefit transfer from upstream to the downstream communities; sacrificing upstream communities (increase the risk of upstream communities), which were compensated (positive discrimination) by downstream communities; funding transfer from downstream to upstream communities.

There are potentially at least three potential classes of barriers that may need to be overcome and which therefore account for these differences in adoption:

212

- Cultural and social relational; for example, in England, the cultural understanding is that all
  existing land uses must be protected from flooding and hence the sacrifice of some land to
  protect other land, and the retreating of defences to free up space for the river, has not so
  far been publically acceptable as a mainstream measure.
- Compensating those whose land will now occasionally be flooded. The implementation of flood storage is that both farmers and government want to convert intermittent large losses and payments into smaller fixed annual amounts.
- Administrative boundaries. Most countries have written Constitutions which define powers to different layers of administration within define boundaries, commonly provincial and municipal. Undertaking works in one administrative unit in order to benefit another consequently is a problem to be resolved.

#### 7.1.3. Stakeholder engagement

In the implementation process of new flood risk management strategies, the empirical results from all study sites in Austria and England highlighted the inclusion of further key stakeholders; including land owners and inhabitants in floodplain areas as well as those stakeholders who are indirectly affected by flood events or those who are paying for the costs of flood risk management. From the empirical results, the different stakeholder groups in the selected countries are not homogenous, inside these groups exist a wide range of different interests and objectives, which reflect the multitude of stakeholders in the planning and decision-making practices (see also section 6.3.2). In general, the involved actors, stakeholders and citizens can are those living at risk of flooding (directly affected by past events) or those living in the catchment or community but not living at risk. Examining the stakeholder engagement in the policy decision-process as well as flood risk management planning I can observe substantial differences between the study sites. In particular, the two English study sites, Morpeth and Cockermouth, are well endowed with the engagement of local actors and stakeholders. The key reason for this high degree of integration is the presence and effectiveness of local grassroots organisations. Also the societal acceptance to engage in the policy and flood risk management planning process seem to be higher in the wealthy rural English study sites than in the Austrian study sites or in Bridgwater. On the other hand, the involvement of private actors it emerged from the interviews that the negotiation process was longer and more complicated as a result, because of the different interests. Most of the time the exclusion of private actors as a member in the partnership are based on lack of political willingness to involve and to enforce private actors to contribute (i9aut). However, the influence of stakeholder in the flood risk management planning and decision-making practices depends on the local capacity as well as trust and openness in the public administration. The involvement of stakeholders implies a longer negotiation process than in the case of individual local authority based solutions. For example, regional authorities of Vorarlberg (Austria) have spent around 1,200 hours in 2010 and the directorate of the III-Walgau co-operation around 800 hours in 2010. However, the transaction costs will reduce at medium-longer term perspective. One main reason is the communication and planning process with just one partner instead of 27 partners (like the case in the Aist). The third aspect includes the risk of a failure of the negotiation process, especially for the re-organisation phase. The key barrier is the different interests of each stakeholder (flood defence benefit transfer to downstream communities and financial transfer to upstream communities). Castells (1977) described this as one explanation 'why the membership of groups is not stable' (ibid: 78). Most of the different actors have strong interdependent interests, such as economic growth vs. restriction in land use management, which cause conflicts between them (see also figures 6.23 and 6.25). Conflicts mainly rise due to funding, especially the amounts of the individual contributions show a high potential of conflicts between the different actors. The Austrian interviews show this to be a key barrier in the development of inter-local co-operation (i5aut; i19aut; i11aut; i9aut). A key solution has been the implementation of a 'fair' financial distribution between the different actors and stakeholders.

This study observed five main incentives for local contributions in flood risk management. First, a strong incentive for the local contribution was the recent flood history, for example in Cockermouth

214

2009 which affected the whole town. Second key factor was the local grassroots organisations. In the English case study, the non Government actors and stakeholders played an important role in the partnership funding agreement. The partnership integrates different private actors in the decision process, mainly households and small-medium business owners. The interplay between the different organisations required a new orientation in the inter- and intra-organisational relationships. The consultation process, however, progressed slowly. Problems of misunderstandings, lack of trust and asymmetric information levels are unavoidable between the different actors. This led to longer and more intensive negotiation processes compared to the 'old' system, especially a key factor was the common understanding about problem definition, objectives and possible solutions between the actors. Especially this study showed the changes in the governance structure in the English flood risk management policy. Under the 'old' funding system mainly public authorities took part in the flood risk management discussion. With the possibility of private actors to contribute to the flood defence scheme, their influence has increased in the overall process. Today, the private actors take part in the planning and decision process. The introduction of partnership funding has to take into account the collaboration, similar to the new spatial planning approach (Baker and Wong, 2013), even though the processes were initiated by local contributors, such as Cockermouth. In particular, the funding policy shifted the financial risk from the EA towards local actors and stakeholders. In practice, local actors and stakeholders have to contribute to additionally needed funding (i43eng), because the EA is providing a fixed allowance.

On the other hand, the Triesting-Tal implemented strong barriers preventing local stakeholders from getting involved in the policy discussion (Tseng and Penning-Rowsell, 2012). The national and regional authorities are the key leaders in the management process (see also chapter 4). They hold a monopoly in the planning of new flood defence measures as well as the initiation of new projects. In contrast, the Triesting-Tal shows a passive role in the implementation process. The local scales not engaged in the decision-making practices, which leads to a strong top-down flood risk management system without local initiatives. The main consequence was that the local scale was not able to take

215

over the responsibility and tasks from the regional and national authorities, because the lack of local capacities and interests. Similar results could be found also in the integration of new members in the inter-local co-operation. This was organised by regional authorities mainly the WLV. Therefore, in terms of organisation and decision-making structures the Triesting-Tal has a strong hierarchical organisation structure with top-down decisions, communication and delegation policy (Watson et al., 2009). Consequently, the scalar arrangement creates only a shadow of balance of power between the national, regional and local scale. In particular, the regional scale in Austria tried to avoid encouraging the meta-governance structures. In the Aist study site, the grass-roots organisation *lnitative fuer oekologischen und nachhaltigen Hochwasserschutz Aist* show a great interest in the current flood risk management debate. The influence of the grass roots organisation was the evaluation of the catchment management concept by the University of Natural Resources and Life Sciences at Vienna. The evaluation foresees the better integration of the use of structural and non-structural measures in the catchment management plan. In sum, the grassroots organisations encourage a communication process between regional and local authorities with citizens in the catchment (i15aut; i12aut).

#### 7.1.4. Risk communication

Risk communication is interpreted as relational proximity between the different members (see also section 3.3). The analysis of the risk communication structure is based on the published work by Höppner et al. (2010) and McCarthy et al. (2008). The data shows similar results between the different study sites in terms of the mode of communication, tools and purposes (table 7.25).
#### Table 7.25: Risk communication in flood risk management in selected study sites

	Aist	III-Walgau	Triesting-Tal	Bridgwater	Cockermouth	Morpeth
Actors	WLV, BWV, local municipalities, consultant group, university, non- governmental group	BWV, local municipalities, private consultant group, utility companies	WLV, BWV, local municipalities, district (first phase)	EA, County Council	FAG, EA, County Council	EA, FAG, County Council
Purposes	ir	nprove relationshi	p, involve actors in c	decision making, incr	ease knowledge	
Modes	few-to-few (regional authorities and members of steering group), communication between regional authorities, few-to-many (general assembly meetings), dialogical two- way	few-to-few (regional authorities and members of steering group), communication between regional authorities, few-to-many (general assembly meetings), dialogical two- way	few-to-few (regional authorities and members of steering group), communication between regional authorities, few- to-many (general assembly meetings)	few-to-few, dialogical two- way (between County Council and EA), non- dialogical two- way communication (towards citizens)	few-to-few, dialogical two-way	few-to-few, dialogical two-way
Tools	flood risk assessment, flood modelling, meetings, scientific reports	flood risk assessment, flood modelling, meetings	flood risk assessment, flood modelling, meetings	flood risk assessment, flood modelling, meetings	flood risk assessment, flood modelling, meetings	flood risk assessment, flood modelling, meetings
Messages	expert- dominant, transparency (internal: yes; external: yes)	expert- dominant, transparency (internal: yes; external: partly)	expert- dominant, transparency (internal: partly; external: no)	expert- dominant, transparency (internal: yes; external: yes)	expert- dominant, transparency (internal: yes; external: yes)	expert- dominant, transparency (internal: yes; external: yes)

Source: own development

The main actors in the communication process are the technocrats from the regional authorities (in England: the EA and in Austria: the WLV and the BWV). The main reasons for this are that these organisations lead the overall flood risk management process (planning, implementation, funding and co-ordination between the different actors and authorities) and have the technical knowledge and expertise. This has an impact also for the type of communication. The communication process is important in the relationship between local and regional actors. Nevertheless, the interviews show

strong conflicts and barriers as well as misunderstandings between local and regional actors. A core problem is that the local actors have difficulties in the understanding the technical language, such as flood return periods. In the Austrian case study, the WLV and the BWV shows different approaches and concepts for the flood risk management, for example the use of different return periods for the standard protection (BWV: 1:100; WLV: 1:125) (Holub and Fuchs, 2009). Furthermore, the interviews show strong barriers and conflicts in the discussion between regional professionals and local politicians, based on lack of technology knowledge and expertise (politicians), but also lack of a common approach between the two organisations (i15aut). In summary, both organisations show strong interdependences. However, the implementation of the EU Floods Directive into national law requires some harmonisation between both organisations, especially regarding the return periods and flood defence standards.

In Austria, I identified clear misgivings from local actors towards regional-national actors because of the use of the expert language. Local actors do not have the knowledge, skills and understanding of the technical terminology, the concepts and ideas provided by the experts from regional or national offices. Therefore, the cases are characterized by a strong top-down communication approach. These conflicts and barriers have a negative impact for the interaction with the local actors in flood risk management. In particular, this has been a clear hindrance in the empowerment of local actors in the ongoing policy discussion. In Austria, for example, main barriers and conflicts were observed in the communication process between the two public authorities (WLV and BWV) with local authorities (see following quotes).

Participant: 'I wonder whatever the system we have is clever that you have a [name] and a [name], because sometimes both organisations have some different understandings, meanings an views there (...) but I find it funny that even various technical data and results have been seen different between them. As a layman I imagine flood risk management is flood risk management and 1:100 flood event is a 1:100 flood event and that there is still are nuances here I do not understand' (Local Authority, Austria).

Participant: '(...) in 2002, the HQ-100 has been re-defined with 320 m<sup>2</sup>. You can here discuss for a long time how we ever come to this shift from 240 m<sup>2</sup> to 320 m<sup>2</sup>. We spent hours of discussion with the regional authorities, because nobody explained us these dramatically changes in flood risk management policy' (Local Authority, Austria).

The modes of conversations (see table 7.25) are mainly organised by few-to-few numbers of actors, such as in Austria the main communication processes are between regional authorities and selected members of the steering group (e.g. chairman and director). Nonetheless, a second important role is the communication process between the regional authorities and the steering groups (in general, based on regular meetings every 5-6 weeks). Overall, the communication process shows a broader-dialogue, where both groups of actors exchange their idea, interests and concepts.

In contrast, I identified a fundamental change in the English cases. The English flood risk communication process has shifted from a non-dialogical communication process towards a more dialogue-two-way approach (see following quote).

Participant: '(...) in [town] where they asked this question, well what if we have a lower standard of protection (...) because we might say to them well if ... to provide a 1 in 75, yes, the scheme is only going to cost £10m. And they were saying of right, well let's ... so they might say well actually if it's only £10m to provide a 1 in 50, why don't we do that and then we don't have to pay anything, but no, because you're not getting the benefits, and as you're now only moving these properties to there, you're not moving these properties at all and those ones not at all, so you're only moving those, so you immediately strike out that £2m... sorry, that £2m is going to be less because the benefits are less (...) so they're saying well can we look at a lower standard of protection, well yes, but you get less money from us' (National Authority, England).

The change was encouraged by the introduction of Partnership Funding and the development and interaction of FAG at local levels. The local actors forced the EA to start a two- way approach, where the local actors can bring in their views and ideas. In the communication process, the main tools are technical models and assessments, which often are produced by private consultant groups. In particular, in the study sites Aist and Cockermouth the actors mainly used models and scientific reports for the internal communication process. Local actors are disadvantaged in the communication process, if they do not have the level of expertise and knowledge to understand it. A second problem refers to the transparency and external communication process. The English study sites show a high degree of transparency in the communication process. A second justification refers to the strong accountancy rules and based discipline in England, especially to quantify and benchmark between the different communities (Swyngedouw, 2011). Overall, the key contributions are the level

of knowledge: skills and interest at local level in the risk communication process as well as the form of dialogue and use of language between the different actors (see also chapter 6). The communication process is more equally based, when local actors show a high degree of knowledge, like the case of Cockermouth, Morpeth or Aist. In these three study sites, local actors collaborate with technical consultant companies to achieve a more 'adequate' educational-professional background. Stakeholder engagement depends on transparency and 'adequate' language in the policy decision-making and flood risk management planning. The usages of highly technical language ensure only the power geometry between public administration and local actors, stakeholders and citizens. However, local stakeholders and actors have to 'learn' the technical language in flood risk management to discuss equally with the public administration. Nevertheless, this capacity strongly depends on local social and cultural capitals which have wider implications, such as uneven development within the country. In comparison to the Austrian study sites, the English study sites, especially Morpeth and Cockermouth, have a high degree of communication interactions between public administration and non-state actors, stakeholders and citizens. The key reason is the strong involvement of grassroots organisation in the policy decision and flood risk management process. In both study sites, the various actors and stakeholders have to be more active in the communication process to inform and to engage the population.

## 7.2. Dependency in partnership approaches

I identified various formalised frameworks which influence the stability of a network. The Austrian study sites show various regulatory conditions, which are top-down defined, but have a positive influence in the stability of the network, e.g. for the partnership funding scheme. The partnership funding rules are a critical top-down condition in the network discussion, where the vertical network displaces the horizontal networks (Cox, 2013). A key condition is the formalised contract between the different actors. The legal contracts provide a high security, especially for local municipalities, who never worked together in the past or for a long time framework, for example in the Aist co-operation. The contract includes various conditions, for example also in preventing the drop-out of

individual members from the network (i19aut). My findings show that the implementation of a steering group performs a high stabilisation factor for the organization itself. First, the steering group lead the communication process and interaction inside the network (i12aut). It undertakes various meetings with the different actors to ensure the survival of the network. The role of local actors depends on the self-realisation in the network. Active local networks, the director and the chairman of the inter-local co-operation play an important role in the cohesion between the members. Second, the shift of power and responsibility from local level towards an inter-local level ensure a more effective way in the interaction between the different actors, the planning discussion and the implementation process. The steering group acts as a buffer zone between the different interests and developments. The key aim is to reduce the level of frustration between each actor. Further regulatory factors are the common interests between the different actors (Castells, 1977). Flood risk management includes a wide involvement of the different actors at local and regional level. Key drivers are economic arguments, mainly the fear for economic prosperity in the region. This is a core aspect for peri-urban and rural areas, because of the lack of economic substitution. Consequently, the different actors are strongly interested in keeping the actual number of people and businesses in the region. The stability of the relationship also depends on the dependency / independency of each actor. In general scholars distinguished between power and resource dependency (Lee et al., 2012; Rössl, 2005a; Drogendijk and Andersson, 2013). Within the introduction of a standardised approach in the organisation structure of the inter-local cooperations as well as fixed rule system for administration processes and approaches, especially in relation to planning, approval and funding, the national Government introduced a framework which encourages the implementation of an inter-local flood risk management strategy. On the other hand, countries, such as England, without this administrative standard hindrance the development of catchment-wide flood risk management strategies and concepts. The implications are that lack of a clear legal construction may cause lack of unclarity and despotism in the country.

#### 7.2.1. Power dependency

Further, I identified the power relationship between the different actors. The aspect of power is a substantial term in the network structure (Allen, 2009). This study analysed power dependency relations between local and regional or/and national actors. Overall, local actors are strongly dependent on the regional authorities. Nevertheless, the possession of power depends on the position of the actor in the interaction with the others (Cox, 2013; Sayre, 2005; Bues and Theesfeld, 2012). Theesfeld (2011) has made an important contribution in the power debate in policy discussion. She added the aspect of consequences of power in the bargaining process, which can be distinguished between relative bargaining power and the distributional effects on the bargaining. According to Dencker (2009), relative bargaining power is understood as the outcome of the relationship between different actors, which depend 'of one of the parties to this relationship on the other party to secure needed resources' (ibid: 453). Particularly in the study sites, I identified a strong dependence relationship between the different actors in terms of power and resources (table 7.26)

#### Table 7.26: Power dependency resources and impact in Austria and England

Characteristics		Consequences on bargaining process	Austrian study sites		English study sites	
			Upstream municipalities	Downstream municipalities	Middle class	Working class
Depending on natural resources	Elements at risk-risk behaviour	Relative bargaining power		++		
	Preference	Relative bargaining power	-	+		
	Exist costs	Relative bargaining power		++		
Depending on government support	Potential power	Relative bargaining power	++	-	+	
	Network power	Relative bargaining power	-	++	++	
	Sanction power	Distributional consequences	+	-	+	
Education and knowledge	Access to information	Distributional consequences	-	++	++	
	Knowledge	Distributional consequences	-	++	++	
	Cultural capacity	Distributional consequences	-	++	++	

Source: adapted from Bues and Theesfold (2012): 279; --very weak position; - weak position; + strong position; ++very

## strong position

The dependency on natural resources (here mainly availability of retention area) has a key role in the flood risk management discussion. As mentioned above, the Austrian study sites follow an interlocal approach to reduce vulnerability in the catchment. A key objective is the implementation of flood storages in the river catchment, especially in the upper part. The main resources of upstream municipalities are the availability of land to store a large capacity of water during a flood event. As a result the downstream municipalities are strongly dependent on the municipalities in the upstream area. Similar results have been found in the aspect of time preferences. The negotiation processes between the different actors can have a long time framework. The main reason for this is that the upstream municipalities have demonstrated large concerns about the projects including the lack of benefits for the upstream municipalities, as well as financial risks and costs and unknown interactions with new actors. The result is that the upstream municipalities have a lower interest and preference to engage in the co-operation compared to the downstream municipalities. It means that the upper part of the catchment has a clear advantage position than the lower part of the catchment (Bues and Theesfeld, 2012). Both variables have a direct impact towards the costs of leaving the partnership (exit costs). Exit costs arise alongside the negotiation process between the different actors (Meyer, 2010; Bues and Theesfeld, 2012). In general, high exit costs restrain co-operation. In this line, the different actors – especially those in a weaker position – try to send incentives to reduce the risk of failure and subsequent the exit costs. For example, the downstream municipalities transfer financial resources towards the upstream municipalities as incentive to co-operate and to ensure a fair financial arrangement between the different actors. The study sites show a long term bargaining process: the Triesting-Tal co-operation took over 10 years. This includes high transaction costs between the involved actors and conflicts between the local municipalities. The results show that the downstream municipalities show higher exit costs, because of their dependency on the upstream municipalities.

The English study sites show minor interests in the dependency in natural resources, because their key focus is on local-linear flood alleviation schemes insteadly the use of multiple flood storages alongside the catchment. In this case, this research study is focussed on the differences between local authorities with or without the involvement of local citizens in the flood risk management process. The results show that 'community support groups' – like FAG – include a key role in the engagement process of local citizens. Therefore, I distinguished between middleclass and working-class communities. In general, I noticed a greater lack of local community support groups in working-class communities. They are more focussed in wealthy-rural middle class communities (i33eng). The consequences are that local authorities as well as local citizens are less involved in the management

process. Consequently, local authorities with local support groups show a privileged position. The consequences are that communities without local support groups tend to move the responsibility and the involvement towards county council level, for example the cases of Bridgwater. In Bridgwater, the key actors were the county councils (Sedgemoor for Bridgwater) in the negotiation process (i40eng; i41eng).

The second type of power dependency refers to the aspect of government support. The regional and national authorities have in the six selected study sites a key role in the management process. The technocratic experts support the local actors in various issues. An overall key objective is to reduce the vulnerability in the catchments as well as economic growth. As already mentioned the local actors are strongly dependent on regional and national authorities in terms of knowledge, planning permission and funding. In Austria, for example, the downstream municipalities have strong support from the government bodies. The government is committed to building a co-operation and management plan for the catchment to reduce the transaction costs with each individual solution (i10aut). Local actors in Morpeth and Cockermouth have been able to build a network relationship with Member of Parliament to ensure their needs and interests.

The third issue refers to education and knowledge characteristics. Both features play an important role in the interaction with regional and national authorities. Also other scholars underlined the importance of knowledge and education or social capacity to interact and to ensure their interests (Kuhlicke et al., 2011; Bues and Theesfield, 2011). Good knowledge, cultural and social capitals have been a key point in the interaction with the regional and national authorities. This has been shown in several empirical results.

Participant: 'they were quite intelligent people, professionals in their own right with different skill sets, and I had a meeting with them and they presented to me a cost proposal for constructing an overflow channel to take out a meander just downstream of the town and reduce flows that way, and it was quite professionally put together (...) there was a bit of a turning point that came about and rather than being a sort of lobbying group they almost became ... well they did become members of the project team, and we ran an appraisal and produced a business case' (National Authority, England).

The study sites with community support groups, such as FAG, show a higher interest and knowledge in the ongoing flood risk management discussion (see also section 6.4). This shows their interaction and network with the national Government and its bodies. Similar results have been found in education and knowledge characteristics, which play a key role in the social structure in flood risk management discussion. A key actor is Defra for capacity building at local level in England and Wales. However, the training oriented mainly to local authorities (technocrats), especially LLFA is to manage surface-runoff and not in improving the capacity of local citizens (Defra, 2010a). In this line, the community support group has a more critical role in the ongoing discussion. Subsequently, communities with local grassroots organisations tend to co-decide the local flood risk management plan, which is a key issue in partnership funding policy. The outcome will be that some local authorities are able to engage local citizens to get involved in the flood risk management process and to ensure their interests. The main challenge refers to the 'space of resistance'. The involvement of local citizens includes also the capacity and power to change the local flood risk management schemes, for example in Cockermouth. Similar developments and problems were identified with the new spatial and land use management planning policy in England (see also Haughton et al., 2013).

A further key aspect in the power discussion is the use of power in the decision-making practice. The use of power depends on the tasks, objective and decision-making practices in the policy discussion and social interaction (Theesfield, 2011; Sehring, 2009). The following table summarised the main results from the empirical investigations.

Table 7.27: Maii	n questions in t	he social interaction	between the	different act	ors and stakeholders
------------------	------------------	-----------------------	-------------	---------------	----------------------

Actors	Tasks	Specific decisions
Upstream-downstream municipalities	Flood storage	Where to build flood storage?
Inter-local co-operation-land owners	Land	Compensation for land use restrictions
	Clearing debris after flood event	
Between members of the inter-local co-operation	Financing	Who is paying, who pays less?
Inter-local co-operation with individual municipalities	Spatial and land use management planning	Restriction in the use of local spatial and land use management plans
Inter-local co-operation – regional and national authorities	Funding	Who is paying, who pays less?
Steering group – inter-local co- operation	Planning process	Who is involved? How are operational rules defined?
Land owner	Releasing water into the storage areas/water stored in flood storage area	Who is responsible for water released? Who is favoured?
Inter-local co-operation-municipalities	Local-linear flood alleviation scheme	Who refers from paying, who pays less?
County council – private actors	funding	Who refers from paying, who pays less?
County council – FAG	Planning process	Who is involved? How are operational rules defined?
Local actors – national authority	funding	Who pays, who pays less?
Private - private	Funding	Who refers from paying, who pays less?

Source: adapted from Theesfield (2011): 92

Local actors need the support (financial and knowledge transfer) from the regional or national authorities. Furthermore, the discussed study sites show a high dependency of regional support in terms of negotiation with local land owners. The dependency is based on formal and informal rules. The regional authorities lead the communication and decision-making process. The domination changed with the self-realisation of local actors in the flood risk management discussion. In Aist for example, the interaction between the different local and regional actors is based on a more equal level, because the local actors show interest and involvement in the decision-making process. The situation in the Triesting-Tal study site is different, where the regional authorities dominate the discussion and the decision-making process (see section 6.4). On the other hand, regional authorities

have the permissive power for flood risk management. The local authorities depend on the final decision from the regional authorities. Without the permission from the regional authorities, the local actors have a limited scope in the flood risk management discussion (see section 4.2.3). Power dependence shows a key factor for the creation of partnership approaches for the creation of partnership approaches. This places implicates communities in a weak power position, for example downstream communities in an inter-local-co-operation. Implications are that the partnership is more costly for the downstream communities.

### 7.2.2. Resource dependency

In the Austrian study sites, the downstream municipalities strongly depend on the availability of land in the upstream municipalities. Lee et al. (2012) and Rössl (2005a) argued the resource dependence as a key factor in the relationship between the different actors. A shortage of land for the implementation of flood storages has been an important stimulation in the development of the inter-local networks (i19aut; i5aut; i12aut; i15aut). The resource dependency increases, when the inter-local flood risk management plans transfer security benefits from the upstream towards the downstream local municipalities. A second resource dependency was identified in the technical knowledge. Here, the local actors are strongly dependent on the regional authorities (see also section 6.4). A third source is the economic dependency in the region. The selected study sties show 'classical' peri-urban / rural economic difficulties. In the regions there are only a few medium and large-sized enterprises. In Schwertberg (Aist study site), for example, the 2002 flood hardly affected the two main local enterprises, which have an inter-local economic influence. This affected the starting of the inter-local network (see also section 5.1). Similar results were identified in England, where insurance challenges stimulated the collaboration between local and regional actors.

## 7.3.Barriers and limitations of adopting a partnership approach

Main barriers are the conflicts of provision of land in the upstream area for flood protection for the downstream municipalities and compensation payments. The Austrian study sites, for example, show the 'classical' upstream-downstream conflicts as a main barrier and challenges in the

development of partnership approaches in flood risk management. A key problem is lack of win-win negotiations. The upstream municipalities gain no or less benefit from the planned and implemented flood alleviation schemes. On the other hand, downstream municipalities are more willing to implement local flood alleviation schemes instead to compensate the upstream municipalities. Similar to the English study sites, the question of funding has been a central point in the creation of partnership approach. The second key barrier refers to the mistrust and concerns between local and regional-national authorities. In particular, the EA had strong concerns to co-operate more intensely with local actors and stakeholders in the flood risk management planning. However, the partnership funding requested a change in the interaction between the EA and local actors and stakeholders. The results have demonstrated that the interactions are still strongly organised in a top-down way. Examples are the actual national contribution (FDGiA) and top-down regulation in neighbourhood planning and new community infrastructure levy. On the other hand, the partnership funding and neighbourhood management planning allows local communities to gain a stronger position in the decision making process. A key problem refers to the 'local trap' or local lock in with respect to decisions at local level as well as uneven developments between communities with a local management plan and communities without any organisational structures. Furthermore, an important influence has been the local interests and risk perceptions. Local authorities have no or fwe interests or low understanding of flood risk. In summary, the risk perception shows a negative correlation in the negotiation process. A fourth problem refers to the co-operation itself. The partnership approach shows no day-to-day business for the involved partners. The co-operation requests high transaction costs, especially in the creation and beginning phase, because the members of the co-operation had never worked together in a previous project or had had no common historical developments and overlaps. These factors show strong barriers towards the creation of a partnership approach in flood risk management.

## 7.4.Key findings

The empirical results are currently characterized by a lack of sufficient partnership approaches with the different actors and stakeholders. Some of the study sites show a higher integration in one field than in others. Furthermore, the interviews show strong barriers and conflict in the discussion between regional professionals and local politicians, based on lack of technology knowledge and expertise (politicians). Third aspect refers to the initiation process. The Aist, Morpeth and Cockermouth study site, in contrast to the other study sites, include a strong bottom-up approach in the development of a partnership approach. This answers the aspect, why there the partnership approach shows a high interest in the overall strategy planning process as a successful factor in the implementation process. Policy actors should try to encourage the co-operation to take over more responsibility as well as to close the gaps and barriers within the co-operation. However, the results have highlighted a splintering of the flood risk management system. Graham and Marvin (2001) defined splintering of infrastructure as a 'spatially selective' (ibid: 197) policy and development. The authors argued that the splintering infrastructure policy influenced also the social relations, mainly by an exclusion of areas with low-income from flood risk management. The outcomes are 'infrastructural bypasses', which 'refer to the development of new parallel or substitute infrastructure networks that facilitate the development of contestable markets' (ibid: 168). Consequently, the results are that we have a 'splintering' nation, where municipalities with local resources, such as knowledge, experiences, income or language, have been privileged in flood risk management policy. Graham and Marvin (2001) described the transformation of a homogenising policy towards the creation of 'hubs' across the country. In England, for example, flood risk projects were realised mainly because of high interest for the County Council or strong local pressure group whilst excluding areas missing necessary frameworks. This challenge increases when the demand is weak or declining (in shrinking processes).

In this chapter, I presented the results of analysing the proximity in flood risk management in both countries. The main findings are:

- A central obstacle is the low involvement of the Federal States in Austria in local spatial planning concepts as an advisory to control the spatial and land use management acts for flood risk management. Consequences have been limited exchange and harmonisation of local spatial and land use management plans.
- Two of the selected study sites in England developed local neighbourhood land use management plans with local actors and stakeholders. However, the focus is oriented towards housing, local economy, heritage and local environment. In general, the aspect of floods plays a minor role in the overall plans.
- In Austria based on the political structure, this thesis demonstrated various spatial misfits between the flood risk management plans and flood hazard.
- The three study sites in Austria follow a clear inter-local flood defence strategy in contrast to the English empirical results. Analysing the catchment management plans, they encourage the achievement of multiple benefits from the strategic proposals, e.g. by increasing biodiversity or by improving the river continuity. In respect to flood risk management, the overall goals are the implementation of flood storages in the catchment. However, the three study sites still have gaps in the implementation of post-flooding measures.
- The involvement of private actors in the co-operation process has encouraged other local authorities to take part of the inter-local co-operation. However, from the interviews it emerged that the negotiation process was longer and more complicated, because of the different interests. Throughout, past co-operation between the regional authorities, local authorities and private actors were given in the region. These experiences helped to identify and to encourage the private actors to take part in the inter-local co-operation.

## 8. Conclusion

Flood risk management has changed over the past decades. The 'classical' flood management approach includes a strong focus on engineering solutions at local level. The changes in the management system - from a security-based approach towards a risk-based approach - includes a combination of structural and non-structural measures, interdisciplinary working and catchmentbased approaches. This enforced also a new focus on the implementation of flood alleviation schemes, especially to take into consideration their spillover-effects in flood risk management, such as linkages to other intra- and inter-regional networks. The aim has been to achieve a more holistic policy (Kubal et al., 2009; Johnson and Penning-Rowsell, 2010). The latest changes include a shift in flood risk management perspectives in Austrian and England. Key developments, such as financial and economic developments and ideologies, show a strong influence on the actual discussion and policy framework, especially the shift from the priority on agricultural productivity and food security towards economic growth and as latest step outsourcing of the flood risk management. Overall, policy ideologies influenced the transformation process in flood risk management and the governance structures in the countries as shown in this thesis. With the downscaling of power and responsibility in the flood risk policy towards local actors, I observed new scalar architectures and new governance structures in state regulation (Brenner, 2004). The empirical results show new governance structures, mainly based on the new politics of scale (see also chapter 7). In England, the national organisations (Defra and EA), have dominated the flood risk management system in terms of funding, planning, implementing and maintenance. Partnership Funding, for example, has enforced a re-design of the involved actors and stakeholders in the policy-decision practises and question of responsibility. I found similar developments in the Austrian study sites, where the interlocal partnership approach changed the decision-making practice. A key aim is to develop an interlocal flood risk management approach, using agricultural land use for flood protection. Subsequently, the number of actors has increased in the decision-process from two actors (one local authority and regional authority) to several state and non-state actors (see also section 4.2). However, the actual

scalar arrangements in the Austrian and English flood risk management policy demonstrate that the central state remains completely dominant, in terms of funding and setting the rules.

The new policy requests a stronger involvement of local actors in the actual policy discussion. I have argued that the dynamics in the production of scales allows also the changes of interaction of local actors (scale jumping), especially by the FAGs. In terms of Cox (1998), the space of depending for the FAG is parish council level, where the members define and influence their specific conditions, e.g. design of flood defence scheme; and the regionally and nationally level as the space of engagement, where the FAG interact to ensure their interests. New actors have closed the political vacuum to ensure and to satisfy their needs and interests. Subsequently, the network as well as the knowledge and the habitus have become a central factor in the policy process. The consequences were that the introduction of the partnership funding scheme led to a shift in the state spatial process (Brenner, 2004), away from a single-equal spatial strategy towards a multiple – individualised approach, where capability of local actors can secure their interests and needs.

An interesting observation is that the local authorities, especially the downstream communities, enforced an overall flood risk management policy discussion in the catchment; based on a strong pressure from the local businesses for the relocation of their businesses in area (see also section 4.2.4). Consequently, the relocation of these businesses would cause strong negative socioeconomic impacts to the region. Therefore, a key driver was the downstream community Schwertberg in the Aist study site. Based on the Aist regional studies published in 2004, the regional authorities in co-operation with the downstream communities started to re-organise their inter-local co-operation. The catchment management plan foresees the implementation of inter-local flood defence measures, e.g. flood storages, forest management plan. The inter-local co-operation shows in contrast to the other two study sites a more active role in the planning and the strategic processes. Especially the steering group shows an active role in the overall implementation process. Moreover, they initiated individual strategies and planning concepts (see section 6.4). However, in respect to the identification and integration of new members, a symmetric relationship between the regional and local authorities is present. In the initiation phase, the regional authorities as well as some downstream communities have forced the communities to take part in the inter-local co-operation. The co-operation process, however, does not include any changes in the formal legislative competences, but a shift in the power structure. In England, for example, I identified a shift of power structure from local politicians towards local actors like FAG. On the other hand, the Austrian inter-communal co-operation enforced a shift in the power structure from local municipalities towards the inter-communal board. This has been especially the case for negotiations with local land owners, communication and contract design. Nevertheless, the inter-local co-operation has no power in the local spatial and land use management plans (i13aut). I have found that political mobilisation at the local level is often driven by economic reasons, such as business interests (e.g. Schwertberg), new residential properties (e.g. Leobersdorf or Bridgwater) or flood insurance (e.g. Cockermouth or Morpeth). At the same time, the engagement of local actors, stakeholders and citizens depends on the economic, cultural and social capital, such as habitus, or house demand.

In England, I identified a shift of the power structure from local politicians towards local actors like FAG. The Austrian inter-communal co-operation enforced a shift in the power structure from local municipalities towards the inter-communal board. This has applied especially for negotiations with local land owners, communication and contract design. Nevertheless, the inter-local co-operation has no power in the local spatial and land use management plans (i13aut). I have found that political mobilisation at local level is often driven by economic reasons, such as business interests (e.g. Schwertberg), new residential properties (e.g. Leobersdorf or Bridgwater) or flood insurance (e.g. Cockermouth or Morpeth). At the same time, the engagement is cultural and social embedded, such as habitus at local level or house demand. However, future limitations in the localism approach refer to the introduction of the new community infrastructure levy by 2014. The new community infrastructure levy includes various restrictions in the introduction of a new local levy, e.g. 'roof taxes'. Furthermore, the infrastructure levy includes using the levy for all infrastructure projects, e.g. streets, schools, affordable housing etc., instead of one propose like the Bridgwater 'roof tax' for

flood risk management. Additionally, within the introduction of the new levy system the amount of the levy is based on the economic viability of the local authorities and it will be independently reviewed. In the case of Bridgwater the economic viability is very low. This may increase the fear that the Government suggests to request quite low levels of contribution from the citizens and no charges from any commercial developments, which would stop the tax collection for the town. These difficulties in respect to the community infrastructure levy the Government follows a strict top-down localism with stronger restrictions for the local authorities. With the fear of a step backwards towards 'top-down' philosophies like the New Labour model (Haughton and Allmendinger, 2013). In both countries the rescaling process emphasizes a shift across the already existing scales in a minor sense. The main changes were observed in the introduction of new scales, such as meta-district governance arrangement in III-Walgau or the Aist co-operation or new scales like Bridgwater, Cockermouth or Morpeth. The consequences and impacts were a change in the regulatory settings and administration practice. The regional authorities, for example, had to develop catchment-wide management plans instead of local solutions, which need different modelling techniques and data (Evans et al., 2002). Influentially, the impact was also the interaction with more than one actor in the planning, design and implementation process instead of one actor (one local authority).

## 8.1. Answering the research questions

The following section addresses the research questions (see section 1.4).

#### Question 1: Which mechanism and action influence the change of the actual scalar arrangements?

In the past two decades, we observed a shift in the scalar re-arrangements in flood risk management policy. Due to the recent developments the role of the state and individual responsibilities for flood risk management are now under review (Defra, 2012). This review has been driven by various developments: financial and economic crises (Defra, 2012; Featherstone et al., 2012), implementation of new legal requirements (EC, 2000; 2007) and recent flood events in England and

Wales (Pitt, 2008; Defra, 2005). The new policy agenda tried to enhance responsibilities of different actors in flood risk management and reduce the controlling role of central governments. This implies a change in the relationship between the different actors (Pearce and Ayres, 2012) including the creation of a 'new state space' in the policy discussions (Brenner, 2004). I identified, the re-scaling process enforced a change in the social structure and interaction between the different actors with new working relations and interactions. First, the selected countries show an increase in the number of actors in the ongoing policy discussion (equalling a new governance framework). Second, the new scalar re-arrangements empower local actors. Third, the local actors are more involved in the decision-making process. However, I have showed that the outcomes and effectiveness of this process strongly depends on local capacity to use these opportunities. In England, for example, I have identified that especially the middle class professional groups with expertise and skills (e.g. planners or lawyers) have gained from the new policy direction. In summary, the scale jump empowers local actors to get stronger involved in the policy discussion. However, this strongly depends on the local interests, identification with the project and idea and resources (capabilities). This can be observed in both countries. The escapes from the scalar fixes put the local actors in a stronger position in the management discussion. However, the co-operations are not fully using this power and position. One reason is the interdependences between the different actors regarding to interests, objectives and capability as well as the problems of parochial and particularised local policy directions (Kitchin and Wilton, 2003). On the other hand, Morpeth and Cockermouth are using the scale jumping in the reorganisation of social interactions to gain power from national level to achieve their desired interests. The actors show a high active role in the actual policy interaction.

Furthermore, the scalar re-arrangement encouraged a new organisational culture, skills, knowledge and competences (Clifford, 2013). The remaining question is if the new territorial scale in flood risk management is more than only a decentralisation of responsibility. In sum, the new territorial scale encourages local actors to get involved in the ongoing decision-process. This included a new space creation for local actors, which included a shifting of power towards non-state actors such as FAG (Allen and Cochrane, 2010). Subsequently, the funding system in England foresees the promotion of wealthy-rural areas, where local actors want and have the experience and capacity to wield it. Nevertheless, this study shows that the 'successful' implementation of new policy strategies refer to questions of local capacity and legitimacy of the involved actors, especially to the FAG (McCarthy and Prudham, 2004). The change is that communities who fail to achieve these criteria will lose out from this new policy (MacLeod and Goodwin, 1999).

## Question 2: How cross-scale dynamics have influenced national flood risk management strategy?

The English results show a strong local involvement in the flood risk management planning in terms of funding, planning and decision-making practices. The localities show high interest and capacity to have a strong voice in the decision-making practices. In particular, the non-state actors have achieved a very important role in the flood risk management system (outscaling). The empirical results show that local actors and stakeholders use network connections at regional and national level to intervene to ensure their interests and needs are realised (scale jumping). The outcome has been a change in the power construction in favour of local elite groups. The impact strongly depends on the study sites and local circumstances. The Austrian results focus more on changes in the boundaries construction and flexibility in the planning process. The use of 'soft spaces' play a crucial role in the planning and implementation process, because of the challenges and conflicts with local land owners. The regional authorities and inter-local association have to update regularly their concepts and management plans. On the other hand, 'fuzzy boundaries' are a central aspect in the Austrian study sites. The objective is to implement inter-local flood alleviation schemes to reduce the vulnerability in the sub-catchments. The Austrian flood risk management policy at regional level favours formalised sub-catchment flood risk management plans, contrary to England. However, also here it depends strongly on the Federal State. Although the national Government provides legal frameworks for inter-local co-operations, the national authorities do not only support catchmentwide flood risk management.

National governments play a central role in new scalar re-arrangements. The central government defined the change of role and relationship between different scales, however failed to answer questions of regulatory capacity and power at local level (Brenner and Theodore, 2002). Until today, new territorial scales debates are focused on emphasising lower public authorities to encourage their role and responsibility in policy process as a main characteristic. These include a shift of responsibility and duties between national and local bodies, but often without a change of legal framework or shift of additional resources as well as power between different scales (Coaffee and Johnston, 2005).

In summary, both countries are characterized by main differences in the direction of networks (vertical-horizontal). Main dissimilarities between Austrian and England are especially based on the institutional natures, above all in the formal rule settings and administration practices. First, the Austrian study sites include a fixed formalised rule system for the creation of partnership approaches. Secondly, the regional authorities (WLV and BWV) provide various instruments for inter-local solutions in flood risk management as can be seen in the administration practices, such as the Regional Study as a planning instrument. Third, a key issue is the lack of interest of the EA to introduce inter-local flood risk management plans to avoid conflicts and discussions with local authorities. In England, the politics of change enforced a new policy strategy for the national authorities (especially EA) to interact with local actors and stakeholders. Also, partnership funding enforced the local scales to get more involved in the decision-making practises. This meant that local actors and stakeholders influenced the policy and implementation process.

# Question 3: What are the barriers and limitations of new scalar re-arrangements in terms of network governance, and what are their social and political functions and consequences?

Local engagement strongly depends on their social capacities, such as knowledge, motivation/selfinterest, economic, network, organisation and procedural (Kuhlicke et al., 2011). Nevertheless, the results show generally that the role of local partners has become more relevant in the discussion. The EA had to open their discussion to the local stakeholders more than by a consultant approach. The new partnership funding offers the possibility and opportunity to bring more local solutions and ideas into account (e.g. Cockermouth). The new funding policy can be seen as an important driver in the developing of a vertical partnership approach, where the policy engages more effectively nonstate actors in the process like the new spatial planning framework (Haughton and Allmendinger, 2013). Furthermore, partnership funding is well embedded in the political ideology of new localism approach as well as 'hollowing out' policy of the Government.

Co-operations can be seen as an ideal instrument for stakeholder engagement in flood risk management. Throughout, with current pressures on local authorities to reduce the spending and in parallel, a reduction in the central state's resources, co-operations has been seen as a possibility to both increase the 'value' of budgets available and to increase efficiency in using current public funds and resources. Furthermore, they may allow direct and indirect benefits including harmonisation of spatial and land use management plans. From the interviews most mentioned the advantages (table 8.28) are the access to additional resources (e.g. funding, land).

First advantages mentioned from the interviews are the possibilities to pool the budgets of local authorities. Due to the reduction of local spending, a co-operation allows savings to be made. Furthermore, the financial burdens are generally lower than for individual solutions. Furthermore, the overall financial volume is higher, which leads to a higher economic power, pressure and a higher potential to realise more flood alleviation schemes at the same time. In respect to small communities, the inter-local co-operation approach is the only possibility to realise any flood defence measures because of the contribution of local authorities in partnership funding scheme. Secondly, the co-operation can ask for additional financial resources more easily. The Aist-co-operation, for example, had access to European – Regional Development Funds. However, rural development programs need a formal inter-local co-operation. The third advantage is the more efficient use of the financial resources and planning instruments. In the catchment-based planning or sub-catchment based management planning, regional and local authorities are more keen to

avoid contract defence measures, equal for spatial and land use management plans as well as emergency management plans. For example, the inter-local co-operation has the possibility to ensure natural retention or flood storage areas in the upstream communities, which is less expensive than individual solutions for each community.

Advantages of inter-local co-operation	Disadvantages of inter-local co-operation
Enable a pooling and sharing of monetary resources	Only re-imagination of fair sharing of balance of power
Access to additional funding schemes	Cost-intensive and time-consuming (higher transaction costs)
Efficiency use of financial resources	Risk of failure and insure outcome
	Disaster driven
	Imagination of fair sharing of costs and risk burdens

I UNIC 0.20. Advantases and disadvantases of coloscilations in nood hisk management	Table 8	.28:	Advantages a	d disadvantages	of co-op	erations in	flood risk	management
---	---------	------	--------------	-----------------	----------	-------------	------------	------------

Source: own development

Nevertheless, besides the advantages of co-operations, the literature and results from the interviews listed a wide range of potential risks and disadvantages. In general, a co-operation is not a 'business as usual' situation for the different members due to the high risk for each member (van Huijstee et al., 2007). Firstly problems related to a fair sharing of power between the different actors. The key problem within a co-operational approach is the shift of power to a steering group, who manage the co-operation. For example, usually 'large' communities are members of the steering group in the inter-local co-operation agreement. The consequence is that 'small' members often have less power in the overall decision process. A key aspect is to reduce the power of 'large' members, e.g. supermajorities are required for certain decisions. A second problem refers to higher transaction costs (Hagspiel, 2001), especially in the short period (see also section 7.1.3).

Question 4: What is the role of proximity in shaping the opportunities and barriers for partnership activities in flood risk management, and what are their social and political functions and consequences in flood risk management?

By analysing the empirical results, scalar arrangements also influence the interaction and engagement between the different actors, stakeholders and citizens in the policy decision-practices. Nevertheless, the empirical results are currently characterized by a lack of sufficient co-operation between the 'local' actors, stakeholders and citizens as well as with the regional authorities.

While examining the partnership approaches in the different study sites, the example of Cockermouth showed a move beyond the participation or partnership approach. Specifically, using the term co-production adopted by Albrechts (2013), Cockermouth now follows a co-production instead of a partnership arrangement. 'Coproduction looked upon as the organized practice that constructs and managers a 'civic realm' (Roy's term, 2009: 160) equips citizens' organizations to challenge, resist and comply state governmentality' (Albrechts, 2013: 51). The main outcomes are common knowledge of production and exchanges, new settings for self-rule norms and populist mediation, where organisations like FAG act and present itself as 'bridging organisations' between state and civil society (Roy, 2009). From the empirical results, I found few changes in the power construction between national, regional and local actors and stakeholders. In particular, study sites with no or little local engagement had minor knowledge transfers between the different scales. The objective was to keep a hierarchical relationship based on specific – technical knowledge at national or regional levels to ensure their position in flood risk management policy. The national and regional authorities established a powerful position. The outcome was to use specific languages and knowledge with the consequences to stabilise power positions by national and regional authorities as well as organisational justification and legitimating to act. In the study sites the discussion between national, regional and local scales changed to a more equalised interplay between the different actors. Also between the new scales I observed different power constructions. Scales with 'correct' habitus (see section 6.4), which is the nature of practices in my case in decision-making processes, influenced the planning process for their purpose and interest (see section 7.1). Binder (2012) identified cultural and social capitals as resources, where actors and stakeholders are qualified to interact for to interact with 'externals' at different scales.

Thus, an asymmetric relationship seems to be dominated by the monopoly position of the regional authorities in relation to technical knowledge and power. The empirical results indicate a low willingness to request more tasks (responsibility and power) from the regional authorities. One reason is that the local authorities expected that the regional authorities are to be responsible for defining and organising the flood risk management in the region with the impact that the rescaling process enforced new boundaries, but without engaging with the local scales which shows only an imagination of more democratic processes (Fatti and Patel, 2013; Razafindrabe et al., 2014; Allen and Cochrance, 2007). Nevertheless, in the literature the main barriers to efficient rescaling processes include the gap in the transfer of power and resources to ensure the local capacities to manage the new tasks (McCarthy and Prudham, 2004). However, the empirical results demonstrated that a further key barrier is the lack of interest at local scales to use the possibility to engage in flood risk management policy. In sum, all study sites show an asymmetric relationship between the national and regional authorities (EA, Defra, WLV and BWV) and members of the inter-local cooperation. Jessop (2004) referred mainly to two key aspects: (1) society, especially different agents, acts interdependently from each other and (2) possibility of different linkages between different scales (vertical, horizontal, and diagonal). The regional authorities have the key role in the overall planning and strategic management process. Within this concentration of power towards the regional actors, the relationship is dominated by the monopoly position of the regional authorities based on technical knowledge and power (Foucault, 1982). With respect to floodplain dwellers, flood risk management is not on everyone's top-priority list. In general, flood risk management plays a less important role in current society debates, also in relation to stakeholder processes in flood risk management (i19aut; i26aut). This includes the minor interest of engagements in public policy debates. Other key problems are the mistrust in public administration as well as asymmetric information levels. However, 'community based disaster risk reduction initiatives always depend on such established community owned institutions' (Surjan and Shaw, 2009: 430). The authors underlined the focus on the given organisational and institutional structure to understand the success or failure of stakeholder engagement process.

## 8.2. Social impact and further research

A number of problems, concerns and challenges have been addressed in this thesis. A key problem refers to the question of fair sharing of risk, power and responsibility between the different actors, stakeholders and citizens. This remains often unclear or in favour of national or regional authorities, such as financial risk-sharing in the English partnership policy. In particular, the English study sites show a lack of transparency in the financial transactions. A central question refers to the use of local financial contributions. At the moment, the financial contributions from the localities have been mainly used for capital works. The strong focus on capital works and financial squeeze for maintenance shows a greater focus on new flood alleviation schemes and less on future requirements. In overall, I found no evidence that local policy decisions are more economically efficient (e.g. Cockermouth or inter-local co-operation) or social equitable (e.g. Morpeth) in comparison to regional or national decision-making practises. A clear hindrance and barrier is local level itself, which has become a central policy target in the new politics of scale. This included also a stronger financial contribution of local and non-state actors in flood risk management, especially the empirical results from England show a 'hollowed-out' effect based on local and (especially) non-state contribution means a crowed-out effect of national contribution, because of the expectation from the national Government that the private sector will close the financial gap. In sum, an increase in the private contribution can increase the national expectation to reduce the national budget with the outcome that the private sector has 'crowded-out' the national Government. Nevertheless, the actual development shows no evidence that the local and non-state sectors are able to fully fill the financial gap based on the fiscal squeeze and spending cuts (see latest announcement from the national government (HM Treasury, 2013)). The empirical results from the two selected countries demonstrated an explorative approach to understanding the new politics of scale development in Europe. In sum, I observed two main developments in flood risk management, mainly vertical and horizontal partnership approaches. The selection of partnership approaches and directions are mainly based on institutional settings and frameworks as well as national and regional flood risk policy. Flood risk management can be described as a result of political discussion and outcomes broader than technical and engineering solutions, concepts and ideas.

The partnership concept is a central challenge in the discussion and realisation. Key challenges refer to the questions of who is the promoter of the co-operation, who is monitoring the co-operation approach and how to engage local stakeholders into the process. In particular, no or little interest of the community includes a strong barrier in the building and involving in the co-operation. The selected study sites are often unclear about how to engage local communities to participate in the decision-making practices. The empirical results confirmed that regional or national authorities have not developed any effective instrument for community engagements; mainly the local stakeholders engage themselves. Communities with no or low interests, such as the Bridgwater or the Triesting-Tal co-operation, are not involved in the decision-making practices. The outcome is that the regional and national authorities need a local community with existing structures and interests. Some of the selected study sites show the capacity of local actors and stakeholders to understand and to organise local initiatives. However, key preconditions are local social and cultural capitals. Lack of these conditions implies great difficulties, if regional or national authorities downscale tasks and responsibilities to lower public authorities. In Austria, the study sites do not show any engagement of local stakeholders in the policy decision, except in a few cases in the Aist study site. In England, mainly local elite groups have been involved under the umbrella of FAG. This raises the question of legitimacy, especially for the non-democratic elected FAG in England. A further challenge refers to the involvement of local ideas and concepts in the policy process. In England, for example, the localism policy is more rhetoric instead of including local solutions in the flood risk management. Examples are the restriction of the development of new financial instruments or development of neighbourhood spatial and land use management planning, where both instruments need the formal approval from national authorities. In fact, the new localism policy limits local freedom, how to

organise and to ensure the local contribution or development of local strategies. Especially low income communities have not been allowed to introduce or increase taxes or not to follow an economic growth strategy. Moreover, the administration burdens for local actors and stakeholders. Problems of accountancy are based on quantitative data for benchmarking processes (Swyngedouw, 2011). A key question refers to social justice and equity, which plays a central role in the development of partnership funding. In Morpeth, for example, the County Council has closed the financial gap by placing financial restrictions of further projects. The County Council are not planning to close every partnership funding project planned in the area. This is a clear problem for communities with a lower average income compared to Morpeth. The main reason for this is the strong interlink between local FAG in Morpeth and Northumberland County Council. Finally, the partnership approaches have various advantages and disadvantages, but often policy makers and academic scholars see them optimistically. Regional and local authorities have a more realistic view of the negative effects of a co-operation.

The empirical results show that some scales were privileged, which can be interpreted as scale gains from the scalar re-arrangements to ensure – in terms of Allmendinger and Haughton (2009) – their hegemonic status, mainly by using specific strategy in negotiation process with national and regional authorities. Following Allmendinger and Haughton (2009) a further central discussion point is not only the questions of 'gainers' and 'losers' in scalar re-arrangement, but also the interaction between the scales. In this thesis, I analysed the interaction between scales in terms of governance. The scalar re-arrangements, especially in England, involved in the policy decision-making practices of various public stakeholders, which normally were never involved in the past. This included more resource intensive negotiation interactions because of an increased number of actors and stakeholders acted as 'partners' or lobbying organisations to influence the decision-making process for their 'community' with no wider exploration at regional level. The scalar re-arrangement has had a strong influence in the governance arrangement and settings, but also in the decision-making

system, where different actors and stakeholders have used new 'strategic and tactical manoeuvre' (Allmendinger and Haughton, 2009: 631) to ensure their interests. An interesting observation from the empirical results is the involvement of grassroots organisations in the planning process. The involvement of the FAG, for example, has had a strongly influence in the planning process, where the new scalar arrangement has moved from a 'classical' hierarchical planning approach towards exercises of 'soft space experiments' (Haughton et al., 2013). The outcome was a shift from elected politicians decisions towards non-state actors decision making-processes with disadvantages of highly localised planning process, increased charity hazard problems and the potential of a environmental racism policy as a consequence (Elliott and Pais, 2006), where low-income and minority communities, which lack social and cultural capital and capacity as well as economic power or national strategic interests were excluded from the policy process or not fully integrated.

Overall, the general lessons are:

- communities with higher socio-economic standards are more likely to ensure their interests in the new scalar arrangements, because of their social and cultural capital capacities – leading to social inequity;
- local communities without this capacity have a lower voice in the decision-making process with the consequences to get less attention by the regional or national government;
- local champions play a limit role, because of the few or no interaction with other communities or inter-local co-operations are given; the study sites have no or limit exchange with others; and
- central challenge is the connection and integration of localities in the policy discussion and decision-making practice.

Nevertheless, there is need for further research studies in this field. A key question refers to the impact of modes of governance towards the creation of partnership approaches in flood risk management. Also the aspect of social justice, transaction costs, accountability and legitimacy in

partnership approaches need more in-depth studies. In reference to the implementation of flood storages, there is a strong need of further research in the understanding of endowment – effects between public authorities and farmers as well as their motivation, barriers and challenges of the implementation of flood storages at their land.

# References

ABI (2005): *ABI Statement of principles on the provision of flood insurance*. Update version, November 2005. http://www.salford.gov.uk/d/abistatementofprinciples2005.pdf [21/01/2014].

Abrahamsen, R. (2004): The power of partnership in global governance. *Third World Quarterly*, 25 (8): 1453-1467.

Adger, W. N., Quinn, T., Lorenzoni, I., Murphy, C., Sweeney, J. (2013): Changing social contracts in climatechange adaptation. *Nature Climate Change*, 3 (4): 330-333.

Albrechts, L. (2013): Reframing strategic spatial planning by using a coproduction perspective. *Planning Theory*, 12 (1): 46-62.

Aligica, P. D. (2006): Institutional and stakeholder mapping: frameworks for policy analysis and institutional change. *Public Organisation Review*, 6 (1): 79-90.

Allen, J. (2009): Three spaces of power: territory, networks, plus a topological twist in the tale of domination and authority. *Journal of Power*, 2 (2): 197-212.

Allen, J., Cochrane, A. (2007): Beyond the territorial fix: regional assemblages, politics and power. *Regional Studies*, 41 (9): 1161-1175.

Allen, J., Cochrane, A. (2010): Assemblages of state power: topological shifts in the organisation of government and politics. *Antipode*, 42 (5): 1071-1089.

Allmendinger, P. (1998): Planning practice and the post-modern debate. *International Planning Studies*, 3 (2): 227-248.

Allmendinger, P.; Haughton, G. (2007): The fluid scales and scope of UK spatial planning. *Environment and Planning A*, 39 (6): 1478-1496.

Allmendinger, P., Haughton, G. (2009a): Soft spaces, fuzzy boundaries, and metagovernance: the new spatial planning in the Thames Gateway. *Environment and Planning A*, 41(3): 617-633.

Allmendinger, P.; Haughton, G. (2009b): Critical reflections on spatial planning. *Environment and Planning A*, 41 (11): 2544-2549.

Allmendinger, P., Haughton, G. (2010): Spatial planning, devolution, and new planning spaces. *Environment and Planning C: Government and Policy*, 28(34): 803-818.

Allmendiner, P., Haughton, G. (2012): Post-political spatial planning in England: a crisis of consensus? *Transactions of the Institute of British Geographers*, 37 (1): 89-103.

Amdam, J. (2001): Structure and strategy for regional learning and innovation, in Voigt, A., Walchhofer, H. P. (eds.): *Interkommunale Planung*. Institut für Örtliche Raumplanung, TU Wien, Vienna, Austria: 5-22.

Amin, A. (2005): Local community on trial. Economy and Society, 34 (4): 612-633.

Ansell, C., Gash, A. (2008): Collaborative governance in theory and practice. *Journal of Public Administration Research*, 18 (4): 543-571.

Arrighi, C., Ferrario, R. (2008): Abductive reasoning, interpretation and collaborative processes. *Foundations and Science*, 13 (1): 75-87.

Asheim, B. (2012): The changing role of learning regions in the globalizing knowledge economy: a theoretical re-examination. *Regional Studies*, 46 (8): 993-1004.

Bache, I., Flinders, M. (2004): Multi-level governance and British politics, in Bache, I., Flinders, M. (eds.): *Multi-level governance*. Oxford University Press, Oxford, United Kingdom: 93-106.

Bachrach, P., Baratz, M. S. (1962): Two faces of power. The American Political Science Review, 56 (4): 947-952

Bäckerstrand, K. (2006): Multi-stakeholder partnerships for sustainable development: rethinking legitimacy, accountability and effectiveness. *European Environment*, 16 (5): 290-306.

Baker, M., Wong, C. (2013): The delusion of strategic spatial planning: what's left after the Labour government's English regional experiment? *Planning Practice & Research*, 28 (1): 83-103.

Ballinger, R. C. (1999): The evolving organizational framework for integrated coastal management in England and Wales. *Marine Policy*, 23 (4-5): 501-523.

Baumol, W. (1952): *Welfare economics and the theory of the state*. Cambridge, MA: Harvard University Press, United States.

Baxter, J., Eyles, J. (1997): Evaluating qualitative research in social geography: establishing 'rigour' in interview analysis. *Transactions of the Institute of British Geographers*, 22 (4): 505-525.

Benson, D., Jordan, A. (2010): The Scaling of Water Governance Tasks: A Comparative Federal Analysis of the European Union and Australia. *Environmental Management*, 46 (1):7-16.

Berkes, F. (2010): Devolution of environment and resources governance: trends and future. *Environmental Conservation*, 37 (4): 489-500.

Beutl, H. (2010): *Regional Governance und Regionalplanung. Zwei Fallbeispiele aus Niederösterreich*. Universität Wien, Vienna, Austria.

Bevir, M., Rhodes, R. A. W., Weller, P. (2003): Traditions of Governance: interpreting the changing role of the public sector. *Public Administration*, 81 (1): 1-17.

Bialasiewicz, L., Giaccaria, P., Jones, A., Minca, C. (2013): Re-scaling 'EU' rope: EU macro-regional fantasies in the Mediterranean. *European Urban and Regional Studies*, 20 (1): 59-76.

Binder, G. (2012): Theory(izing)/practice: The model of recursive cultural adaptation. *Planning Theory*, 11 (3): 221-241.

Birner, R., Wittmer, H. (2004): On the ,efficient boundaries of the state': the contribution of transaction-costs economics to the analysis of decentralization and devolution in natural resource management. *Environment and Planning C: Government and Policy*, 22 (5): 667-685.

Blagescu, M., de Las Casas, L., Lloyd, R. (2005): *Pathways to Accountability*. The GAP Framework. One World Trust, London.

Blaikie, N.W.H. (1991): A critique of the use of triangulation in social research. *Quality and Quantity*, 25 (2): 115-136.

Blatchford, P. (2005): A Multi-method Approach to the Study of School Class Size Differences. *International Journal of Social Research Methodology*, 8 (3): 195-205.

BMLFUW (2010): Leben mit Naturgefahren. Ratgeber für die Eigenvorsorge bei Hochwasser, Muren, Lawinen, Steinschlag und Rutschungen. BMLFUW, Vienna, Austria.

BMU (2005): *Gesetz zur Verbesserung des vorbeugenden Hochwasserschutzes*. Bundesgesetzblatt Jahrgang 2005 Teil I Nr. 26, ausgegeben zu Bonn am 9 Mai 2005, Bonn, Germany.

Borrini-Feyerabend, G. (1996): *Collaborative Management of PRO-TECTED Areas: Tailoring the Approach to the Context IUCN*. Gland, Switzerland.

Boschma, R. (2005): Proximity and Innovation: A Critical Assessment. Regional Studies, 39 (1): 61-74.

Boschma, R., Fornahl, D. (2011): Cluster evolution and a roadmap for future research. *Regional Studies*, 45 (1): 1295-1298.

Boudreau, J. A. (2007): Making new political spaces: mobilising spatial imaginaries, instrumentalising spatial practices, and strategically using spatial tools. *Environment and Planning A*, 39 (11): 2593-2611.

Bourdieu, P. (1986): The forms of capital, in Richardson, J. (eds.): *Handbook of Theory and Research for the Sociology of Education*. Greenwood, New York, United States: 241-258.

Boyle, D., Harris, M. (2009): The challenge of co-production. Howe equals partnerships between professionals and the public are crucial to improving public services. Discussion paper, New Economics Foundation (NEF), London.

Brenner, N. (1997): State territorial restructuring and the production of spatial scale. Urban and regional planning in the Federal Republic of Germany, 1960-1990. *Political Geography*, 16 (4): 273-306.

Brenner, N. (1999): Beyond state-centrism? Space, territoriality, and geographical scale in globalization studies. *Theory and Society*, 28 (1): 39-78.

Brenner, N. (2001): The limits to scale? Methodological reflections on scalar structuration. *Progress in Human Geography*, 25 (4): 591-614.

Brenner, N. (2004): *New State Spaces. Urban Governance and the Rescaling of Statehood*. Oxford Univ. Press, Oxford, United Kingdom.

Brenner, N., Theodore, N. (2002): Preface: From the "New Localism" to the Spaces of Neoliberalism. *Antipode*, 34 (3): 341-347.

Brinkerhoff, J. M. (2002): Assessing and improving partnership relationships and outcomes: a proposed framework. *Evaluation and Program Planning*, 25 (3): 215-231.

Brown, J. C., Purcell, M. (2005): There's nothing inherent about scale: political ecology, the local trap, and the politics of development in the Brazilian Amazon. *Geoforum*, 36 (5): 607-624.

Brown, J. D., Damery, S. L., (2002): Managing flood risk in the UK: towards an integration of social and technical perspectives. *Transactions of the Institute of British Geographers*, 27 (4): 412-426.

Bues, A., Theesfeld, I. (2012): Water Grabbing and the Role of Power: Shifting Water Governance in the Light of Agricultural Foreign Direct Investment. *Water Alternatives*, 5 (2): 266-283.

Butler, C., Pidgeon, N. (2011): From 'flood defence' to 'flood risk management': Exploring governance, responsibility, and blame. *Environment and Planning C: Government and Policy* 29 (3):533-547.

Butler, R. (2007): From where I write: the place of positionality in qualitative writing' (2001) in Limb, M. and Dwyer, C. (eds.): *Qualitative Methodologies for Geographers Arnold*, London: 264-276.

Cannon, T., Müller-Mahn, D. (2010): Vulnerability, resilience and development discourses in context of climate change. *Natural Hazards* 55 (3):621-635.

Capano, G. (2009): Understanding Policy Change as an Epistemological and Theoretical Problem. *Journal of Comparative Policy Analysis: Research and Practice*, 11 (1): 7-31.

Cash, D. W., Adger, W. N., Berkes, F., Garden, P., Lebel, L., Olsson, P., Pritchard, L., Young, O. (2006): Scale and Cross-Scale Dynamics: Governance and Information in a Multilevel World. *Ecology and Society*, 11 (2): 8.

Castells, M. (1977): The Urban Question: A Marxist Approach. Edward Arnold, London, United Kingdom.

CCC (2012): Climate change – is the UK preparing for flooding and water scarcity? Adaptation Sub-Committee Progress Report 2012. http://www.preventionweb.net/files/27607\_cccascreport2012bookmarked2.pdf [26/09/2013].

Chappel, L. (2010): Comparative gender and institutions: directions for research. Perspectives on Politics, 8 (1): 183-189.

Chatterton, J. B., Viavattene, C., Morris, J., Penning-Rowsell, E., Tapsell, S. (2010): *The costs of the summer 2007 floods in England*. Environment Agency, Bristol, United Kingdom.

Ciriacy-Wantrup, S. (1971): The Economics of Environmental Policy. Land Economics, 47 (1): 36-45.

Clarke, N. (2009): In what sense 'spaces of neoliberalism'? The new localism, the new politics of scale, and town twinning. *Political Geography*, 28 (8): 496-507.

Clarke, N., Cochrane, A. (2013): Geographies and politics of localism: The localism of the United Kingdom's coalition government. *Political Geography*, 34: 10-23.

Clarno, A. (2013): Rescaling White Space in Post-apartheid Johannesburg. Antipode, 45 (2): 1190-1212.

CLG (2012): National Planning Policy Framework. Communities and Local Government. Department for Communities and Local Government, London, United Kingdom.

Clifford, B. P. (2013): Reform on the Frontline: Reflections on Implementing Spatial Planning in England, 2004-2008. *Planning Practice and Research*, 28 (4): 361-383.

Coaffee, J. (2005): New localism and the management of regeneration. *The International Journal of Public Sector Management*, 18 (2): 108-113.

Coaffee, J., Johnston, L. (2005): The management of local government modernisation. Area decentralisation and pragmatic localism. *The International Journal of Public Sector Management*, 18 (2): 164-177.

Cochrane, A. (2012): Making up a region: the rise and fall of the 'South East of England' as a political territory. *Environment and Planning C: Government and Policy*, 30 (1): 95-108.

Collinge, C. (1999): Self-organisation of society by scale: a spatial reworking of regulation theory. *Environment* and Planning D: Society and Space, 17 (5): 557-574.

Colomb, C. (2007): Unpacking new labour's 'Urban Renaissance' agenda: Towards a socially sustainable reurbanization of British cities? *Planning Practice & Research*, 22 (1): 1-24.

Comstock, D. (1982): A method for critical research, in Bredo, E., Feinberg, W. (eds.): *Knowledge and Values in Social and Educational Research*. Temple University Press, Philadelphia: 370-390.

Cook, B. R., Kesby, M., Fazey, I., Spray, C. (2013): The persistence of 'normal' catchment management despite the participatory turn: Exploring the power effects of competing frames of reference. *Social Studies of Sciences*, 43 (5): 754-779.

Corkindale, J. (2007): The potential role of market-based instruments in the economic optimisation of investment in flood risk management. Report for the EA. Bristol, United Kingdom.

Coulthard, T. J., Frostick, L. E. (2010): The Hull floods of 2007: implications for the governance and management of urban drainage systems. *Journal of Flood Risk Management*, 3 (3): 223-231.

Cowell, R., Owens, S. (2006): Governing space: planning reform and the politics of sustainability. *Environment and Planning C: Government and Policy*, 24 (3): 403-421.

Cox, K. R. (1998): Spaces of dependence, spaces of engagement and the politics of scale, or: looking for local politics. *Political Geography*, 17 (1): 1-23.

Cox, K. R. (2009): 'Rescaling the state' in question. *Cambridge Journal of Regions, Economy and Society*, 2 (1): 107-121.

Cox, K. R. (2013): Territory, Scale, and Why Capitalism Matters. Territory, Politics, Governance, 1 (1): 46-61

Cox, K. R., Mair, A. (1989): Levels of abstraction in locality studies. Antipode, 21 (2): 121-132.

Cumbria County Council (2013): News 14/3/2013 – Cockermouth to learn more about Main Street improvements. http://www.cumbria.gov.uk/news/2013/March/\_14\_03\_2013-1517551.asp [26/09/2013].

Cumbria Resilience (2011): *Cumbria Floods November 2009. Learning from experience recovery phase debrief report.* Cumbria Resilience Forum, Carlisle, United Kingdom.

Dahl, R. A. (1957): The Concept of Power. Behavioural Science, 2 (3): 201-215.

Davies, J. S. (2002): The Governance of Urban Regeneration: A Critique of the 'Governing without Government' Thesis. *Public Administration*, 80 (2): 301-322.

de Leeuw, S., Vis, I. F. A., Jonkman, S. N. (2012): Exploring Logistics Aspects of Flood Emergency Measures. *Journal of Contingencies and Crisis Management*, 20 (3): 166-179.

de Marchi, B. (1995): Environmental problems, policy decisions and risk communication. *Science and Policy*, 30 (3): 171-176.

de Marchi, B., Scolobig, A. (2012): The views of experts and residents on social vulnerability to flash floods in an Alpine region of Italy. *Disasters*, 36 (2): 316-327.

Deas, I. (2013): Towards Post-political Consensus in Urban Policy? Localism and the Emerging Agenda for Regeneration under the Cameron Government. *Planning Practice and Research*, 28 (1): 65-82.

Defra (2005): Making Space for Water. Taking forward a new Government strategy for flood and coastal erosion risk management in England. First Government response to the autumn 2004 Making space for water consultation exercise. Defra, London.

Defra (2007): Summary of responses to consultation on Outcome Measures and prioritisation approaches for flood and coastal erosion risk management. Defra, London.

Defra (2010a): *Draft strategy for skills and capacity building in local authorities for local flood risk management*. Defra, London.

Defra (2010b): Future funding for flood and coastal erosion risk management. Consultation on the future Capital Grant-In-Aid allocation process in England. Defra, London.

Defra (2011): Flood and Coastal Resilience Partnership Funding. Defra policy statement on an outcome-focused, partnership approach to funding flood and coastal erosion risk management. London, United Kingdom.

Defra (2012): *Principles for implementing flood and coastal resilience funding partnerships*. Environment Agency, Bristol.

Delaney, D., Leitner, H. (1997): The political construction of scale. *Political Geography*, 16 (2): 93-97.

Dencker, J. C. (2009): Relative Bargaining Power, Corporate Restructuring, and Managerial Incentives. *Administrative Science Quarterly*, 54 (3): 453-485

Devon County Council (2012): Local Levy and Flood Risk Funding. PTE/12/79 Cabinet 14 November 2012, Exeter.

Dicke, W., Meijerink, S. (2008): The Public-Private Divide in Flood Management. *International Journal of Water Resources Development*, 24 (4): 495-497.

Doak, J., Parker, G., Raco, M. (2004): *Community Strategies in England: reshaping spaces of governance?* Department of Real Estate and Planning, The University of Reading Working Papers in Real Estate & Planning 01/04, Reading.

Dooley, L. M. (2002): Case Study Research and Theory Building. *Advances in Developing Human Resources*, 4 (3): 335-354.

Doorewaard, H. (2010): Conceptual Model in a Qualitative Research project, in Mills, A.J., Durepos, G., Wiebe, E. (eds.): *Encyclopaedia of Case Study Research*. Sage Publications, London: 207-208.

Dore, J., Lebel, L. (2010): Deliberation and Scale in Mekong Region Water Governance. *Environmental Management*, 46 (1): 60-80.

Downward, P., Mearman, A. (2007): Retroduction as mixed-methods triangulation in economic research: reorienting economics into social science. *Cambridge Journal of Economics*, 31 (1): 77-99.

Driessen, P. P. J., Dieperink, C., van Laerhoven, F., Runhaar, H. A. C., Vermeulen, W. J. V. (2012): Towards a Conceptual Framework for The Study of Shifts in Modes of Environmental Governance – Experiences From the Netherlands. *Environmental Policy and Governance*, 22 (3): 143-160.

Drogendijk, R., Andersson, U. (2013): Relationship development in Greenfield expansions. *International Business Review*, 22 (2): 381-391.

Dunn, W. N. (1994): Public policy analysis: An introduction. Second Edition, Prentice Hall, New Jersey.

EA (2008): Benefiting people and the environment. Proposed prioritisation of flood and coastal erosion risk management projects. Environment Agency, Bristol.

EA (2009a): Investing for the future. Flood and coastal risk management in England. A long-term investment strategy. Environment Agency, Bristol.

EA (2009b): Environment Agency Flood and Coastal Risk Management Modelling Data. Environment Agency, Bristol.

EA (2011): Memorandum relating to capital grants for local authorities and internal drainage boards in England under the Flood and Water Management Act 2010, Coast Protection Act 1949, Land Drainage Act 1991 (April 2011 (Amended 2012)). Environment Agency, Bristol.
EA (2012a): Estimating Outcome Measure contributions and using the FDGiA Funding calculator for Flood and Coastal Erosion Risk Management projects. Environment Agency, Bristol.

EA (2012b): Memorandum relating to capital grants for local authorities and internal drainage boards in England under the Flood and Water Management Act 2010, Coast Protection Act 1949, Land Drainage Act 1991 (April 2012). Applicable for studies & projects commencing post 1 April 2012. Environment Agency, Bristol.

EA (2013): New flood defence construction starts in England in 2012/13. http://www.environment-agency.gov.uk/static/documents/Research/Flood Schemes Construction starts.pdf [21/02/2014].

EC (2000): Directive 2000/60/EC of the European Parliament and of the Council of 23 October 2000 establishing a framework for Community action in the field of water policy. European Commission and Parliament, Brussels.

EC (2007): Directive 2007/60/EC of the European Parliament and of the Council of 23 October 2000 on the assessment and management of flood risks. European Commission and Parliament, Brussels.

Edina Digimap (2012): Edina Digimap Ordinance Survey Service. http://edina.ac.uk/digimap [27/03/2013].

Edwards, B., Goodwin, M., Pemberton, S., Woods, M. (2001): Partnerships, power, and scale in rural governance. *Environment and Planning C: Government and Policy*, 19 (2): 289-310.

Eisenhardt, K. M. (1989): Building Theories from Case Study Research. *The Academy of Management Review*, 14 (4): 532-550.

Elliott, J.R., Pais, J. (2006): Race, class, and Hurricane Katrina: Social differences in human responses to disaster. *Social Science Research*, 35 (2): 295-321.

Elster, J. (1992): Local Justice. How Institutions Allocate Scarce Goods and Necessary Burdens. Russel Sage Foundation, New York.

Erdlenbruch, K., Thoyer, S., Grelot, F., Kast, R., Enjolras, G. (2009): Risk-sharing policies in the context of the French Flood Prevention. *Journal of Environmental Management*, 91 (2): 363-369.

Escobar, M. P., Demeritt, D. (2014): Flooding and the framing of risk in British broadsheets, 1985-2010. *Public Understanding of Science*, 23 (4): 454-471.

Evans, E. P., Ramsbotton, D. M., Wicks, J. M., Packman, J. C., Penning-Rowsell, E. C. (2002): Catchment flood management plans and the modelling and decision support framework. *Proceedings of ICE Civil Engineering*, 150: 43-48.

Evers, D., de Vries, J. (2013): Explaining Governance in Five Mega-City Regions: Rethinking the Role of Hierarchy and Government. *European Planning Studies*, 21(4): 536-555

Fainstein, S. S. (1999): Power and geographic scale: comments on Morrill. Political Geography, 18 (1): 39-43.

Fairtlough, G. (2007): The three ways of getting things done: hierarchy, heterarcy and responsible autonomy in organisations. Triarchy Press, Axminster.

Fatti, C. E., Patel, Z. (2013): Perceptions and responses to urban flood risk: Implications for climate governance in the South. *Applied Geography*, 36: 13-22.

Featherstone, D., Ince, A., MacKinnon, D., Strauss, K., Cumbers, A. (2012): Progressive localism and the construction of political alternatives. *Transactions of the Institute of British Geographers*, 37 (2): 177–182.

Fenwick, J., Johnston Miller, K., McTavish, D. (2012): Co-governance or meta-bureaucracy? Perspectives of local governance 'partnership' in England and Scotland. *Policy and Politics*, 40 (3): 405-422.

Fereday, J., Muir-Cochrane, E. (2006): Demonstrating Rigor Using Thematic Analysis: A Hybrid Approach of Inductive and Deductive Coding and Theme Development. *International Journal of Qualitative Methods*, 5 (1): https://ejournals.library.ualberta.ca/index.php/IJQM/article/view/4411/3530 [21/02/2014].

Fiedling, J., Fiedling, N. (2009): Synergy and Synthesis: Integrating Qualitative and Quantitative Data, in Alasuutari, P., Bickman, L., Brannen, J. (eds.): *The SAGE Handbook of Social Research Methods*. SAGE, London: 555-571.

Filatova, T., Mulder, J. P. M., van der Veen, A. (2011): Coastal risk management: How to motivate individual economic decisions to lower flood risk? *Ocean and Coastal Management*, 54 (2): 164-172.

Flick, U. (1998): An introduction to qualitative research. Sage Publication Ltd., London.

Flyvbjerg, B. (2006): Five Misunderstandings about case-study research. Qualitative Inquiry, 12 (2): 219-245.

Fontana, A., and Frey, J. (2000): The interview: From structured questions to negotiated text, in Denzin, N., Lincoln, Y (eds.): *Handbook of qualitative research (2nd Ed.)*. Sage Publication Ltd, London: 645-672.

Foucault, M. (1982): The Subject and Power. Critical Inquiry, 8 (4): 777-795.

Fowler, A. F. (1998): Authentic NGDO Partnerships in the New Policy Agenda for International Aid: Dead End or Light Ahead? *Development and Change*, 29 (1): 137-159.

Frederickson, H. G. (2004): Whatever Happened to Public Administration? Governance, Governance Everywhere. Working Paper QU/GOV/3/2004.

Fürst, D. (2003): Steuerung auf regionaler Ebene versus Regional Governance. *Informationen zur Raumentwicklung*, 8/9: 441-450.

Fürst, D. (2006): The role of experimental regionalism in rescaling the German state. *European Planning Studies*, 14 (7): 923-938.

Garrelts, H., Lange, H. (2011): Path Dependencies and Path Change in Complex Fields of Action: Climate Adaptation Policies in Germany in the Realm of Flood Risk Management. *Ambio*, 40 (2): 200-209

Gärtner, S. (2011): Corporations and Regions: Capturing Multiple Vicinity Spaces in European Regions. *European Planning Studies*, 19 (11): 1931-1950.

Geeraerts, G. (1994): Analysing Non-State Actors in World Politics: A Conceptual Approach, *Pole-Papers* 1 (4). 36 blz.

Gerring, J. (2006): Case Study Research. Principles and Practices. Cambridge University Press, Cambridge.

Gibbs, D., Jonas, A. E. G. (2000): Governance and regulation in local environmental policy: the utility of a regime approach. *Geoforum*, 31 (3): 299-313.

Gibbs, D., Jonas, A. E. G. (2001): Rescaling and regional governance: the English Regional Development Agencies and the environment. *Environment and Planning C: Government and Policy*, 19 (2):269-288.

Gibson, C. C., Ostrom, E., Ahn, T. K. (2000): The concept of scale and the human dimensions of global change: a survey. *Ecological Economics*, 32 (2): 217-239.

Giddens, A. (1999): Risk and Responsibility. The Modern Law Review, 62 (1): 1-10.

Glasbergen, P. (2011): Understanding Partnerships for Sustainable Development Analytically: the Ladder of Partnership Activity as a Methodological Tool. *Environmental Policy and Governance*, 21 (1): 1-13.

Glavovic, B. C., Saunders, W. S. A., Becker, J. S. (2009): Land-use planning for natural hazards in New Zealand: the setting, barriers, 'burning issues' and priority actions. *Natural Hazards*, 54 (3): 679-706.

Graham, M., Ernstson, H. (2012): Comanagement at the Frings: Examining Stakeholder Perspectives at Macassar Dunes, Cape Town, South Africa – at the Intersection of High Biodiversity, Urban Poverty, and Inequality. *Ecology and Society*, 17 (3): 34.

Graham, S., Marvin, S. (2001): *Splintering Urbanism: Networked Infrastructures, Technological Mobilities and the Urban Condition*. Routledge, London.

Green, C. (2010): The practise of power: governance and flood risk management, in Pender, G., Faulkner, H. (eds.): *Flood Risk Science and Management*. Wiley-Blackwell, West Sussex: 359-371.

Green, C. (2011): *Mapping the field. The landscape of governance*. Switch report, Flood Hazard Research Centre, Middlesex University, London.

Green, C., Penning-Rowsell, E. (2010): Stakeholder engagement in flood risk management, in Pender, G., Faulkner, H. (eds.): *Flood Risk Science and Management*. Wiley-Blackwell, West Sussex: 372-385.

Grimble, R., Wellard, K. (1997): Stakeholder Methodologies in Natural Resource Management: a Review of Principles, Contexts, Experiences and Opportunities. *Agricultural Systems*, 55 (2): 173-193.

Grimm, D.W. (2013): Fracturing Hegemony: Regionalism and State Rescaling in South Korea, 1961-71. *International Journal of Urban and Regional Research*, 37 (4): 1147-1167.

Gualini, E. (2006): The Rescaling of Governance in Europe: New Spatial and Institutional Rationales. *European Planning Studies*, 14 (7): 882-904.

Habermas, J. (1989): The Structural Transformation of the Public Sphere: An Inquiry into a Category of Bourgeois Society. Polity, Cambridge.

Habersack, H., Burgel, J., Petraschek, A. (2004): *Analyse der Hochwasserereignisse vom August 2002 – Floodrisk, Synthese*. Eigenverlag Bundesministerium für Land- und Forstwirtschaft, Umwelt und Wasserwirtschaft, Wien.

Hagspiel, E. (2001): Interkommunale Planung im regionalen Kontext – Merkmale und Eigenschaften, in Voigt, A., Walchhofer, H. P. (eds.): *Interkommunale Planung*. Institut für Örtliche Raumplanung, TU Wien, Vienna: 63-72.

Hanger, S., Pfenninger, S., Dreyfus, M., Patt, A. (2013): Knowledge and information needs of adaptation policymakers: A European study. *Regional Environmental Change* 13 (1):91-101.

Hanley, N., Spash, C. L. (1994): Cost-Benefit Analysis and the Environment. Edward Elgar, Cheltenham.

Harrison, E. (2002): 'The Problem with the Locals': Partnership and Participation in Ethiopia. *Development and Change*, 33 (4): 587-610.

Harrison, J. (2012): Life after Regions? The Evolution of City-regionalism in England. *Regional Studies*, 46 (9): 1243-1259.

Harvey, L. (1990): Critical Social Research. Unwin Hyman, London.

Haughton, G., Allmendinger, P. (2007): 'Soft spaces' in planning. Town & Country Planning, 76 (9): 306-308.

Haughton, G., Allmendinger, P. (2008): The soft spaces of local economic development. *Local Economy*, 23 (2): 138-148.

Haughton, G., Allmendinger, P. (2013): Spatial Planning and the New Localism. *Planning Practice and Research*, 28 (1): 1-5.

Haughton, G., Allmendinger, P., Oosterlynck, S. (2013): Spaces of neoliberal experimentation: soft spaces, postpolitics, and neoliberal governmentality. *Environment and Planning A*, 45(1): 217-234.

Haupter, B., Heiland, P., Neumüller, J. (2005): Interregional and Transnational Co-operation in River Basins – Chances to Improve Flood Risk Management? *Natural Hazards*, 36 (1-2): 5-24.

Heley, J, Jones, M. (2012): Devolution and Wales – fiddling with spatial governance whilst the economy burns, in Ward, M., Hardy, S. (eds.): *Changing Gear – is Localism the New Regionalism?* The Smith Institute, London: 25-33.

Heley, J. (2013): Soft Spaces, Fuzzy Boundaries and Spatial Governance in Post-devolution Wales. *International Journal of Urban and Regional Research*, 37(4): 1325-1348.

Hildreth, P., Bailey, D. (2013): The economics behind the move to 'localism' in England. *Cambridge Journal of Regions, Economy and Society*, 6 (2): 233-249.

HM Treasury (2013): Investing in Britain's future. HM Treasury, London [26/09/2013].

Hodgson, G. M. (2006): What Are Institutions? Journal of Economic Issues, 40 (1): 1-25.

Holub, M., Fuchs, S. (2009): Mitigating mountain hazards in Austria – legislation, risk transfer, and awareness building. *Natural Hazards Earth System*, 9 (2): 523-537.

Holub, M.; Suda, J.; Fuchs, S. (2012): Mountain hazards: reducing vulnerability by adapted building design. *Environmental Earth Sciences*, 66 (7): 1853-1870.

Hooghe, L., Marks, G. (2003): Unraveling the Central State, but How? Types of Multi-level Governance. *American Political Science Review*, 97 (2): 233-243.

Höppner, C., Bründl, M., Buchecker, M. (2010): *Risk Communication and Natural Hazards*. CapHaz-Net WP5 Report, Swiss Federal Research Institute WSL.

House of Commons (1998): *The current administrative framework for flood and coastal defence in England and Wales. Select Committee on Agriculture, Sixth Report.* House of Commons, London.

House of Commons (2008): Environment, Food and Rural Affairs – Fifth Report. London, United Kingdom.

House of Commons (2009): Written Answers – 24 February 2009. http://www.publications.parliament.uk/pa/cm200809/cmhansrd/cm090224/text/90224w0014.htm [26/09/2013].

House of Commons (2013a): *Environment, Food and Rural Affairs Committee. Flood Funding, written evidence*. House of Commons, London.

House of Commons (2013b): Environment, Food and Rural Affairs Committee. Managing Flood Risk. Third Report of Session 2013-14. House of Commons, London.

Howitt, R. (1993): "A world in a grain of sand": towards a reconceptualisation of geographical scale. *Australian Geographer*, 24 (1): 33-44.

Howitt, R. (1998): Scale as relation: musical metaphors of geographical scale. Area, 30 (1): 49-58.

Howitt, R. (2003): Scale, in Agnew, J., Mitchell, K., Gerard, T. (eds.): A comparison to political geography. Blackwell Publishing, Oxford: 138-157.

Huber, F. (2012): On the Role and Interrelationship of Spatial, Social and Cognitive Proximity: Personal Knowledge Relationships of R&D Workers in the Cambridge Information Technology Cluster. *Regional Studies*, 46 (9): 1169-1182.

Hung, H. V., Shaw, R., Kobayashi, M. (2010): Flood risk management for the riverside urban areas of Hanoi. The need for synergy in urban development and risk management policies. *Disaster Prevention and Management*, 19 (1): 103-118.

Huntjens, P. Lebel, L., Pahl-Wostl, C., Camkin, J., Schulze, R., Kranz, N. (2012): Institutional design propositions for the governance of adaptation to climate change in the water sector. *Global Environmental Change*, 22 (1): 67-81.

Huntjens, P., Pahl-Wostl, C., Grin, J. (2010): Climate change adaptation in European rivers basins. *Regional Environmental Change*, 10 (4): 263-284.

Hutter, G. (2014): Goal-oriented networks and capacity building for natural hazards – examples in the Dresden region. *Natural Hazards and Earth System Science*, 14 (1): 81-94.

ICE (2008): Flooding: Engineering resilience. Institute of Civil Engineers (ICE), London.

Ingold, K., Balsiger, J., Hirschi, C. (2010): Climate change in mountain regions: how local communities adapt to extreme events. *Local Environment: The International Journal of Justice and Sustainability*. 15 (7): 651-661.

Ingstrup, M. B., Damgaard, T. (2013): Cluster facilitation from a cluster life cycle perspective. *European Planning Studies*, 21 (4): 556-574.

Innes, J. E., Booher, D. E. (2004): Reframing public participation: strategies for the 21st century. *Planning Theory & Practice*, 5 (4): 419-436.

IRGC (2008): An introduction to IRGC Risk Governance Framework. IRGC, Geneva.

Irvin, R. A., Stansbury, J. (2004): Citizen Participation in Decision Making: Is It Worth the Effort? *Public Administration Review*, 64 (1): 55-65.

Jackson, K. M., Trochim, W. M. (2002): Concept Mapping as an Alternative Approach for the Analysis of Open-Ended Survey Responses. *Organizational Research Methods*, 5 (4): 307-336.

Jackson, P. M. (2001): Public sector added value: Can bureaucracy deliver? Public Administration, 79 (1): 5-28.

Janssen, A. E. B. (2008): On peaks and politics; governance analysis of flood risk management cooperation between Germany and the Netherlands. *International Journal of River Basin Management*, 6 (4): 349-355.

JBA (2011): *Castle Morpeth Borough Council. Flooding in Castle Morpeth 6 & 7 September 2008*. Independent Review. Pitsford.

Jessop, B. (1990): *State theory: putting the capitalist state in its place.* Pennsylvania State University Press, University Park.

Jessop, B. (1998): The rise of governance and the risks of failure: the case of economic development. *International Social Science Journal*, 50 (155): 29-45.

Jessop, B. (2004): Multi-level governance and multi-level metagovernance changes in the European Union as integral moments in the transformation and reorientation of contemporary statehood, in: Bache, I., Flinders, M. (eds.): *Multi-level Governance*, Oxford: Oxford University Press: 49-74.

Jessop, B., Brenner, N., Jones, M. (2008): Theorizing sociospatial relations. *Environment and Planning D:* Society and Space, 26 (3): 389-401.

Johnson, C. L., Penning-Rowsell, E. (2010): What really determines policy? An evaluation of outcome measures for prioritising flood and coastal risk management investment in England. *Journal of Flood Risk Management*, 3 (1): 25-32.

Johnson, C. L., Penning-Rowsell, E., Parker, D. (2007): Natural and imposed injustices: the challenges in implementing 'fair' flood risk management policy in England. *The Geographical Journal*, 173 (4):374-390.

Johnson, C. L., Priest, S. J. (2008): Flood Risk Management in England: A Changing Landscape of Risk Responsibility. *Water Resource Development*, 24 (4):513-525.

Johnson, C., Tunstall, S., Priest, S., McCarthy, S., Penning-Rowsell, E. (2008): Social justice in the context of flood and coastal risk management: a review of policy and practice. Defra, London.

Jones, K. T. (1998): Scale as epistemology. Political Geography, 17 (1): 25-28.

Jones, M. D., Beth, M. K. (2010): A Narrative Policy Framework: Clear Enough to Be Wrong? *Policy Studies Journal*, 38 (2): 329-353.

Jones, M., Jessop, B. (2010): Thinking State/Space Incompossibly. Antipode, 42 (5): 1119-1149.

Jordan, A. (2008): The governance of sustainable development: taking stock and looking forwards. *Environment and Planning C: Government and Policy*, 26 (1): 17-33.

Jordan, A., Wurzel, R. K. W., Zito, A. (2005): The Rise of 'New' Policy Instruments in Comparative Perspective: Has Governance Eclipsed Government? *Political Studies*, 53 (3): 477-496.

Joss, S. (2010): Accountable Governance, Accountable Sustainability? A Case Study of Accountability in the Governance for Sustainability. *Environmental Policy and Governance*, 20 (6): 408-421.

Juntti, M., Russel, D., Turnpenny, J. (2009): Evidence, politics and power in public policy for the environment. *Environmental Science and Policy*, 12(3): 207-215

Kanonier, A. (2006): Raumplanungsrechtliche Regelungen als Teil des Naturgefahrenmanagements, in Fuchs, S., Khakzadeh, L., Weber, K. (eds.): *Recht in Naturgefahrenmanagement*. Studienverlag, Innsbruck: 123-153.

Keessen, A.M., Hamer, J.M., van Rijswick, H.F.M.W., Wiering, M. (2013): The concept of resilience from a normative perspective: examples from Dutch adaptation strategies. *Ecology and Society*, 18 (2): 45.

Kerstetter, K. (2012): Insider, outsider, or somewhere in between: the impact of researchers' identities on the community-based research process. *Journal of Rural Social Sciences*, 27 (2): 99-117.

Kettl, D. F. (2000): The Transformation of Governance: Globalization, Devolution, and the Role of Government. *Public Administration Review*, 60 (6): 488-497.

Kitchin, R., Wilton, R. (2003): Disability activism and the politics of scale. Canadian Geographer, 47 (2): 97-115.

Klink, J. (2013): Development Regimes, Scales and State Spatial Restructuring: Change and Continuity in the Production of Urban Space in Metropolitan Rio de Janeiro, Brazil. *International Journal of Urban and Regional Research*, 37 (4): 1168-1187.

Kohler-Koch, B., Rittberger, B. (2006): Review Article: The 'Governance Turn' in EU Studies. *Journal of Common Market Studies*, 44 (Suppl. 1): 27-49.

Kooiman, J. (2003): Governing as Governance. Sage, London.

Kramer, J. P., Revilla Diez, J. (2012): Catching the Local Buzz by Embedding? Empirical insights on the Regional Embeddedness of Multinational Enterprises in Germany and the UK. *Regional Studies*, 46 (10): 1303-1317.

Kubal, C., Haase, D., Meyer, V., Scheuer, S. (2009): Integrated urban flood risk assessment – adapting a multicriteria approach to a city. *Natural Hazards Earth System*, 9 (6): 1881-1895.

Kuhlicke, C., Steinfuehrer, A., Begg, C., Bianchizza, C., Bruendl, M., Buchecker, M., De Marchi, B., Di Masso Tarditti, M., Hoeppner, C., Komac, B., Lemkow, L., Luther, J., McCarthy, S.S., Pellizzoni, L., Renn, O., Scolobig, A., Supramaniam, M., Tapsell, S., Wachinger, G., Walker, G., Whittle, R., Zorn, M., Faulkner, H. (2011): Perspectives on social capacity building for natural hazards: outlining an emerging field of research and practice in Europe. *Environmental and Policy*, 14 (7): 804-814.

Kumar, R. C. (1999): Research Methodology: A Step-by-Step Guide for Beginners. Sage Publication Ltd., London.

Kumar, R. C. (2008): Research Methodology. APH Publishing Corporation, New Delhi.

Kundzewicz, Z. W., Pinskwar, I., Brakenridge, G. R. (2013): Large floods in Europe, 1985-2009. *Hydrological Sciences Journal*, 58 (1): 1-7.

Kythreotis, A. P., Jonas, A. E. G. (2012): Scaling sustainable development? How voluntary groups negotiate spaces of sustainability governance in the United Kingdom. *Environment and Space D: Society and Space*, 30 (3): 381-399.

Lamers, M., Ottow, B., Francois, G., von Korff, Y. (2010): Beyond Dry Feet? Experiences from a Participatory Water-Management Planning Case in The Netherlands. *Ecology and Society*, 15 (1).

Lammond, J. E., Proverbs, D. G., Hammond, F. N. (2009): Accessibility of flood risk insurance in UK: Confusion, competition and complacency. *Journal of Risk Research*, 12 (6): 825-841.

Larson, A. M., Pacheco, P., Toni, F., Vallejo, M. (2007): The Effects of Forestry Decentralization on Access to Livelihood Assets. *The Journal of Environment Development*, 16 (3): 251-268.

Larson, A. M., Soto, F. (2008): Decentralization of Natural Resource Governance Regimes. *Annual Review of Environment and Resources*, 33: 213-239.

Latane, B. (1996): Dynamic Social Impact: The Creation of Culture by Communication. *Journal of Communication*, 46 (4): 13-25.

Le Gales, P. (1998): Regulations and Governance in European Cities. *International Journal of Urban and Regional Research*, 22 (3): 482-506.

Lebensministerium (2007): *Hydrologischer Atlas von Österreich*. Bundesministerium für Land- und Forstwirtschaft, Umwelt und Wasserwirtschaft, Abteilung Wasserhaushalt, Vienna.

Lee, Y, Lee, I. W., Feiock, R. C. (2012): Interorganizational Collaboration Networks in Economic Development Policy: An Exponential Random Graph Model Analysis. *Policy Studies Journal*, 40 (3): 547-573.

Lefebrve, H. (2007): *The Production of Space*. Blackwell Publishing, Maiden.

Leitner, H., Miller, B. (2007): Scale and the limitations of ontological debate: a commentary on Marston, Jones and Woodward. *Transactions of the Institute of British Geographers*, 32 (1): 116-125.

Lemos, M. C., Agrawal, A. (2006): Environmental Governance. *Annual Review of Environment and Resources*, 31: 297-325.

Leon, S. (2010): Who is responsible for what? Clarity of responsibilities in multilevel states: The case of Spain. *European Journal of Political Research*, 50 (1): 80-109.

Leroy, P., Arts, B. (2006): Institutional Dynamics in Environmental Governance, in Arts, B., Leroy, P. (eds.): *Institutional Dynamics in Environmental Governance*. Springer, New York: 1-19.

Leventhal, G. S. (1980): What should be done with equity theory? New approaches to the study of fairness in social relationships, in Gergen, K., Greenberg, M., Willis, R. (eds.): *Social exchange*. Plenum Press, New York: 27-55.

Ley, D., Mountz, A (2001): Interpretation, representation, positionality: issues in field research in human geography, in Limb, M, Dwyer, C (eds.): *Qualitative methods for geographers: Issues and debates*. Arnold, London. Chapter 15.

Li, Y., Wu, F. (2012): The transformation of regional governance in China: The rescaling of statehood. *Progress in Planning*, 78 (2): 55-99.

Lofstedt, R. (2003): Risk communication: pitfalls and promises. European Review, 11 (3): 417-435.

Lukes, S. (2005): Power. A radical view. Second Edition. Palgrave Macmillan, London.

Lundquist, K. J., Trippl, M. (2013): Distance, proximity and types of cross-border innovation systems. A conceptual analysis. *Regional Studies*, 47 (3): 450-460.

Mabry, L. (2009): Case Study in Social Research, in Alasuutari, P., Bickman, L., Brannen, J. (eds.): *The SAGE Handbook of Social Research Methods*. SAGE, London: 214-227.

MacKinnon, D. (2011): Reconstructing scale: Towards a new scalar politics. *Progress of Human Geography*, 35 (1): 21-36.

MacLeavy, J. (2007): The six dimensions of New Labour: structures, strategies, and languages of neoliberal legitimacy. *Environment and Planning A*, 39 (7): 1715-1734.

MacLeod, G., Goodwin, M. (1999): Space, scale and state strategy: rethinking urban and regional governance. *Progress in Human Geography*, 23 (4): 503-527.

Macpherson, I., Brooker, R., Ainsworth, P.: (2000): Case study in the contemporary world of research: Using notions of purpose, place, process and product to develop some principles for practice. *International Journal of Social Research Methodology*, 3 (1): 49-61.

MAFF (1993): Strategy for Flood and Coastal Defence in England and Wales. MAFF, London.

Malova, D., Sivakova, D. (1996): The national council of the Slovak republic: Between democratic tradition and national state-building. *The Journal of Legislative Studies*, 2 (1): 108-132.

Margerum, R. D. (2002): Collaborative Planning: Building Consensus and Building a Distinct Model for Practice. *Journal of Planning Education and Research*, 21 (3): 237-253.

Margerum, R. D. (2008): A Typology of Collaboration Efforts in Environmental Management. *Environmental Management*, 41 (4): 487-500.

Marston, S. A. (2000): The social construction of scale. Progress in Human Geography, 24 (2): 219-242.

Marston, S. A., Jones III, J. P., Woodward, K. (2005): Human geography without scale. *Transactions of the Institute of British Geographers*, 30 (4): 416-432.

Martin, G. P. (2011): The Third Sector, User Involvement and Public Service Reform: A case study in the Co-Governance of Health Service Provision. *Public Administration*, 89 (3): 909-932.

Martin, R., Sunley, P. (2011): Conceptualizing Cluster Evolution: Beyond the Life Cycle Model? *Regional Studies*, 45 (10): 1299-1318.

Martinez-Vazquez, J., Timofeev, A. (2008): *Fiscal Perspective of State Rescaling*. International Studies Program Working Paper 08-06, Georgia State University, Atlanta.

Marvasti, A. B. (2004) Qualitative Research in Sociology. Sage, London.

Mason, J. (1996): Qualitative Researching. Sage Publications Ltd., London.

Mason, S. A., Muller, A. (2007): Analyzing economic market interactions as conflicts: New concepts to assess market-based policy instruments. *Ecological Economics*, 61 (1): 81-90.

Massard, N., Mehier, C. (2009): Proximity and Innovation through an 'Accessibility to Knowledge, Lens. *Regional Studies*, 43 (1): 77-88.

Massey, D. (1992): Politics and Space/Time. New Left Review, 196: 65-84.

Maxwell, S., Riddell, R. (1998): Conditionality or contract: perspectives on partnership for development. *Journal of International Development*, 10 (2): 257-268.

May, P. J. (1992): Policy Learning and Failure. Journal of Public Policy, 12 (4): 331-354.

May, V. (2007): Use of Mixed Methods in a Study of Residence and Contact Disputes between Divorced and Separated Parents. *International Journal of Social Research Methodology*, 10 (4): 295-306.

McCann, L., Colby, B., Easter, K. W., Kasterine, A., Kuperan, K. V. (2005): Transaction cost measurement for evaluating environmental policies. *Ecological Economics*, 52 (4): 527-542.

McCarthy, J., Prudham, S. (2004): Neoliberal nature and the nature of neoliberalism. *Geoforum*, 35 (3): 275-283.

McCarthy, S. (2004). *Definition and experience of flooding: Residents' and officials' perspectives*. PhD thesis University of Surrey, Surrey.

McCarthy, S., Tunstall, S., Faulkner, H. (2008): *Risk Communication: Inter-professional Flood Risk Management*. RPA Stakeholder and Policy Final Report WP 7.3 FRMRC, London.

McCarthy, S., Viavattene, C., Green, C. (2013): Mirror *Case Study: Encouraging flood storage in England and Wales*. EPI – Water Report. Flood Hazard Research Centre, Middlesex University, London.

McCay, B. J. (1996): *Forms of property rights and the impacts of changing ownership*. Proceedings, Form Foundation National Public Policy Education conference, Providence, RI, 16-18 September. http://www.farmfoundation.org/news/articlefiles/40-prforms.pdf [28/05/2013].

Medeiros, E. (2011): (Re)defining the Euroregion Concept. European Planning Studies, 19 (1): 141-158.

Mees, H. L. P., Driessen, P. P. J., Runhaar, H. A. C. (2012): Exploring the Scope of Public and Private Responsibilities for Climate Adaptation. *Journal of Environmental Policy and Planning*, 14 (3): 305-330.

Mees, H. L. P., Driessen, P. P. J., Runhaar, H. A. C. (2014): Legitimate adaptive flood risk governance beyond the dikes: the cases of Hamburg, Helsinki and Rotterdam. *Regional Environmental Change*, 14 (2): 671-682.

Meijerink, S., Dicke, W. (2008): Shifts in the Public-Private Divide in Flood Management. *International Journal of Water Resources Development*, 24 (4): 499-512.

Mendelow, A. (1981): Environmental scanning: the impact of stakeholder concept, in Ross, C. A. (eds.): *Proceedings of the 2<sup>nd</sup> International Conference on Information Systems*, 7-9 December, Cambridge, MA: 407-417.

Mens, M. J. P., Erlich, M., Gaume, E., Lumbroso, D., Moreda, Y., van der Vat, M., Versini, P. A. (2007): *Frameworks for flood event management*. FLOODsite research project, T19-07-03.

Merton, R.K. (1972): Insiders and Outsiders: A Chapter in the Sociology of Knowledge. *American Journal of Sociology*, 78 (1): 9-47.

Metzger, J., Schmitt, P. (2012): When soft spaces harden: the EU strategy for the Baltic Sea Region. *Environment and Planning A*, 44 (2): 263-280.

Meyer, T. L. (2010): Power, Exit Costs, and Renegotiation in International Law. *Harvard International Law Journal*, 51 (2): 379-425.

Mill, J. S. [1910] (2010): Utilitarianism, Liberty and Representative Government. Wildside Press, Milton Keynes.

Moellenkamp, S., Lamers, M., Ebenhoeh, E. (2008): Institutional elements for adaptive water management regimes. Comparing two regional water management regimes in the Rhine basin, in Pahl-Wostl, C., Kabat, P., Moeltgen, J. (eds.): *Adaptive and Integrated Water Management*, Springer, Berlin, Heidelberg: 147-166.

Moodysson, J., Jonsson, O. (2007): Knowledge Collaboration and Proximity: The Spatial Organization of Biotech Innovation Projects. *European Urban and Regional Studies*, 14 (2): 115-131.

Moore, A. (2008): Rethinking scale as a geographical category: from analysis to practice. *Progress of Human Geography*, 32 (2): 203-225.

Morgan, P., Taschereau, S. (1996): Capacity and Institutional Assessment: Frameworks, Methods and Tools for Analysis. Prepared for CIDA Policy Branch, Quebec.

Morris, J. Bannister, N., Hess, T. M., Gowing, D. J. G., Leeds-Harrison, P. B., Vivash, R., Wade, M. (2004): *Integrated Washlands for Flood Defence and Biodiversity*. Report to English Nature and Defra, No. 986. English Nature, Peterborough.

Moss, T. (2012): Spatial Fit, from Panacea to Practice: Implementing the EU Water Framework Directive. *Ecology and Society*, 17 (3): 2.

Moss, T., Newig, J. (2010): Multi-Level Water Governance and Problems of Scale. Setting the Stage for a Broader Debate. *Environmental Management*, 46 (1):1-6.

Mostert, E. (2006): Integrated Water Resources Management in The Netherlands: How Concepts Function. *Journal of Contemporary Water Research and Education*, 135: 19-27.

Naturschutzbund (2010): Naturschutzbund Österreich. Naturschutzbung, Salzburg.

Neal, S. (2013): Transition culture: Politics, localities and ruralities. Journal of Rural Studies, 32: 60-69.

Nelissen, N. (2002): The Administrative Capacity of New Types of Governance. *Public Organization Review: A Global Journal*, 2 (1): 5-22.

Neumann, R. P. (2009): Political ecology: theorizing scale. Progress of Human Geography, 33 (3): 398-406.

Neuvel, J. M. M., van der Knaap, W. (2010): A Spatial Planning Perspective for Measures Concerning Flood Risk Management. *International Journal of Water Resources Development*, 26 (2): 283-296.

NFF (n.d.): What we do. http://nationalfloodforum.org.uk/?page\_id=12 [21/02/2014].

Niskanen, W. (1973): Bureaucracy - Servant or Master? Institute of Economic Affairs, London.

NÖ (2013): Aktuelle Abgeordnete, Ausschüsse & Politiker. http://www.landtag-noe.at/personen-ausschuesse/aktuelle-abgeordnete-ausschuesse [28/05/2013].

North, D. (1990): *Institutions, institutional change and economic performance*. Cambridge University Press, Cambridge.

Northumberland County Council (2011): Morpeth Neighbourhood Planning Front Runners Project. Submission to the Department for Communities and Local Government from Northumberland County Council. Morpeth.

Novy, A. (2002): Die Methodologie interpretativer Sozialforschung. *SRE - Discussion Papers*, 2002/01. Institut für Wirtschaftsgeographie, Abt. Stadt- und Regionalentwicklung, WU Vienna University of Economics and Business, Vienna.

NPPF (2012): National Planning Policy Framework. Department for Communities and Local Government, London.

O'Riordan, T. (1980): A case study in the politics of land drainage. *Disasters*, 4 (4): 393-410.

Olander, S., Landin, A. (2005): Evaluation of stakeholder influence in the implementation of construction projects. *International Journal of Project Management*, 23 (4): 321-328.

Onwuegbuzie, A.J., Leech, N.L. (2007): Sampling Designs in Qualitative Research: Making the Sampling Process More Public. The Qualitative Report, Vol. 12 Number 2 June 2007. http://www.nova.edu/ssss/QR/QR12-2/onwuegbuzie1.pdf [21/02/2014].

OÖ (2012): Landesrecht Oberösterreich: Gesamte Rechtsvorschrift für Oö. Katastrophenschutzgesetz, Fassung vom 20.05.2013. Linz.

OÖ (2013): Die Abgeordneten des Oö. Landtags. http://www.landoberoesterreich.gv.at/cps/rde/xchg/ooe/hs.xsl/12748\_DEU\_HTML.htm [28/05/2013].

Paasi, A. (2004): Place and region: looking through the prism of scale. *Progress in Human Geography*, 28 (4): 536-546.

Paavola, J. (2005): Environmental Conflicts and Institutions as Conceptual Cornerstones of Environmental Governance Research. CSERGE Working Paper EDM 05-01, University of East Anglia, Norwich.

Pardoe, J., Penning-Rowsell, E., Tunstall, S. (2011): Floodplain conflicts: regulation and negotiation. *Natural Hazards Earth System*, 11 (10): 2889-2902.

Parker, D.; Sewell, W. R. D. (1988): Evolving Water Institutions in England and Wales: an assessment of two decades of experiences. *Natural Resources Journal*, 28 (4): 751-785.

Patton, M. (1990): Qualitative evaluation and research methods. Sage, Beverly Hills, CA.

Pearce, G., Ayres, S. (2012): Back to the Local? Recalibrating the Regional Tier of Governance in England. *Regional and Federal Studies*, 22 (1): 1-24.

Peck, J. (2002): Political Economics of Scale: Fast Policy, Interscalar Relations, and Neoliberal Workfare. *Economic Geography*, 78 (3): 331-360.

Peck, J., Tickell, A. (2002): Neoliberalizing Space. Antipode, 34 (3): 380-404.

Peck, J., Tickell, A. (2012): Apparitions of neoliberalism: revisting 'Jungle law breaks out'. Area, 44 (2): 245-249.

Pellizzoni, L. (2004): Responsibility and Environmental Governance. Environmental Politics, 13 (3): 541-565.

Pellizzoni, L. (2011): Governing through disorder: Neoliberal environmental governance and social theory. *Global Environmental Change*, 21 (3): 795-803.

Penning-Rowsell, E., Johnson, C., Tunstall, S., Tapsell, S., Morris, J., Chatterton, J., Green, C. (2005): *The Benefits of Flood and Coastal Risk Management: A Handbook of Assessment Techniques.* Flood Hazard Research Centre, Middlesex University Press.

Penning-Rowsell, E., Johnson, C.L., Tunstall, S. (2006): 'Signals' from pre-crises discourse: lessons from UK flooding for global environmental policy change? *Global Environmental Change A*, 16 (4): 323-339.

Perkmann, M. (2007): Construction of new territorial scales: a framework and case study of the EUREGIO cross-border region. *Regional Studies*, 41 (2): 253-266.

Petit, S. J. (1999): The statutory approach to coastal defence in England and Wales. *Marine Policy*, 23 (4-5): 465-477.

Pierre, J. (2005): Comparative Urban Governance: Uncovering Complex Causalities. *Urban Affairs Review*, 40 (4): 446-462.

Pitt, M. (2008): The Pitt Review. Lessons learned from the 2007 summer floods. Cabinet Office, London.

Plummer, R. (2009): The Adaptive Co-Management Process: an Initial Synthesis of Representative Models and Influential Variables. *Ecology and Society*, 14 (2): 24.

Plummer, R., Fitzgibbon, J. (2004): Co-Management of Natural Resources: A Proposed Framework. *Environmental Management*, 33 (6): 876-885.

Pomeroy, R. S., Katon, B. M., Harkes, I. (2001): Conditions affecting the success of fisheries co-management: lessons from Asia. *Marine Policy*, 25 (3): 197-208.

Porse, E. C. (2013): Stormwater Governance and Future Cities. Water, 5 (1): 29-52.

Porter, J., Demeritt, D. (2012): Flood Risk Management, Mapping and Planning: The Institutional Politics of Decision-Support in England. *Environment and Planning A*, 44 (10): 2359-2378.

Posthumus, H., Morris, J. (2007): Flood Risk Management Policy Issues – Volume 1 – Rural. FRMRC report, Cranfield University, Cranfield.

Prager, K. (2010): Local and Regional Partnerships in Natural Resource Management: The Challenge of Bridging Institutional Levels. *Environmental Management*, 46 (5): 711-724.

Puchinger, F., Henle, A. (2007): Regionalplanungen – ein Instrument zur Umsetzung nachhaltiger Schutzkonzepte. *Wildbach- und Lawinenverbau*, 71 (156): 90-99.

Pugalis, L., Townsend, A. (2013): Rescaling of Planning and Its Interface with Economic Development. *Planning Practice and Research*, 28 (1): 104-121.

Ragin, C. C.; Strand, S. I. (2008): Using qualitative comparative analysis to study causal order: comment on Caren and Panofsky (2005). *Sociological Methods & Research*, 36 (4): 431-441.

Rawls, J. (1972): A Theory of Justice. Clarendon Press, Oxford.

Rayner, J. (2009): Understanding Policy Change as a Historical Problem. *Journal of Comparative Policy Analysis: Research and Practice*, 11 (1): 83-96.

Razafindrabe, B. H. N., Kada, R., Arima, M., Inoue, S. (2014): Analysing flood risk and related impacts to urban communities in central Vietnam. *Mitigation and Adaptation Strategies for Global Change*,19 (2): 177-198.

Reed, M. G., Bruyneel, S. (2010): Rescaling environmental governance, rethinking the state: A threedimensional review. *Progress in Human Geography*, 34 (5): 646-653.

Reed, M. S., Graves, A., Dandy, N., Posthumus, H., Hubacek, K., Morris, J., Prell, C., Quinn, C. H., Stringer, L. C. (2009): Who's in and why? Who's in and why? A typology of stakeholder analysis methods for natural resource management. *Journal of Environmental Management*, 90 (5): 1933-1949.

Renn, O. (2008): *Risk Governance Coping With Uncertainty in a Complex World*. The Earthscan Risk in Society Series, Taylor and Francis Group, London.

Republik Österreich (1959): Wasserrechtsgesetz 1959 (WRG 1959) idF BGBI I Nr 14 2011, Wien.

Republik Österreich (1975): Forstgesetz 1975, BGBl. 440/1975. Wien.

Republik Österreich (1985): Wasserbautenförderungsgesetzes, BGBI. 148/1985. Wien.

Rhodes, R. A. W. (1990): Policy networks: a British perspective. Journal of Theoretical Politics, 2 (3): 293-317

Rhodes, R. A. W. (1996): The New Governance: Governing without Government. *Political Studies*, 44(4): 652-667.

Roggero, M., Fritsch, O. (2010): Mind the Costs: Rescaling and Multi-Level Environmental Governance in Venice Lagoon. *Environmental Management*, 46 (1): 17-28.

Romanelli, A., Massone, H. E., Escalante, A. H. (2011): Stakeholder analysis and social-biophysical interdependencies for common pool resource management: La Brava wetland (Argentina) as a case study. *Environmental Management*, 48 (3): 462-474.

Rössl, D. (2005)a: Relationship Management mit Beiträgen aus Wissenschaft und Praxis. WUV Verlag, Vienna.

Rössl, D. (2005)b: Family Businesses and Interfirm Cooperation. Family Business Review, 18 (3): 203-214.

Rothschild, K. W. (2002): The absence of power in contemporary economic theory. *Journal of Socio-Economics*, 31 (5): 433-442.

Rouquette, J. R., Posthumus, H., Morris, J., Hess, T. M., Dawson, Q. L., Gowing, D. J. G. (2011): Synergies and trade-offs in the management of lowland rural floodplains: an ecosystem services approach. *Hydrological Sciences Journal*, 56 (8): 1566-1581.

Roy, A. (2009): Civic governmentality: The politics of inclusion in Beirut and Mumbai. Antipode, 41 (1): 159-179.

Royal Commission (1927): *Report of the Royal Commission on Land Drainage in England and Wales*. Presented to Parliament by Command of His Majesty December, 1927. London.

Rutten, R., Boekema, F. (2007): Regional social capital: Embeddedness, innovation networks and regional economic development. *Technological Forecasting & Social Change*, 74 (9): 1834-1846.

Ryan, G. W., Bernard, R. (2005): Data management and analysis methods, in Denzin, N. K., Lincoln, Y. (eds.): *Handbook of Qualitative Research*. Second Edition, Sage, London: 769-802.

Ryan, L., Kofman, E., Aaron, P. (2011): Insiders and outsiders: working with peer researchers in researching Muslim communities. *International Journal of Social Research Methodology*, 14 (1): 49-60.

Sabatier, P. (2007): The neede for better theories, in Sabatier, P. (eds.): *Theories of the policy process*. Westview Press, Boulder.

Sandström, C. (2009): Institutional Dimensions of Comanagement: Participation, Power, and Process. Society and Natural Resources: An International Journal, 22 (3): 230-244.

Sapountzaki, K., Wanczura, S., Caseratno, G., Greiving, S., Xanthopoulus, G., Ferrara, F. F. (2011): Disconnected policies and actors and the missing role of spatial planning throughout the risk management cycle. *Natural Hazards*, 59 (3): 1445-1474.

Sartori, G. (1970): Concept misformation in comparative politics. *The American Political Science Review*, 64 (4): 1033-1053.

Sartori, G. (1991): Comparing and miscomparing. Journal of Theoretical Politics, 3 (3): 243-257.

Sayer, A. (2004): Seeking the geographies of power. Economy and Society, 33 (2): 255-270.

Sayre, N. F. (2005): Ecological and geographical scale: parallels and potential for integration. *Progress of Human Geography*, 29 (3): 276-290.

Scharpf, F. W. (1994): Games real actors could play. Positive and negative coordination in embedded negotiations. *Journal of Theoretical Politics*, 6 (1): 27-53.

Schlenker, B. R., Britt, T. W., Pennington, J., Murphy, R., Doherty, K. (1994): The Triangle Model of Responsibility. *Psychological Review*, 101 (4): 632-652.

Schwarze, R., Wagner, G. G. (2007): The political economy of natural disaster insurance: lessons from the failure of a proposed compulsory insurance scheme in Germany. *Environmental Policy and Governance*, 17 (6): 403-415.

Scrase, J. I., Sheate, W. R. (2005): Re-framing Flood Control in England and Wales. *Environmental Values*, 14 (1): 113-137.

Seawright, J., Gerring, J. (2008): Case Selection Techniques in Case Study Research: A Menu of Qualitative and Quantitative Options. *Political Research Quarterly*, 61 (2): 294-308.

Sedgemoor (2009)a: Bridgwater Strategic Flood Defence Infrastructure Planning. Final Report, Bridgwater.

Sedgemoor (2009)b: Local Development Framework. Bridgwater Strategic Flood Defence Tariff SPD. Adopted September 2009. Sedgemoor District Council, Bridgwater.

Sedgemoor (2009)c: Sedgemoor District Council Level 1 Strategic Flood Risk Assessment. Revision Schedule. Sedgemoor District Council, Bridgwater.

Seher, W., Berger, H. (2009): Praktische Umsetzung künftiger Strategien risikoarmer Raumnutzung – Absiedlung und Flächenwidmung. Projektteil Interkommunale Kooperation. Endbericht, University of Life Sciences, Vienna.

Sehring, J. (2009): Path Dependencies and Institutional Bricolage in Post-Soviet Water Governance. *Water Alternatives*, 2 (1): 61-81.

Selin, S., Chavez, D. (1995): Developing a Collaborative Model for Environmental Planning and Management. *Environmental Management*, 19 (2): 189-195.

Sendzimir, J., Flachner, Z., Pahl-Wostl, C., Knieper, C. (2010): Stalled regime transition in the upper Tisza River Basin: the dynamics of linked action situations. *Environmental Science and Policy*, 13 (7): 604-619.

Silvern, S. E. (1999): Scales of justice: Law, American Indian treaty rights and the political construction. *Political Geography*, 18 (6): 639-668.

Smith, A. (2000): Policy networks and advocacy coalitions: explaining policy change and stability in UK industrial pollution policy? *Environment and Planning C: Government and Policy*, 18 (1): 95-114.

Smith, K., Petley, D. N. (2009): *Environmental hazards assessing risk and reducing disaster*. Routledge – Taylor & Francis Group, London.

Smith, N. (1990): Uneven development. Nature, capital, and the Production of Space. Second Edition. Basil Blackwall, Oxford.

Smith, N. (1992): Contours of a Spatialized Politics: Homeless Vehicles and the Production of Geographical Scale. *Social Text*, 33: 54-81.

Smith, N. (2001a): Marxism in Geography. Zeitschrift für Literatur und Diskussion, 3 (2): 5-21.

Smith, N., Ward, D. (1987): The Restructuring of Geographical Scale: Coalescence and Fragmentation of the Northern Core Region. *Economic Geography*, 63 (2): 160-182.

Smith, S. (2001b): Doing qualitative research: from interpretation to action, in Limb, M., Dwyer, C. (eds.): *Qualitative methodologies for geographers: Issues and debates*. Arnold, London: 23-40.

Soane, E., Schubert, I., Challenor, P., Lunn, R., Narendran, S., Pollard, S. (2010): Flood perception and mitigation: the role of severity, agency, and experience in the purchase of flood protection, and the communication of flood information. *Environment and Planning A*, 42 (12): 3023-3038.

Somerville, P. (2011): Multiscalarity and Neighbourhood Governance. *Public Policy and Administration*, 26 (1): 81-105.

Somerville, P., Haines, N. (2008): Prospects for local co-governance. Local Government Studies, 34 (1): 61-79.

Sorensen, E. (2006): Metagovernance: The Changing Role of Politicians in Processes of Democratic Governance. *American Review of Public Administration*, 36 (1): 98-114.

Späth, P., Rohrbacher, H. (2013): Beyond localism: The spatial scale and scaling in energy transitions, in Padt, F. (eds): *Scale-sensitive Governance of the Environment*. Wiley-Blackwell, in press.

Statistik Austria (2004): Volkszählung Berufspendler. Statistik Austria. Vienna.

Steurer, R., Bauer, A., Feichtinger, J. (2010): *Analytical framework for the stock taking survey of governance approaches in 10 OECD countries*. InFER Report, University of Life Sciences, Vienna.

Stockhammer, E. (2012): The Euro crises, European neoliberalism and the need for a European welfare state. *Soundings*, 50, 121-130.

Stoker, G. (1998): Governance as theory: five propositions. International Social Science Journal, 50 (155): 17-28.

Stoker, G. (2006): Public Value Management: A Narrative for Networked Governance? *The American Review of Public Administration*, 36 (1): 41-57.

Stoker, G. (2011): Was Local Governance such a Good Idea? A Global Comparative Perspective. *Public Administration*, 89 (1): 15-31.

Suchman, M. C. (1995): Managing Legitimacy: Strategic and Institutional Approaches. *The Academy of Management Review*, 20 (3): 571-610.

Sugden, R. (2004): Developing the Cost-Benefit Framework for the Appraisal of Flood and Coastal Erosion Risk Management Projects. Joint Defra/EA Flood and Coastal Erosion Risk Management R&D Programme. London, United Kingdom.

Surjan, A., Shaw, R. (2009): Enhancing disaster resilience through local environment management. Case of Mumbai, India. *Disaster Prevention and Management*, 18 (4): 418-433.

Swyngedouw, E. (1997): Neither Global nor Local. "Glocalization" and the Politics of Scale, in Cox, K. R. (eds.): *Spaces of Globalization. Reasserting the Power of the Local*. The Guilford Press, New York: 137-200.

Swyngedouw, E. (2000): Authoritarian governance, power, and the politics of rescaling. *Environment and Planning D: Society and Space*, 18 (1): 63-76.

Swyngedouw, E. (2004): Globalisation or 'glocalisation'? Networks, territories and rescaling. *Cambridge Review* of International Affairs, 17 (1): 25-48.

Swyngedouw, E. (2007): Technonatural revolutions: the scalar politics of Franco's hydro-social dream for Spain, 1939-1975. *Transactions of the Institute of British Geographers*, 32 (1): 9-28.

Swyngedouw, E. (2011): Interrogating post-democratization: Reclaiming egalitarian political spaces. *Political Geography*, 30: 370-380

Tashohkori, A., Teddlie, C. (1998): *Mixed Methodology: Combining Qualitative and Quantitative Approaches. Applied Social Research Methods Series, Vol.* 46. Sage Publisher, London.

Teamey, K. (2010): Research on relationships between government agencies and non-state providers of basic services: A discussion on the methods, theories and typologies used and ways forward. Whose Public Action? Analysing Inter-sectoral Collaboration for Service Delivery. NGPARP Working Paper No. 38, London School of Economics, London.

Thaler, T. (2010): Siedlungswasserwirtschaft in öffentlicher oder privater Hand. England/Wales, die Niederlande und Porto Alegre (Brasilien) als Fallbeispiele. Informationen zur Umweltpolitik, Kammer für Arbeiter und Angestellte für Wien, Vienna.

Theesfeld, I. (2011): Perceived Power Resources in Situations of Collective Action. *Water Alternatives*, 4 (1): 86-103

Thiel, A. (2009): Europeanisation and the Rescaling of Water Services: Agency and State Spatial Strategies in the Algarve, Portugal. *Water Alternatives*, 2 (2): 225-244.

Thiel, A. (2010): Constructing a Strategic, National Resource: European Politics and the Up-Scaling of Water Services in the Algarve, Portugal. *Environmental Management*, 46 (1): 44-59.

Thiel, A., Egerton, C. (2011): Re-scaling of resource governance as institutional change: the case of water governance in Portugal. *Journal of Environmental Planning and Management*, 54 (3): 383-402.

Tödtling, F., Grillitsch, M., Höglinger, C. (2012): Knowledge Sourcing and Innovation in Austrian ICT Companies – How Does Geography Matter? *Industry and Innovation*, 19 (4): 327-348.

Tödtling, F., Lehner, P., Trippl, M. (2006): Innovation in knowledge intensive industries: The nature and geography of knowledge links. *European Planning Studies*, 14 (8): 1035-1058.

Torre, A., Gilly, J. P. (2000): On the Analytical Dimension of Proximity Dynamics. *Regional Studies*, 34 (2): 169-180.

Torre, A., Rallet, A. (2005): Proximity and Localization. Regional Studies, 39 (1): 47-59.

Treib, O., Bähr, H., Falkner, G. (2007): Modes of governance: towards a conceptual clarification. *Journal of European Public Policy*, 14 (1): 1-20.

Trippl, M., Tödtling, F., Lengauer, L. (2009): Knowledge Sourcing Beyond Buzz and Pipelines: Evidence from the Vienna Software Cluster. *Economic Geography*, 85 (4): 443-462.

Tseng, C. P., Penning-Rowsell, E. (2012): Micro-political and related barriers to stakeholder engagement in flood risk management. *The Geographical Journal*, 178 (3): 253-269.

Turnpenny, J., Nilsson, M., Russel, D., Jordan, A., Hertin, J., Nykvist, B. (2008): Why is integrating policy assessment so hard? A comparative analysis of the institutional capacities and constraints. *Journal of Environmental Planning and Management*, 51 (6): 759-775.

UK Government (2004): Planning and Compulsory Purchase Act 2004. London, United Kingdom.

UK Legislation (1980): *Highway Act 1980*. UK Parliament, London.

UK Legislation (1989): Water Act 1989. UK Parliament, London.

UK Legislation (2009): The Flood Risk Regulations 2009. UK Parliament, London.

UK Legislation (2010): Flood and Water Management Act 2010. 2010 Chapter 29. UK Parliament, London.

UK Legislation (2011): The Environment Agency (Levies) (England and Wales) Regulations 2011. UK Parliament, London.

UK Legislation (2012): The Town and Country Planning Regulations 2012. UK Parliament, London.

UN/ISDR (2004): *Living with risk. A global review of disaster reduction initiatives* (Geneva: United Nations), Geneva.

van Assche, K., Beunen, R., Duineveld, M., de Jong, H. (2013): Co-evolutions of planning and design: Risks and benefits of design perspectives in planning systems. *Planning Theory*, 12(2): 177-198.

van Asselt, M. B. A., Renn, O. (2011): Risk governance. Journal of Risk Research, 14 (4): 431-449.

van Dijk, T.A. (2001): Critical Discourse Analysis, in Tannen, D.; Schiffrin, D.; Hamilton, H. (eds.): Handbook of Discourse Analysis. Blackwell, Oxford: 352-371.

van Huijstee, M., Francken, M., Leroy, P. (2007): Partnerships for sustainable development: a review of current literature. *Environmental Sciences*, 4 (2): 75-89.

van Schendel, W. (2002): Geographies of knowing, geographies of ignorance: jumping scale in Southeast Asia. *Environment and Planning D: Society and Space*, 20 (6): 647-668

Varian, H. R. (1975): Distributive Justice, Welfare Economics, and the Theory of Fairness. *Philosophy and Public Affairs*, 4 (3): 223-247.

Verschuren, P. (2003): Case study as a research study: Some ambiguities and opportunities. *International Journal of Social Research Methodology*, 6 (2): 121-139.

Virno, P. (2004): A Grammar of the Multitude. Semiotext(a). Los Angeles.

Waddock, S. A. (1989): Understanding social partnerships: An evolutionary model of partnership organizations. *Administration and Society*, 21 (1): 78-100.

Walker, G., Burningham, K. (2011): Flood risk, vulnerability and environmental justice: Evidence and evaluation of inequality in a UK context. *Critical Social Policy*, 31 (2): 216-240.

Walker, R. (1985): An Introduction to Applied Qualitative Research, in Walker, R. (eds.): *Applied Qualitative Research*. Gower Pub Co., Aldershot, United Kingdom: 2-26.

Ward, P.J., Pauw, W. P., van Buuren, M. W., Marfai, M. A. (2013): Governace of flood risk management in a time of climate change: the cases of Jakarta and Rotterdam. *Environmental Politics*, 22 (3): 518-536.

Warner, J., Lulofs, K., Bressers, H. (2010): The Fine Art of Boundary Spanning: Making Space for Water in the East Netherlands. *Water Alternatives*, 3 (1): 137-153.

Watson, N., Deeming, H., Treffny, R. (2009): Beyond Bureaucracy? Assessing Institutional Change in the Governance of Water in England. *Water Alternatives*, 2 (3): 448-460.

Weber, M. (1991): Politics as vocation. http://www.sscnet.ucla.edu/polisci/ethos/Weber-vocation.pdf [28/05/2013].

Weber, M., Driessen, P. P. J., Runhaar, H. A. C. (2011): Drivers of and Barriers to Shifts in Governance: Analysing Noise Policy in the Netherlands. *Journal of Environmental Policy and Planning*, 13 (2): 119-137.

Weichhart, P. (2006): Interkommunale Kooperation: Zwischen Notwendigkeit und Verweigerung, in Biwald, P., Hack, H., Wirth. K. (eds.): *Interkommunale Kooperation. Zwischen Tradition und Aufbruch*. NWV Verlag, Vienna: 151-166.

White, I., Howe, J. (2002): Flooding and the role of planning in England and Wales: A critical review. *Journal of Environmental Planning and Management*, 45(5): 735-745.

Whitty, B. (2008): Accountability Principles for Research Organisations. One World Trust, London.

Wiek, A., Larson, K. L. (2012): Water, People, and Sustainability – A Systems Framework for Analysing and Assessing Water Governance Regimes. *Water Resource Management*, 26 (11): 3153-3171.

Wiering, M., Immink, I. (2006): When water management meets spatial planning: a policy-arrangements perspective. *Environment and Planning C: Government and Policy*, 24 (3): 423-438.

Wildavsky, A. (1969), Rescuing policy analysis from Planning-Programming Budgeting Systems. *Public Administration Review*, 29 (2): 189-202.

Wilkinson, M. E.; Quinn, P. F.; Welton, P. (2010): Runoff management during the September 2008 floods in the Belford catchment, Northumberland. *Journal of Flood Risk Management*, 3 (4): 285-295.

Wirth, K., Biwald, P. (2006): Gemeindekooperationen in Österreich – Zwischen Tradition und Aufbruch, in Biwald, P., Hack, H., Wirth, K. (eds.): Interkommunale Kooperation. Zwischen Tradition und Aufbruch. NWV Publisher, Vienna, Graz: 19-34.

Wirth, P., Hutter, G., Schanze, J. (2008): *Regional Governance in Flood Risk Mitigation – The Case of Weißeritz-Regio.* Presentation Regional Studies Association Annual Conference 2008, 27th – 29th May, Prague.

Wissen, M. (2009): Contested terrains: Politics of scale, the national state and struggles for the control over nature. *Review of International Political Economy*, 16 (5): 883-906.

Witt, Harald (2001). Forschungsstrategien bei quantitativer und qualitativer Sozialforschung. Forum Qualitative Sozialforschung / Forum Qualitative Social Research, 2(1), Art. 8, <u>http://nbn-resolving.de/urn:nbn:de:0114-fqs010189</u> [28/05/2013].

Wupperverband (2007): Wasserwirtschaft für Generationen. Limberg Druck, Remscheid.

Yin, R. C. (1994): *Case Study Research: Design and Methods (Applied Social Research Methods). Second Edition.* Sage, London.

Zeller, C. (2004): North Atlantic Innovative Relations of Swiss Pharmaceuticals and the Proximities with Regional Biotech Arenas. *Economic Geography*, 80 (1): 83-111.

Zischg, A., Flury, C., Costa, R., Huber, B., Berger, S. (2011): *Auswirkungen der landwirtschaftlichen Bewirtschaftung auf die Naturgefahren*. Nationale Plattform für Naturgefahren PLANAT, Bern.

Zittoun, P. (2009): Understanding Policy Change as a Discursive Problem. *Journal of Comparative Policy Analysis: Research and Practice*, 11 (1): 65-82.

# Appendix

Appendix 1: Details of intervie	ews
---------------------------------	-----

Code	Organisation interviewed
i1aut	Regional Authority
i2aut	Academic organisation
i3aut	Academic organisation
i4aut	Local Authority
i5aut	Local Authority
i6aut	Local Authority
i7aut	National Authority
i8aut	Regional Authority
i9aut	National Authority
i10aut	Regional Authority
i11aut	Regional Authority
i12aut	Local Authority
i13aut	Local Authority
i14aut	Local Authority
i15aut	Local Authority
i16aut	Private organisation
i17aut	Local Authority
i18aut	Local Authority
i19aut	Regional Authority
i20aut	Regional Authority
i21aut	Regional Authority
i22aut	Regional Authority
i23aut	Regional Authority
i24aut	Regional Authority
i25aut	Academic organisation
i26aut	National Authority
i27aut	Academic organisation
i28aut	Academic organisation
i29eng	National Authority
i30eng	National Authority
i31eng	Private organisation
i32eng	Private organisation
i33eng	Academic organisation
i34eng	Academic organisation
i35eng	Local Authority
i36eng	Private organisation
i37eng	National Authority
i38eng	National Authority
i39eng	National Authority
i40eng	Local Authority
i41eng	Local Authority
i42eng	National Authority
i43eng	Local Authoritiy
i44aut	Regional Authority
i45aut	Local Authority
i46aut	, Private organisation
i47eng	Private organisation
i48eng	National Authority
i49aut	Local Authority

#### **Appendix 2: EU Floods Directive**

The introduction of the EU-Floods Directive opens a new chapter in the flood risk management discussion across Europe, especially in relating to the questions of who and how implement the the requested aspects are implemented. The EU-Floods Directive refers mainly to three implementation steps: (1) preliminary flood risk assessment (Article 4 and 5), (2) developing flood hazard and risk maps (Article 6) and (3) the development of flood risk management plans (Article 7).

#### Step 1: preliminary flood risk assessment

The preliminary flood risk assessment includes general information about the flood risk problems in a specific area. The report includes a general description of historical flood events and its impact to society. Moreover, the report should include a general description of historical flood events comprising a first assessment of impacts to various vulnerability groups. A main objective is to define all potential risks to society, e.g. to develop on description and map of all potential floods occurred in the past, which had a major impact to society; future flood risk events etc. According to the EU-Floods Directive the preliminary flood risk assessment should include three main attributes: (1) hazard determination, i.e. type of flooding (flash flooding, debris flow, pluvial flooding, or tidal flooding, (2) vulnerability determination, i.e. physical assets, systemic assets (critical infrastructure), environmental assets, cultural heritage assets and human beings, and (3) risk determination. Based on this 'general' information, European Member States have to identify and describe potential areas with 'significant' flood risks.

### Step 2: Flood hazard and risk maps

The second step includes the development of flood hazard and risk maps. In the Directive, this needed to be completed by 22 December 2013. The EU-Floods Directive requires more detailed information about potential flood risk events (flood hazard maps) and their potential impacts on society (flood risk maps). This includes a more holistic understanding and in-depth description of flood risk patterns. Article 6 enforces the inclusion of spatial-temporal dynamics into the assessment within different scenarios (low, medium=100 and high probability). The flood hazard and risk maps

are based on the information from Article 4 and 5 (preliminary flood risk assessment), but with more detailed information, e.g. using results from 1D/2D modelling. Here, Member States have to produce two different types of maps, (1) static maps, flood hazard maps with different scenarios for the potentially involved areas and (2) dynamic maps; which are flood risk maps to show potential impacts of a flood event to society and environment.

In the literature, risk appraisal can be divided into two major aspects:

- risk assessment: potential risks to environment, human health, economy, culture heritage,
  i.e. hazard identification and estimation, exposure and vulnerability assessment and risk estimation;
- concern assessment: assessment of related concerns, perceptions as well as socio-economic impacts, i.e. risk perceptions, social concerns, socio-economic impacts (Renn, 2008).

However, the terms hazard, risk and vulnerability include different definitions and approaches how to assess it based on different views and understandings between scholars, policy makers and other stakeholders. The EU-Floods directive (risk characterisation and evaluation) refers to the aspect of definition of the objective how to deal with flood hazards. There is the need is to develop and define a procedural framework to find and understand potential risks and how to manage them. Here, the Member States have to define key objectives and targets about how to manage aspects of flood hazard to reduce their potential impact and consequences towards society and environment. Key points are to define the risk perceptions of the society, i.e. which events are acceptable, tolerable or intolerable and the risk weighting.

#### Step 3: Flood risk management plans

Flood risk management plans select final programmes of measures to reduce the flood impact (preflood reduction), but also to manage effectively and efficiently flood events as well as the recovery phase, such as psychological and economic assistance, support in recovery management, e.g. rebuilding, cleaning etc. Flood risk management includes three main components: (1) pre-flood

271

reduction, i.e. structural and non-structural measures, (2) flood event reduction, i.e. emergency management, and (3) post-flood reduction, i.e. recovery management (Smith and Petley, 2009). Smith and Petley (2009) described risk management as a 'process through which risk is evaluated before strategies are introduced to manage and mitigate the threat' (ibid: 65). However, the main objective is to assess all available information - generated in the previous steps - to define the final objectives and programme of measures. In general, possible measures are done to reduce potential sources of hazards and their negative impacts. Since 1990s/2000s, risk management strategies shifted their focus from structural measures/emergency planning to a broader integrated flood risk management approach, e.g. land use planning, warning system, risk communication, coordination between different organisations (Sapountzaki et al., 2011). These changes in risk management strategies had a strong influence by new paradigm/or change in society on governance structure, recognise of transaction costs into policy decision process and independence. Key challenges and barriers include pressure to reduce public spending, the pressure from society (especially 'green' grassroots organisations) to use a more 'eco-friendly' flood defence scheme, frequent flood events and generational shifts in public administration. The selection process of the final risk management options for stakeholders comprises predefined criteria and used instruments, e.g. multi-criteria analysis (MCA), cost-benefit analysis (CBA), cost-effectiveness analysis (CEA) to finalise their decision.

Further central points includes the following articles: Article 9 (3) (stakeholder participation), Article 10 (risk communication), and Article 14 (updates of Articles 4, 6 and 7 – every six years). Each different step has a strong linkage to the others. A key aim is to exchange and evaluate knowledge and information as well as to coordinate differently involved stakeholders (EC, 2007). A third development is the involvement of non-state actors in the policy process that Howitt (1993; 1998) described as an outwards re-scaling process. Furthermore, Howitt (1993; 1998) attributed to this process as a dialectically correlation between different agents, organisations, and institutions (see also the following figure below). Within the implementation of the EU-Floods Directive, the rescaling process includes all three directions between the scale units: (1) upwards: EU wide policy definition;

(2) downwards: local or regional actors are the key actors in the implementation process and (3) outwards: involvement of non-state actors in the policy discussion (see also chapter 7).



Figure: Dialectical interaction of scales in EU Floods Directive

Source: own development

## Appendix 3: Memorandum capital grants

Memorandum relating to capital grants	Memorandum relating to capital grants (April
(April 2011)	2012)
(2.4) Projects are assessed on the basis of all	(2.4) Projects are assessed on the basis of all costs
costs and benefits. The actual grant payable	and benefits. <i>Prioritisation and approval is</i>
and therefore the approved grant sum is	determined on this basis. The actual grant payable
different to the full cost in the business case,	and therefore the approved grant sum is different
and is determined by reference to grant	and is determined by reference to the tariffs
eligible costs.	associated with outcomes delivered by the
	proposal.
(3.1) The DCLG rules on grant from	(3.1) The DCLG rules on grant from Government
Government Departments state that the	Departments state that the grants provided cannot
grants provided cannot be ringfenced, but are	be ring-fenced, but are bid based grants. <i>Grant will</i>
bid based grants where funding is provided in	only be provided in response to a bid from the
response to an approved bid. There are a	Authority to investigate and remediate a Flood and
number of capital grant mechanisms	Coastal Erosion problem in line with the
operated by the Environment Agency	partnership funding rules under the Floods and
FCERM:	Water Management Act and the National Flood
- Operating Authority capital FCERM studies,	and Coastal Erosion Risk Management Strategy
strategies and schemes	approved by Parliament in July 2011. The
- Property Level Flood Protection &	Environment Agency can make grants for all forms
Resilience schemes (Local Authority Only)	of flooding except sewer flooding.
- Local Authority Quick Wins (to 31/03/2011)	
- Early Actions	
- Coastal Assistance	
(3.4) A "study" may be used to create a	(3.4) A "study" may be used to create a strategy, a
strategy, a management plan, investigate	management plan, investigate options for schemes
options for schemes of works, or investigate	of works, or investigate the environmental impacts
the environmental impacts of FCERM on an	of FCERM on an area of statutory environmental
area of statutory environmental designation.	designation. Studies concluding that a capital
Studies concluding that a capital investment	investment is appropriate would take the evaluation
is appropriate would take the evaluation of	of scheme options to include Environmental
scheme options to include Environmental	Assessment, outline design and outline planning

Assessment, outline design and outline	approval. (Detailed design of a solution is part of the
planning approval. (Detailed design of a	subsequent Project Appraisal Report process). <i>In</i>
solution is part of the subsequent Project	order to assist risk management authorities with
Appraisal Report process)	the initial investigation of a problem the
	Environment Agency has decided to fund 100% of
	the costs of a study as the study progresses. If a
	subsequent capital scheme is approved the whole
	cost including the study will be used to determine
	the partnership funding grant.
(3.7) Grants are available to authorities	(3.7) Grants are available to authorities towards
towards approved capital expenditure	approved capital expenditure incurred on the
incurred on the improvement of existing, or	improvement of existing, or the construction of
the construction of new, flood or erosion risk	new, flood or erosion risk management and water
management and water level management	level management works. Within this definition,
works. Within this definition, grant is	grant is available for studies, projects within
available for studies, projects within	strategies and stand alone projects. Also, amongst
strategies and stand alone projects. Also,	other things grant is available for:
amongst other things grant is available for:	Estuary or beach management projects
Estuary or beach management	(recharge, replenishment, re-cycling and
projects (recharge, replenishment,	monitoring) provided they are part of an
re-cycling and monitoring) provided	agreed long-term beach management
they are part of an agreed long-term	programme,
beach management programme,	Projects designed to manage water levels so
Projects designed to manage water	as to achieve environmental benefits as well
levels so as to achieve environmental	as for those designed to secure other
benefits as well as for those designed	benefits. Such projects may also facilitate
to secure other benefits. Such	surface irrigation through increased water
projects may also facilitate surface	levels. Priority will be given to such projects
irrigation through increased water	in sites of international and national
levels. Priority will be given to such	conservation importance. (Note: Grant is
projects in sites of international and	not available for projects where the prime
national conservation importance.	objective is spray irrigation or water supply).
(Note: Grant is not available for	Purchase of land for habitat compensation
projects where the prime objective is	to allow schemes to comply with the

spray irrigation or water supply).

- Purchase of land for habitat compensation to allow schemes to comply with the Habitats Regulations.
- The preparation of Environmental Statements under the Environmental Impact Assessment (Land Drainage Improvement Works) Regulations 1999, when consultants are used to undertake this work.

Habitats Regulations.

**Coast Protection:** 

(Including CPA) The preparation of Environmental Statements under the With respect to projects that require a marine licence, the Marine Works (Environmental Impact Assessment) Regulations 2007 (the MWR) (as amended by The Marine Works (Environmental Impact Assessment) (Amendment) Regulations 2011) will also apply. The Town and Country Planning (Environmental Impact Assessment) (England and Wales) *Regulations 2011 are also relevant for projects that* require planning permission. The preparation of Environmental Statements under the With respect to projects that require a marine licence, the Marine Works (Environmental Impact Assessment) Regulations 2007 (the MWR) (as amended by The Marine Works (Environmental Impact Assessment) (Amendment) Regulations 2011) will also apply. The Town and Country Planning (Environmental Impact Assessment) (England and Wales) Regulations 2011 are also relevant for projects that require planning permission. (4.1) When considering a proposal for grant, the Environment Agency will assess the following: has the problem been correctly identified?

has the problem been correctly identified? is there a real problem to be addressed?

(4.1) When considering a proposal for grant,

the Environment Agency will assess the

following:

- Can the problem and proposed solution be funded by a capital grant scheme?
- has an acceptable range of options been evaluated?
- Can the problem and proposed solution be funded by a capital grant scheme?

is there a real problem to be addressed?

- has an acceptable range of options been evaluated?
- have the major technical and environmental issues been addressed sufficiently

- have the major technical and environmental issues been addressed sufficiently
- thoroughly?
- has the preferred option been arrived at in a properly logical fashion?
- is the proposal consistent with government policies?
  - Coastal Habitat Management
    Plans (CHaMPS)
  - Catchment Flood
    Management Plans/
    Shoreline Management Plan/
    Water level Management
    Plan
  - o Beach Management Plan
  - o Biodiversity Action Plan
  - o Local Flood Risk Strategy
  - Beneficiaries and
    Contributions policies

And specifically that the proposal is

- eligible within the terms of the Act,
- technically sound and environmentally acceptable in line with FCERM Appraisal Guidance in England)
- economically justified in accordance with FCERM-AG and with the choice of project option based on the FCERM-AG decision rule (see above),
- in line with action plans set out in CFMPs, SMPs and Coastal Habitat Management Plans (CHaMPs), where appropriate.

- thoroughly?
- has the preferred option been arrived at in a properly logical fashion?
- is the proposal consistent with government policies?
  - Water Framework Directive (WFD)
  - Coastal Habitat Management Plans (CHaMPS)
  - Catchment Flood Management
    Plans/ Shoreline Management Plan/
    Water level Management Plan
  - o Beach Management Plan
  - o Biodiversity Action Plan
  - Local Flood Risk Strategy
  - Beneficiaries and Contributions policies

And specifically that the proposal is

- eligible within the terms of the Act,
- technically sound and environmentally acceptable in line with FCERM Appraisal Guidance in England)
- economically justified in accordance with FCERM-AG and with the choice of project option based on the FCERM-AG decision rule (see above),
- in line with action plans set out in CFMPs,
  SMPs and Coastal Habitat Management
  Plans (CHaMPs), where appropriate.
- on the Sanctioned List of projects and studies agreed to be eligible for grant.
- Robust Commitments for non grant funding (match funding) are in place.

on the Sanctioned List of projects and	
studies agreed to be eligible for	
grant.	
	(5.1) Partnership Funding has shifted the
	mechanism for determining grant. Under quantum
	of grant available for a scheme is determined not
	in relation to costs, but on the basis of the
	outcomes delivered. All schemes should seek
	contributions from beneficiaries, but under
	Partnership Funding the quantum of grant
	provided will be capped by reference to the
	outcomes. Partnership Funding guidance can be
	found at:
	http://www.environment-
	agency.gov.uk/research/planning/134732.aspx
(7.2) Administration costs and day to day	(7.2) If the Regional Flood and Coastal Committee
running expenses, which an authority must	agrees to make local levy available to match grant
necessarily incur to discharge its functions	the Committee expects that levy granted would be
will not be admitted for grant, e.g.	for additional costs incurred by the risk
accommodation and related costs, insurance	management authority and that the risk
charges, legal and other professional costs. If	management authority would itself bear the cost
however additional costs are incurred, say for	of general overheads.
site offices and compounds they are	
allowable.	
In cases where expenditure in respect of such	
items as holidays with pay, sick pay, national	
insurance contributions, Public Liability	
Insurance, small tools and light plant etc	
cannot be conveniently apportioned to	
approved schemes, recovery may be made as	
an on-cost on salaries. In respect of stores	
handling the expenditure may be recovered	
by an addition to the cost of stores issued.	
(8.5) A record of condition and plight should	(8.5) A record of condition and plight should be

be taken before any works are started to	taken before any works are started to provide a
provide a basis for subsequent determination	basis for subsequent determination of the
of the compensation to be paid. The fees of	compensation to be paid. (Second part is delete)
independent specialists employed by an	
authority to carry out this exercise will be	
eligible for consideration for grant.	
(8.6) The cost of post-condition surveys may	Delete
be considered for grant, provided they can be	
justified	
(8.7) Only the reasonable fees of the	Delete
claimants' solicitors, surveyors and other	
experts (e.g. valuation specialist) payable by	
an authority in connection with the	
settlement of approved claims are eligible for	
consideration for grant.	
(9.1) Operating authorities are expected to	(9.1) Operating authorities are expected to obtain
obtain contributions wherever possible	contributions wherever possible towards the cost of
towards the cost of a scheme from those who	a scheme from those who will derive benefit from it
will derive benefit from it or whose actions or	or whose actions or requests have led to the
requests have led to the scheme being	scheme being undertaken. <b>Under Partnership</b>
undertaken.	Funding the grant payable will be determined on
	the basis of the outcomes delivered by the scheme.
(9.3) Where contributions are made by other	Delete
FCERM authorities or riparian owners	
towards essential FCERM works (excluding	
betterment) they are normally classed as	
"windfalls" towards the scheme and are not	
deducted before calculating grant (non-	
deductible contributions). The amount of	
grant payable is the scheme cost less the	
contributions received.	
(9.5) Deductible contributions should be	(9.4) Deductible contributions should be sought:
sought:	(i) where part of the works will facilitate new
(i) where part of the works will facilitate new	development, <b>(Development post 1</b>

development,	January 2012 cannot be included in bid for grant)
(ii) in respect of betterment value where a	(ii) in respect of betterment value where a project
project requires existing structures or	requires existing structures or property (e.g. roads,
property (e.g. roads, bridges) to be replaced,	bridges) to be replaced, altered or made good, or
altered or made good, or where advantage is	where advantage is taken by others to improve or
taken by others to improve or alter the	alter the original structure or works,
original structure or works,	(iii) for works in commutation of existing liabilities,
(iii) for works in commutation of existing	or
liabilities, or	(iv) for works undertaken as a result of subsidence
(iv) for works undertaken as a result of	due to coal mining or other extractive operations.
subsidence due to coal mining or other	
extractive operations.	
(10.1) Formal approval represents a	(10.1) A risk management authority will submit a
commitment to the funding for the study,	Project Appraisal Report with recommendations on
strategy or works, subject to the terms of this	costs and options, identifying a preferred option.
Memorandum. A formal approval letter from	The Environment Agency will assess the report and
the Environment Agency with an approved	submit for approval. In order to assist risk
grant eligible sum is necessary before work	management authorities negotiate contributions
can commence (except in emergencies).	the Environment Agency will provide a formal
	technical approval to the proposal. This is not
	approval and allocation of the funding at that
	stage, but enables the partnership funding grant
	estimate to be prepared.
(10.2) For formal approval, submission of a	(10.2) Once the authority confirms that all match
FCERM7 or FCERM2 plus a project appraisal	funding requirements are in place the Environment
report (PAR), signed by officers authorised by	Agency will provide formal financial approval for
the Authority; and undertaking to comply	the grant. Formal financial approval represents a
with grant conditions should be submitted to	commitment to the grant funding for the study,
the Area FCRM Manager. The attention of	strategy or works, subject to the terms of this
consultants employed on a project should be	Memorandum. A formal approval letter from the
drawn to these conditions regarding	Environment Agency with the approved
approval.	partnership grant is necessary before work can
	commence (except in emergencies). The grant offer
	is a capped sum based on the outcomes provided.

	(12.1) The Partnership Funding grant is based on
	outcomes and caps the Exchequers contribution.
	The Environment Agency manages project
	contingency at a programme level and does not
	allocate contingency to the individual projects. The
	allocation made to the risk management authority
	can be increased up to the value of contingency
	included in the approved sum. Requests to increase
	the grant approved should be made using an
	FCERM4.
(12.2) Following the changes in DCLG rules	(12.2) Delete first part.
the approved sum notified to the Local	Immediately it appears likely that expenditure in
Authority is the project approval sum less	excess of the approved sum may have to be
ineligible items and contingency. The	incurred in order to complete a scheme details of
Environment Agency manages contingency at	the reasons and levels of expected extra costs must
a program level and will consider applications	be submitted on form FCERM 4 for formal approval.
for additional financial support within the	The Environment Agency is keen to ensure all
costs identified at project appraisal.	schemes are delivered, will support the risk
Immediately it appears likely that	management authority to secure additional funding
expenditure in excess of the approved sum	and may consider offering additional grant support.
may have to be incurred in order to complete	Failure to seek prior approval for excess
a scheme details of the reasons and levels of	expenditure may result in it not being accepted for
expected extra costs must be submitted on	supplementary grant.
form FCERM 4 for formal approval. The	
Environment Agency is keen to ensure all	
schemes are delivered, will support the risk	
management authority to secure additional	
funding and may consider offering additional	
grant support. Failure to seek prior approval	
for excess expenditure may result in it not	
being accepted for supplementary grant.	
	(17.4) Partnership Funding is made available on the
	basis of whole life costs and therefore includes a
	commuted element for future maintenance. The

	commuted element should be used by the
	Authority to provide future years funding for
	maintenance.
(21.1) IDB rates vary, generally studies and	(21.1) The rate of grant applicable for a scheme
works are grant aided at 45%. Special grant	depends upon the outcomes delivered. Previous
rates are available to IDBs undertaking capital	grant rates are replaced by the outcome based
works (and associated studies) to implement	payment. There is one restriction, studies and
agreed water level management plans	schemes that are proposed outside of a strategic
ineligible items.	FCERM plan for the area will be capped at a grant
Lists of rate applicable	rate of 45% of the approved partnership funding
(a) sites of national importance (50%),	outcomes.
(b) internationally important conservation	
sites (80%) and	
(c) certain priority sites listed by Natural	
England – Natural 2000 sites (100%).	
(21.2) For Local Authority schemes only,	Delete
100% grant rate will be paid on approved	
eligible costs for all types of work.	
(22.1) Subject to an authority carrying out an	(22.1) Grant will be paid at the appropriate rate as
approved scheme in a satisfactory manner	the scheme progresses. Failure to follow grant
and observing all the conditions laid down in	terms and conditions could result in the
this memorandum, grant will be paid at the	Environment Agency withholding grant.
appropriate rate on the actual approved	
eligible expenditure. Failure to follow grant	
terms and conditions could result in the	
Environment Agency withholding grant.	
(22.3) Payments will be made to the	(22.3) If the Regional Flood and Coastal Committee
authority's bankers normally within one	are providing match funding or have determined to
month of the submission of the FCERM 3.	fund 100% of eligible costs of a project the grant
	will be provided using the same route. The
	approval letter will specify the proportions of
	grant. The authority makes a single claim using the
	FCERM3 form and the Environment Agency will
	ensure that the correct apportionment of grant is

made.
(22.5) Grant may be withheld in respect of any
scheme where the start date has not been notified,
or if the grant claim profile changes over more
than one financial year, without the prior
agreement of the Environment Agency.

Source: EA (2011); (2012b)